# UNIVERSITY OF OKLAHOMA GRADUATE COLLEGE

# A COUPLED CHEMO-THERMO-POROELASTIC SYSTEM: WELL-POSEDNESS AND APPROXIMATION FOR CONTROL APPLICATIONS

A DISSERTATION

### SUBMITTED TO THE GRADUATE FACULTY

in partial fulfillment of the requirements for the

Degree of

DOCTOR OF PHILOSOPHY

By

TETYANA M. MALYSHEVA Norman, Oklahoma 2015

### A COUPLED CHEMO-THERMO-POROELASTIC SYSTEM: WELL-POSEDNESS AND APPROXIMATION FOR CONTROL APPLICATIONS

# A DISSERTATION APPROVED FOR THE DEPARTMENT OF MATHEMATICS

BY

Dr. Luther White, Chair

Dr. Musharraf Zaman

Dr. Kevin Grasse

Dr. John Albert

Dr. Christian Remling

© Copyright by TETYANA M. MALYSHEVA 2015 All Rights Reserved.

#### Acknowledgements

First and foremost, I would like to express my sincere and deepest gratitude to my Ph.D. advisor, Professor Luther W. White, for his professional and inspirational guidance, unconditional support and dedication, continuous encouragement, and believing in me throughout my Ph.D. studies. I am very grateful for having had the chance to work with him.

I would like to specially thank Dr. Kevin Grasse, Dr. John Albert, Dr. Christian Remling, and Dr. Musharraf Zaman for agreeing to serve on my Doctoral Advisory Committee, their valuable time, and support.

I am also thankful to all the faculty and staff members of the Department of Mathematics at the University of Oklahoma for providing a supportive environment for my study. I am especially grateful to Dr. Andy Miller, Dr. Kevin Grasse, and Dr. Keri Kornelson for everything they did to help me with my professional growth.

Last but not least, I would like to wholeheartedly thank my mother, Liudmyla Popovichieva, and my husband, Yuri Malyshev, for their unconditional and endless love, incredible support, constant encouragement, and unwavering confidence in my ability to achieve this goal.

# Table of Contents

1	Inti	roduct	ion	1				
<b>2</b>	Coupled TCPu System: Underlying Equations							
	2.1	Two-o	limensional Model	5				
	2.2	Three	-dimensional Model	15				
	2.3	Diago	nalized TCP System with Homogeneous Boundary Condition	. s. 26				
3	We	ll-Pose	dness of the Coupled TCPu Problem	32				
	3.1	TCP	Problem	32				
	3.2	Elasti	c Problem	40				
	3.3	Coupl	ed TCPu Problem	47				
4	Fou	rier-F	inite Element Method for TCPu Problems	53				
	4.1	2D T(	CPu Problem	53				
		4.1.1	Fourier-type approximation of the TCP sytem	53				
		4.1.2	Fourier-type approximation of the elastic system	57				
		4.1.3	Finite element approximation	64				
	4.2	3D T(	CPu Problem for an Inclined Borehole	80				
		4.2.1	Fourier-type approximation of the TCP sytem	81				
		4.2.2	Fourier-type approximation of the elastic system	88				
		4.2.3	Finite element approximation	104				
<b>5</b>	Exp	oerime	ntal Validation of the Fourier-Finite Element					
	Me	$\operatorname{thod}$		149				
6	Cor	nclusio	ns and Future Work	157				
Bi	blio	graphy		159				
$\mathbf{A}_{\mathbf{j}}$	ppen	dices		164				
$\mathbf{A}$	Phy	sical l	Parameters of the Coupled TCPu System	164				

# List of Tables

A.1	Physical properties of the rock/fluid system			•	•	•	•	•	•	•	. 1	165
A.2	Input parameters for experimental validation	•	•	•		•	•				. 1	165

# List of Figures

2.1	Two-dimensional region exterior to the borehole $\ldots \ldots \ldots$
2.2	Three-dimensional region exterior to the borehole
4.1	r-approximation of 2D Dirichlet boundary conditions
4.2	r-approximation of 2D Neumann boundary conditions
4.3	$r\text{-approximation of 3D}$ Dirichlet boundary conditions $\ .\ .\ .\ .\ .\ .$ . 104
4.4	z-approximation of 3D Dirichlet boundary conditions
4.5	$r\text{-}\mathrm{approximation}$ of 3D Neumann boundary conditions
4.6	z-approximation of 3D Neumann boundary conditions
5.1	Temperature distribution around the borehole
5.2	Solute mass fraction distribution around the borehole
5.3	Pore pressure distribution around the borehole
5.4	Total radial stress distribution around the borehole at the polar angle of $\theta = 0$ 151
55	Total tangential stress distribution around the borehole at the polar
0.0	angle of $\theta = 0$ 152
5.6	Total radial stress distribution around the borehole at the polar angle
	of $\theta = 30 \dots $
5.7	Total tangential stress distribution around the borehole at the polar
	angle of $\theta = 30$
5.8	Total radial stress distribution around the borehole at the polar angle
	of $\theta = 45 \dots $
5.9	Total tangential stress distribution around the borehole at the polar
	angle of $\theta = 45$
5.10	Total radial stress distribution around the borehole at the polar angle
	of $\theta = 60 \dots 154$
5.11	Total tangential stress distribution around the borehole at the polar
	angle of $\theta = 60$
5.12	Total radial stress distribution around the borehole at the polar angle
	of $\theta = 90$
5.13	Total tangential stress distribution around the borehole at the polar
	angle of $\theta = 90$

### Abstract

This dissertation addresses well-posedness and approximation problems for coupled parabolic-elliptic systems with applications to geomechanics. This work is motivated by problems of borehole stability in porous formations that involve the modeling of fully coupled thermal, chemical, hydraulic, and mechanical processes. The underlying thermo-chemo-poroelastic (TCPu) model is a system of time dependent parabolic equations modeling thermal, solute, and fluid diffusions coupled with Navier-type elliptic equations that attempt to capture the elastic behavior of rock around the borehole. The obtained results are fundamental to posing optimal control problems using borehole parameters as controls to achieve desired stress distributions in the neighborhood of the borehole.

Sufficient conditions for well-posedness (in the sense of Hadamard) of the coupled parabolic-elliptic initial-boundary value problems are obtained for the two- and three-dimensional TCPu models. These results provide a mathematical basis for the general theory of fully coupled chemical thermo-poroelasticity in homogeneous isotropic porous media saturated by a fluid.

Numerical methods for solving fully coupled parabolic-elliptic initial-boundary value problems for two- and three-dimensional TCPu processes, including radially non-symmetric processes in an inclined borehole, are developed. These methods are based on a hybrid Fourier-finite-element approximation technique proposed in this research. The technique also involves a boundary penalization strategy and captures borehole geometry and mechanics. The numerical methods allow numerical analysis of a control problem in which the temperature and pressure on the borehole boundary are considered as control parameters used to achieve desirable stresses in the neighborhood of the borehole.

#### Chapter 1

### Introduction

Natural systems are usually complex and composed of numerous interacting parts. Such interactions require coupled models that, in turn, create a need for advanced mathematical analysis and control techniques. In this dissertation, we consider initialboundary value problems for coupled parabolic-elliptic systems with applications to geomechanics. This work is motivated by the problems of borehole stability in porous formations saturated by a fluid that involve the modeling of fully coupled thermal, chemical, hydraulic, and mechanical (elastic) processes.

The poroelasticity theory describing the coupled processes of elastic deformation and pore fluid diffusion in fluid-saturated isothermal porous media can be traced back to the pioneering works of von Terzaghi [41, 42] (1925, 1943) and Biot [5] (1941). In 1956, Biot [6] pointed out a complete mathematical analogy between poroelasticity and thermoelasticity with the temperature playing the same role as the fluid pressure and heat conduction corresponding to fluid flow. A complete mathematical analysis of the well-posedness of a general initial-boundary value problem for a system of coupled partial differential equations that describes the Biot consolidation model [5] in poroelasticity, as well as a coupled quasi-static problem in thermoelasticity, has been carried out by Showalter [37, 38]. The corresponding Biot consolidation model has the form

$$-(\lambda + \mu)\nabla(\nabla \cdot \mathbf{u}(x, t)) - \mu\Delta\mathbf{u}(x, t) + \alpha\nabla p(x, t) = \mathbf{f}(x, t)$$
(1.0.1)

$$\frac{\partial}{\partial t}(c_0 p(x,t) + \alpha \nabla \cdot \mathbf{u}(x,t)) - \nabla \cdot k \nabla p(x,t) = h(x,t)$$
(1.0.2)

consisting of the equilibrium equation for momentum (1.0.1) and the diffusion equation for Darcy flow (1.0.2). Here  $\mathbf{u}(x,t)$  is the displacement of the stucture; p(x,t) is the fluid pore pressure; the positive Lamé constants  $\lambda$  and  $\mu$  are the dilation and shear moduli of elasticity, respectively;  $\alpha > 0$  is the Biot-Willis constant that accounts for the pressure-deformation coupling; the constant  $c_0 \ge 0$  combines the porosity of the medium and the compressibility of the fluid; k > 0 involves the permeability of the medium and the viscosity of the fluid; f and g are given data, typically the volume-distributed external forces and the volume-distributed source density, respectively. In the context of the coupled thermoelasticity system, p(x,t)denotes the temperature;  $c_0 > 0$  is the specific heat of the medium; and k > 0 is the conductivity.

Based on the theory of linear degenerate evolution equations in Hilbert spaces, the existence and uniqueness of strong and weak solutions, as well as regularity theory for the system (1.0.1)-(1.0.2) supplemented by initial and boundary conditions, were developed. The earlier works on the mathematical issues of analysis and numerical approximation for the Biot consolidation model include [2, 7, 8, 29, 44, 45, 47]. The a priori analysis of Euler-Galerkin approximations for the Biot consolidation problem is presented in the works of Murad, Loula, and Thomée [30, 31, 32], including the semi-discrete and fully discrete cases and long-time behavior; and the a posteriori error analysis was developed by Ern and Meunier [17].

In 1971, Schiffman [35] first extended the Biot theory for non-isothermal systems to account for the effects of thermal expansion of both the pore fluid and the elastic matrix. Since then, a substantial literature on thermo-poroelasticity theory and the modeling of the coupled hydro-thermo-mechanical behavior of a fluid-saturated porous media has been developed for geomechanics, including petroleum and geothermal borehole stability [1, 4, 10, 14, 23, 24, 26, 28, 33, 39, 43]. Considerable attention has been recently placed on the impact of chemical processes in porous media on drilling and borehole stability. Ghassemi and Diek [20] developed a linear chemoporoelastic model that couples solute transfer in the mud/shale system to stress and pore pressure within the framework of a Biot-like isotropic poroelastic theory. Thereafter, Zhou and Ghassemi [46] and Rafieepour et al. [34] presented linear and nonlinear chemo-thermo-poroelastic models, respectively. However, all those models are not rigorously developed from the first and second laws of thermodynamics. In 2011, Diek et al. [12, 13, 11] first presented fully coupled thermo-poroelastic and chemo-thermo-poroelastic models for drilling in geological formations that satisfy the first and second laws of thermodynamics.

Due to the complexity of cross-coupling mechanisms involved in thermo-poroelastic and chemo-thermo-poroelastic models, the majority of corresponding initialboundary value problems are solved by various numerical and computational techniques [18, 19, 20, 25, 34]. Very few analytical solutions of the fully coupled thermoporoelastic and chemo-poroelastic equations are currently available. Typically, the solutions are derived under the assumptions that some of the couplings can be neglected [10, 23, 24, 28]. For instance, in the works of Kodashima and Kurashige [23, 24], the displacement field is decoupled from the temperature and pore fluid pressure fields; and in Coussy [10], an analytical solution is presented under the assumption that the temperature equation can be decoupled and approximated by a purely diffusive equation. Belotserkovets and Prevost [4], using the Laplace transformation and the residue theorem, developed an analytical method and presented an exact unique solution for a specific case of the fully coupled thermo-hydro-mechanical response of a fluid saturated porous sphere under mechanical pulse load.

In this dissertation, we study the model that is based on equations derived by Diek [11] and constitutes the general theory of fully coupled chemo-thermo-poroelasticity in homogeneous isotropic porous media saturated by a fluid. The theory satisfies the first and second laws of thermodynamics and is based on concepts of irreversible thermodynamics, a novel rock constitutive relation, and Onsager's transport phenomenology [11]. Since the model is derived from physical laws, it requires a mathematical basis for its well-posedness. On the other hand, the geomechanical nature of the problem demands analytical and numerical methods that capture both the geometry and the physical background of the problem. With this in mind, the aims of the dissertation are to establish the mathematical theory of well-posedness for coupled linear parabolic-elliptic systems modeling fully coupled mechanical, hydraulic, chemical, and thermal processes in homogeneous isotropic media and to develop and experimentally validate numerical methods for approximation and boundary control problems in geomechanical applications.

The dissertation is organized in chapters as follows. Chapter 2 describes the underlying fully coupled thermo-chemo-poroelastic (TCPu) model for fluid-saturated porous rock formations and formulates the corresponding coupled parabolic-elliptic initial-boundary value problems for two- and three-dimensional regions. Chapter 3 addresses the well-posedness theory for the initial-boundary value problems introduced in Chapter 2. Chapter 4 develops numerical methods for approximation and boundary control problems for the two- and three-dimensional TCPu models. Chapter 5 presents experimental validation of the numerical methods developed in Chapter 4. Chapter 6 contains the conclusions and potential directions for further research.

#### Chapter 2

### Coupled TCPu System: Underlying Equations

The underlying TCPu model [11] is a system of time-dependent parabolic partial differential equations (PDEs) coupled with Navier-type elliptic PDEs with time,  $t \in (0, t_f)$ , as a parameter. The parabolic equations represent heat, solute, and fluid diffusions, and the Navier-type elliptic equations attempt to capture the elastic behavior of rock, while incorporating thermal, chemical, and porous media effects. The equations are developed in terms of the following independent variables: the absolute temperature T, the solute mass fraction C, the pore pressure p, and the vector of solid displacements  $\vec{u}$ . The coupled partial differential equations supplemented by the appropriate initial and boundary conditions constitute an initial-boundary value problem defined in a region exterior to the borehole. Although this region is infinite, it is specified with a far-field radius such that, without loss of generality, we may assume that, on the far-field boundary, the absolute temperature, solute mass fraction, pore pressure, and displacements are time-independent, and the displacement is negligibly small.

#### 2.1 Two-dimensional Model

The initial-boundary value problem that constitutes our model is defined in a region  $\Omega$  in  $\mathbb{R}^2$  exterior to the borehole. We assume that  $\Omega$  is a bounded open domain with a sufficiently smooth boundary  $\Gamma$ . The boundary  $\Gamma$  is expressed as the union of disjoint sets  $\Gamma_B$  and  $\Gamma_F$  having nonempty interiors relative to the set  $\Gamma$ . Specifically,  $\Omega \subset \mathbb{R}^2$  is an annular region with an inner radius  $R_B$  and an outer radius  $R_F$ .

Let (X, Y) be a borehole Cartesian coordinate system such that

x -axis is in the direction of maximum horizontal principal stress  $\sigma_H$  and

y -axis is in the direction of minimum horizontal principal stress  $\sigma_h$ .

A point  $\bar{x}$  in  $\mathbb{R}^2$  is associated with position vector  $\bar{x} = x\bar{i} + y\bar{j}$ , where  $\bar{i}$  and  $\bar{j}$  are the basis vectors of the Cartesian coordinate system (X, Y). Displacements of points within  $\Omega$  are described in terms of a vector-valued displacement function of the position  $\bar{x}$  and time t as  $\vec{u}(\bar{x}, t) = u(\bar{x}, t)\vec{i} + v(\bar{x}, t)\vec{j}$ .

The boundary  $\Gamma$  of  $\Omega$  is  $\Gamma = \Gamma_B \cup \Gamma_F$ , where

$$\Gamma_B = \left\{ \bar{x} = (x, y) \in \mathbb{R}^2 : \sqrt{x^2 + y^2} = R_B \right\}$$
  
$$\Gamma_F = \left\{ \bar{x} = (x, y) \in \mathbb{R}^2 : \sqrt{x^2 + y^2} = R_F \right\}$$





Figure 2.1: Two-dimensional region exterior to the borehole

The initial-boundary value problem that constitutes the two-dimensional radially nonsymmetric TCPu model is the following.

Thermal diffusion:

$$\Lambda \dot{T} + \Sigma \dot{C} + \Phi \dot{p} - \frac{k^T}{T_F} \nabla^2 T - \frac{\rho_f R T_F D^T}{M^s (1 - C_F)} \nabla^2 C + K^T \nabla^2 p = -\zeta (\nabla \cdot \dot{\vec{u}}) \qquad \text{in } \Omega \times (0, t_f)$$

$$(2.1.1)$$

with boundary conditions

$$T(\bar{x},t) = T_B(t) \qquad \text{on } \Gamma_B \times [0,t_f) \tag{2.1.2}$$

$$T(\bar{x},t) = T_F(\bar{x}) \qquad \text{on } \Gamma_F \times [0,t_f) \tag{2.1.3}$$

and initial conditions

$$T(\bar{x},0) = \tilde{T}_I(\bar{x}) \qquad \text{in } \Omega \tag{2.1.4}$$

Solute diffusion:

$$\phi \dot{C} - C_F D^T \nabla^2 T - D \nabla^2 C + \frac{k \Re}{\eta} \nabla^2 p = 0 \qquad \text{in } \Omega \times (0, t_f)$$
(2.1.5)

with boundary conditions

$$C(\bar{x},t) = C_B(t) \qquad \text{on } \Gamma_B \times [0,t_f) \tag{2.1.6}$$

$$C(\bar{x},t) = C_F(\bar{x}) \qquad \text{on } \Gamma_F \times [0,t_f) \tag{2.1.7}$$

and initial conditions

$$C(\bar{x},0) = \tilde{C}_I(\bar{x}) \qquad \text{in } \Omega \tag{2.1.8}$$

Fluid diffusion:

$$\Gamma \dot{T} + \chi \dot{C} + \Psi \dot{p} + K^T \nabla^2 T + \frac{\rho_f \Re \Omega k}{\eta} \nabla^2 C - \frac{k}{\eta} \nabla^2 p = -\alpha (\nabla \cdot \dot{\vec{u}}) \quad \text{in } \Omega \times (0, t_f) \quad (2.1.9)$$

with boundary conditions

$$p(\bar{x},t) = p_B(t) \qquad \text{on } \Gamma_B \times [0,t_f) \tag{2.1.10}$$

$$p(\bar{x},t) = p_F(\bar{x}) \qquad \text{on } \Gamma_F \times [0,t_f) \tag{2.1.11}$$

and initial conditions

$$p(\bar{x},0) = \tilde{p}_I(\bar{x}) \qquad \text{in } \Omega \tag{2.1.12}$$

Navier-type elastic equations:

$$\left(K + \frac{G}{3}\right)\nabla(\nabla \cdot \dot{\vec{u}}) + G\nabla^2 \dot{\vec{u}} - \tilde{\zeta}\nabla \dot{T} + \xi\nabla \dot{C} - \tilde{\alpha}\nabla \dot{p} = 0 \quad \text{in } \Omega \times (0, t_f) \quad (2.1.13)$$

with boundary conditions

$$\vec{u}(\bar{x},t) \approx 0$$
 on  $\Gamma_F \times (0,t_f)$  (2.1.14)

$$\dot{\vec{u}}(\bar{x},t) \approx 0$$
 on  $\Gamma_F \times (0,t_f)$  (2.1.15)

and initial conditions

$$\vec{u}(\bar{x},0) = 0 \qquad \text{in } \bar{\Omega} \tag{2.1.16}$$

The description and values of physical constants are presented in Appendix A.

It is convenient to consider the thermal diffusion, solute diffusion, and fluid diffusion as a system (TCP system). Hence, we introduce the following matrices and vectors:

$$M = \begin{bmatrix} \Lambda & \Sigma & \Phi \\ 0 & \phi & 0 \\ \Gamma & \chi & \Psi \end{bmatrix}, \quad A = \begin{bmatrix} \frac{k^T}{T_F} & \frac{\rho_f R T_F D^T}{M^s (1 - C_F)} & -K^T \\ C_F D^T & D & -\frac{k \Re}{\eta} \\ -K^T & -\frac{\rho_f \Re \Omega k}{\eta} & \frac{k}{\eta} \end{bmatrix}, \quad \vec{b}_0 = \begin{bmatrix} \zeta \\ 0 \\ \alpha \end{bmatrix} \quad (2.1.17)$$

$$\bar{V}(\bar{x},t) = \begin{bmatrix} T\\ C\\ p \end{bmatrix} (\bar{x},t), \qquad V_B = \begin{bmatrix} T_B\\ C_B\\ p_B \end{bmatrix}, \qquad V_F = \begin{bmatrix} T_F\\ C_F\\ p_F \end{bmatrix}$$
(2.1.18)

With the notations (2.1.17) and (2.1.18), the initial-boundary value problem (2.1.1)-(2.1.12) has the form

$$M\dot{\bar{V}} - A\nabla^2 \bar{V} = -\vec{b}_0 (\nabla \cdot \dot{\vec{u}}) \qquad \text{in } \Omega \times (0, t_f)$$
(2.1.19)

with boundary conditions

$$\bar{V}(\bar{x},t) = \begin{cases} V_B(t) & \text{on } \Gamma_B \times [0,t_f) \\ V_F(\bar{x}) & \text{on } \Gamma_F \times [0,t_f) \end{cases}$$
(2.1.20)

and initial conditions

$$\bar{V}(\bar{x},0) = \bar{V}_I(\bar{x}) \qquad \text{in } \Omega \tag{2.1.21}$$

Note that  $\vec{b}_0$  is the factor that couples the displacement velocity to the absolute temperature, the pore pressure, the solute mass fraction (TCP), and their velocities.

By the principle of minimum total potential energy, the region  $\Omega$  shall displace to a position that minimizes the total potential energy; it is the stable configuration for equilibrium.

The total potential energy of our system is

$$\mathcal{V}(\vec{u}) = \mathcal{V}_S(\vec{u}) - W_b(\vec{u}) - W_S(\vec{u}) \tag{2.1.22}$$

where  $\mathcal{V}_S(\vec{u})$  is the elastic energy of the system;  $W_b(\vec{u})$  is work done by body forces due to the absolute temperature, the solute mass fraction, and the pore pressure; and  $W_S(\vec{u})$  is work done by applied boundary stress.

Now we specify  $\mathcal{V}_S(\vec{u})$ ,  $W_b(\vec{u})$ , and  $W_S(\vec{u})$ . Here and in the following we will suppress the time dependence of the displacement vector for the sake of brevity.

Given the displacement

$$\vec{u} = u\vec{i} + v\vec{j}$$

the linearized strain is a second order symmetric tensor

$$\varepsilon = \begin{bmatrix} \varepsilon_{11} & \varepsilon_{12} \\ \varepsilon_{21} & \varepsilon_{22} \end{bmatrix} = \frac{1}{2} (\nabla \vec{u} + \nabla \vec{u}^T)$$

where

$$\varepsilon_{11} = u_x, \qquad \varepsilon_{12} = \varepsilon_{21} = \frac{1}{2}(u_y + v_x), \qquad \varepsilon_{22} = v_y$$

In the linear elasticity theory, a relation between the stress tensor  $\tau$  and the linearized strain tensor  $\varepsilon(\vec{u})$  is given by

$$\tau_{ij}(\vec{u}) = a_{ijkh}\varepsilon_{kh}(\vec{u}) \tag{2.1.23}$$

where  $a_{ijkh}$  are the coefficients of elasticity, independent of the strain tensor.

The coefficients of elasticity have properties of symmetry

$$a_{ijkh} = a_{jihk} = a_{khij} \tag{2.1.24}$$

and of ellipticity: There exists a constant  $\alpha_1 > 0$  such that

$$a_{ijkh}\varepsilon_{ij}\varepsilon_{kh} \ge \alpha_1\varepsilon_{ij}\varepsilon_{ij}, \quad \forall \ \varepsilon_{ij}.$$
 (2.1.25)

In equations (2.1.23)-(2.1.25), we made use of the Einstein summation convention concerning repeated indices.

For a homogeneous elastic isotropic medium, the stress-strain relation (2.1.23) in terms of the bulk modulus K and the shear modulus G has the form

$$\tau = \begin{bmatrix} \tau_{11} & \tau_{12} \\ \tau_{21} & \tau_{22} \end{bmatrix} = 2G\varepsilon + \left(K - \frac{2}{3}G\right)(\operatorname{tr}\varepsilon)I \tag{2.1.26}$$

That is,

$$\tau_{11} = 2Gu_x + \left(K - \frac{2}{3}G\right)\left(u_x + v_y\right)$$
  

$$\tau_{12} = \tau_{21} = G\left(u_y + v_x\right)$$
  

$$\tau_{22} = 2Gv_y + \left(K - \frac{2}{3}G\right)\left(u_x + v_y\right)$$
  
(2.1.27)

The elastic strain energy of the system is

$$\mathcal{V}_{S}(\vec{u}) = \frac{1}{2} \int_{\Omega} \tau_{11}(\vec{u}) \varepsilon_{11}(\vec{u}) + 2\tau_{12}(\vec{u}) \varepsilon_{12}(\vec{u}) + \tau_{22}(\vec{u}) \varepsilon_{22}(\vec{u}) \,\mathrm{d}\Omega \tag{2.1.28}$$

The work done by body forces due to the absolute temperature, the solute mass fraction, and the pore pressure is

$$W_b(\vec{u}) = \int_{\Omega} f(\operatorname{tr}\varepsilon(\vec{u})) \,\mathrm{d}\Omega \qquad (2.1.29)$$

where

$$f = \vec{b}_1^T \bar{V}, \qquad \vec{b}_1 = [\tilde{\zeta} - \xi \ \tilde{\alpha}]^T, \qquad \bar{V} = [T \ C \ p]^T$$
(2.1.30)

Note that  $\vec{b}_1$  is the factor that couples the absolute temperature, the pore pressure, and the solute mass fraction (TCP) to the displacement.

The work done by applied boundary stress is

$$W_S(\vec{u}) = \int_{\Gamma} (\hat{\sigma}\vec{n}) \cdot \vec{u} \,\mathrm{d}\Gamma \tag{2.1.31}$$

where  $\hat{\sigma}$  is applied boundary stress and  $\vec{n}$  is the outward unit normal vector on the boundary.

Applying (2.1.28), (2.1.29), and (2.1.31) to (2.1.22), we have

$$\mathcal{V}(\vec{u}) = \frac{1}{2} \int_{\Omega} \tau_{11}(\vec{u}) \varepsilon_{11}(\vec{u}) + 2\tau_{12}(\vec{u}) \varepsilon_{12}(\vec{u}) + \tau_{22}(\vec{u}) \varepsilon_{22}(\vec{u}) \,\mathrm{d}\Omega - \int_{\Omega} f\left(\mathrm{tr}\varepsilon(\vec{u})\right) \,\mathrm{d}\Omega - \int_{\Gamma} (\hat{\sigma}\vec{n}) \cdot \vec{u} \,\mathrm{d}\Gamma$$
(2.1.32)

To aid in giving a mathematical formulation, let

$$H = L^{2}(\Omega; \mathbb{R}^{2}) = \left\{ L^{2}(\Omega; \mathbb{R}) \right\}^{2}$$
$$V = H^{1}(\Omega; \mathbb{R}^{2}) = \left\{ H^{1}(\Omega; \mathbb{R}) \right\}^{2}$$
$$V_{0} = \left\{ \varphi \in V : \varphi \Big|_{\Gamma_{F}} = 0 \right\}$$
$$\vec{\Phi}(\bar{x}) = \phi(\bar{x})\vec{i} + \psi(\bar{x})\vec{j}$$

At this point, we wish to express the elastic strain energy  $\mathcal{V}_S(\vec{u})$  as a functional on V. To that end, define the bilinear form on V by

$$a_E(\vec{u},\vec{\Phi}) = \int_{\Omega} \tau_{11}(\vec{u})\varepsilon_{11}(\vec{\Phi}) + 2\tau_{12}(\vec{u})\varepsilon_{12}(\vec{\Phi}) + \tau_{22}(\vec{u})\varepsilon_{22}(\vec{\Phi}) \,\mathrm{d}\Omega, \qquad \forall \ \vec{u},\vec{\Phi} \in V$$

$$(2.1.33)$$

Then from (2.1.28) and (2.1.33),

$$\mathcal{V}_S(\vec{u}) = \frac{1}{2} a_E(\vec{u}, \vec{u})$$
 (2.1.34)

and, from (2.1.32) and (2.1.34), the total potential energy has the form

$$\mathcal{V}(\vec{u}) = \frac{1}{2} a_E(\vec{u}, \vec{u}) - \int_{\Omega} f\left(\mathrm{tr}\varepsilon(\vec{u})\right) \mathrm{d}\Omega - \int_{\Gamma} (\hat{\sigma}\vec{n}) \cdot \vec{u} \,\mathrm{d}\Gamma$$
(2.1.35)

Define the following vectors:

$$\vec{\tau}_1 = [\tau_{11} \quad \tau_{12}]^T, \qquad \vec{\tau}_2 = [\tau_{21} \quad \tau_{22}]^T$$

Then

$$a_{E}(\vec{u},\vec{\Phi}) = \int_{\Omega} \tau_{11}\phi_{x} + \tau_{12}\phi_{y} + \tau_{21}\psi_{x} + \tau_{22}\psi_{y} d\Omega$$
  

$$= \int_{\Omega} \vec{\tau}_{1} \cdot \nabla\phi + \vec{\tau}_{2} \cdot \nabla\psi d\Omega$$
  

$$= \int_{\Omega} \nabla \cdot (\phi\vec{\tau}_{1}) - \phi\nabla \cdot \vec{\tau}_{1} + \nabla \cdot (\psi\vec{\tau}_{2}) - \psi\nabla \cdot \vec{\tau}_{2} d\Omega$$
  

$$= \int_{\Gamma} \phi(\vec{\tau}_{1} \cdot \vec{n}) + \psi(\vec{\tau}_{2} \cdot \vec{n}) d\Gamma - \int_{\Omega} \phi\nabla \cdot \vec{\tau}_{1} + \psi\nabla \cdot \vec{\tau}_{2} d\Omega$$
  

$$= \int_{\Gamma} (\tau\vec{n}) \cdot \vec{\Phi} d\Gamma - \int_{\Omega} [\nabla \cdot \vec{\tau}_{1} \quad \nabla \cdot \vec{\tau}_{2}]^{T} \cdot \vec{\Phi} d\Omega, \quad \forall \vec{\Phi} \in V \quad (2.1.36)$$

Thus, we obtain the following Green's formula:

$$a_E(\vec{u},\vec{\Phi}) = \int_{\Gamma} (\tau \vec{n}) \cdot \vec{\Phi} \,\mathrm{d}\Gamma - \int_{\Omega} [\nabla \cdot \vec{\tau_1} \quad \nabla \cdot \vec{\tau_2}]^T \cdot \vec{\Phi} \,\mathrm{d}\Omega, \qquad \forall \ \vec{\Phi} \in V \qquad (2.1.37)$$

Referring to (2.1.14) and (2.1.15), we assume that  $\vec{u}$  and  $\dot{\vec{u}}$  are negligible on  $\Gamma_F$ and therefore, the set of admissible displacements is

$$\mathcal{U}_{ad} = \left\{ \vec{u} \in V_0 : \dot{\vec{u}} \in V_0 \right\}$$

Using the principle of minimum total potential energy, the displacement  $\vec{u}$  that the region  $\Omega$  undergoes is given by

$$D\mathcal{V}(\vec{u})\vec{\Phi} = 0, \qquad \forall \ \vec{\Phi} \in V_0 \tag{2.1.38}$$

where

$$D\mathcal{V}(\vec{u})\vec{\Phi} = \frac{d}{d\delta}\mathcal{V}(\vec{u} + \delta\vec{\Phi})\Big|_{\delta=0}$$

is the Gâteaux differential of  $\mathcal{V}$  with increment  $\vec{\Phi}$ .

From (2.1.35), (2.1.38)

$$a_E(\vec{u},\vec{\Phi}) - \int_{\Omega} f(\operatorname{tr}\varepsilon(\vec{\Phi})) \,\mathrm{d}\Omega - \int_{\Gamma_B} (\hat{\sigma}\vec{n}) \cdot \vec{\Phi} \,\mathrm{d}\Gamma = 0, \qquad \forall \ \vec{\Phi} \in V_0 \tag{2.1.39}$$

From Green's formula (2.1.37) we have

$$a_E(\vec{u},\vec{\Phi}) = \int_{\Gamma_B} (\tau \vec{n}) \cdot \vec{\Phi} \,\mathrm{d}\Gamma - \int_{\Omega} [\nabla \cdot \vec{\tau}_1 \quad \nabla \cdot \vec{\tau}_2]^T \cdot \vec{\Phi} \,\mathrm{d}\Omega, \qquad \forall \vec{\Phi} \in V_0 \qquad (2.1.40)$$

Now

$$\begin{split} \int_{\Omega} f\left(\mathrm{tr}\varepsilon(\vec{\Phi})\right) \mathrm{d}\Omega &= \int_{\Omega} f(\nabla \cdot \vec{\Phi}) \mathrm{d}\Omega \\ &= \int_{\Omega} \nabla \cdot (f\vec{\Phi}) - \nabla f \cdot \vec{\Phi} \mathrm{d}\Omega \\ &= \int_{\Gamma_B} f\vec{\Phi} \cdot \vec{n} \, \mathrm{d}\Gamma - \int_{\Omega} \nabla f \cdot \vec{\Phi} \, \mathrm{d}\Omega \\ &= \int_{\Gamma_B} (fI\vec{n}) \cdot \vec{\Phi} \, \mathrm{d}\Gamma - \int_{\Omega} \nabla f \cdot \vec{\Phi} \, \mathrm{d}\Omega, \qquad \forall \ \vec{\Phi} \in V_0 \qquad (2.1.41) \end{split}$$

Applying (2.1.40) and (2.1.41) to (2.1.39), we have

$$\begin{split} \int_{\Gamma_B} (\tau \vec{n}) \cdot \vec{\Phi} \, \mathrm{d}\Gamma &- \int_{\Omega} [\nabla \cdot \vec{\tau}_1 \quad \nabla \cdot \vec{\tau}_2]^T \cdot \vec{\Phi} \, \mathrm{d}\Omega - \\ &- \int_{\Gamma_B} (fI\vec{n}) \cdot \vec{\Phi} \, \mathrm{d}\Gamma + \int_{\Omega} \nabla f \cdot \vec{\Phi} \, \mathrm{d}\Omega - \int_{\Gamma_B} (\hat{\sigma} \vec{n}) \cdot \vec{\Phi} \, \mathrm{d}\Gamma = 0, \qquad \Rightarrow \end{split}$$

$$\int_{\Gamma_B} (\tau - fI - \hat{\sigma}) \vec{n} \cdot \vec{\Phi} \, \mathrm{d}\Gamma + \int_{\Omega} \left( - \begin{bmatrix} \nabla \cdot \vec{\tau}_1 & \nabla \cdot \vec{\tau}_2 \end{bmatrix}^T + \nabla f \right) \cdot \vec{\Phi} \, \mathrm{d}\Omega = 0, \qquad \forall \ \vec{\Phi} \in V_0$$

It follows that

$$\begin{bmatrix} \nabla \cdot \vec{\tau}_1 & \nabla \cdot \vec{\tau}_2 \end{bmatrix}^T = \nabla f \quad \text{in } \Omega \tag{2.1.42}$$

$$\tau \vec{n} = (fI + \hat{\sigma})\vec{n}$$
 on  $\Gamma_B$  (2.1.43)

From (2.1.27),

$$\nabla \cdot \vec{\tau}_{1} = \tau_{11_{x}} + \tau_{12_{y}} = \left(K + \frac{G}{3}\right)u_{xx} + \left(K - \frac{2G}{3}\right)v_{yx} + Gu_{xx} + Gu_{yy} + Gv_{xy}$$
$$= \left(K + \frac{G}{3}\right)u_{xx} + \left(K + \frac{G}{3}\right)v_{yx} + Gu_{xx} + Gu_{yy}$$
$$\nabla \cdot \vec{\tau}_{2} = \tau_{21_{x}} + \tau_{22_{y}} = Gu_{yx} + Gv_{xx} + \left(K + \frac{G}{3}\right)v_{yy} + \left(K - \frac{2G}{3}\right)u_{xy} + Gv_{yy}$$
$$= \left(K + \frac{G}{3}\right)u_{xy} + \left(K + \frac{G}{3}\right)v_{yy} + Gv_{xx} + Gv_{yy} \Rightarrow$$

$$\begin{bmatrix} \nabla \cdot \vec{\tau_1} \\ \nabla \cdot \vec{\tau_2} \end{bmatrix} = \left( K + \frac{G}{3} \right) \begin{bmatrix} (u_x + v_y)_x \\ (u_x + v_y)_y \end{bmatrix} + G \begin{bmatrix} (u_{xx} + u_{yy}) \\ (v_{xx} + v_{yy}) \end{bmatrix} \Rightarrow$$

$$\begin{bmatrix} \nabla \cdot \vec{\tau}_1 & \nabla \cdot \vec{\tau}_2 \end{bmatrix}^T = \left( K + \frac{G}{3} \right) \nabla (\nabla \cdot \vec{u}) + G \nabla^2 \vec{u}$$
(2.1.44)

From (2.1.42)-(2.1.44) and (2.1.15),

$$\left(K + \frac{G}{3}\right)\nabla(\nabla \cdot \vec{u}) + G\nabla^2 \vec{u} = \nabla f \qquad \text{in } \Omega \times (0, t_f) \tag{2.1.45}$$

$$\tau \vec{n} = (fI + \hat{\sigma})\vec{n}$$
 on  $\Gamma_B \times (0, t_f)$  (2.1.43)

$$\vec{u} \approx 0$$
 on  $\Gamma_F \times (0, t_f)$  (2.1.14)

Differentiating (2.1.45) with respect to time and applying (2.1.30), we obtain the system (2.1.13). On the other hand, since  $\dot{\vec{u}} \in V_0$ , from (2.1.39) we have

$$a_E(\dot{\vec{u}},\vec{\Phi}) - \int_{\Omega} f(\mathrm{tr}\varepsilon(\vec{\Phi})) \,\mathrm{d}\Omega - \int_{\Gamma_B} (\hat{\sigma}\vec{n}) \cdot \vec{\Phi} \,\mathrm{d}\Gamma = 0, \qquad \forall \ \vec{\Phi} \in V_0 \qquad (2.1.46)$$

Using the same argument as above and applying (2.1.30), we obtain the equivalence of systems (2.1.45), (2.1.43), (2.1.14) and (2.1.13)-(2.1.15).

Remark 2.1 ([3]). The condition for mechanical equilibrium is

$$\nabla \cdot \sigma = 0$$

where  $\sigma = \tau - fI$  is the poroelastic stress tensor.

From this relation, (2.1.27), and (2.1.30), we obtain the system (2.1.45) that constitutes the equations of equilibrium.

#### 2.2 Three-dimensional Model

The initial-boundary value problem that constitutes our model is defined in a bounded open domain  $\Omega$  in  $\mathbb{R}^3$  exterior to the borehole, with a sufficiently smooth boundary  $\Gamma$ . Specifically,  $\Omega \subset \mathbb{R}^3$  is an inclined cylindrical region with an inner radius  $R_B$ and an outer radius  $R_F$ .

Let (X, Y, Z) be a Cartesian coordinate system such that

x -axis is in the direction of maximum horizontal principal stress  $\sigma_H(z)$ ,

y -axis is in the direction of minimum horizontal principal stress  $\sigma_h(z)$ , and

z -axis is in the direction of overburden stress  $\sigma_V$  and

 $(X_b, Y_b, Z_b)$  be a borehole Cartesian coordinate system.

A point  $\bar{x}$  in  $\mathbb{R}^3$  is associated with position vector  $\bar{x} = x\bar{i} + y\bar{j} + z\bar{k}$ , where  $\vec{i}$ ,  $\vec{j}$ , and  $\vec{k}$  are the basis vectors of the borehole Cartesian coordinate system  $(X_b, Y_b, Z_b)$ . Displacements of points within  $\Omega$  are described in terms of a vectorvalued displacement function of the position  $\bar{x}$  and time t as  $\vec{u}(\bar{x}, t) = u(\bar{x}, t)\bar{i} + v(\bar{x}, t)\bar{j} + w(\bar{x}, t)\bar{k}$ . The boundary  $\Gamma$  of  $\Omega$  is  $\Gamma = \Gamma_B \cup \Gamma_F \cup \tilde{\Gamma}_T \cup \tilde{\Gamma}_B$ , where

$$\Gamma_B = \left\{ (x, y, z) \in \mathbb{R}^3 : \sqrt{x^2 + y^2} = R_B, \ z_T \le z \le z_B \right\}$$
$$\Gamma_F = \left\{ (x, y, z) \in \mathbb{R}^3 : \sqrt{x^2 + y^2} = R_F, \ z_T \le z \le z_B \right\}$$
$$\tilde{\Gamma}_T = \left\{ (x, y, z_T) \in \mathbb{R}^3 : R_B \le \sqrt{x^2 + y^2} \le R_F \right\}$$
$$\tilde{\Gamma}_B = \left\{ (x, y, z_B) \in \mathbb{R}^3 : R_B \le \sqrt{x^2 + y^2} \le R_F \right\}$$

Figure 2.2 shows the two-dimensional region  $\Omega$  in  $\mathbb{R}^3$  exterior to the borehole.



Figure 2.2: Three-dimensional region exterior to the borehole

Now we formulate the initial-boundary value problem that constitutes the threedimensional radially nonsymmetric TCPu model. The thermal diffusion, solute diffusion, and fluid diffusion equations, as well as the Navier-type elastic equations for the three-dimensional problem are the same as in the two-dimensional case and thus, the problem is the following. Thermal diffusion:

$$\Lambda \dot{T} + \Sigma \dot{C} + \Phi \dot{p} - \frac{k^T}{T_F} \nabla^2 T - \frac{\rho_f R T_F D^T}{M^s (1 - C_F)} \nabla^2 C + K^T \nabla^2 p = -\zeta (\nabla \cdot \dot{\vec{u}}) \qquad \text{in } \Omega \times (0, t_f)$$

$$(2.2.1)$$

with boundary conditions

$$T(\bar{x},t) = T_B(z,t) \qquad \text{on } \Gamma_B \times [0,t_f) \qquad (2.2.2a)$$

$$T(\bar{x},t) = T_F(\bar{x})$$
 on  $\Gamma_F \times [0,t_f)$  (2.2.2b)

$$T(\bar{x},t) = \tilde{T}_T(x,y,t) \qquad \text{on } \tilde{\Gamma}_T \times [0,t_f)$$
(2.2.2c)

$$T(\bar{x},t) = \tilde{T}_B(x,y,t)$$
 on  $\tilde{\Gamma}_B \times [0,t_f)$  (2.2.2d)

compatibility conditions

$$\tilde{T}_T(t)\big|_{\Gamma_B} = T_B(t)\big|_{\tilde{\Gamma}_T}$$
(2.2.3a)

$$\tilde{T}_T(t)\big|_{\Gamma_F} = T_F\big|_{\tilde{\Gamma}_T} \tag{2.2.3b}$$

$$\tilde{T}_B(t)\big|_{\Gamma_B} = T_B(t)\big|_{\tilde{\Gamma}_B}$$
(2.2.3c)

$$\tilde{T}_B(t)\big|_{\Gamma_F} = T_F\big|_{\tilde{\Gamma}_T} \tag{2.2.3d}$$

and initial conditions

$$T(\bar{x},0) = \tilde{T}_I(\bar{x}) \qquad \text{in } \Omega \tag{2.2.4}$$

Solute diffusion:

$$\phi \dot{C} - C_F D^T \nabla^2 T - D \nabla^2 C + \frac{k \Re}{\eta} \nabla^2 p = 0 \qquad \text{in } \Omega \times (0, t_f)$$
(2.2.5)

with boundary conditions

$$C(\bar{x},t) = C_B(z,t) \qquad \text{on } \Gamma_B \times [0,t_f) \qquad (2.2.6a)$$
$$C(\bar{x},t) = C_B(\bar{x}) \qquad \text{on } \Gamma_B \times [0,t_f) \qquad (2.2.6b)$$

$$C(\bar{x},t) = C_F(\bar{x}) \qquad \text{on } \Gamma_F \times [0,t_f) \qquad (2.2.6b)$$

$$C(\bar{x},t) = \tilde{C}_T(x,y,t) \qquad \text{on } \tilde{\Gamma}_T \times [0,t_f)$$
(2.2.6c)

$$C(\bar{x},t) = \tilde{C}_B(x,y,t) \quad \text{on } \tilde{\Gamma}_B \times [0,t_f)$$
(2.2.6d)

compatibility conditions

$$\tilde{C}_T(t)\big|_{\Gamma_B} = C_B(t)\big|_{\tilde{\Gamma}_T}$$
(2.2.7a)

$$\tilde{C}_T(t)\big|_{\Gamma_F} = C_F\big|_{\tilde{\Gamma}_T} \tag{2.2.7b}$$

$$\tilde{C}_B(t)\big|_{\Gamma_B} = C_B(t)\big|_{\tilde{\Gamma}_B}$$
(2.2.7c)

$$\tilde{C}_B(t)\big|_{\Gamma_F} = C_F\big|_{\tilde{\Gamma}_T} \tag{2.2.7d}$$

and initial conditions

$$C(\bar{x},0) = \tilde{C}_I(\bar{x}) \qquad \text{in } \Omega \tag{2.2.8}$$

Fluid diffusion:

$$\Gamma \dot{T} + \chi \dot{C} + \Psi \dot{p} + K^T \nabla^2 T + \frac{\rho_f \Re \Omega k}{\eta} \nabla^2 C - \frac{k}{\eta} \nabla^2 p = -\alpha (\nabla \cdot \dot{\vec{u}}) \quad \text{in } \Omega \times (0, t_f) \quad (2.2.9)$$

with boundary conditions

$$p(\bar{x},t) = p_B(z,t) \qquad \text{on } \Gamma_B \times [0,t_f) \qquad (2.2.10a)$$

$$p(\bar{x},t) = p_F(\bar{x})$$
 on  $\Gamma_F \times [0,t_f)$  (2.2.10b)

$$p(\bar{x},t) = \tilde{p}_T(x,y,t) \qquad \text{on } \tilde{\Gamma}_T \times [0,t_f) \tag{2.2.10c}$$

$$p(\bar{x},t) = \tilde{p}_B(x,y,t) \qquad \text{on } \tilde{\Gamma}_B \times [0,t_f) \tag{2.2.10d}$$

compatibility conditions

$$\tilde{p}_T(t)\big|_{\Gamma_B} = p_B(t)\big|_{\tilde{\Gamma}_T}$$
(2.2.11a)

$$\tilde{p}_T(t)\big|_{\Gamma_F} = p_F\big|_{\tilde{\Gamma}_T} \tag{2.2.11b}$$

$$\tilde{p}_B(t)\big|_{\Gamma_B} = p_B(t)\big|_{\tilde{\Gamma}_B} \tag{2.2.11c}$$

$$\tilde{p}_B(t)\big|_{\Gamma_F} = p_F\big|_{\tilde{\Gamma}_T} \tag{2.2.11d}$$

and initial conditions

$$p(\bar{x},0) = \tilde{p}_I(\bar{x}) \qquad \text{in } \Omega \tag{2.2.12}$$

Navier-type elastic equations:

$$\left(K + \frac{G}{3}\right)\nabla(\nabla \cdot \dot{\vec{u}}) + G\nabla^2 \dot{\vec{u}} - \tilde{\zeta}\nabla \dot{T} + \xi\nabla \dot{C} - \tilde{\alpha}\nabla \dot{p} = 0 \quad \text{in } \Omega \times (0, t_f) \quad (2.2.13)$$

with boundary conditions

$$\vec{u}(\bar{x},t) \approx 0$$
 on  $\Gamma_F \times (0,t_f)$  (2.2.14)

$$\dot{\vec{u}}(\bar{x},t) \approx 0$$
 on  $\Gamma_F \times (0,t_f)$  (2.2.15)

and initial conditions

$$\vec{u}(\bar{x},0) = 0$$
 in  $\bar{\Omega} = \Omega \cup \Gamma$  (2.2.16)

The description and values of physical constants are presented in Appendix A.

Using the matrices and vectors (2.1.17) and (2.1.18) defined for the two-dimensional case, we have the following TCP initial-boundary value problem:

$$M\dot{\bar{V}} - A\nabla^2 \bar{V} = -\vec{b}_0 (\nabla \cdot \dot{\vec{u}}) \qquad \text{in } \Omega \times (0, t_f)$$
(2.2.17)

with boundary conditions

$$\bar{V}(\bar{x},t) = \begin{cases} V_B(z,t) & \text{on } \Gamma_B \times [0,t_f) \\ V_F(\bar{x}) & \text{on } \Gamma_F \times [0,t_f) \\ \tilde{V}_T(x,y,t) & \text{on } \tilde{\Gamma}_T \times [0,t_f) \\ \tilde{V}_B(x,y,t) & \text{on } \tilde{\Gamma}_B \times [0,t_f) \end{cases}$$
(2.2.18)

compatibility conditions

$$\tilde{V}_T(t)\big|_{\Gamma_B} = V_B(t)\big|_{\tilde{\Gamma}_T}$$
(2.2.19a)

$$\tilde{V}_T(t)\big|_{\Gamma_F} = V_F\big|_{\tilde{\Gamma}_T} \tag{2.2.19b}$$

$$\tilde{V}_B(t)\big|_{\Gamma_B} = V_B(t)\big|_{\tilde{\Gamma}_B}$$
(2.2.19c)

$$\tilde{V}_B(t)\big|_{\Gamma_F} = V_F\big|_{\tilde{\Gamma}_T} \tag{2.2.19d}$$

and initial conditions

$$\bar{V}(\bar{x},0) = \tilde{V}_I(\bar{x}) \qquad \text{in } \Omega \tag{2.2.20}$$

As before,  $\vec{b}_0$  is the factor that couples the displacement velocity to the absolute temperature, the pore pressure, the solute mass fraction (TCP), and their velocities.

Next, using the principle of minimum total potential energy, we will show that the three-dimensional Navier-type elastic system (2.2.13)-(2.2.15) is equivalent to the system

$$\left(K + \frac{G}{3}\right)\nabla(\nabla \cdot \vec{u}) + G\nabla^2 \vec{u} = \nabla f \qquad \text{in } \Omega \times (0, t_f)$$
(2.2.21)

$$\tau \vec{n} = (fI + \hat{\sigma})\vec{n}$$
 on  $\Gamma \setminus \Gamma_F \times (0, t_f)$  (2.2.22)

$$\vec{u} \approx 0$$
 on  $\Gamma_F \times (0, t_f)$  (2.2.14)

where f is given by (2.1.30).

Here and in the following we will suppress the time dependence of the displacement vector for the sake of brevity. Given the displacement

$$\vec{u} = u\vec{i} + v\vec{j} + w\vec{k}$$

the linearized strain is a second order symmetric tensor

$$\varepsilon = \begin{bmatrix} \varepsilon_{11} & \varepsilon_{12} & \varepsilon_{13} \\ \varepsilon_{21} & \varepsilon_{22} & \varepsilon_{23} \\ \varepsilon_{31} & \varepsilon_{32} & \varepsilon_{33} \end{bmatrix} = \frac{1}{2} \left( \nabla \vec{u} + \nabla \vec{u}^T \right)$$

where

$$\varepsilon_{11} = u_x, \qquad \varepsilon_{12} = \varepsilon_{21} = \frac{1}{2} (u_y + v_x), \qquad \varepsilon_{22} = v_y$$
  
$$\varepsilon_{13} = \varepsilon_{31} = \frac{1}{2} (u_z + w_x), \qquad \varepsilon_{23} = \varepsilon_{32} = \frac{1}{2} (v_z + w_y), \qquad \varepsilon_{33} = w_z$$

For a homogeneous elastic isotropic medium, the stress-strain relation (2.1.23) in terms of the bulk modulus K and the shear modulus G has the form

$$\tau = \begin{bmatrix} \tau_{11} & \tau_{12} & \tau_{13} \\ \tau_{21} & \tau_{22} & \tau_{23} \\ \tau_{13} & \tau_{23} & \tau_{33} \end{bmatrix} = 2G\varepsilon + \left(K - \frac{2}{3}G\right)(\operatorname{tr}\varepsilon)I \qquad (2.2.23)$$

That is,

$$\tau_{11} = 2Gu_x + \left(K - \frac{2}{3}G\right)\left(u_x + v_y + w_z\right)$$
  

$$\tau_{12} = \tau_{21} = G\left(u_y + v_x\right)$$
  

$$\tau_{22} = 2Gv_y + \left(K - \frac{2}{3}G\right)\left(u_x + v_y + w_z\right)$$
  

$$\tau_{13} = \tau_{31} = G\left(u_z + w_x\right)$$
  

$$\tau_{23} = \tau_{32} = G\left(v_z + w_y\right)$$
  

$$\tau_{33} = 2Gw_z + \left(K - \frac{2}{3}G\right)\left(u_x + v_y + w_z\right)$$
(2.2.24)

The total potential energy of our system is

$$\mathcal{V}(\vec{u}) = \mathcal{V}_S(\vec{u}) - W_b(\vec{u}) - W_S(\vec{u}) \tag{2.2.25}$$

where

$$\mathcal{V}_{S}(\vec{u}) = \frac{1}{2} \int_{\Omega} \tau_{11}(\vec{u}) \varepsilon_{11}(\vec{u}) + 2\tau_{12}(\vec{u}) \varepsilon_{12}(\vec{u}) + \tau_{22}(\vec{u}) \varepsilon_{22}(\vec{u}) + 2\tau_{13}(\vec{u}) \varepsilon_{13}(\vec{u}) + 2\tau_{23}(\vec{u}) \varepsilon_{23}(\vec{u}) + \tau_{33}(\vec{u}) \varepsilon_{33}(\vec{u}) \,\mathrm{d}\Omega \qquad (2.2.26)$$

is the elastic strain energy of the system;

$$W_b(\vec{u}) = \int_{\Omega} f(\operatorname{tr}\varepsilon(\vec{u})) \,\mathrm{d}\Omega \qquad (2.2.27)$$

with f given by (2.1.30), is work done by body forces due to the absolute temperature, the solute mass fraction, and the pore pressure ( $\vec{b}_1$  is the factor that couples the absolute temperature, the pore pressure, and the solute mass fraction (TCP) to the displacement); and

$$W_S(\vec{u}) = \int_{\Gamma} (\hat{\sigma}\vec{n}) \cdot \vec{u} \,\mathrm{d}\Gamma \qquad (2.2.28)$$

is the work done by applied boundary stress  $\hat{\sigma}$ . Here  $\vec{n}$  is the outward unit normal vector on the boundary.

Applying (2.2.26), (2.2.27), and (2.2.28) to (2.2.25), we have

$$\mathcal{V}(\vec{u}) = \frac{1}{2} \int_{\Omega} \tau_{11}(\vec{u}) \varepsilon_{11}(\vec{u}) + 2\tau_{12}(\vec{u}) \varepsilon_{12}(\vec{u}) + \tau_{22}(\vec{u}) \varepsilon_{22}(\vec{u}) + 2\tau_{13}(\vec{u}) \varepsilon_{13}(\vec{u}) + 2\tau_{23}(\vec{u}) \varepsilon_{23}(\vec{u}) + \tau_{33}(\vec{u}) \varepsilon_{33}(\vec{u}) \,\mathrm{d}\Omega - \int_{\Omega} f\left(\mathrm{tr}\varepsilon(\vec{u})\right) \,\mathrm{d}\Omega - \int_{\Gamma} (\hat{\sigma}\vec{n}) \cdot \vec{u} \,\mathrm{d}\Gamma$$

$$(2.2.29)$$

To aid in giving a mathematical formulation, let

$$H = L^{2}(\Omega; \mathbb{R}^{3}) = \left\{ L^{2}(\Omega; \mathbb{R}) \right\}^{3}$$
$$V = H^{1}(\Omega; \mathbb{R}^{3}) = \left\{ H^{1}(\Omega; \mathbb{R}) \right\}^{3}$$
$$V_{0} = \left\{ \varphi \in V : \varphi \right|_{\Gamma_{F}} = 0 \right\}$$
$$\vec{\Phi}(\bar{x}) = \phi(\bar{x})\vec{i} + \psi(\bar{x})\vec{j} + \omega(\bar{x})\vec{k}$$

Define the bilinear form on V by

$$a_{E}(\vec{u},\vec{\Phi}) = \int_{\Omega} \tau_{11}(\vec{u})\varepsilon_{11}(\vec{\Phi}) + 2\tau_{12}(\vec{u})\varepsilon_{12}(\vec{\Phi}) + \tau_{22}(\vec{u})\varepsilon_{22}(\vec{\Phi}) + 2\tau_{13}(\vec{u})\varepsilon_{13}(\vec{\Phi}) + 2\tau_{23}(\vec{u})\varepsilon_{23}(\vec{\Phi}) + \tau_{33}(\vec{u})\varepsilon_{33}(\vec{\Phi}) \,\mathrm{d}\Omega, \quad \forall \, \vec{u}, \vec{\Phi} \in V$$
(2.2.30)

Then from (2.2.26) and (2.2.30)

$$\mathcal{V}_{S}(\vec{u}) = \frac{1}{2} a_{E}(\vec{u}, \vec{u})$$
(2.2.31)

and, from (2.2.29) and (2.2.31), the total potential energy has the form

$$\mathcal{V}(\vec{u}) = \frac{1}{2} a_E(\vec{u}, \vec{u}) - \int_{\Omega} f\left(\mathrm{tr}\varepsilon(\vec{u})\right) \mathrm{d}\Omega - \int_{\Gamma} (\hat{\sigma}\vec{n}) \cdot \vec{u} \,\mathrm{d}\Gamma$$
(2.2.32)

Define the following vectors:

$$\vec{\tau}_1 = \begin{bmatrix} \tau_{11} & \tau_{12} & \tau_{13} \end{bmatrix}^T, \quad \vec{\tau}_2 = \begin{bmatrix} \tau_{21} & \tau_{22} & \tau_{23} \end{bmatrix}^T, \quad \vec{\tau}_3 = \begin{bmatrix} \tau_{31} & \tau_{32} & \tau_{33} \end{bmatrix}^T$$

Then

$$\begin{aligned} a_E(\vec{u}, \vec{\Phi}) &= \int_{\Omega} \tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z + \tau_{21} \psi_x + \tau_{22} \psi_y + \tau_{23} \psi_z + \\ &+ \tau_{31} \omega_x + \tau_{32} \omega_y + \tau_{33} \omega_z \, \mathrm{d}\Omega \\ &= \int_{\Omega} \vec{\tau}_1 \cdot \nabla \phi + \vec{\tau}_2 \cdot \nabla \psi + \vec{\tau}_3 \cdot \nabla \omega \, \mathrm{d}\Omega \\ &= \int_{\Omega} \nabla \cdot (\phi \vec{\tau}_1) - \phi \nabla \cdot \vec{\tau}_1 + \nabla \cdot (\psi \vec{\tau}_2) - \psi \nabla \cdot \vec{\tau}_2 + \nabla \cdot (\omega \vec{\tau}_3) - \omega \nabla \cdot \vec{\tau}_3 \, \mathrm{d}\Omega \\ &= \int_{\Gamma} \phi(\vec{\tau}_1 \cdot \vec{n}) + \psi(\vec{\tau}_2 \cdot \vec{n}) + \omega(\vec{\tau}_3 \cdot \vec{n}) \, \mathrm{d}\Gamma - \\ &- \int_{\Omega} \phi \nabla \cdot \vec{\tau}_1 + \psi \nabla \cdot \vec{\tau}_2 + \omega \nabla \cdot \vec{\tau}_3 \, \mathrm{d}\Omega \\ &= \int_{\Gamma} (\tau \vec{n}) \cdot \vec{\Phi} \, \mathrm{d}\Gamma - \int_{\Omega} [\nabla \cdot \vec{\tau}_1 \quad \nabla \cdot \vec{\tau}_2 \quad \nabla \cdot \vec{\tau}_3]^T \cdot \vec{\Phi} \, \mathrm{d}\Omega, \qquad \forall \ \vec{u}, \ \vec{\Phi} \in V \\ \end{aligned}$$

$$(2.2.33)$$

Thus, we obtain the following Green's formula:

$$a_E(\vec{u},\vec{\Phi}) = \int_{\Gamma} (\tau \vec{n}) \cdot \vec{\Phi} \,\mathrm{d}\Gamma - \int_{\Omega} [\nabla \cdot \vec{\tau}_1 \quad \nabla \cdot \vec{\tau}_2 \quad \nabla \cdot \vec{\tau}_3]^T \cdot \vec{\Phi} \,\mathrm{d}\Omega, \qquad \forall \, \vec{u}, \, \vec{\Phi} \in V \ (2.2.34)$$

Referring to (2.2.14) and (2.2.15), we assume that  $\vec{u}$  and  $\dot{\vec{u}}$  are negligible on  $\Gamma_F$  and therefore, the set of admissible displacements is

$$\mathcal{U}_{ad} = \left\{ \vec{u} \in V_0 : \dot{\vec{u}} \in V_0 \right\}$$

Using the principle of minimum total potential energy, the displacement  $\vec{u}$  that the region  $\Omega$  undergoes is given by

$$D\mathcal{V}(\vec{u})\vec{\Phi} = 0, \qquad \forall \ \vec{\Phi} \in V_0 \tag{2.2.35}$$

where

$$D\mathcal{V}(\vec{u})\vec{\Phi} = \frac{d}{d\delta}\mathcal{V}(\vec{u} + \delta\vec{\Phi})\Big|_{\delta=0}$$

is the Gâteaux differential of  $\mathcal{V}$  with increment  $\vec{\Phi}$ .

From (2.2.32), (2.2.35)

$$a_E(\vec{u},\vec{\Phi}) - \int_{\Omega} f\left(\mathrm{tr}\varepsilon(\vec{\Phi})\right) \mathrm{d}\Omega - \int_{\Gamma \setminus \Gamma_F} (\hat{\sigma}\vec{n}) \cdot \vec{\Phi} \,\mathrm{d}\Gamma = 0, \qquad \forall \ \vec{\Phi} \in V_0 \qquad (2.2.36)$$

From Green's formula (2.2.34) we have

$$a_E(\vec{u},\vec{\Phi}) = \int_{\Gamma \setminus \Gamma_F} (\tau \vec{n}) \cdot \vec{\Phi} \,\mathrm{d}\Gamma - \int_{\Omega} [\nabla \cdot \vec{\tau}_1 \quad \nabla \cdot \vec{\tau}_2 \quad \nabla \cdot \vec{\tau}_3]^T \cdot \vec{\Phi} \,\mathrm{d}\Omega, \quad \forall \vec{\Phi} \in V_0 \ (2.2.37)$$

Now

$$\begin{split} \int_{\Omega} f\left(\mathrm{tr}\varepsilon(\vec{\Phi})\right) \mathrm{d}\Omega &= \int_{\Omega} f(\nabla \cdot \vec{\Phi}) \mathrm{d}\Omega \\ &= \int_{\Omega} \nabla \cdot (f\vec{\Phi}) - \nabla f \cdot \vec{\Phi} \mathrm{d}\Omega \\ &= \int_{\Gamma \setminus \Gamma_F} f\vec{\Phi} \cdot \vec{n} \, \mathrm{d}\Gamma - \int_{\Omega} \nabla f \cdot \vec{\Phi} \, \mathrm{d}\Omega \\ &= \int_{\Gamma \setminus \Gamma_F} (fI\vec{n}) \cdot \vec{\Phi} \, \mathrm{d}\Gamma - \int_{\Omega} \nabla f \cdot \vec{\Phi} \, \mathrm{d}\Omega, \qquad \forall \ \vec{\Phi} \in V_0 \qquad (2.2.38) \end{split}$$

Applying (2.2.37) and (2.2.38) to (2.2.36), we have

$$\int_{\Gamma \setminus \Gamma_F} (\tau \vec{n}) \cdot \vec{\Phi} \, \mathrm{d}\Gamma - \int_{\Omega} [\nabla \cdot \vec{\tau}_1 \quad \nabla \cdot \vec{\tau}_2 \quad \nabla \cdot \vec{\tau}_3]^T \cdot \vec{\Phi} \, \mathrm{d}\Omega$$
$$- \int_{\Gamma \setminus \Gamma_F} (fI\vec{n}) \cdot \vec{\Phi} \, \mathrm{d}\Gamma + \int_{\Omega} \nabla f \cdot \vec{\Phi} \, \mathrm{d}\Omega - \int_{\Gamma \setminus \Gamma_F} (\hat{\sigma}\vec{n}) \cdot \vec{\Phi} \, \mathrm{d}\Gamma = 0$$

and so,

$$\int_{\Gamma \setminus \Gamma_F} (\tau - fI - \hat{\sigma}) \vec{n} \cdot \vec{\Phi} \, \mathrm{d}\Gamma + \int_{\Omega} \left( - \begin{bmatrix} \nabla \cdot \vec{\tau}_1 & \nabla \cdot \vec{\tau}_2 & \nabla \cdot \vec{\tau}_3 \end{bmatrix}^T + \nabla f \right) \cdot \vec{\Phi} \, \mathrm{d}\Omega = 0, \quad \forall \ \vec{\Phi} \in V_0$$

It follows that

$$\begin{bmatrix} \nabla \cdot \vec{\tau}_1 & \nabla \cdot \vec{\tau}_2 & \nabla \cdot \vec{\tau}_3 \end{bmatrix}^T = \nabla f \quad \text{in } \Omega \tag{2.2.39}$$
$$\tau \vec{n} = (fI + \hat{\sigma})\vec{n} \quad \text{on } \Gamma \setminus \Gamma_F \tag{2.2.40}$$

From (2.2.24),

$$\nabla \cdot \vec{\tau_1} = \tau_{11_x} + \tau_{12_y} + \tau_{13_z}$$

$$= \left(K + \frac{G}{3}\right)u_{xx} + \left(K - \frac{2G}{3}\right)(v_{yx} + w_{zx}) + Gu_{xx} + Gu_{yy} + Gv_{xy} + Gu_{zz} + Gw_{xz}$$

$$= \left(K + \frac{G}{3}\right)u_{xx} + \left(K + \frac{G}{3}\right)(v_{yx} + w_{zx}) + G(u_{xx} + u_{yy} + u_{zz})$$

$$= \left(K + \frac{G}{3}\right)(\nabla \cdot \vec{u})_x + G\nabla^2 u$$

$$\begin{aligned} \nabla \cdot \vec{\tau_2} &= \tau_{21x} + \tau_{22y} + \tau_{23z} \\ &= G(u_{yx} + v_{xx}) + \left(K + \frac{G}{3}\right)v_{yy} + \left(K - \frac{2G}{3}\right)(u_{xy} + w_{zy}) + \\ &+ Gv_{yy} + G(v_{zz} + w_{yz}) \\ &= \left(K + \frac{G}{3}\right)(u_{xy} + v_{yy} + w_{zy}) + G(v_{xx} + v_{yy} + v_{zz}) \\ &= \left(K + \frac{G}{3}\right)(\nabla \cdot \vec{u})_y + G\nabla^2 v \end{aligned}$$

$$\nabla \cdot \vec{\tau}_{3} = \tau_{31_{x}} + \tau_{32_{y}} + \tau_{33_{z}}$$

$$= G(u_{zx} + w_{xx}) + G(v_{zy} + w_{yy}) + \left(K + \frac{G}{3}\right)w_{zz} + \left(K - \frac{2G}{3}\right)(u_{xz} + v_{yz}) + Gw_{zz}$$

$$= \left(K + \frac{G}{3}\right)(u_{xz} + v_{yz} + w_{zz}) + G(w_{xx} + w_{yy} + w_{zz})$$

$$= \left(K + \frac{G}{3}\right)(\nabla \cdot \vec{u})_{z} + G\nabla^{2}w$$

It follows that

$$\begin{bmatrix} \nabla \cdot \vec{\tau}_1 \\ \nabla \cdot \vec{\tau}_2 \\ \nabla \cdot \vec{\tau}_3 \end{bmatrix} = \begin{pmatrix} K + \frac{G}{3} \end{pmatrix} \begin{bmatrix} (\nabla \cdot \vec{u})_x \\ (\nabla \cdot \vec{u})_y \\ (\nabla \cdot \vec{u})_z \end{bmatrix} + G \begin{bmatrix} \nabla^2 u \\ \nabla^2 v \\ \nabla^2 w \end{bmatrix} \Rightarrow$$
$$[\nabla \cdot \vec{\tau}_1 \quad \nabla \cdot \vec{\tau}_2 \quad \nabla \cdot \vec{\tau}_3]^T = \begin{pmatrix} K + \frac{G}{3} \end{pmatrix} \nabla (\nabla \cdot \vec{u}) + G \nabla^2 \vec{u} \qquad (2.2.41)$$

From (2.2.39)-(2.2.41) and (2.2.14)

$$\left(K + \frac{G}{3}\right)\nabla(\nabla \cdot \vec{u}) + G\nabla^2 \vec{u} = \nabla f \quad \text{in } \Omega \times (0, t_f)$$
 (2.2.42)

$$\tau \vec{n} = (fI + \hat{\sigma})\vec{n}$$
 on  $\Gamma \setminus \Gamma_F \times (0, t_f)$  (2.2.40)

$$\vec{u} \approx 0$$
 on  $\Gamma_F \times (0, t_f)$  (2.2.14)

Differentiating (2.2.42) with respect to time and applying (2.1.30), we obtain the system (2.2.13). On the other hand, since  $\dot{\vec{u}} \in V_0$ , from (2.2.36) we have

$$a_E(\dot{\vec{u}},\vec{\Phi}) - \int_{\Omega} f(\operatorname{tr}\varepsilon(\vec{\Phi})) \,\mathrm{d}\Omega - \int_{\Gamma \setminus \Gamma_F} (\hat{\sigma}\vec{n}) \cdot \vec{\Phi} \,\mathrm{d}\Gamma = 0, \qquad \forall \, \vec{\Phi} \in V_0 \qquad (2.2.43)$$

Using the same argument as above and applying (2.1.30), we obtain the equivalence of systems (2.2.42), (2.2.40), (2.2.14) and (2.2.13)-(2.2.15).

*Remark* 2.2. From the condition of mechanical equilibrium given in Remark 2.1, (2.2.24), and (2.1.30), it follows that the system (2.2.42) constitutes the equation of equilibrium.

#### 2.3 Diagonalized TCP System with Homogeneous Boundary Conditions

In the next chapter, we will derive the weak formulation of the TCP initial-boundary value problem. To aid in the formulation of the weak TCP problem, in this section we obtain the diagonalized two- and three-dimensional TCP systems with homogeneous boundary conditions. First, we obtain the two-dimensional TCP system with homogeneous boundary conditions that is equivalent to (2.1.19)-(2.1.21). To that end, we define the function

$$W(\bar{x},t) = V_B(t) + \frac{|\bar{x}| - R_B}{R_F - R_B} (V_F(\bar{x}) - V_B(t))$$
(2.3.1)

and the vector

$$V(\bar{x},t) = \bar{V}(\bar{x},t) - W(\bar{x},t)$$
(2.3.2)

Then (2.1.19)-(2.1.21) has the form

$$M\dot{V} - A\nabla^2 V = -M\dot{W} + A\nabla^2 W - \vec{b}_0(\nabla \cdot \dot{\vec{u}}) \qquad \text{in } \Omega \times (0, t_f)$$

with boundary conditions

$$V(\bar{x},t) = 0$$
 on  $\Gamma \times [0,t_f)$ 

and initial conditions

$$V(\bar{x},0) = V_0(\bar{x}) \qquad \text{in } \Omega$$

where

$$V_0(\bar{x}) = \tilde{V}_I(\bar{x}) - \frac{R_F - |\bar{x}|}{R_F - R_B} V_B(0) - \frac{|\bar{x}| - R_B}{R_F - R_B} V_F(\bar{x})$$
(2.3.3)

Now we obtain the three-dimensional TCP system with homogeneous boundary conditions that is equivalent to (2.2.17)-(2.2.20). Define the function

$$W^{0}(\bar{x},t) = \frac{R_{F} - \sqrt{x^{2} + y^{2}}}{R_{F} - R_{B}} V_{B}(z,t) + \frac{\sqrt{x^{2} + y^{2}} - R_{B}}{R_{F} - R_{B}} V_{F}(\bar{x})$$
(2.3.4)

and the vector

$$\bar{V}^{0}(\bar{x},t) = \bar{V}(\bar{x},t) - W^{0}(\bar{x},t)$$
(2.3.5)

Then (2.2.17)-(2.2.20) has the form

$$M\dot{\bar{V}}^0 - A\nabla^2 \bar{V}^0 = -M\dot{W}^0 + A\nabla^2 W^0 - \vec{b}_0(\nabla \cdot \dot{\vec{u}}) \qquad \text{in } \Omega \times (0, t_f)$$

with boundary conditions

$$\bar{V}^{0}(\bar{x},t) = \begin{cases} 0 & \text{on } (\Gamma_{B} \cup \Gamma_{F}) \times [0,t_{f}) \\ \tilde{V}_{T}(x,y,t) - W^{0}(x,y,z_{T},t) & \text{on } \tilde{\Gamma}_{T} \times [0,t_{f}) \\ \tilde{V}_{B}(x,y,t) - W^{0}(x,y,z_{B},t) & \text{on } \tilde{\Gamma}_{B} \times [0,t_{f}) \end{cases}$$

and initial conditions

$$\bar{V}^0(\bar{x},0) = \tilde{V}_I(\bar{x}) - W^0(\bar{x},0)$$
 in  $\Omega$ 

Next define

$$\hat{V}_T(x, y, t) = \tilde{V}_T(x, y, t) - W^0(x, y, z_T, t)$$
(2.3.6)

$$\hat{V}_B(x, y, t) = \tilde{V}_B(x, y, t) - W^0(x, y, z_B, t)$$
(2.3.7)

and

$$W^{z}(\bar{x},t) = \frac{z - z_{T}}{z_{B} - z_{T}} \hat{V}_{B}(x,y,t) + \frac{z_{B} - z}{z_{B} - z_{T}} \hat{V}_{T}(x,y,t)$$
(2.3.8)

$$V = \bar{V}^0 - W^z = \bar{V} - W^0 - W^z$$
, so  
 $V = \bar{V} - W$ , where  $W = W^0 + W^z$  (2.3.9)

Then (2.2.17)-(2.2.20) has the form

$$M\dot{V} - A\nabla^2 V = -M\dot{W} + A\nabla^2 W - \vec{b}_0(\nabla \cdot \dot{\vec{u}})$$
 in  $\Omega \times (0, t_f)$ 

with boundary conditions

$$V(\bar{x},t) = 0$$
 on  $\Gamma \times [0,t_f)$ 

and initial conditions

$$V(\bar{x},0) = V_0(\bar{x}) \qquad \text{in } \Omega$$
where

$$V_0(\bar{x}) = \tilde{V}_I(\bar{x}) - W(\bar{x}, 0) \tag{2.3.10}$$

Note that both the two-dimensional TCP system (2.1.19)-(2.1.21) and the threedimensional TCP system (2.2.17)-(2.2.20) are equivalent to the following parabolic system with homogeneous boundary conditions:

$$M\dot{V} - A\nabla^2 V = -M\dot{W} + A\nabla^2 W - \vec{b}_0(\nabla \cdot \dot{\vec{u}}) \qquad \text{in } \Omega \times (0, t_f)$$
(2.3.11)

with boundary conditions

$$V(\bar{x},t) = 0 \qquad \text{on } \Gamma \times [0,t_f) \tag{2.3.12}$$

and initial conditions

$$V(\bar{x},0) = V_0(\bar{x}) \qquad \text{in } \Omega \tag{2.3.13}$$

Now we diagonalize the TCP system (2.3.11)-(2.3.13). To that end, we make the following assumption.

Assumption 2.1. The matrix M is nonsingular and the matrix

$$\mathcal{A} = M^{-1}A \tag{2.3.14}$$

has three distinct positive eigenvalues  $\lambda_1 > \lambda_2 > \lambda_3 > 0$ .

*Remark* 2.3. Under Assumption 2.1, the matrix  $\mathcal{A}$  is diagonalizable:

$$\mathcal{A} = PDP^{-1} \tag{2.3.15}$$

where

$$D = \begin{bmatrix} \lambda_1 & 0 & 0 \\ 0 & \lambda_2 & 0 \\ 0 & 0 & \lambda_3 \end{bmatrix}, \qquad P = \begin{bmatrix} \vec{e}_1 & \vec{e}_2 & \vec{e}_3 \end{bmatrix},$$

 $\vec{e_i}$  is the eigenvector of  $\mathcal{A}$  corresponding to the eigenvalue  $\lambda_i$ ,  $||\vec{e_i}|| = 1$ ,  $1 \le i \le 3$ .

Let

$$U = \begin{bmatrix} U_1 \\ U_2 \\ U_3 \end{bmatrix} = P^{-1}V \tag{2.3.16}$$

Using (2.3.14)-(2.3.16), the TCP system (2.3.11)-(2.3.13) has the form

$$\dot{U} - D\nabla^2 U = -P^{-1}\dot{W} + DP^{-1}\nabla^2 W - P^{-1}M^{-1}\vec{b}_0(\nabla \cdot \dot{\vec{u}}) \quad \text{in } \Omega \times (0, t_f) \quad (2.3.17)$$

with boundary conditions

$$U(\bar{x},t) = 0 \qquad \text{on } \Gamma \times [0,t_f) \tag{2.3.18}$$

and initial conditions

$$U(\bar{x}, 0) = P^{-1}V_0(\bar{x})$$
 in  $\Omega$  (2.3.19)

Define

$$F_U = -P^{-1}\dot{W} + DP^{-1}\nabla^2 W - P^{-1}M^{-1}\vec{b}_0(\nabla \cdot \dot{\vec{u}})$$
(2.3.20)

$$U_0(\bar{x}) = U(\bar{x}, 0) = P^{-1} V_0(\bar{x})$$
(2.3.21)

Then we have the following diagonalized TCP system:

$$\dot{U} - D\nabla^2 U = F_U \qquad \text{in } \Omega \times (0, t_f) \tag{2.3.22}$$

with homogeneous boundary conditions

$$U(\bar{x},t) = 0 \qquad \text{on } \Gamma \times [0,t_f) \tag{2.3.23}$$

and initial conditions

$$U(\bar{x},0) = U_0(\bar{x}) \qquad \text{in } \Omega \tag{2.3.24}$$

In this chapter, we presented a fully coupled parabolic-elliptic initial-boundary value problem that models fully coupled thermal, chemical, hydraulic, and mechanical

processes in homogeneous isotropic chemically active porous media saturated by a fluid. It was shown that, under appropriate assumptions on the matrices of diffusion coefficients, the parabolic initial-boundary value problem is equivalent to the diagonalized one with homogeneous boundary conditions. Using the principle of the minimum total potential energy, we derived the equivalent formulation of the Navier-type elliptic initial-boundary value problem. While the original elliptic system is developed in terms of velocity, the derived system is formulated in terms of the displacement vector and contains compatibility boundary conditions on stresses. The obtained results play a key role in well-posedness analysis presented in the next chapter.

## Chapter 3

## Well-Posedness of the Coupled TCPu Problem

We begin by defining what we mean by a well-posed problem.

**Definition 3.1** (R. Showalter [36]). A problem is called well-posed if for each set of data there exists exactly one solution and this dependence of the solution on the data is continuous.

First, we will discuss the well-posedness of the parabolic initial-boundary value problem for the TCP system and the elliptic initial-boundary value problem for the elastic system considering coupling terms as data. Then we will investigate the existence and uniqueness of a solution to the coupled TCPu problem.

## 3.1 TCP Problem

To aid in giving a mathematical formulation of the problem, we make the following assumption.

Assumption 3.1.  $V_B(t)$  and  $\dot{V}_B(t)$  are  $H^1([0, t_f]; \mathbb{R}^3)$ -functions;  $V_B(z, t)$  and  $\dot{V}_B(z, t)$ are  $H^1(\Gamma_B \times [0, t_f]; \mathbb{R}^3)$ -functions;  $V_F(\bar{x})$  is an  $H^2(\Omega; \mathbb{R}^3)$ -function;  $\tilde{V}_T(\bar{x}, t)$  and  $\tilde{V}_B(\bar{x}, t)$  are  $L^2$ -integrable functions from  $[0, t_f]$  into  $H^2(\Omega; \mathbb{R}^3)$ ; and  $\dot{\tilde{V}}_T(\bar{x}, t)$  and  $\dot{\tilde{V}}_B(\bar{x}, t)$  are  $L^2$ -integrable functions from  $[0, t_f]$  into  $L^2(\Omega; \mathbb{R}^3)$ .

We start with the weak formulation of the system (2.3.22)-(2.3.24). Suppose that U is a classical solution of the system (2.3.22)-(2.3.24), say  $U \in C^2(\bar{\Omega} \times [0, t_f])$ .

Let 
$$\varphi(\bar{x}) = \begin{bmatrix} \varphi_1 \\ \varphi_2 \\ \varphi_3 \end{bmatrix} (\bar{x}) \in C_0^{\infty}(\Omega; \mathbb{R}^3)$$
. Then from (2.3.22) we have

$$\int_{\Omega} \dot{U} \cdot \varphi \, \mathrm{d}\Omega - \int_{\Omega} (D\nabla^2 U) \cdot \varphi \, \mathrm{d}\Omega = \int_{\Omega} F_U \cdot \varphi \, \mathrm{d}\Omega \qquad \Rightarrow$$
$$\int_{\Omega} \dot{U} \cdot \varphi \, \mathrm{d}\Omega - \sum_{k=1}^{3} \int_{\Omega} \lambda_k \nabla^2 U_k \varphi_k \, \mathrm{d}\Omega = \int_{\Omega} F_U \cdot \varphi \, \mathrm{d}\Omega$$

Applying Green's first identity,

$$\int_{\Omega} \dot{U} \cdot \varphi \, \mathrm{d}\Omega + \sum_{k=1}^{3} \int_{\Omega} \lambda_k (\nabla U_k \cdot \nabla \varphi_k) \, \mathrm{d}\Omega = \int_{\Omega} F_U \cdot \varphi \, \mathrm{d}\Omega$$

Since

$$\int_{\Omega} \dot{U} \cdot \varphi \, \mathrm{d}\Omega = \frac{d}{dt} \int_{\Omega} U \cdot \varphi \, \mathrm{d}\Omega,$$

it follows that

$$\frac{d}{dt} \int_{\Omega} U \cdot \varphi \, \mathrm{d}\Omega + \sum_{k=1}^{3} \int_{\Omega} \lambda_k (\nabla U_k \cdot \nabla \varphi_k) \, \mathrm{d}\Omega = \int_{\Omega} F_U \cdot \varphi \, \mathrm{d}\Omega \tag{3.1.1}$$

This equation holds  $\forall \varphi \in C_0^{\infty}(\Omega; \mathbb{R}^3)$  and, by continuity, it holds  $\forall \varphi \in H_0^1(\Omega; \mathbb{R}^3)$ .

Define the following spaces:

$$\mathcal{H} = L^2(\Omega; \mathbb{R}^3) = \left\{ L^2(\Omega) \right\}^3$$

with the scalar product

$$(u,v)_{\mathcal{H}} = \sum_{k=1}^{3} (u_k, v_k)_{L^2(\Omega)} = \int_{\Omega} u \cdot v \, \mathrm{d}\Omega, \qquad u = \begin{bmatrix} u_1 \\ u_2 \\ u_3 \end{bmatrix}, v = \begin{bmatrix} v_1 \\ v_2 \\ v_3 \end{bmatrix} \in \mathcal{H} \qquad (3.1.2)$$

and the norm  $||u||_{\mathcal{H}} = (u, u)_{\mathcal{H}}^{1/2}$  and

$$\mathcal{V}_0 = H_0^1(\Omega; \mathbb{R}^3) = \left\{ H_0^1(\Omega) \right\}^3$$

with the scalar product

$$(u,v)_{\mathcal{V}_0} = \sum_{k=1}^3 (u_k, v_k)_{H_0^1(\Omega)} = \sum_{k=1}^3 \int_{\Omega} \nabla u_k \cdot \nabla v_k \, \mathrm{d}\Omega, \quad u = \begin{bmatrix} u_1 \\ u_2 \\ u_3 \end{bmatrix}, v = \begin{bmatrix} v_1 \\ v_2 \\ v_3 \end{bmatrix} \in \mathcal{V}_0$$
(3.1.3)

and the norm  $||u||_{\mathcal{V}_0} = (u, u)_{\mathcal{V}_0}^{1/2}$ .

The space  $\mathcal{H}$  is the Cartesian product of three separable Hilbert spaces  $L^2(\Omega)$ and the space  $\mathcal{V}_0$  is the Cartesian product of three separable Hilbert spaces  $H_0^1(\Omega)$ , therefore  $\mathcal{H}$  and  $\mathcal{V}_0$  are also separable Hilbert spaces. Since  $H_0^1(\Omega)$  is dense in  $L^2(\Omega)$  and the injection  $H_0^1(\Omega) \hookrightarrow L^2(\Omega)$  is continuous, it follows that  $\mathcal{V}_0$  is dense in  $\mathcal{H}$  and the injection  $\mathcal{V}_0 \hookrightarrow \mathcal{H}$  is continuous. Also,  $L^2(\Omega)$  is identified with its dual space  $L^2(\Omega)'$  and  $L^2(\Omega)'$  is dense in  $H_0^1(\Omega)' = H^{-1}(\Omega)$ . It follows that  $\mathcal{H}$  is identified with its dual space  $\mathcal{H}'$  and  $\mathcal{H}'$  is dense in  $\mathcal{V}'_0 = \{H^{-1}(\Omega)\}^3 = H^{-1}(\Omega; \mathbb{R}^3)$ . We have

$$\mathcal{V}_0 \subset \mathcal{H} = \mathcal{H}' \subset \mathcal{V}'_0 \tag{3.1.4}$$

Also, the scalar product in  $\mathcal{H}$  of  $u \in \mathcal{H}$  and  $v \in \mathcal{V}_0$  is the same as the scalar product of u and v in the duality between  $\mathcal{V}'_0$  and  $\mathcal{V}_0$ , denoted by  $\langle u, v \rangle$ :

$$\langle u, v \rangle = (u, v)_{\mathcal{H}}, \qquad u \in \mathcal{H}, v \in \mathcal{V}_0$$

$$(3.1.5)$$

Define the bilinear form  $a: \mathcal{V}_0 \times \mathcal{V}_0 \to \mathbb{R}$  by

$$a(u,v) = \sum_{k=1}^{3} \lambda_k \int_{\Omega} \nabla u_k \cdot \nabla v_k \,\mathrm{d}\Omega \tag{3.1.6}$$

 $a(\cdot, \cdot)$  is symmetric:

$$a(u, v) = a(v, u)$$
 (3.1.7)

$$a(u,u) = \sum_{k=1}^{3} \int_{\Omega} \lambda_{k} \nabla u_{k} \cdot \nabla u_{k} \, \mathrm{d}\Omega = \sum_{k=1}^{3} \int_{\Omega} \lambda_{k} |\nabla u_{k}|^{2} \, \mathrm{d}\Omega \ge$$
$$\geq \left\{ \min_{k} \lambda_{k} \right\} \sum_{k=1}^{3} \int_{\Omega} |\nabla u_{k}|^{2} \, \mathrm{d}\Omega = \lambda_{3} ||u||_{\mathcal{V}_{0}}^{2}$$

Thus,

$$a(u,u) \ge \lambda_3 ||u||_{\mathcal{V}_0}^2, \qquad \lambda_3 = \min_k \lambda_k > 0 \tag{3.1.8}$$

Also,

$$|a(u,v)| = \left| \sum_{k=1}^{3} \int_{\Omega} \lambda_{k} \nabla u_{k} \cdot \nabla v_{k} \, \mathrm{d}\Omega \right| \leq \left\{ \max_{k} \lambda_{k} \right\} \sum_{k=1}^{3} |(u_{k}, v_{k})_{H_{0}^{1}(\Omega)}|$$
  
$$\leq \lambda_{1} \sum_{k=1}^{3} ||u_{k}||_{H_{0}^{1}(\Omega)} ||v_{k}||_{H_{0}^{1}(\Omega)} \leq \lambda_{1} \sum_{i=1}^{3} ||u_{i}||_{H_{0}^{1}(\Omega)} \sum_{j=1}^{3} ||v_{j}||_{H_{0}^{1}(\Omega)}$$
  
$$\leq 3\lambda_{1} ||u||_{V_{0}} ||v||_{V_{0}}, \qquad \text{so}$$
  
$$|a(u, v)| \leq 3\lambda_{1} ||u||_{V_{0}} ||v||_{V_{0}} \qquad (3.1.9)$$

From (3.1.7)-(3.1.9) it follows that  $a(\cdot, \cdot)$  defines a scalar product on  $\mathcal{V}_0$ . We denote this scalar product by

$$(u, v)_a = a(u, v), \qquad u, v \in \mathcal{V}_0$$
 (3.1.10)

and the corresponding (energy) norm by  $||u||_a = (u, u)_a^{1/2}$ . Furthermore, the norm  $|| \cdot ||_{\nu_0}$ .

For each  $u \in \mathcal{V}_0$ , the form  $u \in \mathcal{V}_0 \to (u, v)_a \in \mathbb{R}$  is linear and continuous on  $\mathcal{V}_0$ . Therefore, there exists an element  $Au \in \mathcal{V}'_0$  such that

$$(u, v)_a = \langle Au, v \rangle, \qquad \forall \ v \in \mathcal{V}_0 \tag{3.1.11}$$

From (3.1.9)-(3.1.11), the mapping  $u \mapsto Au$  is linear and continuous, that is,  $A \in \mathcal{L}(\mathcal{V}_0, \mathcal{V}'_0)$ . Also, by the Riesz representation theorem, for every  $Au \in \mathcal{V}'_0$ , there exists a unique  $u \in \mathcal{V}_0$  such that (3.1.11) holds. Thus,  $A : \mathcal{V}_0 \to \mathcal{V}'_0$  is an isomorphism from  $\mathcal{V}_0$  onto  $\mathcal{V}'_0$ .

Let X be a Hilbert space. We introduce the space  $L^2(0, t_f; X)$  of  $L^2$ -integrable functions from  $[0, t_f]$  into X with the norm

$$||u||_{L^2(0,t_f;\mathbf{X})} = \left[\int_0^{t_f} ||u(t)||_{\mathbf{X}}^2 \,\mathrm{d}t\right]^{1/2}$$

Note that if  $u \in L^2(0, t_f; \mathcal{V}_0)$ , then  $t \mapsto Au(t) \in \mathcal{V}'_0$ , that is,

$$A \in \mathcal{L}\left(L^2(0, t_f; \mathcal{V}_0), L^2(0, t_f; \mathcal{V}_0')\right)$$

From Assumption 3.1, (3.1.1), (3.1.2), (3.1.5), (3.1.6), and (3.1.10), we obtain the following weak formulation of the problem (2.3.22)-(2.3.24):

Given  $F_U \in L^2(0, t_f; \mathcal{V}'_0)$  and  $U_0 \in \mathcal{H}$ , find  $U \in L^2(0, t_f; \mathcal{V}_0)$  such that

$$\frac{d}{dt}(U,\varphi)_{\mathcal{H}} + (U,\varphi)_a = \langle F_U,\varphi \rangle, \qquad \forall \varphi \in \mathcal{V}_0$$
(3.1.12)

and

$$U(0) = U_0 \qquad \text{in } \mathcal{H} \tag{3.1.13}$$

The meaning of (3.1.13) will be explained below.

By (3.1.5) and (3.1.10), the equation (3.1.12) can be written as

$$\frac{d}{dt} < U, \varphi > + < AU, \varphi > = < F_U, \varphi >, \qquad \forall \varphi \in \mathcal{V}_0$$

and so

$$\frac{d}{dt} < U, \varphi > = < F_U - AU, \varphi >, \qquad \forall \varphi \in \mathcal{V}_0$$
(3.1.14)

Since  $F_U \in L^2(0, t_f; \mathcal{V}'_0)$  and  $AU \in L^2(0, t_f; \mathcal{V}'_0)$ , it follows that  $F_U - AU \in L^2(0, t_f; \mathcal{V}'_0)$ . From [40],

$$\frac{d}{dt} < U, \varphi > = <\dot{U}, \varphi >, \qquad \forall \varphi \in \mathcal{V}_0$$
(3.1.15)

in the scalar distribution sense on  $(0, t_f)$  and, from (3.1.14) and (3.1.15),  $\dot{U} \in L^2(0, t_f; \mathcal{V}'_0)$ .

We introduce the following Hilbert space [40, 27]:

$$\mathcal{W}(0, t_f; \mathcal{V}_0) = \left\{ u : \ u \in L^2(0, t_f; \mathcal{V}_0), \ \dot{u} \in L^2(0, t_f; \mathcal{V}'_0) \right\}$$
(3.1.16)

with the norm

$$||u||_{\mathcal{W}} = \left[\int_0^{t_f} ||u(t)||_{\mathcal{V}_0}^2 \,\mathrm{d}t + \int_0^{t_f} ||\dot{u}(t)||_{\mathcal{V}_0'}^2 \,\mathrm{d}t\right]^{1/2} \tag{3.1.17}$$

Remark 3.1 ([40]). Each  $u \in \mathcal{W}(0, t_f; \mathcal{V}_0)$  is almost everywhere equal to a function continuous from  $[0, t_f]$  into  $\mathcal{H}$ , and the following inequality holds in the scalar distribution sense on  $(0, t_f)$ :

$$2 < \dot{u}, u >= \frac{d}{dt} ||u||_{\mathcal{H}}^2$$
(3.1.18)

The initial condition (3.1.13) now makes sense.

From (3.1.14) and (3.1.15),

$$\langle \dot{U}, \varphi \rangle = \langle F_U - AU, \varphi \rangle, \quad \forall \varphi \in \mathcal{V}_0,$$
so  
 $\dot{U} + AU = F_U$ 

Thus, the alternative formulation of the weak problem (3.1.12)-(3.1.13) is the following.

Given  $F_U \in L^2(0, t_f; \mathcal{V}'_0)$  and  $U_0 \in \mathcal{H}$ , find  $U \in \mathcal{W}(0, t_f; \mathcal{V}_0)$  such that

$$\dot{U} + AU = F_U$$
 on  $(0, t_f)$  (3.1.19)

and

$$U(0) = U_0 \qquad \text{in } \mathcal{H} \tag{3.1.13}$$

The well-posedness of the problem (3.1.19), (3.1.13) is proved in [40, 27] and we have the following result.

**Theorem 3.1.** Problem (3.1.19), (3.1.13) admits a unique solution in  $\mathcal{W}(0, t_f; \mathcal{V}_0)$ . Furthermore, the solution depends continuously on the data: the bilinear map

$$F_U, U_0 \mapsto U$$

is continuous from  $L^2(0, t_f; \mathcal{V}'_0) \times \mathcal{H}$  to  $\mathcal{W}(0, t_f; \mathcal{V}_0)$ .

**Corollary 3.1.** The TCP system (2.1.19)-(2.1.21) admits a unique weak solution in  $L^2(0, t_f; H^1(\Omega, \mathbb{R}^3))$  and this solution depends continuously on the data. *Proof.* The result follows immediately from Assumption 3.1, (2.3.1), (2.3.2), and (2.3.16).

**Corollary 3.2.** The TCP system (2.2.17)-(2.2.19) admits a unique weak solution in  $L^2(0, t_f; H^1(\Omega, \mathbb{R}^3))$  and this solution depends continuously on the data.

*Proof.* The result follows immediately from Assumption 3.1, (2.3.4)-(2.3.9), and (2.3.16).

We finish this section with the following theorem which will be needed later.

**Theorem 3.2.** If  $F_U \in L^2(0, t_f; \mathcal{H})$  and  $U_0 \in \mathcal{V}_0$ , then  $\dot{U} \in L^2(0, t_f, \mathcal{H})$  and

$$||\dot{U}||_{L^{2}(0,t_{f},\mathcal{H})} \leq \sqrt{\lambda_{1}}||U_{0}||_{\mathcal{V}_{0}} + ||F_{U}||_{L^{2}(0,t_{f},\mathcal{H})}$$
(3.1.20)

*Proof.* Since  $\mathcal{V}_0$  is separable, there exists a countable set  $w_1, \ldots, w_m, \ldots$  which is dense in  $\mathcal{V}_0$ . For each m, define an approximate solution  $U_m(\bar{x}, t)$  as

$$U_m(\bar{x},t) = \sum_{i=1}^m g_{im}(t)w_i(\bar{x})$$
(3.1.21)

where  $g_{im}(t)$ ,  $1 \le i \le m$ , are the scalar functions defined on  $[0, t_f]$ ,  $U_m(\bar{x}, t)$ satisfies

$$(\dot{U}_m, w_j)_{\mathcal{H}} + (U_m, w_j)_a = \langle F_U, w_j \rangle, \qquad j = 1, \dots, m$$
 (3.1.22)

$$U_m(\bar{x},0) = U_{0_m} = \sum_{i=1}^m \xi_{im} w_i(\bar{x}) \xrightarrow[m \to \infty]{} U_0 \quad \text{in } \mathcal{H}$$
(3.1.23)

and

$$U_m(0) = U_{0_m} (3.1.24)$$

where  $U_{0_m}$  is the orthogonal projection in  $\mathcal{H}$  of  $U_0$  on the space spanned by  $w_1, \ldots, w_m$ .

We multiply (3.1.22) by  $\dot{g}_{jm}(t)$  and add for j:

$$||\dot{U}_m(t)||_{\mathcal{H}}^2 + (U_m(t), \dot{U}_m(t))_a = \langle F_U(t), \dot{U}_m(t) \rangle$$

Using (3.1.23), (3.1.24), and

$$(U_m(t), \dot{U}_m(t))_a = \frac{1}{2} \frac{d}{dt} ||U_m(t)||_a^2$$

and integrating over  $(0, t_f)$ , we have

$$2\int_{0}^{t_{f}} ||\dot{U}_{m}(t)||_{\mathcal{H}}^{2} dt + \int_{0}^{t_{f}} \frac{d}{dt} ||U_{m}(t)||_{a}^{2} dt = \int_{0}^{t_{f}} 2 \langle F_{U}(t), \dot{U}_{m}(t) \rangle dt \qquad \Rightarrow$$

$$2\int_{0}^{t_{f}} ||\dot{U}_{m}(t)||_{\mathcal{H}}^{2} dt + ||U_{m}(t_{f})||_{a}^{2} - ||U_{m}(0)||_{a}^{2} \leq$$

$$\leq \int_{0}^{t_{f}} ||F_{U}(t)||_{\mathcal{H}}^{2} dt + \int_{0}^{t_{f}} ||\dot{U}_{m}(t)||_{\mathcal{H}}^{2} dt \qquad \Rightarrow$$

$$||\dot{U}_{m}||_{L^{2}(0,t_{f};\mathcal{H})}^{2} \leq ||U_{0_{m}}||_{a}^{2} + ||F||_{L^{2}(0,t_{f};\mathcal{H})}^{2} \qquad (3.1.25)$$

Note that

$$||U||_a^2 = a(U,U) = \sum_{k=1}^3 \int_{\Omega} \lambda_k |\nabla U_k|^2 \,\mathrm{d}\Omega \le \left\{ \max_k \lambda_k \right\} \sum_{k=1}^3 \int_{\Omega} |\nabla U_k|^2 \,\mathrm{d}\Omega = \lambda_1 ||U||_{\mathcal{V}_0}^2$$

and so

$$||U||_a \le \sqrt{\lambda_1} ||U||_{\mathcal{V}_0} \tag{3.1.26}$$

Applying (3.1.26) to (3.1.25), we obtain

$$||\dot{U}_m||^2_{L^2(0,t_f;\mathcal{H})} \le \lambda_1 ||U_0||^2_{\mathcal{V}_0} + ||F||^2_{L^2(0,t_f;\mathcal{H})}$$

Thus,

$$||\dot{U}_m||_{L^2(0,t_f;\mathcal{H})} \le \left[\lambda_1 ||U_0||_{\mathcal{V}_0}^2 + ||F||_{L^2(0,t_f;\mathcal{H})}^2\right]^{1/2}$$

This shows that the sequence  $\dot{U}_m$  ranges in a bounded set of  $L^2(0, t_f; \mathcal{H})$ . Therefore,  $\dot{U} \in L^2(0, t_f; \mathcal{H})$  and

$$||\dot{U}||_{L^{2}(0,t_{f},\mathcal{H})} \leq \sqrt{\lambda_{1}}||U_{0}||_{\mathcal{V}_{0}} + ||F_{U}||_{L^{2}(0,t_{f},\mathcal{H})}$$

## 3.2 Elastic Problem

Using notations from Chapter 2, let

$$H = L^{2}(\Omega; \mathbb{R}^{n}) = \left\{ L^{2}(\Omega; \mathbb{R}) \right\}^{n}$$
$$V = H^{1}(\Omega; \mathbb{R}^{n}) = \left\{ H^{1}(\Omega; \mathbb{R}) \right\}^{n}$$
$$V_{0} = \left\{ \varphi \in V : \varphi \right|_{\Gamma_{F}} = 0 \right\}$$

where n = 2, 3 is the spatial dimensionality of the problem.

We start with the weak formulation of the two-dimensional Navier-type elastic system

$$\left(K + \frac{G}{3}\right)\nabla(\nabla \cdot \vec{u}) + G\nabla^2 \vec{u} = \nabla f \quad \text{in } \Omega \times (0, t_f)$$
 (2.1.45)

$$\tau \vec{n} = (fI + \hat{\sigma})\vec{n}$$
 on  $\Gamma_B \times (0, t_f)$  (2.1.43)

$$\vec{u} \approx 0$$
 on  $\Gamma_F \times (0, t_f)$  (2.1.14)

Take an arbitrary  $\vec{\Phi}(\bar{x}) = \phi(\bar{x})\vec{i} + \psi(\bar{x})\vec{j} \in V_0$ . From (2.1.27) and (2.1.45) we have

$$\left(K + \frac{G}{3}\right) \int_{\Omega} \nabla(\nabla \cdot \vec{u}) \cdot \vec{\Phi} \, \mathrm{d}\Omega + G \int_{\Omega} (\nabla^2 \vec{u}) \cdot \vec{\Phi} \, \mathrm{d}\Omega = \int_{\Omega} \nabla f \cdot \vec{\Phi} \, \mathrm{d}\Omega \qquad \Rightarrow$$
$$\left(K + \frac{G}{3}\right) \int_{\Omega} (u_x + v_y)_x \phi + (u_x + v_y)_y \psi \, \mathrm{d}\Omega +$$

$$+G\int_{\Omega}(u_{xx}+u_{yy})\phi + (v_{xx}+v_{yy})\psi \,\mathrm{d}\Omega = \int_{\Omega}\nabla f \cdot \vec{\Phi} \,\mathrm{d}\Omega \qquad \Rightarrow$$

$$\int_{\Omega} \left[ \left( K + \frac{G}{3} \right) u_{xx} + \left( K - \frac{2G}{3} \right) v_{yx} + G u_{xx} + G u_{yy} + G v_{xy} \right] \phi + \left[ \left( K + \frac{G}{3} \right) v_{yy} + \left( K - \frac{2G}{3} \right) u_{xy} + G v_{yy} + G v_{xx} + G u_{yx} \right] \psi \, \mathrm{d}\Omega =$$

$$\begin{split} &= \int_{\Omega} \left\{ \left[ \left( K + \frac{G}{3} \right) u_x + \left( K - \frac{2G}{3} \right) v_y + G u_x \right]_x + G (u_y + v_x)_y \right\} \phi + \\ &+ \left\{ \left[ \left( K + \frac{G}{3} \right) v_y + \left( K - \frac{2G}{3} \right) u_x + G v_y \right]_y + G (u_y + v_x)_x \right\} \psi \, \mathrm{d}\Omega = \\ &= \int_{\Omega} (\tau_{11x} + \tau_{12y}) \phi + (\tau_{21x} + \tau_{22y}) \psi \, \mathrm{d}\Omega = \int_{\Omega} \nabla f \cdot \vec{\Phi} \, \mathrm{d}\Omega \qquad \Rightarrow \\ &\int_{\Omega} (\nabla \cdot \vec{\tau}_1) \phi + (\nabla \cdot \vec{\tau}_2) \psi \, \mathrm{d}\Omega = \int_{\Omega} \nabla f \cdot \vec{\Phi} \, \mathrm{d}\Omega \qquad \Rightarrow \\ &\int_{\Omega} \nabla \cdot (\phi \vec{\tau}_1) - \vec{\tau}_1 \cdot \nabla \phi + \nabla \cdot (\psi \vec{\tau}_2) - \vec{\tau}_2 \cdot \nabla \psi \, \mathrm{d}\Omega = \int_{\Omega} \nabla \cdot (f \vec{\Phi}) - f \nabla \cdot \vec{\Phi} \, \mathrm{d}\Omega \qquad \Rightarrow \\ &\int_{\Gamma_B} (\phi \vec{\tau}_1 \cdot \vec{n} + \psi \vec{\tau}_2 \cdot \vec{n}) \, \mathrm{d}\Gamma - \int_{\Omega} \vec{\tau}_1 \cdot \nabla \phi + \vec{\tau}_2 \cdot \nabla \psi \, \mathrm{d}\Omega = \int_{\Gamma_B} f \vec{\Phi} \cdot \vec{n} \, \mathrm{d}\Gamma - \int_{\Omega} f \nabla \cdot \vec{\Phi} \, \mathrm{d}\Omega \\ &\Rightarrow \qquad \int_{\Gamma_B} (\tau \vec{n}) \cdot \vec{\Phi} \, \mathrm{d}\Gamma - \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{21} \psi_x + \tau_{22} \psi_y) \, \mathrm{d}\Omega = \\ &= \int_{\Gamma_B} (f \vec{\Phi}) \cdot \vec{n} \, \mathrm{d}\Gamma - \int_{\Omega} f (\phi_x + \psi_y) \, \mathrm{d}\Omega \quad \Rightarrow \\ &\int_{\Omega} (\tau_{11} \phi_x + \tau_{12} (\phi_y + \psi_x) + \tau_{22} \psi_y) \, \mathrm{d}\Omega - \int_{\Omega} f (\phi_x + \psi_y) \, \mathrm{d}\Omega + \end{split}$$

$$\int_{\Omega} (\tau \vec{n}) \cdot \vec{\Phi} \, \mathrm{d}\Gamma + \int_{\Gamma_B} (fI\vec{n}) \cdot \vec{\Phi} \, \mathrm{d}\Gamma = 0$$

Using (2.1.32), (2.1.38) and (2.1.43),

$$\begin{split} \int_{\Omega} (\tau_{11}\phi_x + \tau_{12}(\phi_y + \psi_x) + \tau_{22}\psi_y) \,\mathrm{d}\Omega &- \int_{\Omega} f(\phi_x + \psi_y) \,\mathrm{d}\Omega - \\ &- \int_{\Gamma_B} (\hat{\sigma}\vec{n}) \cdot \vec{\Phi} \,\mathrm{d}\Gamma = 0 \quad \Rightarrow \end{split}$$

$$\begin{split} \int_{\Omega} \tau_{11} \varepsilon_{11}(\vec{\Phi}) + 2\tau_{12} \varepsilon_{12}(\vec{\Phi}) + \tau_{22} \varepsilon_{22}(\vec{\Phi}) \, \mathrm{d}\Omega - \int_{\Omega} f \mathrm{tr} \varepsilon(\vec{\Phi}) \, \mathrm{d}\Omega - \int_{\Gamma_B} (\hat{\sigma} \vec{n}) \cdot \vec{\Phi} \, \mathrm{d}\Gamma = 0 \\ \Rightarrow \qquad D \mathcal{V}(\vec{u}) \vec{\Phi} = 0, \qquad \forall \ \vec{\Phi} \in V_0 \end{split}$$

Thus,

$$D\mathcal{V}(\vec{u})\vec{\Phi} = 0, \qquad \forall \ \vec{\Phi} \in V_0 \tag{3.2.1}$$

is a weak formulation of the elastic system (2.1.43) and (2.1.45). As before,  $D\mathcal{V}(\vec{u})\vec{\Phi}$  is the Gâteaux differential of  $\mathcal{V}$  with increment  $\vec{\Phi}$ .

Now we obtain the weak formulation of the three-dimensional Navier-type elastic system

$$\left(K + \frac{G}{3}\right)\nabla(\nabla \cdot \vec{u}) + G\nabla^2 \vec{u} = \nabla f \quad \text{in } \Omega \times (0, t_f)$$
 (2.2.43)

$$\tau \vec{n} = (fI + \hat{\sigma})\vec{n}$$
 on  $\Gamma \setminus \Gamma_F \times (0, t_f)$  (2.2.41)

$$\vec{u} \approx 0$$
 on  $\Gamma_F \times (0, t_f)$  (2.2.14)

Take an arbitrary  $\vec{\Phi}(\bar{x}) = \phi(\bar{x})\vec{i} + \psi(\bar{x})\vec{j} + \omega(\bar{x})\vec{k} \in V_0$ . From (2.2.43) we have

$$\left(K + \frac{G}{3}\right) \int_{\Omega} \nabla(\nabla \cdot \vec{u}) \cdot \vec{\Phi} \,\mathrm{d}\Omega + G \int_{\Omega} (\nabla^2 \vec{u}) \cdot \vec{\Phi} \,\mathrm{d}\Omega = \int_{\Omega} \nabla f \cdot \vec{\Phi} \,\mathrm{d}\Omega \qquad \Rightarrow$$

$$\begin{split} \left(K + \frac{G}{3}\right) \int_{\Omega} (u_x + v_y + w_z)_x \phi + (u_x + v_y + w_z)_y \psi + (u_x + v_y + w_z)_z \omega \, \mathrm{d}\Omega + \\ + G \int_{\Omega} (u_{xx} + u_{yy} + u_{zz}) \phi + (v_{xx} + v_{yy} + v_{zz}) \psi + (w_{xx} + w_{yy} + w_{zz}) \omega \, \mathrm{d}\Omega = \\ &= \int_{\Omega} \nabla f \cdot \vec{\Phi} \, \mathrm{d}\Omega \qquad \Rightarrow \end{split}$$

$$\int_{\Omega} \left[ \left( K + \frac{G}{3} \right) u_{xx} + \left( K - \frac{2G}{3} \right) (v_y + w_z)_x + G(u_{xx} + v_{xy} + w_{zx} + u_{yy} + u_{zz}) \right] \phi + \left[ \left( K + \frac{G}{3} \right) v_{yy} + \left( K - \frac{2G}{3} \right) (u_x + w_z)_y + G(v_{yy} + u_{xy} + w_{zy} + v_{xx} + v_{zz}) \right] \psi + \left[ \left( K + \frac{G}{3} \right) v_{yy} + \left( K - \frac{2G}{3} \right) (u_x + w_z)_y + G(v_{yy} + u_{xy} + w_{zy} + v_{xx} + v_{zz}) \right] \psi + \left[ \left( K + \frac{G}{3} \right) v_{yy} + \left( K - \frac{2G}{3} \right) (u_x + w_z)_y + G(v_{yy} + u_{xy} + w_{zy} + v_{xx} + v_{zz}) \right] \psi + \left[ \left( K + \frac{G}{3} \right) v_{yy} + \left( K - \frac{2G}{3} \right) (u_x + w_z)_y + G(v_{yy} + u_{xy} + w_{zy} + v_{xx} + v_{zz}) \right] \psi + \left[ \left( K + \frac{G}{3} \right) v_{yy} + \left( K - \frac{2G}{3} \right) (u_x + w_z)_y + G(v_{yy} + u_{xy} + w_{zy} + v_{xx} + v_{zz}) \right] \psi + \left[ \left( K + \frac{G}{3} \right) v_{yy} + \left( K - \frac{2G}{3} \right) (u_x + w_z)_y + G(v_{yy} + u_{xy} + w_{zy} + v_{xx} + v_{zz}) \right] \psi + \left[ \left( K + \frac{G}{3} \right) v_{yy} + \left( K - \frac{2G}{3} \right) (u_x + w_z)_y + G(v_{yy} + u_{xy} + w_{zy} + v_{xx} + v_{zz}) \right] \psi + \left[ \left( K + \frac{G}{3} \right) v_{yy} + \left( K - \frac{2G}{3} \right) (u_x + w_z)_y + G(v_{yy} + u_{xy} + v_{zy} + v_{xx} + v_{zz}) \right] \psi + \left[ \left( K + \frac{G}{3} \right) v_{yy} + \left( K - \frac{2G}{3} \right) (u_x + w_z)_y + G(v_{yy} + u_{xy} + v_{yz} + v_{xx} + v_{zz}) \right] \psi + \left[ \left( K + \frac{G}{3} \right) v_{yy} + \left( K - \frac{2G}{3} \right) (u_x + v_z)_y + G(v_{yy} + u_{xy} + v_{yz} + v_{yz} + v_{yz}) \right] \psi + \left[ \left( K + \frac{G}{3} \right) v_{yy} + \left( K - \frac{2G}{3} \right) (u_x + v_z)_y + G(v_y + v_{yy} + v_{yy} + v_{yy} + v_{yy} + v_{yy} \right] \psi + \left[ \left( K + \frac{G}{3} \right) v_{yy} + \left( K - \frac{2G}{3} \right) (u_x + v_z) \right] \psi + \left[ \left( K + \frac{G}{3} \right) (u_x + v_z) \right] \psi + \left[ \left( K + \frac{G}{3} \right) (u_x + v_z) \right] \psi + \left[ \left( K + \frac{G}{3} \right) (u_y + v_z) \right] \psi + \left[ \left( K + \frac{G}{3} \right) (u_y + v_z) \right] \psi + \left[ \left( K + \frac{G}{3} \right) (u_y + v_z) \right] \psi + \left[ \left( K + \frac{G}{3} \right) (u_y + v_z) \right] \psi + \left[ \left( K + \frac{G}{3} \right) (u_y + v_z) \right] \psi + \left[ \left( K + \frac{G}{3} \right) (u_y + v_z) \right] \psi + \left[ \left( K + \frac{G}{3} \right) (u_y + v_z) \right] \psi + \left[ \left( K + \frac{G}{3} \right) (u_y + v_z) \right] \psi + \left[ \left( K + \frac{G}{3} \right) (u_y + v_z) \right] \psi + \left[ \left( K + \frac{G}{3} \right) (u_y + v_z) \right] \psi + \left[ \left( K + \frac{G}{3} \right) (u_y + v_z) \right] \psi + \left[ \left( K + \frac{G}{3} \right) (u_y + v$$

$$+\left[\left(K+\frac{G}{3}\right)w_{zz}+\left(K-\frac{2G}{3}\right)(u_{x}+v_{y})_{z}+G(w_{zz}+u_{xz}+v_{yz}+w_{xx}+w_{zz})\right]\omega\,\mathrm{d}\Omega=$$

$$\begin{split} &\int_{\Omega} \left\{ \left[ \left( K + \frac{G}{3} \right) u_x + \left( K - \frac{2G}{3} \right) (v_y + w_z) + G u_x \right]_x + \\ &+ G (u_y + v_x)_y + G (w_x + u_z)_z \right\} \phi + \\ &+ \left\{ \left[ \left( K + \frac{G}{3} \right) v_y + \left( K - \frac{2G}{3} \right) (u_x + w_z) + G v_y \right]_y + \\ &+ G (u_y + v_x)_x + G (w_y + v_z)_z \right\} \psi + \\ &+ \left\{ \left[ \left( K + \frac{G}{3} \right) w_z + \left( K - \frac{2G}{3} \right) (u_x + v_y) + G w_z \right]_z + \\ &+ G (u_z + w_x)_x + G (v_z + w_y)_y \right\} \omega \, \mathrm{d}\Omega = \end{split}$$

$$= \int_{\Omega} (\tau_{11_x} + \tau_{12_y} + \tau_{13_z})\phi + (\tau_{21_x} + \tau_{22_y} + \tau_{23_z})\psi + (\tau_{31_x} + \tau_{32_y} + \tau_{33_z})\omega \,\mathrm{d}\Omega =$$
$$= \int_{\Omega} (\nabla \cdot \vec{\tau_1})\phi + (\nabla \cdot \vec{\tau_2})\psi + (\nabla \cdot \vec{\tau_3})\omega \,\mathrm{d}\Omega = \int_{\Omega} \nabla f \cdot \vec{\Phi} \,\mathrm{d}\Omega \qquad \Rightarrow$$

$$\begin{split} \int_{\Omega} \nabla \cdot (\phi \vec{\tau}_1) - \vec{\tau}_1 \cdot \nabla \phi + \nabla \cdot (\psi \vec{\tau}_2) - \vec{\tau}_2 \cdot \nabla \psi + \nabla \cdot (\omega \vec{\tau}_3) - \vec{\tau}_3 \cdot \nabla \omega \, \mathrm{d}\Omega = \\ = \int_{\Omega} \nabla \cdot (f \vec{\Phi}) - f \nabla \cdot \vec{\Phi} \, \mathrm{d}\Omega \quad \Rightarrow \end{split}$$

$$\begin{split} \int_{\Gamma \setminus \Gamma_F} (\phi \vec{\tau}_1 \cdot \vec{n} + \psi \vec{\tau}_2 \cdot \vec{n} + \omega \vec{\tau}_3 \cdot \vec{n}) \, \mathrm{d}\Gamma &- \int_{\Omega} \vec{\tau}_1 \cdot \nabla \phi + \vec{\tau}_2 \cdot \nabla \psi + \vec{\tau}_3 \cdot \nabla \omega \, \mathrm{d}\Omega = \\ &= \int_{\Gamma \setminus \Gamma_F} f \vec{\Phi} \cdot \vec{n} \, \mathrm{d}\Gamma - \int_{\Omega} f \nabla \cdot \vec{\Phi} \, \mathrm{d}\Omega \quad \Rightarrow \\ &\int_{\Gamma \setminus \Gamma_F} (\tau \vec{n}) \cdot \vec{\Phi} \, \mathrm{d}\Gamma - \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z + \eta_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_x + \tau_{12} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_y + \tau_{13} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega} (\tau_{11} \phi_z) \, \mathrm{d}\Gamma + \int_{\Omega$$

$$+\tau_{21}\psi_x + \tau_{22}\psi_y + \tau_{23}\psi_z + \tau_{31}\omega_x + \tau_{32}\omega_y + \tau_{33}\omega_z)\,\mathrm{d}\Omega =$$
$$= \int_{\Gamma\setminus\Gamma_F} (f\vec{\Phi})\cdot\vec{n}\,\mathrm{d}\Gamma - \int_{\Omega} f(\phi_x + \psi_y + \omega_z)\,\mathrm{d}\Omega \quad \Rightarrow$$

$$\int_{\Omega} (\tau_{11}\phi_x + \tau_{12}(\phi_y + \psi_x) + \tau_{22}\psi_y + \tau_{13}(\phi_z + \omega_x) + \tau_{23}(\psi_z + \omega_y) + \tau_{33}\omega_z) \,\mathrm{d}\Omega - \int_{\Omega} f(\phi_x + \psi_y + \omega_z) \,\mathrm{d}\Omega + \left[ -\int_{\Gamma \setminus \Gamma_F} (\tau \vec{n}) \cdot \vec{\Phi} \,\mathrm{d}\Gamma + \int_{\Gamma \setminus \Gamma_F} (fI\vec{n}) \cdot \vec{\Phi} \,\mathrm{d}\Gamma \right] = 0$$

Using (2.2.30), (2.2.35), and (2.2.41),

$$\begin{split} \int_{\Omega} (\tau_{11}\phi_x + \tau_{12}(\phi_y + \psi_x) + \tau_{22}\psi_y + \tau_{13}(\phi_z + \omega_x) + \tau_{23}(\psi_z + \omega_y) + \tau_{33}\omega_z) \,\mathrm{d}\Omega - \\ &- \int_{\Omega} f(\phi_x + \psi_y + \omega_z) \,\mathrm{d}\Omega - \int_{\Gamma \setminus \Gamma_F} (\hat{\sigma}\vec{n}) \cdot \vec{\Phi} \,\mathrm{d}\Gamma = 0 \quad \Rightarrow \end{split}$$

$$\begin{split} \int_{\Omega} \tau_{11}\varepsilon_{11}(\vec{\Phi}) + 2\tau_{12}\varepsilon_{12}(\vec{\Phi}) + \tau_{22}\varepsilon_{22}(\vec{\Phi}) + 2\tau_{13}\varepsilon_{13}(\vec{\Phi}) + 2\tau_{23}\varepsilon_{23}(\vec{\Phi}) + \tau_{33}\varepsilon_{33}(\vec{\Phi}) \,\mathrm{d}\Omega \\ &- \int_{\Omega} f \mathrm{tr}\varepsilon(\vec{\Phi}) \,\mathrm{d}\Omega - \int_{\Gamma \setminus \Gamma_F} (\hat{\sigma}\vec{n}) \cdot \vec{\Phi} \,\mathrm{d}\Gamma = 0 \\ &\Rightarrow \qquad D\mathcal{V}(\vec{u})\vec{\Phi} = 0, \qquad \forall \ \vec{\Phi} \in V_0 \end{split}$$

Thus,

$$D\mathcal{V}(\vec{u})\vec{\Phi} = 0, \qquad \forall \ \vec{\Phi} \in V_0 \tag{3.2.2}$$

is a weak formulation of the elastic system (2.2.41) and (2.2.43). As before,  $D\mathcal{V}(\vec{u})\vec{\Phi}$  is the Gâteaux differential of  $\mathcal{V}$  with increment  $\vec{\Phi}$ .

From (2.1.35), (2.2.33), (3.2.1), and (3.2.2), with the assumption that the applied boundary stress  $\hat{\sigma}$  is bounded on  $\bar{\Omega} \times [0, t_f]$ , the weak formulation of the two- and three-dimensional elastic problems has the following form: Given  $f \in L^2(0, t_f; H^1(\Omega))$ , find  $\vec{u} \in L^2(0, t_f; V_0)$  such that  $\dot{\vec{u}} \in L^2(0, t_f; V_0)$  and

$$a_E(\vec{u},\vec{\Phi}) - \int_{\Omega} f\left(\nabla \cdot \vec{\Phi}\right) d\Omega - \int_{\Gamma \setminus \Gamma_F} (\hat{\sigma}\vec{n}) \cdot \vec{\Phi} \, d\Gamma = 0, \qquad \forall \vec{\Phi} \in V_0 \tag{3.2.3}$$

Next we prove the well-posedness of the elastic system (3.2.3). Define a continuous linear functional F on V by a pairing

$$\vec{\Phi} \mapsto \langle F, \vec{\Phi} \rangle$$

where

$$\langle F, \vec{\Phi} \rangle = \int_{\Omega} f(\nabla \cdot \vec{\Phi}) \,\mathrm{d}\Omega + \int_{\Gamma \setminus \Gamma_F} (\hat{\sigma}\vec{n}) \cdot \vec{\Phi} \,\mathrm{d}\Gamma, \qquad \forall \ \vec{\Phi} \in V \tag{3.2.4}$$

Since F is continuous on V, there exists a constant  $c_F > 0$  such that

$$|\langle F, \vec{\Phi} \rangle| \le c_F ||\vec{\Phi}||_V, \qquad \forall \vec{\Phi} \in V$$
(3.2.5)

From (3.2.3) and (3.2.4) in follows that

$$a_E(\vec{u}, \vec{\Phi}) = < F, \vec{\Phi} >, \qquad \forall \ \vec{\Phi} \in V \tag{3.2.6}$$

The existence and uniqueness of a solution to the elastic problem is based on Korn's inequality [16]:

**Theorem 3.3.** Let  $\Omega$  be a bounded open set with a regular boundary. Then there exists a constant c > 0 that depends on  $\Omega$  such that

$$\int_{\Omega} \varepsilon_{ij}(\vec{v}) \varepsilon_{ij}(\vec{v}) \,\mathrm{d}\Omega + \int_{\Omega} v_i^2 \,\mathrm{d}\Omega \ge c ||\vec{v}||_V^2, \qquad \forall \ \vec{v} \in V$$

The consequence of Theorem 3.3 is the following result [16]:

**Theorem 3.4.** Let  $\Omega$  be a bounded open set with a regular boundary. Let  $\Gamma_F \subset \Gamma$ and  $\Gamma_F$  have positive measure. Then there exists a constant  $c_E > 0$  such that

$$a_E(\vec{v}, \vec{v}) \ge c_E ||\vec{v}||_V^2, \quad \forall \ \vec{v} \in V_0$$
(3.2.7)

For the sake of convenience, here and in the following we will take

$$\vec{u} = u_1 \vec{i} + u_2 \vec{j}, \quad \vec{v} = v_1 \vec{i} + v_2 \vec{j}, \text{ and } \vec{x} = x_1 \vec{i} + x_2 \vec{j} \in \Omega$$

for the 2D elastic problem and

$$\vec{u} = u_1 \vec{i} + u_2 \vec{j} + u_3 \vec{k}, \quad \vec{v} = v_1 \vec{i} + v_2 \vec{j} + v_3 \vec{k}, \text{ and } \vec{x} = x_1 \vec{i} + x_2 \vec{j} + x_3 \vec{k} \in \Omega$$

for the 3D elastic problem.

Then, from (2.1.23), (2.1.33), and (2.2.31), for every  $\vec{u}, \vec{v} \in V$ ,

$$\begin{split} |a_{E}(\vec{u},\vec{v})| &= \left| \int_{\Omega} a_{ijkl} \varepsilon_{ij}(\vec{u}) \varepsilon_{kl}(\vec{v}) \,\mathrm{d}\Omega \right| \leq \max_{i,j,k,l} \{a_{ijkl}\} \sum_{i,j,k,l=1}^{n} \left| \int_{\Omega} \varepsilon_{ij}(\vec{u}) \varepsilon_{kl}(\vec{v}) \,\mathrm{d}\Omega \right| = \\ &= \frac{1}{4} \max_{i,j,k,l} \{a_{ijkl}\} \sum_{i,j,k,l=1}^{n} \left| \int_{\Omega} \left( \frac{\partial u_{i}}{\partial x_{j}} + \frac{\partial u_{j}}{\partial x_{i}} \right) \left( \frac{\partial v_{k}}{\partial x_{l}} + \frac{\partial v_{l}}{\partial x_{k}} \right) \,\mathrm{d}\Omega \right| \leq \\ &\leq \frac{1}{4} \max_{i,j,k,l} \{a_{ijkl}\} \sum_{i,j,k,l=1}^{n} \left\{ \left\| \frac{\partial u_{i}}{\partial x_{j}} \right\|_{L^{2}(\Omega)} \left\| \frac{\partial v_{k}}{\partial x_{l}} \right\|_{L^{2}(\Omega)} + \left\| \frac{\partial u_{j}}{\partial x_{i}} \right\|_{L^{2}(\Omega)} \right\| \left\| \frac{\partial v_{l}}{\partial x_{k}} \right\|_{L^{2}(\Omega)} + \\ &+ \left\| \frac{\partial u_{i}}{\partial x_{j}} \right\|_{L^{2}(\Omega)} \left\| \frac{\partial v_{l}}{\partial x_{k}} \right\|_{L^{2}(\Omega)} + \left\| \frac{\partial u_{j}}{\partial x_{i}} \right\|_{L^{2}(\Omega)} \left\| \frac{\partial v_{k}}{\partial x_{k}} \right\|_{L^{2}(\Omega)} \right\} = \\ &= \max_{i,j,k,l} \{a_{ijkl}\} \sum_{i,j=1}^{n} \left\| \frac{\partial u_{i}}{\partial x_{j}} \right\|_{L^{2}(\Omega)} \sum_{k,l=1}^{n} \left\| \frac{\partial v_{k}}{\partial x_{l}} \right\|_{L^{2}(\Omega)} \right|^{\frac{1}{2}} \\ &\leq n \cdot \max_{i,j,k,l} \{a_{ijkl}\} \left[ \sum_{i,j=1}^{n} \left\| \frac{\partial u_{i}}{\partial x_{j}} \right\|_{L^{2}(\Omega)} \right]^{\frac{1}{2}} \left[ \sum_{k,l=1}^{n} \left\| \frac{\partial v_{k}}{\partial x_{l}} \right\|_{L^{2}(\Omega)} \right]^{\frac{1}{2}} \\ &\leq n \cdot \max_{i,j,k,l} \{a_{ijkl}\} \left[ \left| \vec{v} \right| |\vec{v}| |\vec{v}| |\vec{v}| |\vec{v}| \\ &\leq n \cdot \max_{i,j,k,l} \{a_{ijkl}\} \left[ \left| \vec{v} \right| |\vec{v}| |\vec{v}| |\vec{v}| |\vec{v}| \\ &= \beta \||\vec{u}||_{V}||\vec{v}||_{V}, \qquad \beta = n \cdot \max_{i,j,k,l} \{a_{ijkl}\}, \qquad n = 2, 3. \end{split}$$

Thus, there is a constant  $\beta > 0$   $(\beta = n \cdot \max_{i,j,k,l} \{a_{ijkl}\})$  such that

$$|a_E(\vec{u}, \vec{v})| \le \beta ||\vec{u}||_V ||\vec{v}||_V, \qquad \forall \ \vec{u}, \vec{v} \in V$$
(3.2.8)

Next we show that  $a_E(\cdot, \cdot)$  is symmetric.

$$a_{E}(\vec{u},\vec{v}) = \int_{\Omega} a_{ijkl} \varepsilon_{ij}(\vec{u}) \varepsilon_{kl}(\vec{v}) \,\mathrm{d}\Omega = \int_{\Omega} a_{klij} \varepsilon_{ij}(\vec{u}) \varepsilon_{kl}(\vec{v}) \,\mathrm{d}\Omega =$$
$$= \int_{\Omega} a_{klij} \varepsilon_{kl}(\vec{u}) \varepsilon_{ij}(\vec{v}) \,\mathrm{d}\Omega = a_{E}(\vec{v},\vec{u}), \quad \forall \ \vec{u}, \vec{v} \in V$$
(3.2.9)

From (3.2.7)-(3.2.9) it follows that  $a_E(\cdot, \cdot)$  is symmetric and continuous on  $V_0$  and there exist constants  $c_E > 0$  and  $\beta > 0$  such that

$$c_E ||\vec{u}||_V^2 \le a_E(\vec{u}, \vec{u}) \le \beta ||\vec{u}||_V^2, \qquad \forall \ \vec{u} \in V_0$$
(3.2.10)

Thus,  $a_E(\cdot, \cdot)$  is an inner product on  $V_0$  that is associated with a topology equivalent to the usual topology on  $V_0$ .

From the Riesz representation theorem and (3.2.5), (3.2.7), we have the following existence and uniqueness result with an a priori estimate.

**Theorem 3.5.** There exists a unique function  $\vec{u} \in V_0$  such that

$$a_E(\vec{u}, \vec{\Phi}) = \langle F, \vec{\Phi} \rangle, \quad \forall \vec{\Phi} \in V_0.$$

Furthermore,

$$||\vec{u}||_V \le \frac{c_F}{c_E}$$

#### 3.3 Coupled TCPu Problem

In Section 3.2, for both the two- and three-dimensional elastic problems we obtained the following weak formulation:

$$a_E(\vec{u},\vec{\Phi}) - \int_{\Omega} f\left(\nabla \cdot \vec{\Phi}\right) d\Omega - \int_{\Gamma \setminus \Gamma_F} (\hat{\sigma}\vec{n}) \cdot \vec{\Phi} \, d\Gamma = 0, \qquad \forall \ \vec{\Phi} \in V_0 \tag{3.2.3}$$

Since  $\dot{\vec{u}} \in V_0$ , we have

$$a_E(\dot{\vec{u}}, \dot{\vec{u}}) = \int_{\Omega} \dot{f}(\nabla \cdot \dot{\vec{u}}) \,\mathrm{d}\Omega + \int_{\Gamma \setminus \Gamma_F} (\dot{\hat{\sigma}} \vec{n}) \cdot \dot{\vec{u}} \,\mathrm{d}\Gamma$$
$$\leq ||\dot{f}||_{L^2(\Omega)} ||\nabla \cdot \dot{\vec{u}}||_{L^2(\Omega)} + ||\dot{\hat{\sigma}} \vec{n}||_{L^2(\Gamma;\mathbb{R}^n)} ||\dot{\vec{u}}||_{L^2(\Gamma;\mathbb{R}^n)}, \quad n = 2, 3.$$
(3.3.1)

Now

$$||\nabla \cdot \dot{\vec{u}}||_{L^{2}(\Omega)}^{2} = \int_{\Omega} \left[\sum_{i=1}^{n} \frac{\partial \dot{u}_{i}}{\partial x_{i}}\right]^{2} \mathrm{d}\Omega \le n \int_{\Omega} \sum_{i=1}^{n} \left|\frac{\partial \dot{u}_{i}}{\partial x_{i}}\right|^{2} \mathrm{d}\Omega \le n ||\nabla \dot{\vec{u}}||_{H}^{2}$$

where

$$||\nabla \dot{\vec{u}}||_{H}^{2} = \int_{\Omega} \sum_{i,j=1}^{n} \left| \frac{\partial \dot{u}_{i}}{\partial x_{j}} \right|^{2} d\Omega, \qquad n = 2, 3.$$

Thus,

$$||\nabla \cdot \vec{u}||_{L^2(\Omega)} \le \sqrt{n} ||\nabla \vec{u}||_H \qquad n = 2, 3.$$
 (3.3.2)

By the trace theorem [36], there exists  $\gamma_1 > 0$  such that

$$||\vec{v}||_{L^2(\Gamma;\mathbb{R}^n)} \le \gamma_1 ||\vec{v}||_V, \qquad \forall \ \vec{v} \in V$$
(3.3.3)

By Poincare-Friedrichs inequality [15], there exists a constant  $\gamma_2 = \gamma_2(\Omega) > 0$  such that

$$||\vec{v}||_H \le \gamma_2 ||\nabla \vec{v}||_H, \qquad \forall \ \vec{v} \in V_0 \tag{3.3.4}$$

From (3.3.3) and (3.3.4) it follows that there exists  $\gamma = \gamma(\Omega) > 0$  such that

$$||\dot{\vec{u}}||_{L^{2}(\Gamma;\mathbb{R}^{n})} \leq \gamma ||\nabla \dot{\vec{u}}||_{H}, \qquad n = 2, 3.$$
 (3.3.5)

Applying (3.3.2) and (3.3.5) to (3.3.1), we have

$$a_E(\vec{u}, \vec{u}) \le \sqrt{n} ||\dot{f}||_{L^2(\Omega)} ||\nabla \vec{u}||_H + \gamma ||\dot{\sigma} \vec{n}||_{L^2(\Gamma;\mathbb{R}^n)} ||\nabla \vec{u}||_H, \qquad n = 2, 3.$$
(3.3.6)

From (2.1.30)

$$||\dot{f}||_{L^{2}(\Omega)} = ||\vec{b}_{1}^{T}\dot{V}||_{L^{2}(\Omega)} \le |\vec{b}_{1}| \cdot ||\dot{V}||_{\mathcal{H}}$$
(3.3.7)

and from (3.2.10), (3.3.6), and (3.3.7) it follows that

$$c_E ||\nabla \dot{\vec{u}}||_H \le \sqrt{n} |\vec{b}_1| \cdot ||\dot{\vec{V}}||_{\mathcal{H}} + \gamma ||\dot{\hat{\sigma}}\vec{n}||_{L^2(\Gamma;\mathbb{R}^n)}, \quad n = 2, 3.$$
(3.3.8)

The last inequality holds for all  $t \in (0, t_f)$  and therefore

$$c_E ||\nabla \dot{\vec{u}}||_{L^2(0,t_f;H)} \le \sqrt{2n} |\vec{b}_1| \cdot ||\dot{\vec{V}}||_{L^2(0,t_f;\mathcal{H})} + \sqrt{2\gamma} ||\dot{\hat{\sigma}}\vec{n}||_{L^2(0,t_f;L^2(\Gamma;\mathbb{R}^n))}, \quad n = 2, 3.$$
(3.3.9)

Now we obtain a priori estimate  $||\dot{V}||_{L^2(0,t_f;\mathcal{H})}$ . Applying (2.3.2) for the 2D problem or (2.3.9) for the 3D problem, (2.3.16), and (2.3.20) to (3.1.20), we have

$$||\bar{V}||_{L^{2}(0,t_{f};\mathcal{H})} \leq ||P|| \cdot ||\dot{U}||_{L^{2}(0,t_{f};\mathcal{H})} + ||\dot{W}||_{L^{2}(0,t_{f};\mathcal{H})} \leq$$

$$\leq ||P|| \Big[ \sqrt{\lambda_1} ||U_0||_{\mathcal{V}_0} + ||P||^{-1} ||\dot{W}||_{L^2(0,t_f,\mathcal{H})} + ||DP^{-1}|| \cdot ||\nabla^2 W||_{L^2(0,t_f,\mathcal{H})} + |P^{-1}M^{-1}\vec{b_0}| \cdot ||\nabla \cdot \dot{\vec{u}}||_{L^2(0,t_f,L^2(\Omega))} \Big] + ||\dot{W}||_{L^2(0,t_f,\mathcal{H})}$$

$$\begin{split} ||\dot{V}||_{L^{2}(0,t_{f};\mathcal{H})} &\leq ||P|| \cdot |P^{-1}M^{-1}\vec{b}_{0}| \cdot ||\nabla \cdot \dot{\vec{u}}||_{L^{2}(0,t_{f},L^{2}(\Omega))} + \\ + ||P|| \Big[\sqrt{\lambda_{1}}||U_{0}||_{\mathcal{V}_{0}} + ||P||^{-1}||\dot{W}||_{L^{2}(0,t_{f},\mathcal{H})} + ||DP^{-1}|| \cdot ||\nabla^{2}W||_{L^{2}(0,t_{f},\mathcal{H})} \Big] + \\ + ||\dot{W}||_{L^{2}(0,t_{f},\mathcal{H})} \end{split}$$

Thus,

$$\|\dot{V}\|_{L^{2}(0,t_{f};\mathcal{H})} \leq \|P\| \cdot |P^{-1}M^{-1}\vec{b}_{0}| \cdot \|\nabla \cdot \dot{\vec{u}}\|_{L^{2}(0,t_{f},L^{2}(\Omega))} + \theta$$
(3.3.10)

where

$$\theta = ||P|| \left[ \sqrt{\lambda_1} ||U_0||_{\mathcal{V}_0} + ||P||^{-1} ||\dot{W}||_{L^2(0,t_f,\mathcal{H})} + ||DP^{-1}|| \cdot ||\nabla^2 W||_{L^2(0,t_f,\mathcal{H})} \right] + ||\dot{W}||_{L^2(0,t_f,\mathcal{H})}$$

Applying (3.3.2) and (3.3.10) to (3.3.9), we have

$$c_{E} ||\nabla \dot{\vec{u}}||_{L^{2}(0,t_{f};H)} \leq \sqrt{2}n|\vec{b}_{1}| \cdot ||P|| \cdot |P^{-1}M^{-1}\vec{b}_{0}| \cdot ||\nabla \dot{\vec{u}}||_{L^{2}(0,t_{f},H)} + \sqrt{2n}|\vec{b}_{1}|\theta + \sqrt{2}\gamma ||\dot{\hat{\sigma}}\vec{n}||_{L^{2}(0,t_{f};L^{2}(\Gamma;\mathbb{R}^{n}))}, \quad n = 2, 3.$$
(3.3.11)

From Corollaries 3.1 and 3.2, Theorem 3.2, Assumption 3.1, (2.3.1), (2.3.2), (2.3.4)-(2.3.9), and (3.3.11), we have the following result:

**Theorem 3.6.** The coupled two- and three-dimensional TCPu systems (2.1.19)-(2.1.21), (2.1.43), (2.1.45) and (2.2.17)-(2.2.22), respectively, have unique solutions  $(\bar{V}, \vec{u}), \ \bar{V} \in L^2(0, t_f; H^1(\Omega, \mathbb{R}^3))$  and  $\ \vec{u} \in L^2(0, t_f; V_0)$  with  $\ \dot{\bar{V}} \in L^2(0, t_f; \mathcal{H})$  and  $\ \dot{\vec{u}} \in L^2(0, t_f; V_0)$  if

$$c_E - \sqrt{2n} ||P|| \cdot |P^{-1}M^{-1}\vec{b}_0| \cdot |\vec{b}_1| > 0$$
(3.3.12)

where  $c_E > 0$  is the Korn's constant,  $\vec{b}_0$  and  $\vec{b}_1$  are coupling vectors given by (2.1.17) and (2.1.30), respectively, and  $M^{-1}A = PDP^{-1}$ .

## 2D Special Case

Remark 3.2. For an annular domain  $\Omega \subset \mathbb{R}^2$  of inner radius  $R_B$  and outer radius  $R_F$ , with  $\delta = \frac{R_B}{R_F} \leq e^{-1}$ , under the following side condition on the displacement  $\vec{u}$ 

$$\int_{\Omega} \sum_{i,j=1}^{2} \left( \frac{\partial u_i}{\partial x_j} - \frac{\partial u_j}{\partial x_i} \right) d\Omega = 0$$
(3.3.13)

the following inequality holds [21]:

$$\frac{D(\vec{u})}{S(\vec{u})} \le 4 \left[ 1 - \left(\frac{3}{\delta^{-2} + 1 + \delta^2}\right)^{1/2} \right]^{-1}$$
(3.3.14)

where

$$D(\dot{\vec{u}}) = \int_{\Omega} \sum_{i,j=1}^{2} \left| \frac{\partial \dot{u}_i}{\partial x_j} \right|^2 \, \mathrm{d}\Omega = ||\nabla \dot{\vec{u}}||_H^2$$

$$S(\dot{\vec{u}}) = \int_{\Omega} \varepsilon_{ij} (\dot{\vec{u}})^2 \,\mathrm{d}\Omega$$

It should be noted that the purpose of the condition (3.3.13) is to eliminate pure rotation. From (3.3.14) we have

$$S(\dot{\vec{u}}) \ge \frac{1}{4} \left[ 1 - \left(\frac{3}{\delta^{-2} + 1 + \delta^2}\right)^{1/2} \right] ||\nabla \dot{\vec{u}}||_H^2$$
(3.3.15)

For homogeneous isotropic medium, the coefficients of elasticity  $a_{ijkl}$ , i, j, k, l = 1, 2, in terms of Lamé parameters  $\lambda$  and  $\mu$  are

$$a_{1111} = a_{2222} = 2\mu + \lambda, \quad a_{1122} = a_{2211} = \lambda, \quad a_{1212} = a_{2121} = a_{1221} = a_{2112} = \mu,$$
  
(3.3.16)

Then from (2.1.23), (2.1.33), and (3.3.16) we have

$$a_E(\dot{\vec{u}}, \dot{\vec{u}}) = \int_{\Omega} (2\mu + \lambda) \varepsilon_{11}^2(\dot{\vec{u}}) + 2\mu \varepsilon_{12}^2(\dot{\vec{u}}) + 2\mu \varepsilon_{21}^2(\dot{\vec{u}}) + 2\lambda \varepsilon_{11}(\dot{\vec{u}}) \varepsilon_{22}(\dot{\vec{u}}) + (2\mu + \lambda) \varepsilon_{22}^2(\dot{\vec{u}}) d\Omega = \int_{\Omega} 2\mu (\varepsilon_{11}^2 + \varepsilon_{12}^2 + \varepsilon_{21}^2 + \varepsilon_{22}^2) + \lambda (\varepsilon_{11} + \varepsilon_{22})^2 d\Omega \geq 2\mu \int_{\Omega} \varepsilon_{11}^2 + \varepsilon_{12}^2 + \varepsilon_{21}^2 + \varepsilon_{22}^2 d\Omega = 2\mu S(\dot{\vec{u}})$$

Thus,

$$a_E(\dot{\vec{u}}, \dot{\vec{u}}) \ge 2\mu S(\dot{\vec{u}}) \tag{3.3.17}$$

where  $\mu = G$  is the shear modulus.

From (3.2.10), (3.3.15), and (3.3.17),

$$c_E = \frac{G}{2} \left[ 1 - \left( \frac{3}{\delta^{-2} + 1 + \delta^2} \right)^{1/2} \right]$$
(3.3.18)

Applying (3.3.18) to (3.3.12), we obtain the following result.

**Corollary 3.3.** For an annular domain  $\Omega \subset \mathbb{R}^2$  of inner radius  $R_B$  and outer radius  $R_F$ , with  $\delta = \frac{R_B}{R_F} \leq e^{-1}$  under side condition (3.3.13), the coupled TCPu system (2.1.19)-(2.1.21), (2.1.43), and (2.1.45) has a unique solution if

$$\frac{G}{2} \left[ 1 - \left(\frac{3}{\delta^{-2} + 1 + \delta^2}\right)^{1/2} \right] - 2\sqrt{2} ||P|| \cdot |P^{-1}M^{-1}\vec{b}_0| \cdot |\vec{b}_1| > 0$$
(3.3.19)

where G is the shear modulus,  $\vec{b}_0$  and  $\vec{b}_1$  are coupling vectors given by (2.1.17) and (2.1.30), respectively, and  $M^{-1}A = PDP^{-1}$ .

In this chapter, we obtained a sufficient condition for the well-posedness in the weak sense of the coupled parabolic-elliptic initial-boundary value problems for the two- and three-dimensional TCPu models. This condition depends on physical parameters of the system, coupling vectors, and the Korn's constant which in turn depends only on the shape of a domain. Evaluation of the Korn's constant becomes necessary for each specific domain and is essential for the proposed well-posedness theory. For a two-dimensional annular region that does not experience pure rotation, the Korn's constant can be expressed explicitly in terms of the ratio of the inner and outer radii of the domain. As a consequence, for such regions, the sufficient condition for the well-posedness of the problem was formulated in terms of physical parameters, coupling vectors, and the ratio of the inner and outer radii.

## Chapter 4

# Fourier-Finite Element Method for TCPu Problems

In this chapter, we will develop the Fourier-finite element method for the two- and three-dimensional radially nonsymmetric TCPu problems discussed in Chapters 2 and 3. The far-field radius can be specified arbitrarily and therefore we assume that far-field boundary conditions are time-independent and the non-zero displacement in the far-field is allowed with a penalty.

### 4.1 2D TCPu Problem

As in Section 2.1, let (X, Y) be a borehole Cartesian coordinate system with the *x*-axis in the direction of the maximum horizontal principal stress  $\sigma_H$  and the *y*-axis in the direction of the minimum horizontal principal stress  $\sigma_h$ . For the sake of convenience, we will make the transformation from the coordinate system (X, Y)to the borehole polar coordinate system  $(r, \theta)$  with its origin in the center of the annular region  $\Omega$ . Then the boundary  $\Gamma$  of  $\Omega$  is  $\Gamma = \Gamma_B \cup \Gamma_F$  with

$$\Gamma_B = \left\{ (R_B, \theta) \in \mathbb{R}^2 : 0 \le \theta < 2\pi \right\}$$
$$\Gamma_F = \left\{ (R_F, \theta) \in \mathbb{R}^2 : 0 \le \theta < 2\pi \right\}$$

#### 4.1.1 Fourier-type approximation of the TCP sytem

In the borehole polar coordinate system  $(r, \theta)$ , the TCP initial-boundary value problem (2.1.19)-(2.1.21) is given by

$$M\dot{\bar{V}} - A\nabla^2 \bar{V} = -\vec{b}_0 (\nabla \cdot \dot{\vec{u}}) \qquad \text{in } \Omega \times (0, t_f)$$
(4.1.1)

with boundary conditions

$$\bar{V}(r,\theta,t) = \begin{cases} V_B(t) & \text{on } \Gamma_B \times [0,t_f) \\ V_F(\theta) & \text{on } \Gamma_F \times [0,t_f) \end{cases}$$
(4.1.2)

and initial conditions

$$\bar{V}(r,\theta,0) = \tilde{V}_I(r,\theta) \tag{4.1.3}$$

where the displacement vector is

$$\vec{u}(r,\theta,t) = u(r,\theta,t)\vec{i}_r + v(r,\theta,t)\vec{i}_\theta$$

The equivalent parabolic TCP-type system with homogeneous boundary conditions, (2.3.11)-(2.3.13), in polar coordinates has the form

$$M\dot{V} - A\nabla^2 V = -M\dot{W} + A\nabla^2 W - \vec{b}_0(\nabla \cdot \vec{u}) \qquad \text{in } \Omega \times (0, t_f)$$
(4.1.4)

with boundary conditions

$$V(r, \theta, t) = 0 \qquad \text{on } \Gamma \times [0, t_f) \tag{4.1.5}$$

and initial conditions

$$V(r,\theta,0) = \tilde{V}_I(r,\theta) - W(r,\theta,0)$$

$$(4.1.6)$$

where

$$W(r,\theta,t) = \frac{R_F - r}{R_F - R_B} V_B(t) + \frac{r - R_B}{R_F - R_B} V_F(\theta)$$
(4.1.7)

$$V = \bar{V} - W \tag{4.1.8}$$

We introduce the following  $\theta$  - approximation:

$$V(r, \theta, t) = V_0(r, t) + V_1(r, t) \sin 2\theta + V_2(r, t) \cos 2\theta$$
$$u(r, \theta, t) = u_0(r, t) + u_1(r, t) \sin 2\theta + u_2(r, t) \cos 2\theta$$

$$v(r, \theta, t) = v_0(r, t) + v_1(r, t) \sin 2\theta + v_2(r, t) \cos 2\theta$$
  

$$V_B(t) = V_{B0}(t)$$
  

$$V_F(\theta) = V_{F0} + V_{F1} \sin 2\theta + V_{F2} \cos 2\theta$$
  

$$\tilde{V}_I(r, \theta) = \tilde{V}_{I0}(r) + \tilde{V}_{I1}(r) \sin 2\theta + \tilde{V}_{I2}(r) \cos 2\theta$$
(4.1.9)

Then

$$W(r,\theta,t) = W_0(r,t) + W_1(r,t)\sin 2\theta + W_2(r,t)\cos 2\theta$$
(4.1.10)

where

$$W_i(r,t) = \frac{R_F - r}{R_F - R_B} V_{Bi}(t) + \frac{r - R_B}{R_F - R_B} V_{Fi}, \qquad i = 0, \dots, 2$$
(4.1.11)

and  $V_{Bi}(t) = 0, \quad i = 1, 2.$ 

With this approximation, the system (4.1.4) has the form:

$$M\dot{V} - A\left[\frac{1}{r}(rV_{r})_{r} + \frac{1}{r^{2}}V_{\theta\theta}\right] = -M\dot{W} + A\left(\frac{1}{r}(rW_{r})_{r} + \frac{1}{r^{2}}W_{\theta\theta}\right) -\vec{b}_{0}\left(\dot{u}_{r} + \frac{1}{r}(\dot{u} + \dot{v}_{\theta})\right)$$
(4.1.12)

where W is given by (4.1.10) and (4.1.11),

$$W_r(r,\theta,t) = W_{0_r}(r,t) + W_{1_r}(r,t)\sin 2\theta + W_{2_r}(r,t)\cos 2\theta$$
(4.1.13)

with

$$W_{i_r}(r,t) = \frac{1}{R_F - R_B} (V_{Fi} - V_{Bi}(t)), \qquad i = 0, \dots, 2$$
(4.1.14)

and V

$$V_{Bi}(t) = 0, \qquad i = 1, 2.$$

$$W_{\theta\theta}(r,\theta,t) = -4W_1(r,t)\sin 2\theta - 4W_2(r,t)\cos 2\theta$$
(4.1.15)

The  $\theta\text{-approximation}$  for the initial conditions  $V(r,\theta,0)$  is

$$V(r, \theta, 0) = V_0(r, 0) + V_1(r, 0) \sin 2\theta + V_2(r, 0) \cos 2\theta$$

where

$$V_{i}(r,0) = \tilde{V}_{Ii}(r) - W_{i}(r,0)$$
  
=  $\tilde{V}_{Ii}(r) - \frac{R_{F} - r}{R_{F} - R_{B}} V_{Bi}(0) - \frac{r - R_{B}}{R_{F} - R_{B}} V_{Fi}, \qquad i = 0, \dots, 2$ (4.1.16)

and  $V_{Bi}(0) = 0, \quad i = 1, 2.$ 

Applying (4.1.13)-(4.1.16) to (4.1.12), multiplying by 1,  $\sin 2\theta$ , and  $\cos 2\theta$ , and integrating over  $(0, 2\pi)$ , we obtain the following three systems.

System I:

$$M\dot{V}_0 - A\frac{1}{r}(rV_{0_r})_r = -M\dot{W}_0 + A\frac{1}{r}(rW_{0_r})_r - \vec{b}_0\left(\dot{u}_{0_r} + \frac{1}{r}\dot{u}_0\right)$$
(4.1.17)

with boundary conditions

$$V_0(r,t) = 0$$
 on  $\Gamma \times [0, t_f)$  (4.1.18)

and initial conditions

$$V_0(r,0) = \tilde{V}_{I0}(r) - W_0(r,0)$$
(4.1.19)

System II:

$$M\dot{V}_{1} - A\left[\frac{1}{r}(rV_{1_{r}})_{r} - \frac{4}{r^{2}}V_{1}\right] = -M\dot{W}_{1} + A\left(\frac{1}{r}(rW_{1_{r}})_{r} - \frac{4}{r^{2}}W_{1}\right)$$
$$-\vec{b}_{0}\left(\dot{u}_{1_{r}} + \frac{1}{r}(\dot{u}_{1} - 2\dot{v}_{2})\right)$$
(4.1.20)

with boundary conditions

$$V_1(r,t) = 0 \qquad \text{on } \Gamma \times [0,t_f) \tag{4.1.21}$$

and initial conditions

$$V_1(r,0) = \tilde{V}_{I1}(r) - W_1(r,0)$$
(4.1.22)

System III:

$$M\dot{V}_{2} - A\left[\frac{1}{r}(rV_{2r})_{r} - \frac{4}{r^{2}}V_{2}\right] = -M\dot{W}_{2} + A\left(\frac{1}{r}(rW_{2r})_{r} - \frac{4}{r^{2}}W_{2}\right)$$
$$-\vec{b}_{0}\left(\dot{u}_{2r} + \frac{1}{r}(\dot{u}_{2} + 2\dot{v}_{1})\right)$$
(4.1.23)

with boundary conditions

$$V_2(r,t) = 0$$
 on  $\Gamma \times [0, t_f)$  (4.1.24)

and initial conditions

$$V_2(r,0) = \tilde{V}_{I2}(r) - W_2(r,0) \tag{4.1.25}$$

# 4.1.2 Fourier-type approximation of the elastic system

In the borehole polar coordinate system  $(r, \theta)$ , the displacement is

$$\vec{u}(r,\theta,t) = u(r,\theta,t)\vec{i}_r + v(r,\theta,t)\vec{i}_\theta$$
  
$$\vec{u} \approx 0 \quad \text{and} \quad \dot{\vec{u}} \approx 0 \quad \text{on} \ \Gamma_F$$
(4.1.26)

and the strain tensor has the form

$$\begin{bmatrix} \varepsilon_{rr} & \varepsilon_{r\theta} \\ \varepsilon_{\theta r} & \varepsilon_{\theta \theta} \end{bmatrix}$$

where

$$\varepsilon_{rr} = u_r, \qquad \varepsilon_{r\theta} = \varepsilon_{\theta r} = \frac{1}{2} \left( v_r + \frac{u_\theta - v}{r} \right), \qquad \varepsilon_{\theta \theta} = \frac{u + v_\theta}{r}$$
(4.1.27)

The stress-strain relations for isotropic homogeneous medium in terms of the shear modulus G and the bulk modulus K are

$$\tau = 2G\varepsilon + \left(K - \frac{2}{3}G\right)(\mathrm{tr}\varepsilon)I \tag{2.1.26}$$

where

$$\tau_{rr} = 2Gu_r + \left(K - \frac{2}{3}G\right)\left(u_r + \frac{u + v_\theta}{r}\right)$$
  
$$\tau_{\theta\theta} = 2G\frac{u + v_\theta}{r} + \left(K - \frac{2}{3}G\right)\left(u_r + \frac{u + v_\theta}{r}\right)$$
  
$$\tau_{r\theta} = \tau_{\theta r} = G\left(v_r + \frac{u_\theta - v}{r}\right)$$
(4.1.28)

The sum of the total potential energy and the penalty term is

$$\Pi(\vec{u}) = \frac{1}{2} \int_{\Omega} \left( \tau_{rr}(\vec{u}) \varepsilon_{rr}(\vec{u}) + 2\tau_{r\theta}(\vec{u}) \varepsilon_{r\theta}(\vec{u}) + \tau_{\theta\theta}(\vec{u}) \varepsilon_{\theta\theta}(\vec{u}) \right) d\Omega - \int_{\Omega} f\left( \operatorname{tr}\varepsilon(\vec{u}) \right) d\Omega - \int_{\Gamma} (\hat{\sigma}\vec{n}) \cdot \vec{u} \, d\Gamma + K_F \int_{\Gamma_F} \vec{u} \cdot \vec{u} \, d\Gamma$$
(4.1.29)

where  $f = \vec{b}_1^T \bar{V} = \vec{b}_1^T (V + W)$  is body force,  $\vec{b}_1^T = \begin{bmatrix} \tilde{\zeta} & -\lambda & \tilde{\alpha} \end{bmatrix}$ ,  $K_F$  is a penalization constant,

$$\hat{\sigma} = \begin{bmatrix} \sigma_r & \sigma_{r\theta} \\ \sigma_{r\theta} & \sigma_{\theta} \end{bmatrix}$$
 is applied boundary stress in the system  $(r, \theta)$ .

Next, we describe stress around the borehole.

$$\sigma_{XY} = \begin{bmatrix} \sigma_x & \sigma_{xy} \\ \sigma_{xy} & \sigma_y \end{bmatrix} = \begin{bmatrix} \sigma_H & 0 \\ 0 & \sigma_h \end{bmatrix}$$
is the stress tensor in  $(X, Y)$  coordinate system and  $\sigma(r, \theta) = \begin{bmatrix} \sigma_r & \sigma_{r\theta} \\ \sigma_{r\theta} & \sigma_{\theta} \end{bmatrix} (r, \theta)$ is the stress tensor in  $(r, \theta)$  coordinate system.

The transformation from  $\sigma_{XY}$  to  $\sigma(r,\theta)$  is given by [9]:

$$\sigma_r = \frac{\sigma_x + \sigma_y}{2} \left( 1 - \frac{R_B^2}{r^2} \right) + p_B \frac{R_B^2}{r^2} + \sigma_{xy} \left( 1 - \frac{4R_B^2}{r^2} + \frac{3R_B^4}{r^4} \right) \sin 2\theta + \frac{\sigma_x - \sigma_y}{2} \left( 1 - \frac{4R_B^2}{r^2} + \frac{3R_B^4}{r^4} \right) \cos 2\theta$$

$$\sigma_{\theta} = \frac{\sigma_x + \sigma_y}{2} \left( 1 + \frac{R_B^2}{r^2} \right) - p_B \frac{R_B^2}{r^2} - \sigma_{xy} \left( 1 + \frac{3R_B^4}{r^4} \right) \sin 2\theta$$
$$- \frac{\sigma_x - \sigma_y}{2} \left( 1 + \frac{3R_B^4}{r^4} \right) \cos 2\theta$$

$$\sigma_{r\theta} = \frac{\sigma_x - \sigma_y}{2} \left( 1 + \frac{2R_B^2}{r^2} - \frac{3R_B^4}{r^4} \right) \sin 2\theta + \sigma_{xy} \left( 1 + \frac{2R_B^2}{r^2} - \frac{3R_B^4}{r^4} \right) \cos 2\theta$$

where  $p_B$  is the well bore pressure and  $\nu$  is Poisson's ratio of the formation. From these equations we obtain applied stress on the boundaries.

On the borehole boundary  $\Gamma_B$ :

$$\sigma_r(R_B, t) = p_B(t)$$
  

$$\sigma_\theta(R_B, \theta, t) = \sigma_x + \sigma_y - p_B(t) - 4\sigma_{xy}\sin 2\theta - 2(\sigma_x - \sigma_y)\cos 2\theta$$
  

$$\sigma_{r\theta}(R_B, \theta) = 0$$

On the far-field boundary  $\Gamma_F$ :

$$\sigma_r(R_F,\theta) = \frac{\sigma_x + \sigma_y}{2} + \sigma_{xy}\sin 2\theta + \frac{\sigma_x - \sigma_y}{2}\cos 2\theta$$
$$\sigma_\theta(R_F,\theta) = \frac{\sigma_x + \sigma_y}{2} - \sigma_{xy}\sin 2\theta - \frac{\sigma_x - \sigma_y}{2}\cos 2\theta$$
$$\sigma_{r\theta}(R_F,\theta) = \frac{\sigma_x - \sigma_y}{2}\sin 2\theta + \sigma_{xy}\cos 2\theta$$

Let  $V = H^1(\Omega; \mathbb{R}^2)$ . Our objective is to find the displacement  $\vec{u}$  such that

$$\vec{u} \in L^2(0, t_f; V), \quad \dot{\vec{u}} \in L^2(0, t_f; V) \quad \text{and}$$
  
$$D\Pi(\vec{u})\vec{\Phi} = 0, \quad \forall \vec{\Phi} \in V \quad (4.1.30)$$

where  $\vec{u}(r,\theta,t) = u(r,\theta,t)\vec{i}_r + v(r,\theta,t)\vec{i}_{\theta}$ ,  $\vec{\Phi}(r,\theta) = \varphi(r,\theta)\vec{i}_r + \psi(r,\theta)\vec{i}_{\theta}$ , and  $D\Pi(\vec{u})\vec{\Phi}$  is the Gâteaux differential of  $\Pi$  with increment  $\vec{\Phi}$ .

Applying (4.1.27) and (4.1.28) to (4.1.30), we have

$$\int_{0}^{2\pi} \int_{R_{B}}^{R_{F}} \left[ \left( 2Gu_{r} + \left(K - \frac{2}{3}G\right) \left(u_{r} + \frac{u + v_{\theta}}{r}\right) \right) \varphi_{r} + G\left(v_{r} + \frac{u_{\theta} - v}{r}\right) \left(\psi_{r} + \frac{\varphi_{\theta} - \psi}{r}\right) \right. \\ \left. + \left( 2G\frac{u + v_{\theta}}{r} + \left(K - \frac{2}{3}G\right) \left(u_{r} + \frac{u + v_{\theta}}{r}\right) \right) \frac{\varphi + \psi_{\theta}}{r} \right] r \, \mathrm{d}r \, \mathrm{d}\theta \\ \left. - \int_{0}^{2\pi} \int_{R_{B}}^{R_{F}} f\left(\varphi_{r} + \frac{\varphi + \psi_{\theta}}{r}\right) r \, \mathrm{d}r \, \mathrm{d}\theta \\ \left. + \int_{0}^{2\pi} \left( \sigma_{r}(R_{B}, \theta) \varphi(R_{B}, \theta) + \sigma_{r\theta}(R_{B}, \theta) \psi(R_{B}, \theta) \right) R_{B} \, \mathrm{d}\theta - \right]$$

$$-\int_{0}^{2\pi} \left( \sigma_r(R_F, \theta) \varphi(R_F, \theta) + \sigma_{r\theta}(R_F, \theta) \psi(R_F, \theta) \right) R_F \, \mathrm{d}\theta + K_F \int_{0}^{2\pi} \left( u(R_F, \theta) \varphi(R_F, \theta) + v(R_F, \theta) \psi(R_F, \theta) \right) R_F \, \mathrm{d}\theta = 0$$
(4.1.31)

The first term of the left-hand side of (4.1.31) is

$$\int_{0}^{2\pi} \int_{R_{B}}^{R_{F}} \left[ \left( K + \frac{4}{3}G \right) u_{r}\varphi_{r}r + \left( K - \frac{2}{3}G \right) u\varphi_{r} + \left( K - \frac{2}{3}G \right) v_{\theta}\varphi_{r} \right. \\ \left. + Gv_{r}\psi_{r}r + Gv_{r}\varphi_{\theta} - Gv_{r}\psi + Gu_{\theta}\psi_{r} - Gv\psi_{r} + Gu_{\theta}\varphi_{\theta}\frac{1}{r} - Gu_{\theta}\psi\frac{1}{r} \right. \\ \left. - Gv\varphi_{\theta}\frac{1}{r} + Gv\psi\frac{1}{r} + \left( K + \frac{4}{3}G \right) u\varphi\frac{1}{r} + \left( K + \frac{4}{3}G \right) u\psi_{\theta}\frac{1}{r} \right. \\ \left. + \left( K + \frac{4}{3}G \right) v_{\theta}\varphi\frac{1}{r} + \left( K + \frac{4}{3}G \right) v_{\theta}\psi_{\theta}\frac{1}{r} + \left( K - \frac{2}{3}G \right) u_{r}\varphi + \left( K - \frac{2}{3}G \right) u_{r}\psi_{\theta} \, dr \, d\theta \right.$$

$$(4.1.32)$$

Applying  $\theta$ -approximation (4.1.9) to (4.1.32) and integrating with respect to  $\theta$  over  $(0, 2\pi)$ , we have

$$\begin{aligned} \pi \int_{R_B}^{R_F} \left[ \left( K + \frac{4}{3}G \right) \left( 2u_{0r}\varphi_{0r}r + u_{1r}\varphi_{1r}r + u_{2r}\varphi_{2r}r \right) \\ &+ \left( K - \frac{2}{3}G \right) \left( 2u_{0}\varphi_{0r} + u_{1}\varphi_{1r} + u_{2}\varphi_{2r} - 2v_{2}\varphi_{1r} + 2v_{1}\varphi_{2r} \right) \\ &+ G \left( 2v_{0r}\psi_{0r}r + v_{1r}\psi_{1r}r + v_{2r}\psi_{2r}r - 2v_{1r}\varphi_{2} + 2v_{2r} - 2u_{2}\psi_{1r} + 2u_{1}\psi_{2r}\varphi_{1} \right) \\ &+ G \left( - 2v_{0r}\psi_{0} - v_{1r}\psi_{1} - v_{2r}\psi_{2} - 2v_{0}\psi_{0r} - v_{1}\psi_{1r} - v_{2}\psi_{2r} \right) \\ &+ G \left( 4u_{2}\varphi_{2}\frac{1}{r} + 4u_{1}\varphi_{1}\frac{1}{r} + 2u_{2}\psi_{1}\frac{1}{r} - 2u_{1}\psi_{2}\frac{1}{r} \right) \\ &+ G \left( 2v_{1}\varphi_{2}\frac{1}{r} - 2v_{2}\varphi_{1}\frac{1}{r} + 2v_{0}\psi_{0}\frac{1}{r} + v_{1}\psi_{1}\frac{1}{r} + v_{2}\psi_{2}\frac{1}{r} \right) \\ &+ \left( K + \frac{4}{3}G \right) \left( 2u_{0}\varphi_{0}\frac{1}{r} + u_{1}\varphi_{1}\frac{1}{r} + u_{2}\varphi_{2}\frac{1}{r} - 2u_{1}\psi_{2}\frac{1}{r} + 2u_{2}\psi_{1}\frac{1}{r} \right) \\ &+ \left( K + \frac{4}{3}G \right) \left( - 2v_{2}\varphi_{1}\frac{1}{r} + 2v_{1}\varphi_{2}\frac{1}{r} + 4v_{2}\psi_{2}\frac{1}{r} + 4v_{1}\psi_{1}\frac{1}{r} \right) \\ &+ \left( K - \frac{2}{3}G \right) \left( 2u_{0r}\varphi_{0} + u_{1r}\varphi_{1} + u_{2r}\varphi_{2} - 2u_{1r}\psi_{2} + 2u_{2r}\psi_{1} \right) \right] dr \quad (4.1.33)
\end{aligned}$$

Combining like terms with respect to the basis functions, (4.1.33) has the following form:

$$\int_{R_B}^{R_F} \left[ 2\left(K + \frac{4}{3}G\right) u_{0_r} \varphi_{0_r} r + 2\left(K - \frac{2}{3}G\right) (u_0 \varphi_{0_r} + u_{0_r} \varphi_0) + 2\left(K + \frac{4}{3}G\right) u_0 \varphi_0 \frac{1}{r} \right] \mathrm{d}r$$
(4.1.34)

 $\varphi_1$  :

 $\varphi_0$  :

$$+\pi \int_{R_B}^{R_F} \left[ \left( K + \frac{4}{3}G \right) u_{1_r} \varphi_{1_r} r + \left( K - \frac{2}{3}G \right) u_1 \varphi_{1_r} - 2\left( K - \frac{2}{3}G \right) v_2 \varphi_{1_r} + 2G v_{2_r} \varphi_1 \right. \\ \left. + \left( K + \frac{16}{3}G \right) u_1 \varphi_1 \frac{1}{r} - 2\left( K + \frac{7}{3}G \right) v_2 \varphi_1 \frac{1}{r} + \left( K - \frac{2}{3}G \right) u_{1_r} \varphi_1 \right] \mathrm{d}r \quad (4.1.35)$$

 $\varphi_2$  :

$$+\pi \int_{R_B}^{R_F} \left[ \left( K + \frac{4}{3}G \right) u_{2r} \varphi_{2r} r + \left( K - \frac{2}{3}G \right) u_2 \varphi_{2r} + 2 \left( K - \frac{2}{3}G \right) v_1 \varphi_{2r} - 2G v_{1r} \varphi_2 \right. \\ \left. + \left( K + \frac{16}{3}G \right) u_2 \varphi_2 \frac{1}{r} + 2 \left( K + \frac{7}{3}G \right) v_1 \varphi_2 \frac{1}{r} + \left( K - \frac{2}{3}G \right) u_{2r} \varphi_2 \right] dr \quad (4.1.36)$$

 $\psi_0$  :

$$+ \pi \int_{R_B}^{R_F} \left[ \underbrace{2Gv_{0_r}\psi_{0_r}r - 2Gv_{0_r}\psi_0 - 2Gv_0\psi_{0_r} + 2Gv_0\psi_0\frac{1}{r}}_{2G\left[\left(v_{0_r} - \frac{v_0}{r}\right)\left(\psi_{0_r} - \frac{\psi_0}{r}\right)\right]r} \right] dr \qquad (4.1.37)$$

 $\psi_1$  :

$$+\pi \int_{R_B}^{R_F} \left[ Gv_{1_r}\psi_{1_r}r - Gv_{1_r}\psi_1 - 2Gu_2\psi_{1_r} - Gv_1\psi_{1_r} + 2\left(K + \frac{7}{3}G\right)u_2\psi_1\frac{1}{r} + \left(4K + \frac{19}{3}G\right)v_1\psi_1\frac{1}{r} + 2\left(K - \frac{2}{3}G\right)u_{2_r}\psi_1 \right] dr \quad (4.1.38)$$

 $\psi_2$  :

$$+\pi \int_{R_B}^{R_F} \left[ Gv_{2_r}\psi_{2_r}r - Gv_{2_r}\psi_2 + 2Gu_1\psi_{2_r} - Gv_2\psi_{2_r} - 2\left(K + \frac{7}{3}G\right)u_1\psi_2\frac{1}{r} + \left(4K + \frac{19}{3}G\right)v_2\psi_2\frac{1}{r} - 2\left(K - \frac{2}{3}G\right)u_{1_r}\psi_2 \right] dr \quad (4.1.39)$$

The  $\sigma$ -approximation of body force f has the form

$$f(r,\theta,t) = f_0(r,t) + f_1(r,t)\sin 2\theta + f_2(r,t)\cos 2\theta$$
(4.1.40)

The second term of the left-hand side of (4.1.31) is

$$-\pi \int_{R_B}^{R_F} \left[ 2f_0 \varphi_{0_r} r + f_1 \varphi_{1_r} r + f_2 \varphi_{2_r} r + 2f_0 \varphi_0 + f_1 \varphi_1 + f_2 \varphi_2 - 2f_1 \psi_2 + 2f_2 \psi_1 \right] dr$$
(4.1.41)

The third term of the left-hand side of (4.1.31) is

$$2\pi p_B \varphi_0(R_B) R_B \tag{4.1.42}$$

The fourth term of the left-hand side of (4.1.31) is

$$-\pi \Big[ 2 \frac{\sigma_x + \sigma_y}{2} \varphi_0(R_F) + \sigma_{xy} \varphi_1(R_F) + \frac{\sigma_x - \sigma_y}{2} \varphi_2(R_F) + \frac{\sigma_x - \sigma_y}{2} \psi_1(R_F) + \sigma_{xy} \psi_2(R_F) \Big] R_F$$
(4.1.43)

The fifth term of the left-hand side of (4.1.31) is

$$\pi K_F \Big[ 2u_0(R_F)\varphi_0(R_F) + u_1(R_F)\varphi_1(R_F) + u_2(R_F)\varphi_2(R_F) \\ + 2v_0(R_F)\psi_0(R_F) + v_1(R_F)\psi_1(R_F) + v_2(R_F)\psi_2(R_F) \Big] R_F \quad (4.1.44)$$

In (4.1.20)-(4.1.39) and (4.1.41)-(4.1.44), we combine terms corresponding to  $\varphi_i$  and  $\psi_i$ ,  $i = 0, \ldots, 2$ , and equal them to 0.

$$\varphi_{0}: \quad (u_{0})$$

$$\int_{R_{B}}^{R_{F}} \left[ \left( K + \frac{4G}{3} \right) u_{0_{r}} \varphi_{0_{r}} r + \left( K - \frac{2G}{3} \right) (u_{0} \varphi_{0_{r}} + u_{0_{r}} \varphi_{0}) + \left( K + \frac{4G}{3} \right) u_{0} \varphi_{0} \frac{1}{r} - \int_{R_{B}}^{R_{F}} f_{0} (\varphi_{0_{r}} r + \varphi_{0}) \, \mathrm{d}r + p_{B} \varphi_{0} (R_{B}) R_{B} - \frac{\sigma_{x} + \sigma_{y}}{2} \varphi_{0} (R_{F}) R_{F} + K_{F} u_{0} (R_{F}) \varphi_{0} (R_{F}) R_{F} = 0 \qquad (4.1.45)$$

$$\varphi_{1}: (u_{1}, v_{2})$$

$$\int_{R_{B}}^{R_{F}} \left[ \left( K + \frac{4G}{3} \right) u_{1r} \varphi_{1r} r + \left( K - \frac{2G}{3} \right) u_{1} \varphi_{1r} + \left( K - \frac{2G}{3} \right) u_{1r} \varphi_{1} \right] \\ + \left( K + \frac{16G}{3} \right) u_{1} \varphi_{1} \frac{1}{r} - 2 \left( K - \frac{2G}{3} \right) v_{2} \varphi_{1r} + 2G v_{2r} \varphi_{1} - 2 \left( K + \frac{7G}{3} \right) v_{2} \varphi_{1} \frac{1}{r} \right] dr$$

$$- \int_{R_{B}}^{R_{F}} f_{1} \left( \varphi_{1r} r + \varphi_{1} \right) dr - \sigma_{xy} \varphi_{1} (R_{F}) R_{F} + K_{F} u_{1} (R_{F}) \varphi_{1} (R_{F}) R_{F} = 0 \qquad (4.1.46)$$

$$\varphi_{2}: (u_{2}, v_{1})$$

$$\int_{R_{B}}^{R_{F}} \left[ \left( K + \frac{4G}{3} \right) u_{2r} \varphi_{2r} r + \left( K - \frac{2G}{3} \right) u_{2} \varphi_{2r} + \left( K - \frac{2G}{3} \right) u_{2r} \varphi_{2} \right]$$

$$+ \left( K + \frac{16G}{3} \right) u_{2} \varphi_{2} \frac{1}{r} + 2 \left( K - \frac{2G}{3} \right) v_{1} \varphi_{2r} - 2G v_{1r} \varphi_{2} + 2 \left( K + \frac{7G}{3} \right) v_{1} \varphi_{2} \frac{1}{r} \right] dr$$

$$- \int_{R_{B}}^{R_{F}} f_{2} \left( \varphi_{2r} r + \varphi_{2} \right) dr - \frac{\sigma_{x} - \sigma_{y}}{2} \varphi_{2} (R_{F}) R_{F} + K_{F} u_{2} (R_{F}) \varphi_{2} (R_{F}) R_{F} = 0$$

$$(4.1.47)$$

$$\psi_0: \quad (v_0)$$

$$\int_{R_B}^{R_F} G\Big[\Big(v_{0_r} - \frac{v_0}{r}\Big)\Big(\psi_{0_r} - \frac{\psi_0}{r}\Big)\Big]r\,\mathrm{d}r + K_F v_0(R_F)\psi_0(R_F)R_F = 0$$

Take  $\psi_0 = v_0$ . Then

$$\int_{R_B}^{R_F} G\left[\left(v_{0_r} - \frac{v_0}{r}\right)^2\right] r \,\mathrm{d}r + K_F R_F v_0^2(R_F) = 0$$

It follows that

$$v_{0_r} - \frac{v_0}{r} = 0, \quad R_B < r < R_F, \quad \text{and} \quad v_0(R_F) = 0$$

and so,

$$v_0(r,t) = C_0(t)r, \quad R_B < r < R_F$$
(4.1.48)

$$\psi_{1}: \quad (u_{2}, v_{1})$$

$$\int_{R_{B}}^{R_{F}} \left[ -2Gu_{2}\psi_{1_{r}} + 2\left(K - \frac{2G}{3}\right)u_{2_{r}}\psi_{1} + 2\left(K + \frac{7G}{3}\right)u_{2}\psi_{1}\frac{1}{r} + Gv_{1_{r}}\psi_{1_{r}}r - Gv_{1}\psi_{1_{r}} - Gv_{1_{r}}\psi_{1} + \left(4K + \frac{19G}{3}\right)v_{1}\psi_{1}\frac{1}{r} \right] dr$$

$$-\int_{R_{B}}^{R_{F}} 2f_{2}\psi_{1} dr - \frac{\sigma_{x} - \sigma_{y}}{2}\psi_{1}(R_{F})R_{F} + K_{F}v_{1}(R_{F})\psi_{1}(R_{F})R_{F} = 0 \quad (4.1.49)$$

$$\psi_{2}: \quad (u_{1}, v_{2})$$

$$\int_{R_{B}}^{R_{F}} \left[ 2Gu_{1}\psi_{2r} - 2\left(K - \frac{2G}{3}\right)u_{1r}\psi_{2} - 2\left(K + \frac{7G}{3}\right)u_{1}\psi_{2}\frac{1}{r} + Gv_{2r}\psi_{2r}r - Gv_{2}\psi_{2r} - Gv_{2r}\psi_{2} + \left(4K + \frac{19G}{3}\right)v_{2}\psi_{2}\frac{1}{r} \right] dr$$

$$+ \int_{R_{B}}^{R_{F}} 2f_{1}\psi_{2} dr - \sigma_{xy}\psi_{2}(R_{F})R_{F} + K_{F}v_{2}(R_{F})\psi_{2}(R_{F})R_{F} = 0 \qquad (4.1.50)$$

# 4.1.3 Finite element approximation

Dirichlet boundary conditions for the TCP system

r-approximation:



Figure 4.1: r-approximation of 2D Dirichlet boundary conditions

Here N is the number of subintervals;  $M_D = N - 1$  is the number of basis functions;  $\bar{b}(r) = \begin{bmatrix} \bar{b}_1(r) & \dots & \bar{b}_{M_D}(r) \end{bmatrix}^T$  is the vector of basis functions; and
$$B(r) = \begin{bmatrix} \bar{b}(r)^T & 0 & 0\\ 0 & \bar{b}(r)^T & 0\\ 0 & 0 & \bar{b}(r)^T \end{bmatrix}_{3 \times 3M_D}$$

Using r-approximation, the coefficients of the  $\theta$ -approximation for  $V(r, \theta, t)$  can be represented as

$$V_i(r,t) = B(r)\bar{c}_i(t)$$
 (4.1.51)

where  $\bar{c}_i(t)$  is a  $(3M_D \times 1)$  vector.

The coefficients of the  $\theta$ -approximation (4.1.9) for boundary conditions (4.1.2) and initial condition (4.1.3) can be represented as

$$V_{B0}(t) = \bar{\nu}_{B0}(t)$$

$$V_{Fi} = \bar{\nu}_{Fi}$$

$$\tilde{V}_{Ii}(r) = B(r)\bar{\nu}_{Ii}, \qquad i = 0, \dots, 2$$
(4.1.52)

where  $\bar{\nu}_{B0}(t)$  and  $\bar{\nu}_{Fi}$  are  $(3 \times 1)$  vectors and  $\bar{\nu}_{Ii}$  is a  $(3M_D \times 1)$  vector.

Neumann boundary conditions for the elastic system

*r*-approximation:



Figure 4.2: r-approximation of 2D Neumann boundary conditions

Here N is the number of subintervals; M = N + 1 is the number of basis functions;  $\bar{\mathbf{b}}(r) = \begin{bmatrix} \bar{\mathbf{b}}_1(r) & \dots & \bar{\mathbf{b}}_M(r) \end{bmatrix}^T$  is the vector of basis functions.

Using r-approximation, the coefficients of the  $\theta$ -approximation (4.1.9) for the dis-

placement coordinates (4.1.51) can be represented as

$$u_i(r,t) = \bar{\mathbf{b}}(r)^T \bar{\mu}_i(t)$$
 (4.1.53)

$$v_i(r,t) = \bar{\mathbf{b}}(r)^T \bar{\eta}_i(t)$$
 (4.1.54)

where  $\bar{\mu}_i(t)$  and  $\bar{\eta}_i(t)$  are  $(M \times 1)$  vectors.

Finite element approximation of the elastic system

Define the following  $M \times M$  matrices:

$$g_2(r) = \int_{R_B}^{R_F} \bar{\mathbf{b}}_r(r) \bar{\mathbf{b}}_r(r)^T r \, \mathrm{d}r$$
$$g_1 = \int_{R_B}^{R_F} \bar{\mathbf{b}}(r) \bar{\mathbf{b}}_r(r)^T \, \mathrm{d}r$$
$$g_0\left(\frac{1}{r}\right) = \int_{R_B}^{R_F} \bar{\mathbf{b}}(r) \bar{\mathbf{b}}(r)^T \frac{1}{r} \, \mathrm{d}r$$
$$g_F = \bar{\mathbf{b}}(R_F) \bar{\mathbf{b}}(R_F)^T$$

Let  $P_3 = \begin{bmatrix} 0 & 0 & 1 \end{bmatrix}^T$ . Then

$$p_B(t) = P_3^T V_B(t) = P_3^T V_{B0}(t) = P_3^T \bar{\nu}_{B0}(t)$$
(4.1.55)

Also, let

$$\sigma_R = \frac{\sigma_x + \sigma_y}{2}$$
 and  $\sigma_D = \frac{\sigma_x - \sigma_y}{2}$  (4.1.56)

As before,

$$f(r,\theta,t) = \vec{b}_1^T \bar{V}(r,\theta,t) = \vec{b}_1^T \big( V(r,\theta,t) + W(r,\theta,t) \big)$$

and from (4.1.9)-(4.1.11), (4.1.13), (4.1.51), and (4.1.52) it follows that

$$f_i(r,t) = \vec{b}_1^T \left( V_i(r,t) + W_i(r,t) \right)$$
  
=  $\vec{b}_1^T B(r) \bar{c}_i(t) + \frac{R_F - r}{R_F - R_B} \vec{b}_1^T \bar{\nu}_{Bi}(t) + \frac{r - R_B}{R_F - R_B} \vec{b}_1^T \bar{\nu}_{Fi}, \quad i = 0, 1, 2 \quad (4.1.57)$ 

and  $\bar{\nu}_{Bi}(t) = 0$ , i = 1, 2.

From (4.1.45)-(4.1.50) we obtain

 $\varphi_0$  :

$$\left[ \left( K + \frac{4G}{3} \right) g_2(r) + \left( K - \frac{2G}{3} \right) \left( g_1^T + g_1 \right) + \left( K + \frac{4G}{3} \right) g_0\left( \frac{1}{r} \right) + K_F R_F g_F \right] \bar{\mu}_0(t) 
= \left[ \int_{R_B}^{R_F} \left( r \,\bar{\mathbf{b}}_r(r) + \bar{\mathbf{b}}(r) \right) \vec{b}_1^T B(r) \,\mathrm{d}r \right] \bar{c}_0(t) 
+ \left\{ \left[ \int_{R_B}^{R_F} \frac{R_F - r}{R_F - R_B} \left( r \,\bar{\mathbf{b}}_r(r) + \bar{\mathbf{b}}(r) \right) \,\mathrm{d}r \right] \vec{b}_1^T - R_B \,\bar{\mathbf{b}}(R_B) P_3^T \right\} \bar{\nu}_{B0}(t) 
+ \left[ \int_{R_B}^{R_F} \frac{r - R_B}{R_F - R_B} \left( r \,\bar{\mathbf{b}}_r(r) + \bar{\mathbf{b}}(r) \right) \,\mathrm{d}r \right] \vec{b}_1^T \bar{\nu}_{F0} + R_F \sigma_R \,\bar{\mathbf{b}}(R_F) \tag{4.1.58}$$

 $\varphi_1$  :

$$\left[ \left( K + \frac{4G}{3} \right) g_2(r) + \left( K - \frac{2G}{3} \right) \left( g_1^T + g_1 \right) + \left( K + \frac{16G}{3} \right) g_0\left(\frac{1}{r}\right) + K_F R_F g_F \right] \bar{\mu}_1(t) -2 \left[ \left( K - \frac{2G}{3} \right) g_1^T - Gg_1 + \left( K + \frac{7G}{3} \right) g_0\left(\frac{1}{r}\right) \right] \bar{\eta}_2(t) = \left[ \int_{R_B}^{R_F} \left( r \,\bar{\mathbf{b}}_r(r) + \bar{\mathbf{b}}(r) \right) \vec{b}_1^T B(r) \,\mathrm{d}r \right] \bar{c}_1(t) + \left[ \int_{R_B}^{R_F} \frac{r - R_B}{R_F - R_B} \left( r \,\bar{\mathbf{b}}_r(r) + \bar{\mathbf{b}}(r) \right) \,\mathrm{d}r \right] \vec{b}_1^T \bar{\nu}_{F1} + R_F \sigma_{xy} \,\bar{\mathbf{b}}(R_F)$$
(4.1.59)

 $\varphi_2$  :

$$\left[ \left( K + \frac{4G}{3} \right) g_2(r) + \left( K - \frac{2G}{3} \right) \left( g_1^T + g_1 \right) + \left( K + \frac{16G}{3} \right) g_0\left(\frac{1}{r}\right) + K_F R_F g_F \right] \bar{\mu}_2(t) + 2 \left[ \left( K - \frac{2G}{3} \right) g_1^T - Gg_1 + \left( K + \frac{7G}{3} \right) g_0\left(\frac{1}{r}\right) \right] \bar{\eta}_1(t) = \left[ \int_{R_B}^{R_F} \left( r \,\bar{\mathbf{b}}_r(r) + \bar{\mathbf{b}}(r) \right) \vec{b}_1^T B(r) \,\mathrm{d}r \right] \bar{c}_2(t) + \left[ \int_{R_B}^{R_F} \frac{r - R_B}{R_F - R_B} \left( r \,\bar{\mathbf{b}}_r(r) + \bar{\mathbf{b}}(r) \right) \,\mathrm{d}r \right] \vec{b}_1^T \bar{\nu}_{F2} + R_F \sigma_D \,\bar{\mathbf{b}}(R_F)$$
(4.1.60)

 $\psi_0$  :

$$\left[Gg_{2}(r) - G\left(g_{1}^{T} + g_{1}\right) + Gg_{0}\left(\frac{1}{r}\right) + K_{F}R_{F}g_{F}\right]\bar{\eta}_{0}(t) = 0$$

It follows that

$$\bar{\eta}_0(t) = 0$$

 $\psi_1$  :

$$2\Big[\Big(K - \frac{2G}{3}\Big)g_1 - Gg_1^T + \Big(K + \frac{7G}{3}\Big)g_0\Big(\frac{1}{r}\Big)\Big]\bar{\mu}_2(t) \\ + \Big[Gg_2(r) - G\Big(g_1^T + g_1\Big) + \Big(4K + \frac{19G}{3}\Big)g_0\Big(\frac{1}{r}\Big) + K_F R_F g_F\Big]\bar{\eta}_1(t) \\ = 2\Big[\int_{R_B}^{R_F} \bar{\mathbf{b}}(r)\vec{b}_1^T B(r)\,\mathrm{d}r\Big]\bar{c}_2(t) \\ + 2\Big[\int_{R_B}^{R_F} \frac{r - R_B}{R_F - R_B}\bar{\mathbf{b}}(r)\,\mathrm{d}r\Big]\vec{b}_1^T\bar{\nu}_{F2} + R_F \sigma_D \bar{\mathbf{b}}(R_F)$$
(4.1.61)

 $\psi_2$  :

$$-2\left[\left(K - \frac{2G}{3}\right)g_{1} - Gg_{1}^{T} + \left(K + \frac{7G}{3}\right)g_{0}\left(\frac{1}{r}\right)\right]\bar{\mu}_{1}(t) \\ + \left[Gg_{2}(r) - G\left(g_{1}^{T} + g_{1}\right) + \left(4K + \frac{19G}{3}\right)g_{0}\left(\frac{1}{r}\right) + K_{F}R_{F}g_{F}\right]\bar{\eta}_{2}(t) \\ = -2\left[\int_{R_{B}}^{R_{F}}\bar{\mathbf{b}}(r)\bar{b}_{1}^{T}B(r)\,\mathrm{d}r\right]\bar{c}_{1}(t) \\ -2\left[\int_{R_{B}}^{R_{F}}\frac{r - R_{B}}{R_{F} - R_{B}}\bar{\mathbf{b}}(r)\,\mathrm{d}r\right]\bar{b}_{1}^{T}\bar{\nu}_{F1} + R_{F}\sigma_{xy}\,\bar{\mathbf{b}}(R_{F})$$
(4.1.62)

Define the following matrices and vectors.

 $M \times M$  matrices:

$$\mathcal{M}_{0} = \left(K + \frac{4G}{3}\right)g_{2}(r) + \left(K - \frac{2G}{3}\right)\left(g_{1}^{T} + g_{1}\right) + \left(K + \frac{4G}{3}\right)g_{0}\left(\frac{1}{r}\right) + K_{F}R_{F}g_{F}$$
$$\mathcal{M}_{1} = \left(K + \frac{4G}{3}\right)g_{2}(r) + \left(K - \frac{2G}{3}\right)\left(g_{1}^{T} + g_{1}\right) + \left(K + \frac{16G}{3}\right)g_{0}\left(\frac{1}{r}\right) + K_{F}R_{F}g_{F}$$
$$\mathcal{M}_{2} = Gg_{2}(r) - G\left(g_{1}^{T} + g_{1}\right) + \left(4K + \frac{19G}{3}\right)g_{0}\left(\frac{1}{r}\right) + K_{F}R_{F}g_{F}$$
$$\mathcal{N} = 2\left[\left(K - \frac{2G}{3}\right)g_{1}^{T} - Gg_{1} + \left(K + \frac{7G}{3}\right)g_{0}\left(\frac{1}{r}\right)\right]$$
(4.1.63)

 $M\times 3M_D$  matrices:

$$\mathcal{Z}_{0} = \int_{R_{B}}^{R_{F}} \left( r \,\bar{\mathbf{b}}_{r}(r) + \,\bar{\mathbf{b}}(r) \right) \vec{b}_{1}^{T} B(r) \,\mathrm{d}r \tag{4.1.64}$$

$$\mathcal{Z}_{1} = 2 \int_{R_{B}}^{R_{F}} \bar{\mathbf{b}}(r) \vec{b}_{1}^{T} B(r) \,\mathrm{d}r$$
(4.1.65)

# $M\times 3$ matrices:

$$Y_{B0} = \left[ \int_{R_B}^{R_F} \frac{R_F - r}{R_F - R_B} \left( r \,\bar{\mathbf{b}}_r(r) + \,\bar{\mathbf{b}}(r) \right) \mathrm{d}r \right] \vec{b}_1^T - R_B \,\bar{\mathbf{b}}(R_B) P_3^T \tag{4.1.66}$$

$$Y_{F0} = \left[ \int_{R_B}^{R_F} \frac{r - R_B}{R_F - R_B} \left( r \,\bar{\mathbf{b}}_r(r) + \,\bar{\mathbf{b}}(r) \right) \,\mathrm{d}r \right] \vec{b}_1^T \tag{4.1.67}$$

$$Y_{F1} = 2 \left[ \int_{R_B}^{R_F} \frac{r - R_B}{R_F - R_B} \,\bar{\mathbf{b}}(r) \,\mathrm{d}r \right] \vec{b}_1^T \tag{4.1.68}$$

 $M\times 1$  vectors:

$$S_R = R_F \sigma_R \,\bar{\mathbf{b}}(R_F) \tag{4.1.69}$$

$$S_{xy} = R_F \sigma_{xy} \,\bar{\mathbf{b}}(R_F) \tag{4.1.70}$$

$$S_D = R_F \sigma_D \,\bar{\mathbf{b}}(R_F) \tag{4.1.71}$$

Applying (4.1.63)-(4.1.71) to (4.1.58)-(4.1.62), we obtain the following matrix equations:

$$\mathcal{M}_0\bar{\mu}_0(t) = \mathcal{Z}_0\bar{c}_0(t) + Y_{B0}\bar{\nu}_{B0}(t) + Y_{F0}\bar{\nu}_{F0} + S_R \tag{4.1.72}$$

$$\begin{bmatrix} \mathcal{M}_1 & -\mathcal{N} \\ -\mathcal{N}^T & \mathcal{M}_2 \end{bmatrix} \begin{bmatrix} \bar{\mu}_1(t) \\ \bar{\eta}_2(t) \end{bmatrix} = \begin{bmatrix} \mathcal{Z}_0 \\ -\mathcal{Z}_1 \end{bmatrix} \bar{c}_1(t) + \begin{bmatrix} Y_{F0} \\ -Y_{F1} \end{bmatrix} \bar{\nu}_{F1} + \begin{bmatrix} 1 \\ 1 \end{bmatrix} S_{xy} \qquad (4.1.73)$$

$$\begin{bmatrix} \mathcal{M}_1 & \mathcal{N} \\ \mathcal{N}^T & \mathcal{M}_2 \end{bmatrix} \begin{bmatrix} \bar{\mu}_2(t) \\ \bar{\eta}_1(t) \end{bmatrix} = \begin{bmatrix} \mathcal{Z}_0 \\ \mathcal{Z}_1 \end{bmatrix} \bar{c}_2(t) + \begin{bmatrix} Y_{F0} \\ Y_{F1} \end{bmatrix} \bar{\nu}_{F2} + \begin{bmatrix} 1 \\ 1 \end{bmatrix} S_D$$
(4.1.74)

Define the following matrices:

$$J = \begin{bmatrix} I_{M \times M} & 0 \\ 0 & -I_{M \times M} \end{bmatrix}$$
$$\mathcal{Q} = \begin{bmatrix} \mathcal{M}_1 & \mathcal{N} \\ \mathcal{N}^T & \mathcal{M}_2 \end{bmatrix} \implies \begin{bmatrix} \mathcal{M}_1 & -\mathcal{N} \\ -\mathcal{N}^T & \mathcal{M}_2 \end{bmatrix} = J\mathcal{Q}J$$
$$\bar{\mathcal{Z}} = \begin{bmatrix} \mathcal{Z}_0 \\ \mathcal{Z}_1 \end{bmatrix} \implies \Rightarrow \begin{bmatrix} \mathcal{Z}_0 \\ -\mathcal{Z}_1 \end{bmatrix} = J\bar{\mathcal{Z}}$$
$$\bar{Y}_F = \begin{bmatrix} Y_{F0} \\ Y_{F1} \end{bmatrix} \implies \Rightarrow \begin{bmatrix} Y_{F0} \\ -Y_{F1} \end{bmatrix} = J\bar{Y}_F$$
$$\bar{S}_{xy} = \begin{bmatrix} S_{xy} \\ S_{xy} \end{bmatrix}, \qquad \bar{S}_D = \begin{bmatrix} S_D \\ S_D \end{bmatrix}$$

$$P_{10} = \begin{bmatrix} 1 & 0 \end{bmatrix} \otimes I_{M \times M}$$
$$P_{01} = \begin{bmatrix} 0 & 1 \end{bmatrix} \otimes I_{M \times M}$$

From (4.1.72)-(4.1.74) we have

$$\bar{\mu}_0(t) = \mathcal{M}_0^{-1} \Big[ \mathcal{Z}_0 \bar{c}_0(t) + Y_{B0} \bar{\nu}_{B0}(t) + Y_{F0} \bar{\nu}_{F0} + S_R \Big]$$
(4.1.75)

$$\bar{\mu}_1(t) = P_{10} \mathcal{Q}^{-1} \Big[ \bar{\mathcal{Z}} \bar{c}_1(t) + \bar{Y}_F \bar{\nu}_{F1} + J \bar{S}_{xy} \Big]$$
(4.1.76)

$$\bar{\eta}_2(t) = -P_{01}\mathcal{Q}^{-1} \Big[ \bar{\mathcal{Z}}\bar{c}_1(t) + \bar{Y}_F\bar{\nu}_{F1} + J\bar{S}_{xy} \Big]$$
(4.1.77)

$$\bar{\mu}_2(t) = P_{10} \mathcal{Q}^{-1} \left[ \bar{\mathcal{Z}} \bar{c}_2(t) + \bar{Y}_F \bar{\nu}_{F2} + \bar{S}_D \right]$$
(4.1.78)

$$\bar{\eta}_1(t) = P_{01} \mathcal{Q}^{-1} \Big[ \bar{\mathcal{Z}} \bar{c}_2(t) + \bar{Y}_F \bar{\nu}_{F2} + \bar{S}_D \Big]$$
(4.1.79)

Finite element approximation of the TCP system

System I (radially symmetric case)

System I is given by (4.1.17)-(4.1.19):

$$M\dot{V}_0 - A\frac{1}{r}(rV_{0_r})_r = -M\dot{W}_0 + A\frac{1}{r}(rW_{0_r})_r - \vec{b}_0\left(\dot{u}_{0_r} + \frac{1}{r}\dot{u}_0\right)$$
(4.1.17)

with boundary conditions

$$V_0(r,t) = 0$$
 on  $\Gamma \times [0, t_f)$  (4.1.18)

and initial conditions

$$V_0(r,0) = \tilde{V}_{I0}(r) - W_0(r,0) \tag{4.1.19}$$

Let

 $\mathcal{A} = M^{-1}A$ 

Then (4.1.17) has the form

$$\dot{V}_0 - \mathcal{A}\frac{1}{r}(rV_{0_r})_r = -\dot{W}_0 + \mathcal{A}\frac{1}{r}(rW_{0_r})_r - M^{-1}\vec{b}_0\left(\dot{u}_{0_r} + \frac{1}{r}\dot{u}_0\right)$$
(4.1.80)

Using (4.1.51), the left-hand side of (4.1.80) is

$$B(r)\dot{\bar{c}}_0(t) - \mathcal{A}\frac{1}{r} \left(rB_r(r)\right)_r \bar{c}_0(t)$$

$$(4.1.81)$$

Multiplying (4.1.81) by  $B(r)^T$  and integrating with respect to r, we have

$$\begin{bmatrix} \int_{R_B}^{R_F} B(r)^T B(r) r \, \mathrm{d}r \end{bmatrix} \dot{\bar{c}}_0(t) - \begin{bmatrix} \int_{R_B}^{R_F} B(r)^T \mathcal{A} \frac{1}{r} \Big( r B_r(r) \Big)_r r \, \mathrm{d}r \Big] \bar{c}_0(t) \\ = \begin{bmatrix} \int_{R_B}^{R_F} B(r)^T B(r) r \, \mathrm{d}r \end{bmatrix} \dot{\bar{c}}_0(t) + \begin{bmatrix} \int_{R_B}^{R_F} B_r(r)^T \mathcal{A} B_r(r) r \, \mathrm{d}r \end{bmatrix} \bar{c}_0(t) \\ = \begin{bmatrix} \int_{R_B}^{R_F} B(r)^T B(r) r \, \mathrm{d}r \end{bmatrix} \dot{\bar{c}}_0(t) + \mathcal{A} \otimes \begin{bmatrix} \int_{R_B}^{R_F} \bar{b}_r(r) \bar{b}_r(r)^T r \, \mathrm{d}r \end{bmatrix} \bar{c}_0(t)$$
(4.1.82)

Using (4.1.53) and (4.1.54), the third term of the right-hand side of (4.1.80) is

$$-M^{-1}\vec{b}_0\left(\dot{u}_{0_r} + \frac{1}{r}\dot{u}_0\right) = -M^{-1}\vec{b}_0\left[\bar{\mathbf{b}}_r(r)^T + \frac{1}{r}\bar{\mathbf{b}}(r)^T\right]\dot{\mu}_0(t)$$
(4.1.83)

Multiplying (4.1.83) by  $B(r)^T$  and integrating with respect to r, we have

$$- \left[ \int_{R_B}^{R_F} B(r)^T M^{-1} \vec{b}_0 \left( r \, \bar{\mathbf{b}}_r(r)^T + \, \bar{\mathbf{b}}(r)^T \right) \mathrm{d}r \right] \dot{\mu}_0(t)$$
(4.1.84)

From (4.1.11) and (4.1.52)

$$W_i(r,t) = \frac{R_F - r}{R_F - R_B} \bar{\nu}_{Bi}(t) + \frac{r - R_B}{R_F - R_B} \bar{\nu}_{Fi}, \quad i = 0, \dots, 2$$
(4.1.85)

and  $\bar{\nu}_{Bi}(t) = 0$ , i = 1, 2.

It follows that

$$\dot{W}_i(r,t) = \frac{R_F - r}{R_F - R_B} \dot{\nu}_{Bi}(t), \quad i = 0, \dots, 2$$
 (4.1.86)

and  $\dot{\bar{\nu}}_{Bi}(t) = 0$ , i = 1, 2.

Multiplying (4.1.86) by  $B(r)^T$  and integrating with respect to r, we have

$$\left[\int_{R_B}^{R_F} B(r)^T \frac{R_F - r}{R_F - R_B} r \,\mathrm{d}r\right] \dot{\nu}_{Bi}(t) \tag{4.1.87}$$

From (4.1.14) and (4.1.52)

$$W_{i_r}(r,t) = -\frac{1}{R_F - R_B}\bar{\nu}_{Bi}(t) + \frac{1}{R_F - R_B}\bar{\nu}_{Fi}, \quad i = 0, \dots, 2$$
(4.1.88)

and  $\bar{\nu}_{Bi}(t) = 0$ , i = 1, 2.

Multiplying  $\mathcal{A}\frac{1}{r}(rW_{i_r})_r$  by  $B(r)^T$  and integrating with respect to r, we have

$$\left[\int_{R_B}^{R_F} B(r)^T \mathcal{A} \frac{1}{R_F - R_B} \,\mathrm{d}r\right] \left(-\bar{\nu}_{Bi}(t) + \bar{\nu}_{Fi}\right), \quad i = 0, \dots, 2$$
(4.1.89)

and  $\bar{\nu}_{Bi}(t) = 0$ , i = 1, 2.

Define the following matrices.

$$3M_D \times 3M_D$$
 matrices:

$$G_0 = \int_{R_B}^{R_F} B(r)^T B(r) r \,\mathrm{d}r$$

$$G_2 = \mathcal{A} \otimes \left[ \int_{R_B}^{R_F} \bar{b}_r(r) \bar{b}_r(r)^T r \, \mathrm{d}r \right]$$

 $3M_D \times M$  matrices:

$$H_0 = \int_{R_B}^{R_F} B(r)^T M^{-1} \vec{b}_0 (r \, \bar{\mathbf{b}}_r(r) + \, \bar{\mathbf{b}}(r))^T \, \mathrm{d}r$$

 $3M_D \times 3$  matrices:

$$\mathcal{R}_B = \int_{R_B}^{R_F} B(r)^T \frac{R_F - r}{R_F - R_B} r \,\mathrm{d}r$$
$$\mathcal{R}_F = \int_{R_B}^{R_F} B(r)^T \frac{r - R_B}{R_F - R_B} r \,\mathrm{d}r$$
$$\mathcal{R} = \Big[\int_{R_B}^{R_F} \frac{1}{R_F - R_B} B(r)^T \,\mathrm{d}r\Big]\mathcal{A}$$

Using these matrices, (4.1.80) multiplied by  $B(r)^T$  and integrated with respect to r, has the form

$$G_0 \dot{\bar{c}}_0(t) + G_2 \bar{c}_0(t) = -H_0 \dot{\bar{\mu}}_0(t) - \mathcal{R}_B \dot{\bar{\nu}}_{B0}(t) - \mathcal{R}\bar{\nu}_{B0}(t) + \mathcal{R}\bar{\nu}_{F0}$$
(4.1.90)

From (4.1.75)

$$\dot{\bar{\mu}}_0(t) = \mathcal{M}_0^{-1} \Big[ \mathcal{Z}_0 \dot{\bar{c}}_0(t) + Y_{B0} \dot{\bar{\nu}}_{B0}(t) \Big]$$
(4.1.91)

Applying (4.1.91) to (4.1.90), we have

$$\mathcal{H}_0 \dot{\bar{c}}_0(t) + G_2 \bar{c}_0(t) = \mathcal{D}_{B0} \dot{\bar{\nu}}_{B0}(t) - \mathcal{R} \bar{\nu}_{B0}(t) + \mathcal{R} \bar{\nu}_{F0}$$
(4.1.92)

where  $\mathcal{H}_0 = G_0 + H_0 \mathcal{M}_0^{-1} \mathcal{Z}_0$ 

$$\mathcal{D}_{B0} = -H_0 \mathcal{M}_0^{-1} Y_{B0} - \mathcal{R}_B$$

Now we approximate initial conditions. From (4.1.11), (4.1.19) and (4.1.52), the coefficients  $V_i(r, 0)$  of  $\theta$ -approximation of initial conditions have the form

$$V_i(r,0) = \tilde{V}_{Ii}(r) - W_i(r,0)$$

where  $\tilde{V}_{Ii}(r) = B(r)\bar{\nu}_{Ii}$ 

$$W_i(r,0) = \frac{R_F - r}{R_F - R_B} \bar{\nu}_{Bi}(0) + \frac{r - R_B}{R_F - R_B} \bar{\nu}_{Fi}, \quad i = 0, \dots, 2$$
(4.1.93)

and  $\bar{\nu}_{Bi}(0) = 0$ , i = 1, 2.

Multiplying  $V_i(r, 0)$  by  $B(r)^T$  and integrating with respect to r, we have

$$\left[\int_{R_B}^{R_F} B(r)^T B(r) r \, \mathrm{d}r\right] \bar{c}_i(0) = \left[\int_{R_B}^{R_F} B(r)^T B(r) r \, \mathrm{d}r\right] \bar{\nu}_{Ii} - \left[\int_{R_B}^{R_F} B(r)^T \frac{R_F - r}{R_F - R_B} r \, \mathrm{d}r\right] \bar{\nu}_{Bi}(0) - \left[\int_{R_B}^{R_F} B(r)^T \frac{r - R_B}{R_F - R_B} r \, \mathrm{d}r\right] \bar{\nu}_{Fi}$$
(4.1.94)

Using matrices defined above, (4.1.94) has the form

$$G_0 \bar{c}_i(0) = G_0 \bar{\nu}_{Ii} - \mathcal{R}_B \bar{\nu}_{Bi}(0) - \mathcal{R}_F \bar{\nu}_{Fi}$$
(4.1.95)

From (4.1.92) and (4.1.95) with i = 0 we obtain the following system of ordinary differential equations (ODEs).

System of ODEs for TCP I (radially symmetric case):

$$\mathcal{H}_0 \dot{\bar{c}}_0(t) + G_2 \bar{c}_0(t) = \mathcal{D}_{B0} \dot{\bar{\nu}}_{B0}(t) - \mathcal{R} \bar{\nu}_{B0}(t) + \mathcal{R} \bar{\nu}_{F0}$$
(4.1.92)

with initial conditions

$$\bar{c}_0(0) = \bar{\nu}_{I0} - G_0^{-1} \left[ \mathcal{R}_B \bar{\nu}_{B0}(0) + \mathcal{R}_F \bar{\nu}_{F0} \right]$$
(4.1.96)

System II (corresponds to  $\sin 2\theta$  approximation)

System II is given by (4.1.20)-(4.1.22):

$$M\dot{V}_{1} - A\left[\frac{1}{r}(rV_{1_{r}})_{r} - \frac{4}{r^{2}}V_{1}\right] = -M\dot{W}_{1} + A\left(\frac{1}{r}(rW_{1_{r}})_{r} - \frac{4}{r^{2}}W_{1}\right)$$
$$-\vec{b}_{0}\left(\dot{u}_{1_{r}} + \frac{1}{r}(\dot{u}_{1} - 2\dot{v}_{2})\right)$$
(4.1.20)

with boundary conditions

$$V_1(r,t) = 0$$
 on  $\Gamma \times [0, t_f)$  (4.1.21)

and initial conditions

$$V_1(r,0) = \tilde{V}_{I1}(r) - W_1(r,0)$$
(4.1.22)

Using  $\mathcal{A} = M^{-1}A$ , (4.1.20) has the form

$$\dot{V}_{1} - \mathcal{A}\left[\frac{1}{r}(rV_{1_{r}})_{r} - \frac{4}{r^{2}}V_{1}\right] = -M^{-1}\vec{b}_{0}\left(\dot{u}_{1_{r}} + \frac{1}{r}\dot{u}_{1} - \frac{2}{r}\dot{v}_{2}\right)$$
$$-\dot{W}_{1} + \mathcal{A}\left[\frac{1}{r}(rW_{1_{r}})_{r} - \frac{4}{r^{2}}W_{1}\right]$$
(4.1.97)

Using (4.1.51), the left-hand side of (4.1.97) is

$$B(r)\dot{\bar{c}}_{1}(t) - \mathcal{A}\frac{1}{r} \Big( rB_{r}(r) \Big)_{r} \bar{c}_{1}(t) + \mathcal{A}\frac{4}{r^{2}} B(r) \bar{c}_{1}(t)$$
(4.1.98)

Multiplying (4.1.98) by  $B(r)^T$  and integrating with respect to r, we have

$$\left[\int_{R_B}^{R_F} B(r)^T B(r) r \,\mathrm{d}r\right] \dot{\bar{c}}_1(t) + \mathcal{A} \otimes \left\{ \left[\int_{R_B}^{R_F} \bar{\bar{b}}_r(r) \bar{\bar{b}}_r(r)^T r \,\mathrm{d}r\right] + 4 \left[\int_{R_B}^{R_F} \bar{\bar{b}}(r) \bar{\bar{b}}(r)^T \frac{1}{r} \,\mathrm{d}r\right] \right\} \bar{c}_1(t)$$
(4.1.99)

Using (4.1.53) and (4.1.54), the third term of the right-hand side of (4.1.97) is

$$-M^{-1}\vec{b}_0\left(\dot{u}_{1r} + \frac{1}{r}\dot{u}_1 - \frac{2}{r}\dot{v}_2\right)$$
  
=  $-M^{-1}\vec{b}_0\left[\bar{\mathbf{b}}_r(r)^T + \frac{1}{r}\bar{\mathbf{b}}(r)^T\right]\dot{\mu}_1(t) + M^{-1}\vec{b}_0\frac{2}{r}\bar{\mathbf{b}}(r)^T\dot{\eta}_2(t)$ 

Multiplying by  $B(r)^T$  and integrating with respect to r, we have

$$- \Big[ \int_{R_B}^{R_F} B(r)^T M^{-1} \vec{b}_0 (r \,\bar{\mathbf{b}}_r(r) + \bar{\mathbf{b}}(r))^T \,\mathrm{d}r \Big] \dot{\mu}_1(t) + 2 \Big[ \int_{R_B}^{R_F} B(r)^T M^{-1} \vec{b}_0 \,\bar{\mathbf{b}}(r)^T \,\mathrm{d}r \Big] \dot{\bar{\eta}}_2(t)$$
(4.1.100)

Multiplying the corresponding term in (4.1.97) by  $B(r)^T$  and integrating with respect to r, we have the following results for  $\dot{W}_1$ : (4.1.87) with  $\dot{\bar{\nu}}_{B1}(t) = 0$  and for  $\mathcal{A}\frac{1}{r}(rW_{1_r})_r$ : (4.1.89) with  $\bar{\nu}_{B1}(t) = 0$ .  $W_1(r,t)$  is given by (4.1.85) with  $\bar{\nu}_{B1}(t) = 0$ .

Multiplying  $\mathcal{A}\frac{4}{r^2}W_i$  by  $B(r)^T$  and integrating with respect to r, we have

$$\left[\int_{R_B}^{R_F} B(r)^T \mathcal{A} \frac{4}{r^2} \frac{r - R_B}{R_F - R_B} r \,\mathrm{d}r\right] \bar{\nu}_{Fi}$$
  
=4 $\left[\int_{R_B}^{R_F} B(r)^T \frac{r - R_B}{R_F - R_B} \frac{1}{r} \,\mathrm{d}r\right] \mathcal{A} \bar{\nu}_{Fi}, \qquad i = 1, 2.$  (4.1.101)

Define the following matrices.

 $3M_D \times 3M_D$  matrix:

$$G_0\left(\frac{1}{r}\right) = \mathcal{A} \otimes \left[\int_{R_B}^{R_F} \bar{b}(r)\bar{b}(r)^T \frac{1}{r} \,\mathrm{d}r\right]$$

 $3M_D \times M$  matrix:

$$H_0(r) = \int_{R_B}^{R_F} B(r)^T M^{-1} \vec{b}_0 \, \bar{\mathbf{b}}(r)^T \, \mathrm{d}r$$

$$3M_D \times 3$$
 matrix:

$$\mathcal{R}_F^{(a)}\Big(\frac{1}{r}\Big) = \Big[\int_{R_B}^{R_F} B(r)^T \frac{r - R_B}{R_F - R_B} \frac{1}{r} \,\mathrm{d}r\Big]\mathcal{A}$$

Using these matrices together with matrices defined for System I, (4.1.97) multiplied by  $B(r)^T$  and integrated with respect to r, has the form

$$G_{0}\dot{\bar{c}}_{1}(t) + \left[G_{2} + 4G_{0}\left(\frac{1}{r}\right)\right]\bar{c}_{1}(t) = -H_{0}\dot{\bar{\mu}}_{1}(t) + 2H_{0}(r)\dot{\bar{\eta}}_{2}(t) + \left[\mathcal{R} - 4\mathcal{R}_{F}^{(a)}\left(\frac{1}{r}\right)\right]\bar{\nu}_{F1}$$

$$(4.1.102)$$

From (4.1.76) and (4.1.77)

$$\dot{\bar{\mu}}_1(t) = P_{10} \mathcal{Q}^{-1} \bar{\mathcal{Z}} \dot{\bar{c}}_1(t) \tag{4.1.103}$$

$$\dot{\bar{\eta}}_2(t) = -P_{01}\mathcal{Q}^{-1}\bar{\mathcal{Z}}\dot{\bar{c}}_1(t)$$
(4.1.104)

Applying (4.1.103) and (4.1.104) to (4.1.102), we have

$$\mathcal{H}_{1}\dot{\bar{c}}_{1}(t) + \mathcal{G}_{1}\bar{c}_{1}(t) = \mathcal{F}_{F1}\bar{\nu}_{F1} \tag{4.1.105}$$

where

$$\mathcal{H}_{1} = G_{0} + \left(H_{0}P_{10} + 2H_{0}(r)P_{01}\right)\mathcal{Q}^{-1}\bar{\mathcal{Z}}$$
  
$$\mathcal{G}_{1} = G_{2} + 4G_{0}\left(\frac{1}{r}\right)$$
  
$$\mathcal{F}_{F1} = \left(\mathcal{R} - 4\mathcal{R}_{F}^{(a)}\left(\frac{1}{r}\right)\right)$$
(4.1.106)

Initial conditions are given by (4.1.95) with i = 1 and  $\bar{\nu}_{B1}(0) = 0$ . From (4.1.106) and (4.1.95) with i = 1 and  $\bar{\nu}_{B1}(0) = 0$  we obtain the following system.

System of ODEs for TCP II (corresponds to  $\sin 2\theta$  approximation):

$$\mathcal{H}_1 \dot{\bar{c}}_1(t) + \mathcal{G}_1 \bar{c}_1(t) = \mathcal{F}_{F1} \bar{\nu}_{F1} \tag{4.1.105}$$

with initial conditions

$$\bar{c}_1(0) = \bar{\nu}_{I1} - G_0^{-1} \mathcal{R}_F \bar{\nu}_{F1} \tag{4.1.107}$$

System III (corresponds to  $\cos 2\theta$  approximation)

System III is given by (4.1.23)-(4.1.25):

$$M\dot{V}_{2} - A\left[\frac{1}{r}(rV_{2r})_{r} - \frac{4}{r^{2}}V_{2}\right] = -M\dot{W}_{2} + A\left(\frac{1}{r}(rW_{2r})_{r} - \frac{4}{r^{2}}W_{2}\right)$$
$$-\vec{b}_{0}\left(\dot{u}_{2r} + \frac{1}{r}(\dot{u}_{2} + 2\dot{v}_{1})\right)$$
(4.1.23)

with boundary conditions

$$V_2(r,t) = 0 \qquad \text{on } \Gamma \times (0,t_f) \tag{4.1.24}$$

and initial conditions

$$V_2(r,0) = \tilde{V}_{I2}(r) - W_2(r,0) \tag{4.1.25}$$

Using  $\mathcal{A} = M^{-1}A$ , (4.1.23) has the form

$$\dot{V}_{2} - \mathcal{A}\left[\frac{1}{r}(rV_{2_{r}})_{r} - \frac{4}{r^{2}}V_{2}\right] = -M^{-1}\vec{b}_{0}\left(\dot{u}_{2_{r}} + \frac{1}{r}\dot{u}_{2} + \frac{2}{r}\dot{v}_{1}\right)$$
$$-\dot{W}_{2} + \mathcal{A}\left[\frac{1}{r}(rW_{2_{r}})_{r} - \frac{4}{r^{2}}W_{2}\right]$$
(4.1.108)

Using (4.1.51), the left-hand side of (4.1.108) is

$$B(r)\dot{\bar{c}}_{2}(t) - \mathcal{A}\frac{1}{r} \left( rB_{r}(r) \right)_{r} \bar{c}_{2}(t) + \mathcal{A}\frac{4}{r^{2}} B(r)\bar{c}_{2}(t)$$
(4.1.109)

Multiplying (4.1.109) by  $B(r)^T$  and integrating with respect to r, we have

$$\left[\int_{R_B}^{R_F} B(r)^T B(r) r \,\mathrm{d}r\right] \dot{\bar{c}}_2(t) + \mathcal{A} \otimes \left\{ \left[\int_{R_B}^{R_F} \bar{b}_r(r) \bar{b}_r(r)^T r \,\mathrm{d}r\right] + 4 \left[\int_{R_B}^{R_F} \bar{b}(r) \bar{b}(r)^T \frac{1}{r} \,\mathrm{d}r\right] \right\} \bar{c}_2(t)$$
(4.1.110)

Using (4.1.53) and (4.1.54), the first term of the right-hand side of (4.1.108) is

$$-M^{-1}\vec{b}_0\left(\dot{u}_{2r} + \frac{1}{r}\dot{u}_2 + \frac{2}{r}\dot{v}_1\right)$$
  
=  $-M^{-1}\vec{b}_0\left[\bar{\mathbf{b}}_r(r)^T + \frac{1}{r}\bar{\mathbf{b}}(r)^T\right]\dot{\mu}_2(t) - M^{-1}\vec{b}_0\frac{2}{r}\bar{\mathbf{b}}(r)^T\dot{\eta}_1(t)$ 

Multiplying by  $B(r)^T$  and integrating with respect to r, we have

$$- \Big[ \int_{R_B}^{R_F} B(r)^T M^{-1} \vec{b}_0 (r \,\bar{\mathbf{b}}_r(r) + \,\bar{\mathbf{b}}(r))^T \,\mathrm{d}r \Big] \dot{\mu}_2(t) -2 \Big[ \int_{R_B}^{R_F} B(r)^T M^{-1} \vec{b}_0 \,\bar{\mathbf{b}}(r)^T \,\mathrm{d}r \Big] \dot{\bar{\eta}}_1(t)$$
(4.1.111)

Multiplying the corresponding term in (4.1.108) by  $B(r)^T$  and integrating with respect to r, we have the following results

for 
$$\dot{W}_{2}$$
: (4.1.87) with  $\dot{\bar{\nu}}_{B2}(t) = 0$   
for  $\mathcal{A}\frac{1}{r}(rW_{2r})_{r}$ : (4.1.89) with  $\bar{\nu}_{B2}(t) = 0$ .  
for  $\mathcal{A}\frac{4}{r^{2}}W_{2}$ : (4.1.101)

Using the matrices defined for Systems I and II, (4.1.108) multiplied by  $B(r)^T$  and integrated with respect to r, has the form

$$G_0 \dot{\bar{c}}_2(t) + \left[G_2 + 4G_0\left(\frac{1}{r}\right)\right] \bar{c}_2(t) = -H_0 \dot{\bar{\mu}}_2(t) - 2H_0(r)\dot{\bar{\eta}}_1(t) + \left[\mathcal{R} - 4\mathcal{R}_F^{(a)}\left(\frac{1}{r}\right)\right] \bar{\nu}_{F2}$$

$$(4.1.112)$$

From (4.1.78) and (4.1.79)

$$\dot{\bar{\mu}}_2(t) = P_{10} \mathcal{Q}^{-1} \bar{\mathcal{Z}} \dot{\bar{c}}_2(t) \tag{4.1.113}$$

$$\dot{\bar{\eta}}_1(t) = P_{01} \mathcal{Q}^{-1} \bar{\mathcal{Z}} \dot{\bar{c}}_2(t)$$
 (4.1.114)

Applying (4.1.113) and (4.1.114) to (4.1.112), we have

$$\mathcal{H}_1 \dot{\bar{c}}_2(t) + \mathcal{G}_1 \bar{c}_2(t) = \mathcal{F}_{F1} \bar{\nu}_{F2} \tag{4.1.115}$$

where the matrices  $\mathcal{H}_1$ ,  $\mathcal{G}_1$ , and  $\mathcal{F}_{F1}$  are given by (4.1.106)

Initial conditions are given by (4.1.95) with i = 2 and  $\bar{\nu}_{B2}(0) = 0$ . From (4.1.115) and (4.1.95) with i = 2 and  $\bar{\nu}_{B2}(0) = 0$  we obtain the following system.

System of ODEs for TCP III (corresponds to  $\cos 2\theta$  approximation):

$$\mathcal{H}_1 \dot{\bar{c}}_2(t) + \mathcal{G}_1 \bar{c}_2(t) = \mathcal{F}_{F1} \bar{\nu}_{F2} \tag{4.1.115}$$

with initial conditions

$$\bar{c}_2(0) = \bar{\nu}_{I2} - G_0^{-1} \mathcal{R}_F \bar{\nu}_{F2} \tag{4.1.116}$$

We have the following three systems of ODEs:

System of ODEs for TCP I (radially symmetric case):

$$\mathcal{H}_0 \dot{\bar{c}}_0(t) + G_2 \bar{c}_0(t) = \mathcal{D}_{B0} \dot{\bar{\nu}}_{B0}(t) - \mathcal{R} \bar{\nu}_{B0}(t) + \mathcal{R} \bar{\nu}_{F0} \tag{4.1.92}$$

with initial conditions

$$\bar{c}_0(0) = \bar{\nu}_{I0} - G_0^{-1} \left[ \mathcal{R}_B \bar{\nu}_{B0}(0) + \mathcal{R}_F \bar{\nu}_{F0} \right]$$
(4.1.96)

System I gives coefficients  $V_0$  and  $u_0$  of the  $\theta$ -approximation (4.1.9).

System of ODEs for TCP II (corresponds to  $\sin 2\theta$  approximation):

$$\mathcal{H}_{1}\dot{\bar{c}}_{1}(t) + \mathcal{G}_{1}\bar{c}_{1}(t) = \mathcal{F}_{F1}\bar{\nu}_{F1} \tag{4.1.105}$$

with initial conditions

$$\bar{c}_1(0) = \bar{\nu}_{I1} - G_0^{-1} \mathcal{R}_F \bar{\nu}_{F1} \tag{4.1.107}$$

System II gives  $\sin 2\theta$ -coefficients  $V_1$ ,  $u_1$  and  $\cos 2\theta$ -coefficient  $v_2$  of the  $\theta$ -approximation (4.1.9).

System of ODEs for TCP III (corresponds to  $\cos 2\theta$  approximation):

$$\mathcal{H}_{1}\dot{\bar{c}}_{2}(t) + \mathcal{G}_{1}\bar{c}_{2}(t) = \mathcal{F}_{F1}\bar{\nu}_{F2} \tag{4.1.115}$$

with initial conditions

$$\bar{c}_2(0) = \bar{\nu}_{I2} - G_0^{-1} \mathcal{R}_F \bar{\nu}_{F2} \tag{4.1.116}$$

System III gives  $\sin 2\theta$ -coefficient  $v_1$  and  $\cos 2\theta$ -coefficients  $V_2$ ,  $u_2$  of the  $\theta$ -approximation (4.1.9).

#### 4.2 3D TCPu Problem for an Inclined Borehole

As in Section 2.2, let (X, Y, Z) be a borehole Cartesian coordinate system with the x-axis in the direction of the maximum horizontal principal stress  $\sigma_H$ , the y-axis in the direction of the minimum horizontal principal stress  $\sigma_h$ , and the z-axis in the direction of overburden stress  $\sigma_V$  and let  $(X_b, Y_b, Z_b)$  be the borehole Cartesian coordinate system. For the sake of convenience, we will make the transformation from the coordinate system  $(X_b, Y_b, Z_b)$  to the borehole cylindrical coordinate system  $(r, \theta, z)$  with the z-axis passing through the central axis of the cylindrical region  $\Omega$ .

Then the boundary  $\Gamma$  of  $\Omega$  is  $\Gamma = \Gamma_B \cup \Gamma_F \cup \tilde{\Gamma}_T \cup \tilde{\Gamma}_B$ :

$$\Gamma_B = \left\{ (R_B, \theta, z) \in \mathbb{R}^3 : 0 \le \theta < 2\pi, z_T \le z \le z_B \right\}$$
  
$$\Gamma_F = \left\{ (R_F, \theta, z) \in \mathbb{R}^3 : 0 \le \theta < 2\pi, z_T \le z \le z_B \right\}$$
  
$$\tilde{\Gamma}_T = \left\{ (r, \theta, z_T) \in \mathbb{R}^3 : R_B \le r \le R_F, 0 \le \theta < 2\pi \right\}$$
  
$$\tilde{\Gamma}_B = \left\{ (r, \theta, z_B) \in \mathbb{R}^3 : R_B \le r \le R_F, 0 \le \theta < 2\pi \right\}$$

# 4.2.1 Fourier-type approximation of the TCP sytem

In the borehole cylindrical coordinate system  $(r, \theta, z)$  the TCP initial-boundary value problem (2.2.17)-(2.2.20) is given by

.

$$M\bar{V} - A\nabla^2 \bar{V} = -\vec{b}_0 (\nabla \cdot \dot{\vec{u}}) \qquad \text{in } \Omega \times (0, t_f)$$
(4.2.1)

with boundary conditions

$$\bar{V}(r,\theta,z,t) = \begin{cases} V_B(z,t) & \text{on } \Gamma_B \times [0,t_f) \\ V_F(\theta,z) & \text{on } \Gamma_F \times [0,t_f) \\ \tilde{V}_T(r,\theta,t) & \text{on } \tilde{\Gamma}_T \times [0,t_f) \\ \tilde{V}_B(r,\theta,t) & \text{on } \tilde{\Gamma}_B \times [0,t_f) \end{cases}$$
(4.2.2)

initial conditions

$$\bar{V}(r,\theta,z,0) = \tilde{V}_I(r,\theta,z) \tag{4.2.3}$$

and compatibility conditions:

$$\tilde{V}_T(R_B, \theta, t) = V_B(z_T, t) \tag{4.2.4a}$$

$$\tilde{V}_T(R_F, \theta, t) = V_F(\theta, z_T)$$
(4.2.4b)

$$\tilde{V}_B(R_B, \theta, t) = V_B(z_B, t) \tag{4.2.4c}$$

$$\tilde{V}_B(R_F, \theta, t) = V_F(\theta, z_B) \tag{4.2.4d}$$

where the displacement vector is

$$\vec{u}(r,\theta,z,t) = u(r,\theta,z,t)\vec{i}_r + v(r,\theta,z,t)\vec{i}_\theta + w(r,\theta,z,t)\vec{i}_z$$

To obtain the equivalent parabolic TCP-type system with homogeneous boundary conditions (2.3.11)-(2.3.13), we first define

$$W^{r}(r,\theta,z,t) = \frac{R_{F} - r}{R_{F} - R_{B}} V_{B}(z,t) + \frac{r - R_{B}}{R_{F} - R_{B}} V_{F}(\theta,z)$$
(4.2.5)

$$\bar{V}^0 = \bar{V} - W^r \tag{4.2.6}$$

Then (4.2.1)-(4.2.3) has the form

$$M\dot{\bar{V}}^{0} - A\nabla^{2}\bar{V}^{0} = -M\dot{W}^{r} + A\nabla^{2}W^{r} - \vec{b}_{0}(\nabla \cdot \dot{\vec{u}}) \quad \text{in } \Omega \times (0, t_{f}) \quad (4.2.7)$$

with boundary conditions

$$\bar{V}^{0}(r,\theta,z,t) = \begin{cases} 0 & \text{on } \Gamma_{B} \times [0,t_{f}) \\ 0 & \text{on } \Gamma_{F} \times [0,t_{f}) \\ \tilde{V}_{T}(r,\theta,t) - W^{r}(r,\theta,z_{T},t) & \text{on } \tilde{\Gamma}_{T} \times [0,t_{f}) \\ \tilde{V}_{B}(r,\theta,t) - W^{r}(r,\theta,z_{B},t) & \text{on } \tilde{\Gamma}_{B} \times [0,t_{f}) \end{cases}$$
(4.2.8)

and initial conditions

$$\bar{V}^{0}(r,\theta,z,0) = \tilde{V}_{I}(r,\theta,z) - W^{r}(r,\theta,z,0)$$
(4.2.9)

Now define

$$\hat{V}_T(r,\theta,t) = \tilde{V}_T(r,\theta,t) - W^r(r,\theta,z_T,t)$$
(4.2.10)

$$\hat{V}_B(r,\theta,t) = \tilde{V}_B(r,\theta,t) - W^r(r,\theta,z_B,t)$$
(4.2.11)

and

$$W^{z}(r,\theta,z,t) = \frac{z - z_{T}}{z_{B} - z_{T}} \hat{V}_{B}(r,\theta,t) + \frac{z_{B} - z}{z_{B} - z_{T}} \hat{V}_{T}(r,\theta,t)$$
(4.2.12)

$$V = \bar{V}^0 - W^z = \bar{V} - W^r - W^z$$
(4.2.13)

$$V = \bar{V} - W(r, \theta, z, t),$$
 where  $W = W^r + W^z$  (4.2.14)

Then (4.2.1)-(4.2.3) has the form

$$M\dot{V} - A\nabla^2 V = -M\dot{W} + A\nabla^2 W - \vec{b}_0(\nabla \cdot \dot{\vec{u}}) \quad \text{in } \Omega \times (0, t_f) \quad (4.2.15)$$

with boundary conditions

$$V(r,\theta,z,t) = 0 \qquad \text{on } \Gamma \times [0,t_f) \tag{4.2.16}$$

and initial conditions

$$V(r, \theta, z, 0) = \tilde{V}_I(r, \theta, z) - W(r, \theta, z, 0)$$
(4.2.17)

We introduce the following  $\theta$  - approximation:

$$V(r, \theta, z, t) = V_0(r, z, t) + V_1(r, z, t) \sin \theta + V_2(r, z, t) \cos \theta$$
$$+ V_3(r, z, t) \sin 2\theta + V_4(r, z, t) \cos 2\theta$$
$$u(r, \theta, z, t) = u_0(r, z, t) + u_1(r, z, t) \sin \theta + u_2(r, z, t) \cos \theta$$
$$+ u_3(r, z, t) \sin 2\theta + u_4(r, z, t) \cos 2\theta$$
$$v(r, \theta, z, t) = v_0(r, z, t) + v_1(r, z, t) \sin \theta + v_2(r, z, t) \cos \theta$$
$$+ v_3(r, z, t) \sin 2\theta + v_4(r, z, t) \cos 2\theta$$

$$w(r,\theta,z,t) = w_0(r,z,t) + w_1(r,z,t)\sin\theta + w_2(r,z,t)\cos\theta$$
$$+ w_3(r,z,t)\sin 2\theta + w_4(r,z,t)\cos 2\theta$$

$$V_B(z,t) = V_{B0}(z,t)$$
$$V_F(\theta,z) = V_{F0}(z) + V_{F1}(z)\sin\theta + V_{F2}(z)\cos\theta$$
$$+ V_{F3}(z)\sin 2\theta + V_{F4}(z)\cos 2\theta$$

$$\tilde{V}_{T}(r,\theta,t) = \tilde{V}_{T0}(r,t) + \tilde{V}_{T1}(r,t)\sin\theta + \tilde{V}_{T2}(r,t)\cos\theta$$

$$+ \tilde{V}_{T3}(r,t)\sin2\theta + \tilde{V}_{T4}(r,t)\cos2\theta$$

$$\tilde{V}_{B}(r,\theta,t) = \tilde{V}_{B0}(r,t) + \tilde{V}_{B1}(r,t)\sin\theta + \tilde{V}_{B2}(r,t)\cos\theta$$

$$+ \tilde{V}_{B3}(r,t)\sin2\theta + \tilde{V}_{B4}(r,t)\cos2\theta$$

$$\tilde{V}_{I}(r,\theta,z) = \tilde{V}_{I0}(r,z) + \tilde{V}_{I1}(r,z)\sin\theta + \tilde{V}_{I2}(r,z)\cos\theta$$

$$+ \tilde{V}_{I3}(r,z)\sin2\theta + \tilde{V}_{I4}(r,z)\cos2\theta \qquad (4.2.18)$$

Then

$$W(r, \theta, z, t) = W_0(r, z, t) + W_1(r, z, t) \sin \theta + W_2(r, z, t) \cos \theta + W_3(r, z, t) \sin 2\theta + W_4(r, z, t) \cos 2\theta$$
(4.2.19)

where

$$W_{i}(r, z, t) = \frac{R_{F} - r}{R_{F} - R_{B}} V_{Bi}(z, t) + \frac{r - R_{B}}{R_{F} - R_{B}} V_{Fi}(z) + \frac{z - z_{T}}{z_{B} - z_{T}} \Big[ \tilde{V}_{Bi}(r, t) - \frac{R_{F} - r}{R_{F} - R_{B}} V_{Bi}(z_{B}, t) - \frac{r - R_{B}}{R_{F} - R_{B}} V_{Fi}(z_{B}) \Big] + \frac{z_{B} - z}{z_{B} - z_{T}} \Big[ \tilde{V}_{Ti}(r, t) - \frac{R_{F} - r}{R_{F} - R_{B}} V_{Bi}(z_{T}, t) - \frac{r - R_{B}}{R_{F} - R_{B}} V_{Fi}(z_{T}) \Big], \qquad i = 0, \dots, 4$$

$$(4.2.20)$$

and  $V_{Bi}(z,t) = 0, \quad z_T \le z \le z_B, \quad i = 1, ..., 4.$ 

With this approximation, the system (4.2.15) has the form:

$$M\dot{V} - A\left[\frac{1}{r}(rV_{r})_{r} + \frac{1}{r^{2}}V_{\theta\theta} + V_{zz}\right] = -M\dot{W} + A\left(\frac{1}{r}(rW_{r})_{r} + \frac{1}{r^{2}}W_{\theta\theta} + W_{zz}\right)$$
$$-\vec{b}_{0}\left(\dot{u}_{r} + \frac{1}{r}(\dot{u} + \dot{v}_{\theta}) + \dot{w}_{z}\right)$$
(4.2.21)

where W is given by (4.2.19) and (4.2.20),

$$W_{r}(r,\theta,z,t) = W_{0_{r}}(r,z,t) + W_{1_{r}}(r,z,t)\sin\theta + W_{2_{r}}(r,z,t)\cos\theta + W_{3_{r}}(r,z,t)\sin2\theta + W_{4_{r}}(r,z,t)\cos2\theta$$
(4.2.22)

with

$$W_{i_r}(r, z, t) = \frac{1}{R_F - R_B} (V_{Fi}(z) - V_{Bi}(z, t)) + \frac{z - z_T}{z_B - z_T} \Big[ \tilde{V}_{Bi_r}(r, t) + \frac{1}{R_F - R_B} (V_{Bi}(z_B, t) - V_{Fi}(z_B)) \Big] + \frac{z_B - z}{z_B - z_T} \Big[ \tilde{V}_{Ti_r}(r, t) + \frac{1}{R_F - R_B} (V_{Bi}(z_T, t) - V_{Fi}(z_T)) \Big], \qquad i = 0, \dots, 4 \quad (4.2.23)$$

and

 $V_{Bi}(z,t) = 0, \qquad z_T \le z \le z_B, \qquad i = 1, \dots, 4;$ 

$$W_{\theta\theta}(r,\theta,z,t) = -W_1(r,z,t)\sin\theta - W_2(r,z,t)\cos\theta - 4W_3(r,z,t)\sin2\theta - 4W_4(r,z,t)\cos2\theta$$
(4.2.24)  
$$W_z(r,\theta,z,t) = W_{0_z}(r,z,t) + W_{1_z}(r,z,t)\sin\theta + W_{2_z}(r,z,t)\cos\theta + W_{3_z}(r,z,t)\sin2\theta + W_{4_z}(r,z,t)\cos2\theta$$
(4.2.25)

(4.2.25)

with

$$W_{i_{z}}(r, z, t) = \frac{R_{F} - r}{R_{F} - R_{B}} V_{Bi_{z}}(z, t) + \frac{r - R_{B}}{R_{F} - R_{B}} V_{Fi_{z}}(z) + \frac{1}{z_{B} - z_{T}} \Big[ \tilde{V}_{Bi}(r, t) - \tilde{V}_{Ti}(r, t) - \frac{R_{F} - r}{R_{F} - R_{B}} \big( V_{Bi}(z_{B}, t) - V_{Bi}(z_{T}, t) \big) - \frac{r - R_{B}}{R_{F} - R_{B}} \big( V_{Fi}(z_{B}) - V_{Fi}(z_{T}) \big) \Big], \qquad i = 0, \dots, 4$$
(4.2.26)

 $V_{Bi_z}(z,t) = 0, \qquad z_T \le z \le z_B, \qquad i = 1, \dots, 4.$ and

The  $\theta\text{-approximation}$  for the initial conditions  $~V(r,\theta,z,0)~(4.2.17)$  is

$$V(r, \theta, z, 0) = V_0(r, z, 0) + V_1(r, z, 0) \sin \theta + V_2(r, z, 0) \cos \theta$$
$$+ V_3(r, z, 0) \sin 2\theta + V_4(r, z, 0) \cos 2\theta$$

where

$$V_i(r, z, 0) = \tilde{V}_{Ii}(r, z) - W_i(r, z, 0) =$$

$$= \tilde{V}_{Ii}(r,z) - \frac{R_F - r}{R_F - R_B} V_{Bi}(z,0) - \frac{r - R_B}{R_F - R_B} V_{Fi}(z) - \frac{z - z_T}{z_B - z_T} \left( \tilde{V}_{Bi}(r,0) - \frac{R_F - r}{R_F - R_B} V_{Bi}(z_B,0) - \frac{r - R_B}{R_F - R_B} V_{Fi}(z_B) \right) - \frac{z_B - z}{z_B - z_T} \left( \tilde{V}_{Ti}(r,0) - \frac{R_F - r}{R_F - R_B} V_{Bi}(z_T,0) - \frac{r - R_B}{R_F - R_B} V_{Fi}(z_T) \right), \qquad i = 0, \dots, 4$$

$$(4.2.27)$$

and  $V_{Bi}(z,0) = 0$ ,  $z_T \le z \le z_B$ , i = 1, ..., 4.

Applying (4.2.22)-(4.2.27) to (4.2.21), multiplying by  $1, \sin \theta, \cos \theta, \sin 2\theta, \text{and } \cos 2\theta$ , and integrating over  $(0, 2\pi)$ , we obtain five systems.

### System I:

$$M\dot{V}_{0} - A\left[\frac{1}{r}(rV_{0_{r}})_{r} + (V_{0_{z}})_{z}\right] = -M\dot{W}_{0} + A\left(\frac{1}{r}(rW_{0_{r}})_{r} + (W_{0_{z}})_{z}\right)$$
$$-\vec{b}_{0}\left(\dot{u}_{0_{r}} + \frac{1}{r}\dot{u}_{0} + \dot{w}_{0_{z}}\right)$$
(4.2.28)

with boundary conditions

$$V_0(r, z, t) = 0$$
 on  $\Gamma \times [0, t_f)$  (4.2.29)

and initial conditions

$$V_0(r, z, 0) = \tilde{V}_{I0}(r, z) - W_0(r, z, 0)$$
(4.2.30)

# System II:

$$M\dot{V}_{1} - A\left[\frac{1}{r}(rV_{1_{r}})_{r} - \frac{1}{r^{2}}V_{1} + (V_{1_{z}})_{z}\right] = -M\dot{W}_{1} + A\left(\frac{1}{r}(rW_{1_{r}})_{r} - \frac{1}{r^{2}}W_{1} + (W_{1_{z}})_{z}\right)$$
$$-\vec{b}_{0}\left(\dot{u}_{1_{r}} + \frac{1}{r}(\dot{u}_{1} - \dot{v}_{2}) + \dot{w}_{1_{z}}\right)$$
(4.2.31)

with boundary conditions

$$V_1(r, z, t) = 0$$
 on  $\Gamma \times [0, t_f)$  (4.2.32)

and initial conditions

$$V_1(r, z, 0) = \tilde{V}_{I1}(r, z) - W_1(r, z, 0)$$
(4.2.33)

System III:

$$M\dot{V}_{2} - A\left[\frac{1}{r}(rV_{2_{r}})_{r} - \frac{1}{r^{2}}V_{2} + (V_{2_{z}})_{z}\right] = -M\dot{W}_{2} + A\left(\frac{1}{r}(rW_{2_{r}})_{r} - \frac{1}{r^{2}}W_{2} + (W_{2_{z}})_{z}\right)$$
$$-\vec{b}_{0}\left(\dot{u}_{2_{r}} + \frac{1}{r}(\dot{u}_{2} + \dot{v}_{1}) + \dot{w}_{2_{z}}\right)$$
(4.2.34)

with boundary conditions

$$V_2(r, z, t) = 0$$
 on  $\Gamma \times [0, t_f)$  (4.2.35)

and initial conditions

$$V_2(r, z, 0) = \tilde{V}_{I2}(r, z) - W_2(r, z, 0)$$
(4.2.36)

System IV:

$$M\dot{V}_{3} - A\left[\frac{1}{r}(rV_{3r})_{r} - \frac{4}{r^{2}}V_{3} + (V_{3z})_{z}\right] = -M\dot{W}_{3} + A\left(\frac{1}{r}(rW_{3r})_{r} - \frac{4}{r^{2}}W_{3} + (W_{3z})_{z}\right)$$
$$-\vec{b}_{0}\left(\dot{u}_{3r} + \frac{1}{r}(\dot{u}_{3} - 2\dot{v}_{4}) + \dot{w}_{3z}\right)$$
(4.2.37)

with boundary conditions

$$V_3(r, z, t) = 0$$
 on  $\Gamma \times [0, t_f)$  (4.2.38)

and initial conditions

$$V_3(r, z, 0) = \tilde{V}_{I3}(r, z) - W_3(r, z, 0)$$
(4.2.39)

System V:

$$M\dot{V}_{4} - A\left[\frac{1}{r}(rV_{4r})_{r} - \frac{4}{r^{2}}V_{4} + (V_{4z})_{z}\right] = -M\dot{W}_{4} + A\left(\frac{1}{r}(rW_{4r})_{r} - \frac{4}{r^{2}}W_{4} + (W_{4z})_{z}\right)$$
$$-\vec{b}_{0}\left(\dot{u}_{4r} + \frac{1}{r}(\dot{u}_{4} + 2\dot{v}_{3}) + \dot{w}_{4z}\right)$$
(4.2.40)

with boundary conditions

$$V_4(r, z, t) = 0$$
 on  $\Gamma \times [0, t_f)$  (4.2.41)

and initial conditions

$$V_4(r,z,0) = \tilde{V}_{I4}(r,z) - W_4(r,z,0)$$
(4.2.42)

### 4.2.2 Fourier-type approximation of the elastic system

In the borehole cylindrical coordinate system  $(r, \theta, z)$ , the displacement is

$$\begin{split} \vec{u}(r,\theta,z,t) &= u(r,\theta,z,t)\vec{i}_r + v(r,\theta,z,t)\vec{i}_\theta + w(r,\theta,z,t)\vec{i}_z \\ \vec{u} &\approx 0 \quad \text{and} \quad \dot{\vec{u}} \approx 0 \quad \text{on} \ \Gamma_F \end{split}$$

and the strain tensor has the form

$$\begin{bmatrix} \varepsilon_{rr} & \varepsilon_{r\theta} & \varepsilon_{rz} \\ \varepsilon_{\theta r} & \varepsilon_{\theta \theta} & \varepsilon_{\theta z} \\ \varepsilon_{zr} & \varepsilon_{z\theta} & \varepsilon_{zz} \end{bmatrix}$$

where

$$\varepsilon_{rr} = u_r, \qquad \varepsilon_{r\theta} = \varepsilon_{\theta r} = \frac{1}{2} \left( v_r + \frac{u_{\theta} - v}{r} \right), \qquad \varepsilon_{\theta \theta} = \frac{u + v_{\theta}}{r}$$

$$\varepsilon_{rz} = \varepsilon_{zr} = \frac{1}{2} (u_z + w_r), \qquad \varepsilon_{\theta z} = \varepsilon_{z\theta} = \frac{1}{2} (\frac{w_\theta}{r} + v_z), \qquad \varepsilon_{zz} = w_z \qquad (4.2.43)$$

The stress-strain relations for isotropic homogeneous medium in terms of the shear modulus G and the bulk modulus K are

$$\tau = 2G\varepsilon + \left(K - \frac{2}{3}G\right)(\mathrm{tr}\varepsilon)I \tag{2.2.23}$$

where

$$\tau_{rr} = 2Gu_r + \left(K - \frac{2}{3}G\right)\left(u_r + \frac{u + v_\theta}{r} + w_z\right)$$

$$\tau_{\theta\theta} = 2G \frac{u + v_{\theta}}{r} + \left(K - \frac{2}{3}G\right) \left(u_r + \frac{u + v_{\theta}}{r} + w_z\right)$$
  

$$\tau_{zz} = 2Gw_z + \left(K - \frac{2}{3}G\right) \left(u_r + \frac{u + v_{\theta}}{r} + w_z\right)$$
  

$$\tau_{r\theta} = \tau_{\theta r} = G \left(v_r + \frac{u_{\theta} - v}{r}\right)$$
  

$$\tau_{rz} = \tau_{zr} = G \left(u_z + w_r\right)$$
  

$$\tau_{\theta z} = \tau_{z\theta} = G \left(\frac{w_{\theta}}{r} + v_z\right)$$
(4.2.44)

The sum of total potential energy and penalty term is

$$\Pi(\vec{u}) = \frac{1}{2} \int_{\Omega} \left( \tau_{rr}(\vec{u}) \varepsilon_{rr}(\vec{u}) + 2\tau_{r\theta}(\vec{u}) \varepsilon_{r\theta}(\vec{u}) + 2\tau_{rz}(\vec{u}) \varepsilon_{rz}(\vec{u}) \right)$$
$$+ \tau_{\theta\theta}(\vec{u}) \varepsilon_{\theta\theta}(\vec{u}) + 2\tau_{\theta z}(\vec{u}) \varepsilon_{\theta z}(\vec{u}) + 2\tau_{zz}(\vec{u}) \varepsilon_{zz}(\vec{u}) \right) d\Omega$$
$$- \int_{\Omega} f\left( \operatorname{tr}\varepsilon(\vec{u}) \right) d\Omega - \int_{\Gamma} (\hat{\sigma}\vec{n}) \cdot \vec{u} \, d\Gamma + K_F \int_{\Gamma_F} \vec{u} \cdot \vec{u} \, d\Gamma \qquad (4.2.45)$$
$$\operatorname{tre} \left[ \vec{f} - \vec{h}_T^T \vec{V} = \vec{h}_T^T (V + W) \text{ is body force } \vec{h}_T^T = [\tilde{c} - \chi - \tilde{c}] \right]$$

where  $f = \vec{b}_1^T \vec{V} = \vec{b}_1^T (V + W)$  is body force,  $b_1^T = [\tilde{\zeta} -\lambda \quad \tilde{\alpha}]$ ,  $\hat{\sigma} = \begin{bmatrix} \sigma_r & \sigma_{r\theta} & \sigma_{rz} \\ \sigma_{r\theta} & \sigma_{\theta} & \sigma_{\theta z} \\ \sigma_{rz} & \sigma_{\theta z} & \sigma_z \end{bmatrix}$  is applied boundary stress in the system  $(r, \theta, Z_b)$ , and

 $K_F$  is a penalization constant.

Next, we describe stress around the borehole. Let

$$\sigma_{XYZ}(z) = \begin{bmatrix} \sigma_H(z) & 0 & 0\\ 0 & \sigma_h(z) & 0\\ 0 & 0 & \sigma_V \end{bmatrix}, \qquad \sigma_{XYZ}^b(z) = \begin{bmatrix} \sigma_x & \sigma_{xy} & \sigma_{xz}\\ \sigma_{xy} & \sigma_y & \sigma_{yz}\\ \sigma_{xz} & \sigma_{yz} & \sigma_{zz} \end{bmatrix} (z)$$
$$\sigma(r, \theta, z) = \begin{bmatrix} \sigma_r & \sigma_{r\theta} & \sigma_{rz}\\ \sigma_{r\theta} & \sigma_{\theta} & \sigma_{\theta z}\\ \sigma_{rz} & \sigma_{\theta z} & \sigma_z \end{bmatrix} (r, \theta, z)$$

be the stress tensors in the (X, Y, Z),  $(X_b, Y_b, Z_b)$ , and  $(r, \theta, z)$  coordinate systems, respectively.

The transformation from  $\sigma_{XYZ}(z)$  to  $\sigma^b_{XYZ}(z)$  is given by [22]:

$$\sigma^b_{XYZ}(z) = L_{\sigma} \cdot \sigma_{XYZ}(z) \cdot L_{\sigma}^T$$

where

$$L_{\sigma} = \begin{bmatrix} l_{11} & l_{12} & l_{13} \\ l_{21} & l_{22} & l_{23} \\ l_{31} & l_{32} & l_{33} \end{bmatrix} = \begin{bmatrix} \cos\varphi_x \cos\varphi_z & \sin\varphi_x \cos\varphi_z & -\sin\varphi_z \\ -\sin\varphi_x & \cos\varphi_x & 0 \\ \cos\varphi_x \sin\varphi_z & \sin\varphi_x \sin\varphi_z & \cos\varphi_z \end{bmatrix}$$

is the transformation matrix,  $\varphi_x$  is the horizontal angle between the x-axis and the borehole axis measures counterclockwise, and  $\varphi_z$  is the vertical angle between the z-axis and the borehole axis measures clockwise.

Equivalently,

$$\begin{bmatrix} \sigma_x \\ \sigma_y \\ \sigma_{zz} \\ \sigma_{xy} \\ \sigma_{yz} \\ \sigma_{xz} \end{bmatrix} (z) = T_{\sigma} \begin{bmatrix} \sigma_H(z) \\ \sigma_h(z) \\ \sigma_V \end{bmatrix} (z), \quad \text{where } T_{\sigma} = \begin{bmatrix} l_{11}^2 & l_{12}^2 & l_{13}^2 \\ l_{21}^2 & l_{22}^2 & l_{23}^2 \\ l_{31}^2 & l_{32}^2 & l_{33}^2 \\ l_{11}l_{12} & l_{12}l_{22} & l_{13}l_{23} \\ l_{21}l_{31} & l_{22}l_{32} & l_{23}l_{33} \\ l_{11}l_{31} & l_{12}l_{32} & l_{13}l_{33} \end{bmatrix}$$

The transformation from  $\sigma^b_{XYZ}(z)$  to  $\sigma(r, \theta, z)$  is given by [9]:

$$\sigma_{r} = \frac{\sigma_{x} + \sigma_{y}}{2} \left( 1 - \frac{R_{B}^{2}}{r^{2}} \right) + p_{B} \frac{R_{B}^{2}}{r^{2}} + \sigma_{xy} \left( 1 - \frac{4R_{B}^{2}}{r^{2}} + \frac{3R_{B}^{4}}{r^{4}} \right) \sin 2\theta + \frac{\sigma_{x} - \sigma_{y}}{2} \left( 1 - \frac{4R_{B}^{2}}{r^{2}} + \frac{3R_{B}^{4}}{r^{4}} \right) \cos 2\theta \sigma_{\theta} = \frac{\sigma_{x} + \sigma_{y}}{2} \left( 1 + \frac{R_{B}^{2}}{r^{2}} \right) - p_{B} \frac{R_{B}^{2}}{r^{2}} - \sigma_{xy} \left( 1 + \frac{3R_{B}^{4}}{r^{4}} \right) \sin 2\theta - \frac{\sigma_{x} - \sigma_{y}}{2} \left( 1 + \frac{3R_{B}^{4}}{r^{4}} \right) \cos 2\theta \sigma_{z} = \sigma_{zz} - 4\nu\sigma_{xy} \frac{R_{B}^{2}}{r^{2}} \sin 2\theta - 2\nu(\sigma_{x} - \sigma_{y}) \frac{R_{B}^{2}}{r^{2}} \cos 2\theta \sigma_{r\theta} = \frac{\sigma_{x} - \sigma_{y}}{2} \left( 1 + \frac{2R_{B}^{2}}{r^{2}} - \frac{3R_{B}^{4}}{r^{4}} \right) \sin 2\theta + \sigma_{xy} \left( 1 + \frac{2R_{B}^{2}}{r^{2}} - \frac{3R_{B}^{4}}{r^{4}} \right) \cos 2\theta$$

$$\sigma_{\theta z} = \left(-\sigma_{xz}\sin\theta + \sigma_{yz}\cos\theta\right) \left(1 + \frac{R_B^2}{r^2}\right)$$
$$\sigma_{rz} = \left(\sigma_{yz}\sin\theta + \sigma_{xz}\cos\theta\right) \left(1 - \frac{R_B^2}{r^2}\right)$$

where  $p_B$  is well bore pressure and  $\nu$  is Poisson's ratio of the formation.

From these equations we obtain applied stress on the boundaries.

On the borehole boundary  $\Gamma_B$ :

$$\sigma_r(R_B, z, t) = p_B(z, t)$$

$$\sigma_\theta(R_B, \theta, z, t) = \sigma_x(z) + \sigma_y(z) - p_B(z, t) - 4\sigma_{xy}(z)\sin 2\theta - 2(\sigma_x(z) - \sigma_y(z))\cos 2\theta$$

$$\sigma_z(R_B, \theta, z) = \sigma_{zz}(z) - 4\nu\sigma_{xy}(z)\sin 2\theta - 2\nu(\sigma_x(z) - \sigma_y(z))\cos 2\theta$$

$$\sigma_{r\theta}(R_B, \theta, z) = 0$$

$$\sigma_{rz}(R_B, \theta, z) = 0$$

$$\sigma_{\theta z}(R_B, \theta, z) = 2(-\sigma_{xz}(z)\sin\theta + \sigma_{yz}(z)\cos\theta)$$

On the far-field boundary  $\Gamma_F$ :

$$\sigma_r(R_F, \theta, z) = \frac{\sigma_x(z) + \sigma_y(z)}{2} + \sigma_{xy}(z)\sin 2\theta + \frac{\sigma_x(z) - \sigma_y(z)}{2}\cos 2\theta$$

$$\sigma_\theta(R_F, \theta, z) = \frac{\sigma_x(z) + \sigma_y(z)}{2} - \sigma_{xy}(z)\sin 2\theta - \frac{\sigma_x(z) - \sigma_y(z)}{2}\cos 2\theta$$

$$\sigma_z(R_F, \theta, z) = \sigma_{zz}(z)$$

$$\sigma_{r\theta}(R_F, \theta, z) = \frac{\sigma_x(z) - \sigma_y(z)}{2}\sin 2\theta + \sigma_{xy}(z)\cos 2\theta$$

$$\sigma_{rz}(R_F, \theta, z) = \sigma_{yz}(z)\sin \theta + \sigma_{xz}(z)\cos \theta$$

$$\sigma_{\theta z}(R_F, \theta, z) = -\sigma_{xz}(z)\sin \theta + \sigma_{yz}(z)\cos \theta$$

On the top boundary  $\tilde{\Gamma}_T$  and the bottom boundary  $\tilde{\Gamma}_B$ :

$$\sigma_r(r,\theta,z_i,t) = \frac{\sigma_x(z_i) + \sigma_y(z_i)}{2} \left(1 - \frac{R_B^2}{r^2}\right) + p_B(z_i,t) \frac{R_B^2}{r^2} + \left(1 - \frac{4R_B^2}{r^2} + \frac{3R_B^4}{r^4}\right) \left(\sigma_{xy}(z_i)\sin 2\theta + \frac{\sigma_x(z_i) - \sigma_y(z_i)}{2}\cos 2\theta\right)$$

$$\sigma_{\theta}(r,\theta,z_{i},t) = \frac{\sigma_{x}(z_{i}) + \sigma_{y}(z_{i})}{2} \left(1 + \frac{R_{B}^{2}}{r^{2}}\right) - p_{B}(z_{i},t) \frac{R_{B}^{2}}{r^{2}} - \left(1 + \frac{3R_{B}^{4}}{r^{4}}\right) \left(\sigma_{xy}(z_{i})\sin 2\theta + \frac{\sigma_{x}(z_{i}) - \sigma_{y}(z_{i})}{2}\cos 2\theta\right) \sigma_{z}(r,\theta,z_{i}) = \sigma_{zz}(z_{i}) - 4\nu \frac{R_{B}^{2}}{r^{2}} \left(\sigma_{xy}(z_{i})\sin 2\theta + \frac{\sigma_{x}(z_{i}) - \sigma_{y}(z_{i})}{2}\cos 2\theta\right) \sigma_{r\theta}(r,\theta,z_{i}) = \left(1 + \frac{2R_{B}^{2}}{r^{2}} - \frac{3R_{B}^{4}}{r^{4}}\right) \left(\frac{\sigma_{x}(z) - \sigma_{y}(z)}{2}\sin 2\theta + \sigma_{xy}(z)\cos 2\theta\right) \sigma_{rz}(r,\theta,z_{i}) = \left(1 - \frac{R_{B}^{2}}{r^{2}}\right) \left(\sigma_{yz}(z_{i})\sin\theta + \sigma_{xz}(z_{i})\cos\theta\right) \sigma_{\theta z}(r,\theta,z_{i}) = \left(1 + \frac{R_{B}^{2}}{r^{2}}\right) \left(-\sigma_{xz}(z_{i})\sin\theta + \sigma_{yz}(z_{i})\cos\theta\right)$$
(4.2.46)

where i = T (top) or i = B (bottom).

Let  $V = H^1(\Omega; \mathbb{R}^3)$ . Our objective is to find the displacement  $\vec{u}$  such that

$$\vec{u} \in L^2(0, t_f; V), \qquad \dot{\vec{u}} \in L^2(0, t_f; V) \qquad \text{and}$$

$$D\Pi(\vec{u})\vec{\Phi} = 0, \qquad \forall \ \vec{\Phi} \in V \qquad (4.2.47)$$

where  $\vec{u}(r,\theta,z,t) = u(r,\theta,z,t)\vec{i}_r + v(r,\theta,z,t)\vec{i}_{\theta} + w(r,\theta,z,t)\vec{i}_z,$  $\vec{\Phi}(r,\theta,z) = \varphi(r,\theta,z)\vec{i}_r + \psi(r,\theta,z)\vec{i}_{\theta} + \omega(r,\theta,z)\vec{i}_z,$ 

 $D\Pi(\vec{u})\vec{\Phi}$  is the Gâteaux differential of  $\Pi$  with increment  $\vec{\Phi}$ .

Then (4.2.47) has the form

$$\int_{z_T}^{z_B} \int_0^{2\pi} \int_{R_B}^{R_F} \left[ \left( 2Gu_r + \left(K - \frac{2G}{3}\right) \left(u_r + \frac{u + v_\theta}{r} + w_z\right) \right) \varphi_r \right]$$
$$+ G\left(v_r + \frac{u_\theta - v}{r}\right) \left(\psi_r + \frac{\varphi_\theta - \psi}{r}\right) + G\left(u_z + w_r\right) \left(\varphi_z + \omega_r\right)$$
$$+ \left( 2G\frac{u + v_\theta}{r} + \left(K - \frac{2G}{3}\right) \left(u_r + \frac{u + v_\theta}{r} + w_z\right) \right) \frac{\varphi + \psi_\theta}{r}$$
$$+ G\left(\frac{w_\theta}{r} + v_z\right) \left(\frac{\omega_\theta}{r} + \psi_z\right) + \left( 2Gw_z + \left(K - \frac{2G}{3}\right) \left(u_r + \frac{u + v_\theta}{r} + w_z\right) \right) \omega_z \right] r \, \mathrm{d}r \, \mathrm{d}\theta \, \mathrm{d}z$$

$$-\int_{z_T}^{z_B}\int_0^{2\pi}\int_{R_B}^{R_F} f\left(\varphi_r + \frac{\varphi + \psi_{\theta}}{r} + \omega_z\right) r \, \mathrm{d}r \, \mathrm{d}\theta \, \mathrm{d}z$$

$$+\int_{z_T}^{z_B}\int_0^{2\pi} \left(\sigma_r(R_B, \theta, z)\varphi(R_B, \theta, z) + \sigma_{r\theta}(R_B, \theta, z)\psi(R_B, \theta, z) + \sigma_{rz}(R_B, \theta, z)\omega(R_B, \theta, z)\right) R_B \, \mathrm{d}\theta \, \mathrm{d}z$$

$$-\int_{z_T}^{z_B}\int_0^{2\pi} \left(\sigma_r(R_F, \theta, z)\varphi(R_F, \theta, z) + \sigma_{r\theta}(R_F, \theta, z)\psi(R_F, \theta, z) + \sigma_{rz}(R_F, \theta, z)\omega(R_F, \theta, z)\right) R_F \, \mathrm{d}\theta \, \mathrm{d}z$$

$$+\int_0^{2\pi}\int_{R_B}^{R_F} \left(\sigma_{rz}(r, \theta, z_T)\varphi(r, \theta, z_T) + \sigma_{\theta z}(r, \theta, z_T)\psi(r, \theta, z_T) + \sigma_{z}(r, \theta, z_B)\omega(r, \theta, z_B)\right) r \, \mathrm{d}r \, \mathrm{d}\theta$$

$$+K_F \int_{z_T}^{z_B}\int_0^{2\pi} \left(u(R_F, \theta, z)\varphi(R_F, \theta, z) + v(R_F, \theta, z)\psi(R_F, \theta, z) + w(R_F, \theta, z)\omega(R_F, \theta, z)\right) R_F \, \mathrm{d}\theta \, \mathrm{d}z = 0 \quad (4.2.48)$$

The first term of the left-hand side of (4.2.48) is

$$\begin{split} \int_{z_T}^{z_B} \int_0^{2\pi} \int_{R_B}^{R_F} \left[ \left( K + \frac{4G}{3} \right) u_r \varphi_r r + \left( K - \frac{2G}{3} \right) u \varphi_r + \left( K - \frac{2G}{3} \right) v_\theta \varphi_r + \left( K - \frac{2G}{3} \right) w_z \varphi_r r \right. \\ \left. + G v_r \psi_r r + G v_r \varphi_\theta - G v_r \psi + G u_\theta \psi_r - G v \psi_r + G u_\theta \varphi_\theta \frac{1}{r} - G u_\theta \psi \frac{1}{r} \right. \\ \left. - G v \varphi_\theta \frac{1}{r} + G v \psi \frac{1}{r} + G u_z \varphi_z r + G u_z \omega_r r + G w_r \varphi_z r + G w_r \omega_r r \right. \\ \left. + \left( K + \frac{4G}{3} \right) u \varphi \frac{1}{r} + \left( K + \frac{4G}{3} \right) u \psi_\theta \frac{1}{r} + \left( K + \frac{4G}{3} \right) v_\theta \varphi \frac{1}{r} + \left( K + \frac{4G}{3} \right) v_\theta \psi_\theta \frac{1}{r} \right. \\ \left. + \left( K - \frac{2G}{3} \right) u_r \varphi + \left( K - \frac{2G}{3} \right) u_r \psi_\theta + \left( K - \frac{2G}{3} \right) w_z \varphi + \left( K - \frac{2G}{3} \right) w_z \psi_\theta \right. \\ \left. + G w_\theta \omega_\theta \frac{1}{r} + G w_\theta \psi_z + G v_z \omega_\theta + G v_z \psi_z r + \left( K + \frac{4G}{3} \right) w_z \omega_z r \right. \\ \left. + \left( K - \frac{2G}{3} \right) u_r \omega_z r + \left( K - \frac{2G}{3} \right) u \omega_z + \left( K - \frac{2G}{3} \right) v_\theta \omega_z \right] dr d\theta dz \quad (4.2.49) \end{split}$$

Applying  $\theta$ -approximation (4.2.18) to (4.2.49), integrating with respect to  $\theta$  over  $(0, 2\pi)$ , and combining like terms with respect to the basis functions, we have  $\varphi_0$ :

$$\pi \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ 2\left(K + \frac{4}{3}G\right) u_{0_r} \varphi_{0_r} r + 2\left(K - \frac{2}{3}G\right) u_0 \varphi_{0_r} + 2\left(K - \frac{2}{3}G\right) w_{0_z} \varphi_{0_r} r \right. \\ \left. + 2Gu_{0_z} \varphi_{0_z} r + 2Gw_{0_r} \varphi_{0_z} r + 2\left(K + \frac{4}{3}G\right) u_0 \varphi_0 \frac{1}{r} \right. \\ \left. + 2\left(K - \frac{2}{3}G\right) u_{0_r} \varphi_0 + 2\left(K - \frac{2}{3}G\right) w_{0_z} \varphi_0 \right] \mathrm{d}r \, \mathrm{d}z \quad (4.2.50)$$

 $\varphi_1$  :

$$+\pi \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ \left( K + \frac{4}{3}G \right) u_{1r} \varphi_{1r} r + \left( K - \frac{2}{3}G \right) u_1 \varphi_{1r} - \left( K - \frac{2}{3}G \right) v_2 \varphi_{1r} \right. \\ \left. + \left( K - \frac{2}{3}G \right) w_{1z} \varphi_{1r} r + G v_{2r} \varphi_1 + \left( K + \frac{7}{3}G \right) u_1 \varphi_1 \frac{1}{r} + G u_{1z} \varphi_{1z} r + G w_{1r} \varphi_{1z} r \right. \\ \left. - \left( K + \frac{7}{3}G \right) v_2 \varphi_1 \frac{1}{r} + \left( K - \frac{2}{3}G \right) u_{1r} \varphi_1 + \left( K - \frac{2}{3}G \right) w_{1z} \varphi_1 \right] dr dz \quad (4.2.51)$$

 $\varphi_2$  :

$$+\pi \int_{z_{T}}^{z_{B}} \int_{R_{B}}^{R_{F}} \left[ \left( K + \frac{4}{3}G \right) u_{2_{r}}\varphi_{2_{r}}r + \left( K - \frac{2}{3}G \right) u_{2}\varphi_{2_{r}} + \left( K - \frac{2}{3}G \right) v_{1}\varphi_{2_{r}} \right. \\ \left. + \left( K - \frac{2}{3}G \right) w_{2_{z}}\varphi_{2_{r}}r - Gv_{1_{r}}\varphi_{2} + \left( K + \frac{7}{3}G \right) u_{2}\varphi_{2}\frac{1}{r} + Gu_{2_{z}}\varphi_{2_{z}}r + Gw_{2_{r}}\varphi_{2_{z}}r \right. \\ \left. + \left( K + \frac{7}{3}G \right) v_{1}\varphi_{2}\frac{1}{r} + \left( K - \frac{2}{3}G \right) u_{2_{r}}\varphi_{2} + \left( K - \frac{2}{3}G \right) w_{2_{z}}\varphi_{2} \right] dr dz \quad (4.2.52)$$

 $\varphi_3$  :

$$+\pi \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ \left( K + \frac{4}{3}G \right) u_{3_r} \varphi_{3_r} r + \left( K - \frac{2}{3}G \right) u_3 \varphi_{3_r} - 2 \left( K - \frac{2}{3}G \right) v_4 \varphi_{3_r} \right. \\ \left. + \left( K - \frac{2}{3}G \right) w_{3_z} \varphi_{3_r} r + 2G v_{4_r} \varphi_3 + \left( K + \frac{16}{3}G \right) u_3 \varphi_3 \frac{1}{r} + G u_{3_z} \varphi_{3_z} r + G w_{3_r} \varphi_{3_z} r \right. \\ \left. - 2 \left( K + \frac{7}{3}G \right) v_4 \varphi_3 \frac{1}{r} + \left( K - \frac{2}{3}G \right) u_{3_r} \varphi_3 + \left( K - \frac{2}{3}G \right) w_{3_z} \varphi_3 \right] dr dz \quad (4.2.53)$$

$$+\pi \int_{z_{T}}^{z_{B}} \int_{R_{B}}^{R_{F}} \left[ \left( K + \frac{4}{3}G \right) u_{4_{r}}\varphi_{4_{r}}r + \left( K - \frac{2}{3}G \right) u_{4}\varphi_{4_{r}} + 2\left( K - \frac{2}{3}G \right) v_{3}\varphi_{4_{r}} \right. \\ \left. + \left( K - \frac{2}{3}G \right) w_{4_{z}}\varphi_{4_{r}}r - 2Gv_{3_{r}}\varphi_{4} + \left( K + \frac{16}{3}G \right) u_{4}\varphi_{4} \frac{1}{r} + Gu_{4_{z}}\varphi_{4_{z}}r + Gw_{4_{r}}\varphi_{4_{z}}r \right. \\ \left. + 2\left( K + \frac{7}{3}G \right) v_{3}\varphi_{4} \frac{1}{r} + \left( K - \frac{2}{3}G \right) u_{4_{r}}\varphi_{4} + \left( K - \frac{2}{3}G \right) w_{4_{z}}\varphi_{4} \right] dr dz \quad (4.2.54)$$

 $\psi_0$  :

$$+\pi \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ 2G\left( \left( v_{0_r} - \frac{v_0}{r} \right) \left( \psi_{0_r} - \frac{\psi_0}{r} \right) + v_{0_z} \psi_{0_z} \right) r \right] \mathrm{d}r \, \mathrm{d}z \tag{4.2.55}$$

 $\psi_1$  :

$$+ \pi \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ Gv_{1_r} \psi_{1_r} r - Gv_{1_r} \psi_1 - Gu_2 \psi_{1_r} - Gv_1 \psi_{1_r} + \left(K + \frac{7}{3}G\right) u_2 \psi_1 \frac{1}{r} \right. \\ \left. + \left(K + \frac{7}{3}G\right) v_1 \psi_1 \frac{1}{r} + \left(K - \frac{2}{3}G\right) u_{2_r} \psi_1 + \left(K - \frac{2}{3}G\right) w_{2_z} \psi_1 \right. \\ \left. - Gw_2 \psi_{1_z} + Gv_{1_z} \psi_{1_z} r \right] dr dz \quad (4.2.56)$$

 $\psi_2$  :

$$+\pi \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ Gv_{2_r} \psi_{2_r} r - Gv_{2_r} \psi_2 + Gu_1 \psi_{2_r} - Gv_2 \psi_{2_r} - \left(K + \frac{7}{3}G\right) u_1 \psi_2 \frac{1}{r} + \left(K + \frac{7}{3}G\right) v_2 \psi_2 \frac{1}{r} - \left(K - \frac{2}{3}G\right) u_{1_r} \psi_2 - \left(K - \frac{2}{3}G\right) w_{1_z} \psi_2 + Gw_1 \psi_{2_z} + Gv_{2_z} \psi_{2_z} r \right] dr dz \quad (4.2.57)$$

 $\psi_3$  :

$$+\pi \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ Gv_{3_r} \psi_{3_r} r - Gv_{3_r} \psi_3 - 2Gu_4 \psi_{3_r} - Gv_3 \psi_{3_r} + 2\left(K + \frac{7}{3}G\right) u_4 \psi_3 \frac{1}{r} \right. \\ \left. + \left(4K + \frac{19}{3}G\right) v_3 \psi_3 \frac{1}{r} + 2\left(K - \frac{2}{3}G\right) u_{4_r} \psi_3 + 2\left(K - \frac{2}{3}G\right) w_{4_z} \psi_3 \right. \\ \left. - 2Gw_4 \psi_{3_z} + Gv_{3_z} \psi_{3_z} r \right] dr dz \quad (4.2.58)$$

 $\varphi_4$  :

$$+\pi \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ Gv_{4_r} \psi_{4_r} r - Gv_{4_r} \psi_4 + 2Gu_3 \psi_{4_r} - Gv_4 \psi_{4_r} - 2\left(K + \frac{7}{3}G\right) u_3 \psi_4 \frac{1}{r} + \left(4K + \frac{19}{3}G\right) v_4 \psi_4 \frac{1}{r} - 2\left(K - \frac{2}{3}G\right) u_{3_r} \psi_4 - 2\left(K - \frac{2}{3}G\right) w_{3_z} \psi_4 + 2Gw_3 \psi_{4_z} + Gv_{4_z} \psi_{4_z} r \right] dr dz \quad (4.2.59)$$

 $\omega_0$  :

$$+\pi \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ 2Gu_{0_z}\omega_{0_r}r + 2Gw_{0_r}\omega_{0_r}r + 2\left(K + \frac{4}{3}G\right)w_{0_z}\omega_{0_z}r + 2\left(K - \frac{2}{3}G\right)u_{0_r}\omega_{0_z}r + 2\left(K - \frac{2}{3}G\right)u_{0}\omega_{0_z}\right] dr dz \quad (4.2.60)$$

 $\omega_1$  :

$$+\pi \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ Gu_{1_z}\omega_{1_r}r + Gw_{1_r}\omega_{1_r}r + Gw_1\omega_1\frac{1}{r} + Gv_{2_z}\omega_1 + \left(K + \frac{4}{3}G\right)w_{1_z}\omega_{1_z}r + \left(K - \frac{2}{3}G\right)u_{1_r}\omega_{1_z}r + \left(K - \frac{2}{3}G\right)u_1\omega_{1_z} - \left(K - \frac{2}{3}G\right)v_2\omega_{1_z}\right] dr dz \quad (4.2.61)$$

 $\omega_2$  :

$$+\pi \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ Gu_{2z}\omega_{2r}r + Gw_{2r}\omega_{2r}r + Gw_2\omega_2\frac{1}{r} - Gv_{1z}\omega_2 + \left(K + \frac{4}{3}G\right)w_{2z}\omega_{2z}r + \left(K - \frac{2}{3}G\right)u_{2r}\omega_{2z}r + \left(K - \frac{2}{3}G\right)u_2\omega_{2z} + \left(K - \frac{2}{3}G\right)v_1\omega_{2z}\right] dr dz \quad (4.2.62)$$

 $\omega_3$  :

$$+\pi \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ Gu_{3_z} \omega_{3_r} r + Gw_{3_r} \omega_{3_r} r + 4Gw_3 \omega_3 \frac{1}{r} + 2Gv_{4_z} \omega_3 + \left(K + \frac{4}{3}G\right) w_{3_z} \omega_{3_z} r + \left(K - \frac{2}{3}G\right) u_{3_r} \omega_{3_z} r + \left(K - \frac{2}{3}G\right) u_3 \omega_{3_z} - 2\left(K - \frac{2}{3}G\right) v_4 \omega_{3_z} \right] dr dz \quad (4.2.63)$$

 $\psi_4$  :

$$+\pi \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ Gu_{4_z} \omega_{4_r} r + Gw_{4_r} \omega_{4_r} r + 4Gw_4 \omega_4 \frac{1}{r} - 2Gv_{3_z} \omega_4 + \left(K + \frac{4}{3}G\right) w_{4_z} \omega_{4_z} r + \left(K - \frac{2}{3}G\right) u_{4_r} \omega_{4_z} r + \left(K - \frac{2}{3}G\right) u_4 \omega_{4_z} + 2\left(K - \frac{2}{3}G\right) v_3 \omega_{4_z} \right] dr dz \quad (4.2.64)$$

The  $\sigma$ -approximation of body force f has the form

$$f(r, \theta, z, t) = f_0(r, z, t) + f_1(r, z, t) \sin \theta + f_2(r, z, t) \cos \theta$$
  
=  $f_3(r, z, t) \sin 2\theta + f_4(r, z, t) \cos 2\theta$  (4.2.65)

The second term of the left-hand side of (4.2.48) is

$$-\pi \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ 2f_0 \varphi_{0_r} r + f_1 \varphi_{1_r} r + f_2 \varphi_{2_r} r + f_3 \varphi_{3_r} r + f_4 \varphi_{4_r} r \right]$$
  
+  $2f_0 \varphi_0 + f_1 \varphi_1 + f_2 \varphi_2 + f_3 \varphi_3 + f_4 \varphi_4 - f_1 \psi_2 + f_2 \psi_1 - 2f_3 \psi_4 + 2f_4 \psi_3$   
+  $2f_0 \omega_{0_z} r + f_1 \omega_{1_z} r + f_2 \omega_{2_z} r + f_3 \omega_{3_z} r + f_4 \omega_{4_z} r \right] dr dz \quad (4.2.66)$ 

The third term of the left-hand side of (4.2.48) is

$$\pi \int_{z_T}^{z_B} 2p_B(z)\varphi_0(R_B, z)R_B \,\mathrm{d}z \tag{4.2.67}$$

The fourth term of the left-hand side of (4.2.48) is

$$-\pi \int_{z_T}^{z_B} \left[ 2 \frac{\sigma_x(z) + \sigma_y(z)}{2} \varphi_0(R_F, z) + \sigma_{xy}(z) \varphi_3(R_F, z) + \frac{\sigma_x(z) - \sigma_y(z)}{2} \varphi_4(R_F, z) \right. \\ \left. + \frac{\sigma_x(z) - \sigma_y(z)}{2} \psi_3(R_F, z) + \sigma_{xy}(z) \psi_4(R_F, z) \right. \\ \left. + \sigma_{yz}(z) \omega_1(R_F, z) + \sigma_{xz}(z) \omega_2(R_F, z) \right] R_F \, \mathrm{d}z \quad (4.2.68)$$

The fifth term of the left-hand side of (4.2.48) is

$$\pi \int_{R_B}^{R_F} \left[ \left( 1 - \frac{R_B^2}{r^2} \right) r \left( \sigma_{yz}(z_T) \varphi_1(r, z_T) + \sigma_{xz}(z_T) \varphi_2(r, z_T) \right) \right. \\ \left. + \left( 1 + \frac{R_B^2}{r^2} \right) r \left( - \sigma_{xz}(z_T) \psi_1(r, z_T) + \sigma_{yz}(z_T) \psi_2(r, z_T) \right) + 2 \sigma_{zz}(z_T) \omega_0(r, z_T) r - \right]$$

 $\omega_4$  :

$$-4\nu \frac{R_B^2}{r} \sigma_{xy}(z_T) \omega_3(r, z_T) - 4\nu \frac{R_B^2}{r} \frac{\sigma_x(z_T) - \sigma_y(z_T)}{2} \omega_4(r, z_T) \bigg] dr \quad (4.2.69)$$

The sixth term of the left-hand side of (4.2.48) is

$$-\pi \int_{R_B}^{R_F} \left[ \left( 1 - \frac{R_B^2}{r^2} \right) r \left( \sigma_{yz}(z_B) \varphi_1(r, z_B) + \sigma_{xz}(z_B) \varphi_2(r, z_B) \right) \right. \\ \left. + \left( 1 + \frac{R_B^2}{r^2} \right) r \left( -\sigma_{xz}(z_B) \psi_1(r, z_B) + \sigma_{yz}(z_B) \psi_2(r, z_B) \right) + 2\sigma_{zz}(z_B) \omega_0(r, z_B) r \right. \\ \left. - 4\nu \frac{R_B^2}{r} \sigma_{xy}(z_B) \omega_3(r, z_B) - 4\nu \frac{R_B^2}{r} \frac{\sigma_x(z_B) - \sigma_y(z_B)}{2} \omega_4(r, z_B) \right] dr \quad (4.2.70)$$

The seventh term of the left-hand side of (4.2.48) is

$$K_{F}\pi \int_{z_{T}}^{z_{B}} \left[ 2u_{0}(R_{F},z)\varphi_{0}(R_{F},z) + u_{1}(R_{F},z)\varphi_{1}(R_{F},z) + u_{2}(R_{F},z)\varphi_{2}(R_{F},z) \right. \\ \left. + u_{3}(R_{F},z)\varphi_{3}(R_{F},z) + u_{4}(R_{F},z)\varphi_{4}(R_{F},z) + 2v_{0}(R_{F},z)\psi_{0}(R_{F},z) \right. \\ \left. + v_{1}(R_{F},z)\psi_{1}(R_{F},z) + v_{2}(R_{F},z)\psi_{2}(R_{F},z) + v_{3}(R_{F},z)\psi_{3}(R_{F},z) \right. \\ \left. + v_{4}(R_{F},z)\psi_{4}(R_{F},z) + 2w_{0}(R_{F},z)\omega_{0}(R_{F},z) + w_{1}(R_{F},z)\omega_{1}(R_{F},z) \right. \\ \left. + w_{2}(R_{F},z)\omega_{2}(R_{F},z) + w_{3}(R_{F},z)\omega_{3}(R_{F},z) + w_{4}(R_{F},z)\omega_{4}(R_{F},z) \right] R_{F} dz \quad (4.2.71)$$

In (4.2.50)-(4.2.64) and (4.2.66)-(4.2.71), we combine terms corresponding to  $\varphi_i$ ,  $\psi_i, \, \omega_i, \, i = 0, \dots, 4$ , and equal them to 0.  $\varphi_0: \quad (u_0, w_0)$ 

$$\int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ \left( K + \frac{4G}{3} \right) u_{0_r} \varphi_{0_r} r + \left( K - \frac{2G}{3} \right) u_0 \varphi_{0_r} + \left( K - \frac{2G}{3} \right) u_{0_r} \varphi_0 \right. \\ \left. + \left( K + \frac{4G}{3} \right) u_0 \varphi_0 \frac{1}{r} + G u_{0_z} \varphi_{0_z} r + \left( K - \frac{2G}{3} \right) w_{0_z} \varphi_{0_r} r + G w_{0_r} \varphi_{0_z} r \right. \\ \left. + \left( K - \frac{2G}{3} \right) w_{0_z} \varphi_0 \right] dr dz - \int_{z_T}^{z_B} \int_{R_B}^{R_F} f_0 (\varphi_{0_r} r + \varphi_0) dr dz \\ \left. + \int_{z_T}^{z_B} p_B(z) \varphi_0 (R_B, z) R_B dz - \int_{z_T}^{z_B} \frac{\sigma_x(z) + \sigma_y(z)}{2} \varphi_0 (R_F, z) R_F dz \\ \left. + K_F \int_{z_T}^{z_B} u_0 (R_F, z) \varphi_0 (R_F, z) R_F dz = 0 \right]$$

$$(4.2.72)$$

# $\varphi_1: (u_1, v_2, w_1)$

$$\int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ \left( K + \frac{4G}{3} \right) u_{1_r} \varphi_{1_r} r + \left( K - \frac{2G}{3} \right) u_1 \varphi_{1_r} + \left( K - \frac{2G}{3} \right) u_{1_r} \varphi_1 + G w_{1_r} \varphi_{1_z} r \right] \\ + \left( K + \frac{7G}{3} \right) u_1 \varphi_1 \frac{1}{r} + G u_{1_z} \varphi_{1_z} r - \left( K - \frac{2G}{3} \right) v_2 \varphi_{1_r} + G v_{2_r} \varphi_1 \\ - \left( K + \frac{7G}{3} \right) v_2 \varphi_1 \frac{1}{r} + \left( K - \frac{2G}{3} \right) w_{1_z} \varphi_{1_r} r + \left( K - \frac{2G}{3} \right) w_{1_z} \varphi_1 \right] dr dz \\ - \int_{z_T}^{z_B} \int_{R_B}^{R_F} f_1 \left( \varphi_{1_r} r + \varphi_1 \right) dr dz + \int_{R_B}^{R_F} \left( 1 - \frac{R_B^2}{r^2} \right) r \sigma_{yz} (z_T) \varphi_1 (r, z_T) dr \\ - \int_{R_B}^{R_F} \left( 1 - \frac{R_B^2}{r^2} \right) r \sigma_{yz} (z_B) \varphi_1 (r, z_B) dr + K_F \int_{z_T}^{z_B} u_1 (R_F, z) \varphi_1 (R_F, z) R_F dz = 0 \\ (4.2.73)$$

$$\varphi_{2}: \quad (u_{2}, v_{1}, w_{2})$$

$$\int_{z_{T}}^{z_{B}} \int_{R_{B}}^{R_{F}} \left[ \left( K + \frac{4G}{3} \right) u_{2r} \varphi_{2r} r + \left( K - \frac{2G}{3} \right) u_{2} \varphi_{2r} + \left( K - \frac{2G}{3} \right) u_{2r} \varphi_{2} + G u_{2z} \varphi_{2z} r \right] \\ + \left( K + \frac{7G}{3} \right) u_{2} \varphi_{2} \frac{1}{r} + \left( K - \frac{2G}{3} \right) v_{1} \varphi_{2r} - G v_{1r} \varphi_{2} + \left( K + \frac{7G}{3} \right) v_{1} \varphi_{2} \frac{1}{r} \\ + G w_{2r} \varphi_{2z} r + \left( K - \frac{2G}{3} \right) w_{2z} \varphi_{2r} r + \left( K - \frac{2G}{3} \right) w_{2z} \varphi_{2} \right] dr dz$$

$$- \int_{z_{T}}^{z_{B}} \int_{R_{B}}^{R_{F}} f_{2} (\varphi_{2r} r + \varphi_{2}) dr dz + \int_{R_{B}}^{R_{F}} \left( 1 - \frac{R_{B}^{2}}{r^{2}} \right) r \sigma_{xz} (z_{T}) \varphi_{2} (r, z_{T}) dr$$

$$- \int_{R_{B}}^{R_{F}} \left( 1 - \frac{R_{B}^{2}}{r^{2}} \right) r \sigma_{xz} (z_{B}) \varphi_{2} (r, z_{B}) dr + K_{F} \int_{z_{T}}^{z_{B}} u_{2} (R_{F}, z) \varphi_{2} (R_{F}, z) R_{F} dz = 0$$

$$(4.2.74)$$

$$\begin{aligned} \varphi_{3}: \quad (u_{3}, v_{4}, w_{3}) \\ \int_{z_{T}}^{z_{B}} \int_{R_{B}}^{R_{F}} \left[ \left( K + \frac{4G}{3} \right) u_{3_{r}} \varphi_{3_{r}} r + \left( K - \frac{2G}{3} \right) u_{3} \varphi_{3_{r}} + \left( K - \frac{2G}{3} \right) u_{3_{r}} \varphi_{3} + G u_{3_{z}} \varphi_{3_{z}} r \right. \\ \left. + \left( K + \frac{16G}{3} \right) u_{3} \varphi_{3} \frac{1}{r} - 2 \left( K - \frac{2G}{3} \right) v_{4} \varphi_{3_{r}} - 2 \left( K + \frac{7G}{3} \right) v_{4} \varphi_{3} \frac{1}{r} \right. \\ \left. + 2G v_{4_{r}} \varphi_{3} + G w_{3_{r}} \varphi_{3_{z}} r + \left( K - \frac{2G}{3} \right) w_{3_{z}} \varphi_{3_{r}} r + \left( K - \frac{2G}{3} \right) w_{3_{z}} \varphi_{3} \right] dr dz \end{aligned}$$

$$-\int_{z_T}^{z_B} \int_{R_B}^{R_F} f_3(\varphi_{3_r}r + \varphi_3) \,\mathrm{d}r \,\mathrm{d}z - \int_{z_B}^{z_T} \sigma_{xy}(z)\varphi_3(R_F, z)R_F \,\mathrm{d}z + K_F \int_{z_T}^{z_B} u_3(R_F, z)\varphi_3(R_F, z)R_F \,\mathrm{d}z = 0$$
(4.2.75)

 $\varphi_4: \quad (u_4, v_3, w_4)$ 

$$\int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ \left( K + \frac{4G}{3} \right) u_{4_r} \varphi_{4_r} r + \left( K - \frac{2G}{3} \right) u_4 \varphi_{4_r} + \left( K - \frac{2G}{3} \right) u_{4_r} \varphi_4 + G u_{4_z} \varphi_{4_z} r \right. \\ \left. + \left( K + \frac{16G}{3} \right) u_4 \varphi_4 \frac{1}{r} + 2 \left( K - \frac{2G}{3} \right) v_3 \varphi_{4_r} + 2 \left( K + \frac{7G}{3} \right) v_3 \varphi_4 \frac{1}{r} \right. \\ \left. - 2G v_{3_r} \varphi_4 + G w_{4_r} \varphi_{4_z} r + \left( K - \frac{2G}{3} \right) w_{4_z} \varphi_{4_r} r + \left( K - \frac{2G}{3} \right) w_{4_z} \varphi_4 \right] dr dz \\ \left. - \int_{z_T}^{z_B} \int_{R_B}^{R_F} f_4 \left( \varphi_{4_r} r + \varphi_4 \right) dr dz - \int_{z_B}^{z_T} \frac{\sigma_x(z) - \sigma_y(z)}{2} \varphi_4(R_F, z) R_F dz \\ \left. + K_F \int_{z_T}^{z_B} u_4(R_F, z) \varphi_4(R_F, z) R_F dz = 0 \right]$$

$$(4.2.76)$$

$$\psi_{0}: \quad (v_{0})$$

$$\int_{z_{T}}^{z_{B}} \int_{R_{B}}^{R_{F}} G\Big[ \Big( v_{0_{r}} - \frac{v_{0}}{r} \Big) \Big( \psi_{0_{r}} - \frac{\psi_{0}}{r} \Big) + v_{0_{z}} \psi_{0_{z}} \Big] r \, \mathrm{d}r \, \mathrm{d}z$$

$$+ K_{F} \int_{z_{T}}^{z_{B}} v_{0}(R_{F}, z) \psi_{0}(R_{F}, z) R_{F} \, \mathrm{d}z = 0$$

Take  $\psi_0 = v_0$ . Then

$$\int_{z_T}^{z_B} \int_{R_B}^{R_F} G\left[\left(v_{0_r} - \frac{v_0}{r}\right)^2 + v_{0_z}^2\right] r \,\mathrm{d}r \,\mathrm{d}z + K_F R_F \int_{z_T}^{z_B} v_0^2(R_F, z) \,\mathrm{d}z = 0$$

It follows that

$$v_{0_r} - \frac{v_0}{r} = 0$$
 and  $v_{0_z} = 0$ ,  $z_T < z < z_B$ ,  $R_B < r < R_F$   
 $v_0(R_F, z) = 0$ ,  $z_T < z < z_B$ 

and so,

$$v_0(r, z, t) = C_0(t)r, \quad R_B < r < R_F$$
(4.2.77)
$$\begin{split} \psi_{1} : & (u_{2}, v_{1}, w_{2}) \\ \int_{z_{T}}^{z_{B}} \int_{R_{B}}^{R_{F}} \left[ -Gu_{2}\psi_{1_{r}} + \left(K - \frac{2G}{3}\right)u_{2_{r}}\psi_{1} + \left(K + \frac{7G}{3}\right)u_{2}\psi_{1}\frac{1}{r} + Gv_{1_{r}}\psi_{1_{r}}r - Gv_{1}\psi_{1_{r}} \right. \\ & -Gv_{1_{r}}\psi_{1} + \left(K + \frac{7G}{3}\right)v_{1}\psi_{1}\frac{1}{r} + Gv_{1_{z}}\psi_{1_{z}}r - Gw_{2}\psi_{1_{z}} + \left(K - \frac{2G}{3}\right)w_{2_{z}}\psi_{1}\right] dr dz \\ & -\int_{z_{T}}^{z_{B}} \int_{R_{B}}^{R_{F}} f_{2}\psi_{1} dr dz - \int_{R_{B}}^{R_{F}} \left(1 + \frac{R_{B}^{2}}{r^{2}}\right)r\sigma_{xz}(z_{T})\psi_{1}(r, z_{T}) dr \\ & +\int_{R_{B}}^{R_{F}} \left(1 + \frac{R_{B}^{2}}{r^{2}}\right)r\sigma_{xz}(z_{B})\psi_{1}(r, z_{B}) dr + K_{F}\int_{z_{T}}^{z_{B}} v_{1}(R_{F}, z)\psi_{1}(R_{F}, z)R_{F} dz = 0 \end{split}$$

$$(4.2.78)$$

$$\psi_{2}: \quad (u_{1}, v_{2}, w_{1})$$

$$\int_{z_{T}}^{z_{B}} \int_{R_{B}}^{R_{F}} \left[ Gu_{1}\psi_{2_{r}} - \left(K - \frac{2G}{3}\right)u_{1_{r}}\psi_{2} - \left(K + \frac{7G}{3}\right)u_{1}\psi_{2}\frac{1}{r} + Gv_{2_{r}}\psi_{2_{r}}r - Gv_{2}\psi_{2_{r}}r -$$

$$\psi_3: (u_4, v_3, w_4)$$

$$\int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ -2Gu_4\psi_{3r} + 2\left(K - \frac{2G}{3}\right)u_{4r}\psi_3 + 2\left(K + \frac{7G}{3}\right)u_4\psi_3\frac{1}{r} + Gv_{3r}\psi_{3r}r - Gv_3\psi_{3r} - Gv_{3r}\psi_3 + \left(4K + \frac{19G}{3}\right)v_3\psi_3\frac{1}{r} + Gv_{3z}\psi_{3z}r - 2Gw_4\psi_{3z} + 2\left(K - \frac{2G}{3}\right)w_{4z}\psi_3\right] dr dz - \int_{z_T}^{z_B} \int_{R_B}^{R_F} 2f_4\psi_3 dr dz - \int_{z_T}^{z_B} \frac{\sigma_x(z) - \sigma_y(z)}{2}\psi_3(R_F, z)R_F dz + K_F \int_{z_T}^{z_B} v_3(R_F, z)\psi_3(R_F, z)R_F dz = 0$$

$$(4.2.80)$$

 $\psi_4: (u_3, v_4, w_3)$ 

$$\int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ 2Gu_3\psi_{4_r} - 2\left(K - \frac{2G}{3}\right)u_{3_r}\psi_4 - 2\left(K + \frac{7G}{3}\right)u_3\psi_4\frac{1}{r} + Gv_{4_r}\psi_{4_r}r - Gv_4\psi_{4_r} - Gv_4\psi_4 + \left(4K + \frac{19G}{3}\right)v_4\psi_4\frac{1}{r} + Gv_{4_z}\psi_{4_z}r + 2Gw_3\psi_{4_z}r - 2\left(K - \frac{2G}{3}\right)w_{3_z}\psi_4 \right] dr dz + \int_{z_T}^{z_B} \int_{R_B}^{R_F} 2f_3\psi_4 dr dz - \int_{z_T}^{z_B} \sigma_{xy}(z)\psi_4(R_F, z)R_F dz + K_F \int_{z_T}^{z_B} v_4(R_F, z)\psi_4(R_F, z)R_F dz = 0 \quad (4.2.81)$$

 $\omega_0:\quad (u_0,w_0)$ 

$$\int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ Gu_{0_z} \omega_{0_r} r + \left( K - \frac{2G}{3} \right) u_{0_r} \omega_{0_z} r + \left( K - \frac{2G}{3} \right) u_0 \omega_{0_z} \right. \\ \left. + \left( K + \frac{4G}{3} \right) w_{0_z} \omega_{0_z} r \right] dr dz \\ \left. - \int_{z_T}^{z_B} \int_{R_B}^{R_F} f_0 \omega_{0_z} r dr dz + \int_{F_B}^{R_F} \sigma_{zz} (z_T) \omega_0 (r, z_T) r dr - \int_{F_B}^{R_F} \sigma_{zz} (z_B) \omega_0 (r, z_B) r dr \right. \\ \left. + K_F \int_{z_T}^{z_B} w_0 (R_F, z) \omega_0 (R_F, z) R_F dz = 0$$

$$(4.2.82)$$

$$\omega_{1}: \quad (u_{1}, v_{2}, w_{1})$$

$$\int_{z_{T}}^{z_{B}} \int_{R_{B}}^{R_{F}} \left[ Gu_{1_{z}} \omega_{1_{r}} r + \left(K - \frac{2G}{3}\right) u_{1_{r}} \omega_{1_{z}} r + \left(K - \frac{2G}{3}\right) u_{1} \omega_{1_{z}} + Gv_{2_{z}} \omega_{1} - \left(K - \frac{2G}{3}\right) v_{2} \omega_{1_{z}} + Gw_{1_{r}} \omega_{1_{r}} r + Gw_{1} \omega_{1} \frac{1}{r} + \left(K + \frac{4G}{3}\right) w_{1_{z}} \omega_{1_{z}} r \right] dr dz$$

$$- \int_{z_{T}}^{z_{B}} \int_{R_{B}}^{R_{F}} f_{1} \omega_{1_{z}} r dr dz - \int_{z_{T}}^{z_{B}} \sigma_{yz}(z) \omega_{1}(R_{F}, z) R_{F} dz$$

$$+ K_{F} \int_{z_{T}}^{z_{B}} w_{1}(R_{F}, z) \omega_{1}(R_{F}, z) R_{F} dz = 0 \qquad (4.2.83)$$

 $\omega_2:\quad (u_2,v_1,w_2)$ 

$$\int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ Gu_{2_z} \omega_{2_r} r + \left( K - \frac{2G}{3} \right) u_{2_r} \omega_{2_z} r + \left( K - \frac{2G}{3} \right) u_2 \omega_{2_z} - Gv_{1_z} \omega_2 \right. \\ \left. + \left( K - \frac{2G}{3} \right) v_1 \omega_{2_z} + Gw_{2_r} \omega_{2_r} r + Gw_2 \omega_2 \frac{1}{r} + \left( K + \frac{4G}{3} \right) w_{2_z} \omega_{2_z} r \right] dr dz \\ \left. - \int_{z_T}^{z_B} \int_{R_B}^{R_F} f_2 \omega_{2_z} r \, dr \, dz - \int_{z_T}^{z_B} \sigma_{xz}(z) \omega_2(R_F, z) R_F \, dz \right. \\ \left. + K_F \int_{z_T}^{z_B} w_2(R_F, z) \omega_2(R_F, z) R_F \, dz = 0$$

$$(4.2.84)$$

 $\omega_3:\quad (u_3,v_4,w_3)$ 

$$\int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ Gu_{3_z} \omega_{3_r} r + \left( K - \frac{2G}{3} \right) u_{3_r} \omega_{3_z} r + \left( K - \frac{2G}{3} \right) u_3 \omega_{3_z} + 2Gv_{4_z} \omega_3 \right. \\ \left. - 2 \left( K - \frac{2G}{3} \right) v_4 \omega_{3_z} + Gw_{3_r} \omega_{3_r} r + 4Gw_3 \omega_3 \frac{1}{r} + \left( K + \frac{4G}{3} \right) w_{3_z} \omega_{3_z} r \right] dr dz \\ \left. - \int_{z_T}^{z_B} \int_{R_B}^{R_F} f_3 \omega_{3_z} r \, dr \, dz - \int_{R_B}^{R_F} 4\nu \frac{R_B^2}{r} \sigma_{xy}(z_T) \omega_3(r, z_T) \, dr \\ \left. + \int_{R_B}^{R_F} 4\nu \frac{R_B^2}{r} \sigma_{xy}(z_B) \omega_3(r, z_B) \, dr + K_F \int_{z_T}^{z_B} w_3(R_F, z) \omega_3(R_F, z) R_F \, dz = 0 \quad (4.2.85) \right]$$

$$\begin{aligned}
\omega_{4}: & (u_{4}, v_{3}, w_{4}) \\
\int_{z_{T}}^{z_{B}} \int_{R_{B}}^{R_{F}} \left[ Gu_{4_{z}} \omega_{4_{r}} r + \left(K - \frac{2G}{3}\right) u_{4_{r}} \omega_{4_{z}} r + \left(K - \frac{2G}{3}\right) u_{4} \omega_{4_{z}} - 2Gv_{3_{z}} \omega_{4} \\
& + 2\left(K - \frac{2G}{3}\right) v_{3} \omega_{4_{z}} + Gw_{4_{r}} \omega_{4_{r}} r + 4Gw_{4} \omega_{4} \frac{1}{r} + \left(K + \frac{4G}{3}\right) w_{4_{z}} \omega_{4_{z}} r \right] dr dz \\
& - \int_{z_{T}}^{z_{B}} \int_{R_{B}}^{R_{F}} f_{4} \omega_{4_{z}} r dr dz - \int_{R_{B}}^{R_{F}} 4\nu \frac{R_{B}^{2}}{r} \frac{\sigma_{x}(z_{T}) - \sigma_{y}(z_{T})}{2} \omega_{4}(r, z_{T}) dr \\
& + \int_{R_{B}}^{R_{F}} 4\nu \frac{R_{B}^{2}}{r} \frac{\sigma_{x}(z_{B}) - \sigma_{y}(z_{B})}{2} \omega_{4}(r, z_{B}) dr + K_{F} \int_{z_{T}}^{z_{B}} w_{4}(R_{F}, z) \omega_{4}(R_{F}, z) R_{F} dz = 0 \\
\end{aligned}$$
(4.2.86)

### 4.2.3 Finite element approximation

Dirichlet boundary conditions for the TCP system

r-approximation:



Figure 4.3: r-approximation of 3D Dirichlet boundary conditions

Here  $N_r$  is the number of subintervals;  $M_{Dr} = N_r - 1$  is the number of basis functions  $\bar{b}(r) = \begin{bmatrix} \bar{b}_1(r) & \dots & \bar{b}_{M_D r}(r) \end{bmatrix}^T$  is the vector of basis functions; and

$$B(r) = \begin{bmatrix} \bar{b}(r)^T & 0 & 0\\ 0 & \bar{b}(r)^T & 0\\ 0 & 0 & \bar{b}(r)^T \end{bmatrix}_{3 \times 3M_{Dr}}$$

z-approximation:



Figure 4.4: z-approximation of 3D Dirichlet boundary conditions

Here  $N_z$  is the number of subintervals;  $M_{Dz} = N_z - 1$  is the number of basis functions  $\hat{b}(z) = \begin{bmatrix} \hat{b}_1(z) & \dots & \hat{b}_{M_D z}(z) \end{bmatrix}^T$  is the vector of basis functions; and

$$\hat{B}(z) = \begin{bmatrix} \hat{b}^T(z) & 0 & 0\\ 0 & \hat{b}(z)^T & 0\\ 0 & 0 & \hat{b}(z)^T \end{bmatrix}_{3 \times 3M_{Dz}} = I_{3 \times 3} \otimes \hat{b}(z)^T$$

Using r-and z-approximations, the coefficients of the  $\theta$ -approximation for  $V(r, \theta, z, t)$ can be represented as

$$V_{i}(r, z, t) = \hat{B}(z) \cdot \begin{bmatrix} \bar{b}(r)^{T} & \dots & 0 \\ & \dots & \\ 0 & \dots & \hat{b}(r)^{T} \end{bmatrix}_{3M_{Dz} \times (M_{Dr} \cdot 3M_{Dz})} \cdot \bar{c}_{i}(t)_{(M_{Dr} \cdot 3M_{Dz}) \times 1}$$
$$i = 0, \dots, 4 \quad (4.2.87)$$

We can rewrite (4.2.87) as

$$V_i(r, z, t) = \left(\hat{b}(z)^T \otimes B(r)\right)\bar{c}_i(t)$$
(4.2.88)

where  $\bar{c}_i(t)$  is a  $(3M_{Dr}M_{Dz} \times 1)$  vector and  $\hat{b}(z)^T \otimes B(r)$  is a  $(3 \times 3M_{Dr}M_{Dz})$  matrix. The coefficients of the  $\theta$ -approximation for boundary conditions (4.2.2) and initial condition (4.2.3) can be represented as

$$V_{B0}(z,t) = \hat{B}(z)\bar{\nu}_{B0}(t)$$

$$V_{Fi}(z) = \hat{B}(z)\bar{\nu}_{Fi}$$

$$\tilde{V}_{Ti}(r,t) = B(r)\bar{\nu}_{\tilde{T}i}(t)$$

$$\tilde{V}_{Bi}(r,t) = B(r)\bar{\nu}_{\tilde{B}i}(t)$$

$$\tilde{V}_{Ii}(r,z) = \left(\hat{b}(z)^T \otimes B(r)\right)\bar{\nu}_{Ii}, \qquad i = 0, \dots, 4$$
(4.2.89)

where  $\bar{\nu}_{B0}(t)$  and  $\bar{\nu}_{Fi}$  are  $(3M_{Dz} \times 1)$  vectors;  $\bar{\nu}_{\tilde{B}i}(t)$  and  $\bar{\nu}_{\tilde{T}i}(t)$  are  $(3M_{Dr} \times 1)$  vectors; and  $\bar{\nu}_{Ii}$  is a  $(3M_{Dr}M_{Dz} \times 1)$  vector.

Neumann boundary conditions for the elastic system

*r*-approximation:



Figure 4.5: r-approximation of 3D Neumann boundary conditions

 $N_r$  is the number of subintervals;  $M_r = N_r + 1$  is the number of basis functions  $\bar{\mathbf{b}}(r) = \begin{bmatrix} \bar{\mathbf{b}}_1(r) & \dots & \bar{\mathbf{b}}_{M_r}(r) \end{bmatrix}^T$  is the vector of basis functions.

z-approximation:



Figure 4.6: z-approximation of 3D Neumann boundary conditions

 $N_z$  is the number of subintervals;  $M_z = N_z + 1$  is the number of basis functions  $\hat{\mathbf{b}}(z) = \begin{bmatrix} \hat{\mathbf{b}}_1(z) & \dots & \hat{\mathbf{b}}_{M_z}(z) \end{bmatrix}^T$  is the vector of basis functions.

Using r-and z-approximations, the coefficients of the  $\theta$ -approximation for the displacement coordinates can be represented as

$$u_i(r, z, t) = \left(\hat{\mathbf{b}}(z)^T \otimes \bar{\mathbf{b}}(r)^T\right) \bar{\mu}_i(t)$$
(4.2.90)

$$v_i(r, z, t) = \left(\hat{\mathbf{b}}(z)^T \otimes \bar{\mathbf{b}}(r)^T\right) \bar{\eta}_i(t)$$
(4.2.91)

$$w_i(r, z, t) = \left(\hat{\mathbf{b}}(z)^T \otimes \bar{\mathbf{b}}(r)^T\right) \bar{\lambda}_i(t)$$
(4.2.92)

where  $\hat{\mathbf{b}}(z)^T \otimes \bar{\mathbf{b}}(r)^T$  is a  $(1 \times M_r M_z)$  vector and  $\bar{\mu}_i(t), \bar{\eta}_i(t), \bar{\lambda}_i(t)$  are  $(M_r M_z \times 1)$  vectors.

Finite element approximation of the elastic system

Define the following matrices.

$$M_r \times M_r$$
 matrices:

$$g_{2}(r) = \int_{R_{B}}^{R_{F}} \bar{\mathbf{b}}_{r}(r) \bar{\mathbf{b}}_{r}(r)^{T} r \, \mathrm{d}r$$
$$g_{1}^{(r)} = \int_{R_{B}}^{R_{F}} \bar{\mathbf{b}}(r) \bar{\mathbf{b}}_{r}(r)^{T} \, \mathrm{d}r$$
$$g_{1}(r) = \int_{R_{B}}^{R_{F}} \bar{\mathbf{b}}(r) \bar{\mathbf{b}}_{r}(r)^{T} r \, \mathrm{d}r$$
$$g_{0}\left(\frac{1}{r}\right) = \int_{R_{B}}^{R_{F}} \bar{\mathbf{b}}(r) \bar{\mathbf{b}}(r)^{T} \frac{1}{r} \, \mathrm{d}r$$
$$g_{0}(r) = \int_{R_{B}}^{R_{F}} \bar{\mathbf{b}}(r) \bar{\mathbf{b}}(r)^{T} r \, \mathrm{d}r$$
$$g_{0}^{(r)} = \int_{R_{B}}^{R_{F}} \bar{\mathbf{b}}(r) \bar{\mathbf{b}}(r)^{T} \, \mathrm{d}r$$

 $M_z \times M_z$  matrices:

$$g_{2}^{(z)} = \int_{z_{T}}^{z_{B}} \hat{\mathbf{b}}_{z}(z) \,\hat{\mathbf{b}}_{z}(z)^{T} \,\mathrm{d}z$$
$$g_{1}^{(z)} = \int_{z_{T}}^{z_{B}} \hat{\mathbf{b}}(z) \,\hat{\mathbf{b}}_{z}(z)^{T} \,\mathrm{d}z$$
$$g_{0}^{(z)} = \int_{z_{T}}^{z_{B}} \hat{\mathbf{b}}(z) \,\hat{\mathbf{b}}(z)^{T} \,\mathrm{d}z$$

 $M_r M_z \times M_r M_z$  matrix:

$$g_F = \int_{z_T}^{z_B} \left( \hat{\mathbf{b}}(z) \otimes \bar{\mathbf{b}}(R_F) \right) \left( \hat{\mathbf{b}}(z) \otimes \bar{\mathbf{b}}(R_F) \right)^T \mathrm{d}z$$

Let  $P_3 = \begin{bmatrix} 0 & 0 & 1 \end{bmatrix}^T$ . Then

$$p_B(z,t) = P_3^T V_B(z,t) = P_3^T V_{B0}(z,t)$$
  
=  $P_3^T \hat{B}(z) \bar{\nu}_{B0}(t)$  (4.2.93)

Also, let

$$\sigma_R(z) = \frac{\sigma_x(z) + \sigma_y(z)}{2} = \hat{\mathbf{b}}(z)^T \bar{\sigma}_R$$
  

$$\sigma_D(z) = \frac{\sigma_x(z) - \sigma_y(z)}{2} = \hat{\mathbf{b}}(z)^T \bar{\sigma}_D$$
  

$$\sigma_{xy}(z) = \hat{\mathbf{b}}(z)^T \bar{\sigma}_{xy}$$
(4.2.94)

As before,

$$f(r,\theta,z,t) = \vec{b}_1^T \bar{V}(r,\theta,z,t) = \vec{b}_1^T (V(r,\theta,z,t) + W(r,\theta,z,t))$$

and from (4.2.18)-(4.2.20), (4.2.65), (4.2.88), and (4.2.89) it follows that

$$f_{i}(r, z, t) = \vec{b}_{1}^{T} \left( V_{i}(r, z, t) + W_{i}(r, z, t) \right)$$

$$= \vec{b}_{1}^{T} \left( \hat{b}(z)^{T} \otimes B(r) \right) \vec{c}_{i}(t)$$

$$+ \frac{R_{F} - r}{R_{F} - R_{B}} \vec{b}_{1}^{T} \left( \hat{B}(z) - \frac{z - z_{T}}{z_{B} - z_{T}} \hat{B}(z_{B}) - \frac{z_{B} - z}{z_{B} - z_{T}} \hat{B}(z_{T}) \right) \vec{\nu}_{Bi}(t)$$

$$+ \frac{r - R_{B}}{R_{F} - R_{B}} \vec{b}_{1}^{T} \left( \hat{B}(z) - \frac{z - z_{T}}{z_{B} - z_{T}} \hat{B}(z_{B}) - \frac{z_{B} - z}{z_{B} - z_{T}} \hat{B}(z_{T}) \right) \vec{\nu}_{Fi}$$

$$+ \frac{z - z_{T}}{z_{B} - z_{T}} \vec{b}_{1}^{T} B(r) \vec{\nu}_{\tilde{B}i}(t) + \frac{z_{B} - z}{z_{B} - z_{T}} \vec{b}_{1}^{T} B(r) \vec{\nu}_{\tilde{T}i}(t), \quad i = 0, \dots, 4 \quad (4.2.95)$$

and  $\bar{\nu}_{Bi}(t) = 0$ ,  $i = 1, \dots, 4$ .

From (4.2.72)-(4.2.86) we obtain

 $\varphi_0$  :

$$\begin{split} & \left[ \left( K + \frac{4G}{3} \right) g_0^{(z)} \otimes g_2(r) + \left( K - \frac{2G}{3} \right) g_0^{(z)} \otimes \left( g_1^{(r)T} + g_1^{(r)} \right) \right. \\ & \left. + \left( K + \frac{4G}{3} \right) g_0^{(z)} \otimes g_0 \left( \frac{1}{r} \right) + G g_2^{(z)} \otimes g_0(r) + K_F R_F g_F \right] \bar{\mu}_0(t) \\ & \left. + \left[ \left( K - \frac{2G}{3} \right) g_1^{(z)} \otimes g_1(r)^T + G g_1^{(z)T} \otimes g_1(r) + \left( K - \frac{2G}{3} \right) g_1^{(z)} \otimes g_0^{(r)} \right] \bar{\lambda}_0(t) \right. \\ & \left. = \int_{z_T}^{z_B} \int_{R_B}^{R_F} f_0 \left[ \hat{\mathbf{b}}(z) \otimes \left( r \, \bar{\mathbf{b}}_r(r) + \bar{\mathbf{b}}(r) \right) \right] \mathrm{d}r \, \mathrm{d}z - \end{split}$$

$$-R_B \Big[ \int_{z_T}^{z_B} (\hat{\mathbf{b}}(z) \otimes \bar{\mathbf{b}}(R_B)) P_3^T \hat{B}(z) \, \mathrm{d}z \Big] \bar{\nu}_{B0}(t) + R_F \Big( g_0^{(z)} \otimes \bar{\mathbf{b}}(R_F) \Big) \bar{\sigma}_R \qquad (4.2.96)$$

 $\varphi_1$  :

$$\left[ \left( K + \frac{4G}{3} \right) g_0^{(z)} \otimes g_2(r) + \left( K - \frac{2G}{3} \right) g_0^{(z)} \otimes \left( g_1^{(r)T} + g_1^{(r)} \right) \right. \\ \left. + \left( K + \frac{7G}{3} \right) g_0^{(z)} \otimes g_0\left(\frac{1}{r}\right) + Gg_2^{(z)} \otimes g_0(r) + K_F R_F g_F \right] \bar{\mu}_1(t) \\ \left. - \left[ \left( K - \frac{2G}{3} \right) g_0^{(z)} \otimes g_1^{(r)T} - Gg_0^{(z)} \otimes g_1^{(r)} + \left( K + \frac{7G}{3} \right) g_0^{(z)} \otimes g_0\left(\frac{1}{r}\right) \right] \bar{\eta}_2(t) \right. \\ \left. + \left[ \left( K - \frac{2G}{3} \right) g_1^{(z)} \otimes g_1(r)^T + Gg_1^{(z)T} \otimes g_1(r) + \left( K - \frac{2G}{3} \right) g_1^{(z)} \otimes g_0^{(r)} \right] \bar{\lambda}_1(t) \right. \\ \left. + \left[ \left( \tilde{K} - \frac{2G}{3} \right) g_1^{(z)} \otimes g_1(r)^T + Gg_1^{(z)T} \otimes g_1(r) + \left( K - \frac{2G}{3} \right) g_1^{(z)} \otimes g_0^{(r)} \right] \bar{\lambda}_1(t) \right. \\ \left. + \left[ \left( \tilde{b}(z_B) \int_{R_B}^{R_F} f_1\left[ \tilde{b}(z) \otimes \left( r \, \bar{b}_r(r) + \bar{b}(r) \right) \right] dr dz \right. \\ \left. + \left\{ \left( \tilde{b}(z_B) \, \tilde{b}(z_B)^T - \tilde{b}(z_T) \, \tilde{b}(z_T)^T \right) \otimes \left[ \int_{R_B}^{R_F} \left( 1 - \frac{R_B^2}{r^2} \right) r \, \bar{b}(r) \, dr \right] \right\} \bar{\sigma}_{yz} \right] \right]$$

 $\varphi_2$  :

$$\left[ \left( K + \frac{4G}{3} \right) g_0^{(z)} \otimes g_2(r) + \left( K - \frac{2G}{3} \right) g_0^{(z)} \otimes \left( g_1^{(r)T} + g_1^{(r)} \right) \right. \\ \left. + \left( K + \frac{7G}{3} \right) g_0^{(z)} \otimes g_0\left(\frac{1}{r}\right) + Gg_2^{(z)} \otimes g_0(r) + K_F R_F g_F \right] \bar{\mu}_2(t) \right. \\ \left. + \left[ \left( K - \frac{2G}{3} \right) g_0^{(z)} \otimes g_1^{(r)T} - Gg_0^{(z)} \otimes g_1^{(r)} + \left( K + \frac{7G}{3} \right) g_0^{(z)} \otimes g_0\left(\frac{1}{r}\right) \right] \bar{\eta}_1(t) \right. \\ \left. + \left[ \left( K - \frac{2G}{3} \right) g_1^{(z)} \otimes g_1(r)^T + Gg_1^{(z)T} \otimes g_1(r) + \left( K - \frac{2G}{3} \right) g_1^{(z)} \otimes g_0^{(r)} \right] \bar{\lambda}_2(t) \right. \\ \left. + \left[ \left( \tilde{h}(z_B) \hat{h}(z_B)^T - \hat{h}(z_T) \hat{h}(z_T)^T \right) \otimes \left[ \int_{R_B}^{R_F} \left( 1 - \frac{R_B^2}{r^2} \right) r \, \tilde{h}(r) \, \mathrm{d}r \right] \right] \right\} \bar{\sigma}_{xz} \quad (4.2.98)$$

$$\left[ \left( K + \frac{4G}{3} \right) g_0^{(z)} \otimes g_2(r) + \left( K - \frac{2G}{3} \right) g_0^{(z)} \otimes \left( g_1^{(r)T} + g_1^{(r)} \right) \right. \\ \left. + \left( K + \frac{16G}{3} \right) g_0^{(z)} \otimes g_0 \left( \frac{1}{r} \right) + G g_2^{(z)} \otimes g_0(r) + K_F R_F g_F \right] \bar{\mu}_3(t) \right. \\ \left. - 2 \left[ \left( K - \frac{2G}{3} \right) g_0^{(z)} \otimes g_1^{(r)T} - G g_0^{(z)} \otimes g_1^{(r)} + \left( K + \frac{7G}{3} \right) g_0^{(z)} \otimes g_0 \left( \frac{1}{r} \right) \right] \bar{\eta}_4(t) \right. \\ \left. + \left[ \left( K - \frac{2G}{3} \right) g_1^{(z)} \otimes g_1(r)^T + G g_1^{(z)T} \otimes g_1(r) + \left( K - \frac{2G}{3} \right) g_1^{(z)} \otimes g_0^{(r)} \right] \bar{\lambda}_3(t) \right. \\ \left. + \left[ \int_{z_T}^{z_B} \int_{R_B}^{R_F} f_3 \left[ \hat{\mathbf{b}}(z) \otimes (r \, \bar{\mathbf{b}}_r(r) + \bar{\mathbf{b}}(r)) \right] \, \mathrm{d}r \, \mathrm{d}z + R_F \left( g_0^{(z)} \otimes \bar{\mathbf{b}}(R_F) \right) \bar{\sigma}_{xy} \right]$$

$$\varphi_4$$
 :

$$\left[ \left( K + \frac{4G}{3} \right) g_0^{(z)} \otimes g_2(r) + \left( K - \frac{2G}{3} \right) g_0^{(z)} \otimes \left( g_1^{(r)T} + g_1^{(r)} \right) \right. \\ \left. + \left( K + \frac{16G}{3} \right) g_0^{(z)} \otimes g_0 \left( \frac{1}{r} \right) + G g_2^{(z)} \otimes g_0(r) + K_F R_F g_F \right] \bar{\mu}_4(t) \right. \\ \left. + 2 \left[ \left( K - \frac{2G}{3} \right) g_0^{(z)} \otimes g_1^{(r)T} - G g_0^{(z)} \otimes g_1^{(r)} + \left( K + \frac{7G}{3} \right) g_0^{(z)} \otimes g_0 \left( \frac{1}{r} \right) \right] \bar{\eta}_3(t) \right. \\ \left. + \left[ \left( K - \frac{2G}{3} \right) g_1^{(z)} \otimes g_1(r)^T + G g_1^{(z)T} \otimes g_1(r) + \left( K - \frac{2G}{3} \right) g_1^{(z)} \otimes g_0^{(r)} \right] \bar{\lambda}_4(t) \right. \\ \left. + \left[ \int_{z_T}^{z_B} \int_{R_B}^{R_F} f_4 \left[ \hat{\mathbf{b}}(z) \otimes (r \, \bar{\mathbf{b}}_r(r) + \bar{\mathbf{b}}(r)) \right] \, \mathrm{d}r \, \mathrm{d}z + R_F \left( g_0^{(z)} \otimes \bar{\mathbf{b}}(R_F) \right) \bar{\sigma}_D \quad (4.2.100) \right.$$

 $\psi_0$  :

$$\left[ Gg_0^{(z)} \otimes g_2(r) - Gg_0^{(z)} \otimes \left( g_1^{(r)T} + g_1^{(r)} \right) + Gg_0^{(z)} \otimes g_0\left(\frac{1}{r}\right) + Gg_2^{(z)} \otimes g_0(r) \right]$$
  
+  $K_F R_F g_F \left] \bar{\eta}_0(t) = 0$ 

It follows that

$$\bar{\eta}_0(t) = 0$$

 $arphi_3$  :

$$\left[ \left( K - \frac{2G}{3} \right) g_0^{(z)} \otimes g_1^{(r)} - Gg_0^{(z)} \otimes g_1^{(r)T} + \left( K + \frac{7G}{3} \right) g_0^{(z)} \otimes g_0 \left( \frac{1}{r} \right) \right] \bar{\mu}_2(t) + \left[ Gg_0^{(z)} \otimes g_2(r) - Gg_0^{(z)} \otimes \left( g_1^{(r)T} + g_1^{(r)} \right) + \left( K + \frac{7G}{3} \right) g_0^{(z)} \otimes g_0 \left( \frac{1}{r} \right) + Gg_2^{(z)} \otimes g_0(r) + K_F R_F g_F \right] \bar{\eta}_1(t) + \left[ - Gg_1^{(z)T} \otimes g_0^{(r)} + \left( K - \frac{2G}{3} \right) g_1^{(z)} \otimes g_0^{(r)} \right] \bar{\lambda}_2(t) = \int_{z_T}^{z_B} \int_{R_B}^{R_F} f_2 \left[ \hat{\mathbf{b}}(z) \otimes \bar{\mathbf{b}}(r) \right] dr dz - \left\{ \left( \hat{\mathbf{b}}(z_B) \hat{\mathbf{b}}(z_B)^T - \hat{\mathbf{b}}(z_T) \hat{\mathbf{b}}(z_T)^T \right) \otimes \left[ \int_{R_B}^{R_F} \left( 1 + \frac{R_B^2}{r^2} \right) r \bar{\mathbf{b}}(r) dr \right] \right\} \bar{\sigma}_{xz}$$
(4.2.101)

 $\psi_2$  :

$$-\left[\left(K - \frac{2G}{3}\right)g_{0}^{(z)} \otimes g_{1}^{(r)} - Gg_{0}^{(z)} \otimes g_{1}^{(r)T} + \left(K + \frac{7G}{3}\right)g_{0}^{(z)} \otimes g_{0}\left(\frac{1}{r}\right)\right]\bar{\mu}_{1}(t) \\ + \left[Gg_{0}^{(z)} \otimes g_{2}(r) - Gg_{0}^{(z)} \otimes \left(g_{1}^{(r)T} + g_{1}^{(r)}\right) + \left(K + \frac{7G}{3}\right)g_{0}^{(z)} \otimes g_{0}\left(\frac{1}{r}\right) \\ + Gg_{2}^{(z)} \otimes g_{0}(r) + K_{F}R_{F}g_{F}\right]\bar{\eta}_{2}(t) \\ - \left[-Gg_{1}^{(z)T} \otimes g_{0}^{(r)} + \left(K - \frac{2G}{3}\right)g_{1}^{(z)} \otimes g_{0}^{(r)}\right]\bar{\lambda}_{1}(t) \\ = -\int_{z_{T}}^{z_{B}}\int_{R_{B}}^{R_{F}} f_{1}\left[\hat{b}(z) \otimes \bar{b}(r)\right] dr dz \\ + \left\{\left(\hat{b}(z_{B})\hat{b}(z_{B})^{T} - \hat{b}(z_{T})\hat{b}(z_{T})^{T}\right) \otimes \left[\int_{R_{B}}^{R_{F}} \left(1 + \frac{R_{B}^{2}}{r^{2}}\right)r \bar{b}(r) dr\right]\right\}\bar{\sigma}_{yz} \quad (4.2.102)$$

 $\psi_3$  :

$$2\Big[\Big(K - \frac{2G}{3}\Big)g_0^{(z)} \otimes g_1^{(r)} - Gg_0^{(z)} \otimes g_1^{(r)T} + \Big(K + \frac{7G}{3}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big)\Big]\bar{\mu}_4(t) \\ + \Big[Gg_0^{(z)} \otimes g_2(r) - Gg_0^{(z)} \otimes \Big(g_1^{(r)T} + g_1^{(r)}\Big) + \Big(4K + \frac{19G}{3}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big) + \Big(6K + \frac{19G}{3}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big) \Big] + \Big[Gg_0^{(z)} \otimes g_2(r) - Gg_0^{(z)} \otimes g_0^{(r)T} + g_1^{(r)}\Big) + \Big(4K + \frac{19G}{3}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big) + \Big(6K + \frac{19G}{3}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big) \Big] + \Big(6K + \frac{19G}{3}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big) + \Big(6K + \frac{19G}{3}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big) \Big] + \Big(6K + \frac{19G}{3}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big) + \Big(6K + \frac{19G}{3}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big) \Big] + \Big(6K + \frac{19G}{3}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big) + \Big(6K + \frac{19G}{3}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big) \Big) + \Big(6K + \frac{19G}{3}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big) + \Big(6K + \frac{19G}{3}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big) + \Big(6K + \frac{19G}{3}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big) \Big) + \Big(6K + \frac{19G}{3}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big) + \frac{19G}{3}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big) + \frac{19G}{3}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big) + \frac{19G}{3}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big) + \frac{19G}{3}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big) + \frac{19G}{3}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big)g_0^{(z)} \otimes g_0\Big(\frac{1}{r}\Big)g_$$

 $\psi_1$  :

$$+ Gg_{2}^{(z)} \otimes g_{0}(r) + K_{F}R_{F}g_{F} \Big] \bar{\eta}_{3}(t) + 2 \Big[ - Gg_{1}^{(z)T} \otimes g_{0}^{(r)} + \Big(K - \frac{2G}{3}\Big)g_{1}^{(z)} \otimes g_{0}^{(r)} \Big] \bar{\lambda}_{4}(t) = 2 \int_{z_{T}}^{z_{B}} \int_{R_{B}}^{R_{F}} f_{4} \Big[ \hat{\mathbf{b}}(z) \otimes \bar{\mathbf{b}}(r) \Big] dr dz + R_{F} \Big( g_{0}^{(z)} \otimes \bar{\mathbf{b}}(R_{F}) \Big) \bar{\sigma}_{D}$$
(4.2.103)

 $\psi_4$  :

$$-2\Big[\Big(K - \frac{2G}{3}\Big)g_{0}^{(z)} \otimes g_{1}^{(r)} - Gg_{0}^{(z)} \otimes g_{1}^{(r)T} + \Big(K + \frac{7G}{3}\Big)g_{0}^{(z)} \otimes g_{0}\Big(\frac{1}{r}\Big)\Big]\bar{\mu}_{3}(t) \\ + \Big[Gg_{0}^{(z)} \otimes g_{2}(r) - Gg_{0}^{(z)} \otimes \Big(g_{1}^{(r)T} + g_{1}^{(r)}\Big) + \Big(4K + \frac{19G}{3}\Big)g_{0}^{(z)} \otimes g_{0}\Big(\frac{1}{r}\Big) \\ + Gg_{2}^{(z)} \otimes g_{0}(r) + K_{F}R_{F}g_{F}\Big]\bar{\eta}_{4}(t) \\ -2\Big[-Gg_{1}^{(z)T} \otimes g_{0}^{(r)} + \Big(K - \frac{2G}{3}\Big)g_{1}^{(z)} \otimes g_{0}^{(r)}\Big]\bar{\lambda}_{3}(t) \\ = -2\int_{z_{T}}^{z_{B}}\int_{R_{B}}^{R_{F}}f_{3}\Big[\hat{\mathbf{b}}(z) \otimes \bar{\mathbf{b}}(r)\Big]\,\mathrm{d}r\,\mathrm{d}z + R_{F}\Big(g_{0}^{(z)} \otimes \bar{\mathbf{b}}(R_{F})\Big)\bar{\sigma}_{xy} \tag{4.2.104}$$

 $\omega_0$  :

$$\left[ \left( K - \frac{2G}{3} \right) g_1^{(z)T} \otimes g_1(r) + G g_1^{(z)} \otimes g_1(r)^T + \left( K - \frac{2G}{3} \right) g_1^{(z)T} \otimes g_0^{(r)} \right] \bar{\mu}_0(t) + \left[ G g_0^{(z)} \otimes g_2(r) + \left( K + \frac{4G}{3} \right) g_2^{(z)} \otimes g_0(r) + K_F R_F g_F \right] \bar{\lambda}_0(t) = \int_{z_T}^{z_B} \int_{R_B}^{R_F} f_0 \left[ \hat{\mathbf{b}}_z(z) \otimes r \, \bar{\mathbf{b}}(r) \right] \, \mathrm{d}r \, \mathrm{d}z + \left\{ \left( \hat{\mathbf{b}}(z_B) \, \hat{\mathbf{b}}(z_B)^T - \, \hat{\mathbf{b}}(z_T) \, \hat{\mathbf{b}}(z_T)^T \right) \otimes \left[ \int_{R_B}^{R_F} r \, \bar{\mathbf{b}}(r) \, \mathrm{d}r \right] \right\} \bar{\sigma}_{zz}$$
(4.2.105)

 $\omega_1$  :

$$\left[\left(K - \frac{2G}{3}\right)g_1^{(z)T} \otimes g_1(r) + Gg_1^{(z)} \otimes g_1(r)^T + \left(K - \frac{2G}{3}\right)g_1^{(z)T} \otimes g_0^{(r)}\right]\bar{\mu}_1(t) - \frac{2G}{3}g_1^{(z)T} \otimes g_0^{(r)} = 0$$

$$-\left[-Gg_{1}^{(z)}\otimes g_{0}^{(r)}+\left(K-\frac{2G}{3}\right)g_{1}^{(z)T}\otimes g_{0}^{(r)}\right]\bar{\eta}_{2}(t) +\left[Gg_{0}^{(z)}\otimes g_{2}(r)+Gg_{0}^{(z)}\otimes g_{0}\left(\frac{1}{r}\right)+\left(K+\frac{4G}{3}\right)g_{2}^{(z)}\otimes g_{0}(r)+K_{F}R_{F}g_{F}\right]\bar{\lambda}_{1}(t) \\ =\int_{z_{T}}^{z_{B}}\int_{R_{B}}^{R_{F}}f_{1}\left[\hat{\mathbf{b}}_{z}(z)\otimes r\,\bar{\mathbf{b}}(r)\right]\mathrm{d}r\,\mathrm{d}z+R_{F}\left(g_{0}^{(z)}\otimes\bar{\mathbf{b}}(R_{F})\right)\bar{\sigma}_{yz}$$
(4.2.106)

 $\omega_2$  :

$$\left[ \left( K - \frac{2G}{3} \right) g_1^{(z)T} \otimes g_1(r) + G g_1^{(z)} \otimes g_1(r)^T + \left( K - \frac{2G}{3} \right) g_1^{(z)T} \otimes g_0^{(r)} \right] \bar{\mu}_2(t) + \left[ - G g_1^{(z)} \otimes g_0^{(r)} + \left( K - \frac{2G}{3} \right) g_1^{(z)T} \otimes g_0^{(r)} \right] \bar{\eta}_1(t) + \left[ G g_0^{(z)} \otimes g_2(r) + G g_0^{(z)} \otimes g_0\left(\frac{1}{r}\right) + \left( K + \frac{4G}{3} \right) g_2^{(z)} \otimes g_0(r) + K_F R_F g_F \right] \bar{\lambda}_2(t) = \int_{z_T}^{z_B} \int_{R_B}^{R_F} f_2 \left[ \hat{\mathbf{b}}_z(z) \otimes r \, \bar{\mathbf{b}}(r) \right] dr \, dz + R_F \left( g_0^{(z)} \otimes \, \bar{\mathbf{b}}(R_F) \right) \bar{\sigma}_{xz}$$
(4.2.107)

 $\omega_3$  :

$$\left[ \left( K - \frac{2G}{3} \right) g_1^{(z)T} \otimes g_1(r) + G g_1^{(z)} \otimes g_1(r)^T + \left( K - \frac{2G}{3} \right) g_1^{(z)T} \otimes g_0^{(r)} \right] \bar{\mu}_3(t) - 2 \left[ -G g_1^{(z)} \otimes g_0^{(r)} + \left( K - \frac{2G}{3} \right) g_1^{(z)T} \otimes g_0^{(r)} \right] \bar{\eta}_4(t) + \left[ G g_0^{(z)} \otimes g_2(r) + 4G g_0^{(z)} \otimes g_0\left(\frac{1}{r}\right) + \left( K + \frac{4G}{3} \right) g_2^{(z)} \otimes g_0(r) + K_F R_F g_F \right] \bar{\lambda}_3(t) = \int_{z_T}^{z_B} \int_{R_B}^{R_F} f_3 \left[ \hat{\mathbf{b}}_z(z) \otimes r \, \bar{\mathbf{b}}(r) \right] dr dz - 4\nu R_B^2 \left\{ \left( \hat{\mathbf{b}}(z_B) \, \hat{\mathbf{b}}(z_B)^T - \, \hat{\mathbf{b}}(z_T) \, \hat{\mathbf{b}}(z_T)^T \right) \otimes \left[ \int_{R_B}^{R_F} \frac{1}{r} \, \bar{\mathbf{b}}(r) \, dr \right] \right\} \bar{\sigma}_{xy}$$
(4.2.108)

 $\omega_4$  :

$$\left[\left(K - \frac{2G}{3}\right)g_1^{(z)T} \otimes g_1(r) + Gg_1^{(z)} \otimes g_1(r)^T + \left(K - \frac{2G}{3}\right)g_1^{(z)T} \otimes g_0^{(r)}\right]\bar{\mu}_4(t) + \frac{2G}{3}g_1^{(z)T} \otimes g_0^{(r)} = \frac{1}{2}g_1^{(z)T} \otimes g_1^{(r)} + \frac{1}{2}g_1^{(r)} \otimes g_1^{(r)} \otimes g_1^{(r)} + \frac{1}{2}g_1^{(r)} \otimes g_1^{(r)} + \frac{1}{2}g_1^{(r)} \otimes g_1^{(r)} \otimes g_1^{(r)} + \frac{1}{2}g_1^{(r)} \otimes g_1^{(r)} \otimes g_1^{(r)} + \frac{1}{2}g_1^{(r)} \otimes g_1^{(r)} \otimes g_1^{(r)} \otimes g_1^{(r)} + \frac{1}{2}g_1^{(r)} \otimes g_1^{(r)} \otimes g$$

$$+2\Big[-Gg_{1}^{(z)}\otimes g_{0}^{(r)} + \left(K - \frac{2G}{3}\right)g_{1}^{(z)T}\otimes g_{0}^{(r)}\Big]\bar{\eta}_{3}(t) \\ +\Big[Gg_{0}^{(z)}\otimes g_{2}(r) + 4Gg_{0}^{(z)}\otimes g_{0}\left(\frac{1}{r}\right) + \left(K + \frac{4G}{3}\right)g_{2}^{(z)}\otimes g_{0}(r) + K_{F}R_{F}g_{F}\Big]\bar{\lambda}_{4}(t) \\ = \int_{z_{T}}^{z_{B}}\int_{R_{B}}^{R_{F}}f_{4}\Big[\hat{b}_{z}(z)\otimes r\,\bar{b}(r)\Big]\,dr\,dz \\ -4\nu R_{B}^{2}\Big\{\Big(\hat{b}(z_{B})\,\hat{b}(z_{B})^{T} - \,\hat{b}(z_{T})\,\hat{b}(z_{T})^{T}\Big)\otimes\Big[\int_{R_{B}}^{R_{F}}\frac{1}{r}\,\bar{b}(r)\,dr\Big]\Big\}\bar{\sigma}_{D} \qquad (4.2.109)$$

Define the following matrices.

$$M_r M_z \times M_r M_z$$
 matrices:

$$\mathcal{M}_{0} = \left(K + \frac{4G}{3}\right)g_{0}^{(z)} \otimes g_{2}(r) + \left(K - \frac{2G}{3}\right)g_{0}^{(z)} \otimes \left(g_{1}^{(r)T} + g_{1}^{(r)}\right) \\ + \left(K + \frac{4G}{3}\right)g_{0}^{(z)} \otimes g_{0}\left(\frac{1}{r}\right) + Gg_{2}^{(z)} \otimes g_{0}(r) + K_{F}R_{F}g_{F}$$

$$\mathcal{M}_{1} = \left(K + \frac{4G}{3}\right)g_{0}^{(z)} \otimes g_{2}(r) + \left(K - \frac{2G}{3}\right)g_{0}^{(z)} \otimes \left(g_{1}^{(r)T} + g_{1}^{(r)}\right) \\ + \left(K + \frac{7G}{3}\right)g_{0}^{(z)} \otimes g_{0}\left(\frac{1}{r}\right) + Gg_{2}^{(z)} \otimes g_{0}(r) + K_{F}R_{F}g_{F}$$

$$\mathcal{M}_{2} = \left(K + \frac{4G}{3}\right)g_{0}^{(z)} \otimes g_{2}(r) + \left(K - \frac{2G}{3}\right)g_{0}^{(z)} \otimes \left(g_{1}^{(r)T} + g_{1}^{(r)}\right) \\ + \left(K + \frac{16G}{3}\right)g_{0}^{(z)} \otimes g_{0}\left(\frac{1}{r}\right) + Gg_{2}^{(z)} \otimes g_{0}(r) + K_{F}R_{F}g_{F}$$

$$\mathcal{M}_{3} = \left(K - \frac{2G}{3}\right)g_{1}^{(z)} \otimes g_{1}(r)^{T} + Gg_{1}^{(z)T} \otimes g_{1}(r) + \left(K - \frac{2G}{3}\right)g_{1}^{(z)} \otimes g_{0}^{(r)}$$
$$\mathcal{N}_{1} = \left(K - \frac{2G}{3}\right)g_{0}^{(z)} \otimes g_{1}^{(r)T} - Gg_{0}^{(z)} \otimes g_{1}^{(r)} + \left(K + \frac{7G}{3}\right)g_{0}^{(z)} \otimes g_{0}\left(\frac{1}{r}\right)$$
$$\mathcal{N}_{2} = Gg_{0}^{(z)} \otimes g_{2}(r) - Gg_{0}^{(z)} \otimes \left(g_{1}^{(r)T} + g_{1}^{(r)}\right) + \left(K + \frac{7G}{3}\right)g_{0}^{(z)} \otimes g_{0}\left(\frac{1}{r}\right)$$
$$+ Gg_{2}^{(z)} \otimes g_{0}(r) + K_{F}R_{F}g_{F}$$

$$\mathcal{N}_{3} = Gg_{0}^{(z)} \otimes g_{2}(r) - Gg_{0}^{(z)} \otimes \left(g_{1}^{(r)T} + g_{1}^{(r)}\right) + \left(4K + \frac{19G}{3}\right)g_{0}^{(z)} \otimes g_{0}\left(\frac{1}{r}\right) + Gg_{2}^{(z)} \otimes g_{0}(r) + K_{F}R_{F}g_{F}$$

$$\mathcal{L}_{0} = Gg_{0}^{(z)} \otimes g_{2}(r) + \left(K + \frac{4G}{3}\right)g_{2}^{(z)} \otimes g_{0}(r) + K_{F}R_{F}g_{F}$$

$$\mathcal{L}_{1} = -Gg_{1}^{(z)T} \otimes g_{0}^{(r)} + \left(K - \frac{2G}{3}\right)g_{1}^{(z)} \otimes g_{0}^{(r)}$$

$$\mathcal{L}_{2} = Gg_{0}^{(z)} \otimes g_{2}(r) + Gg_{0}^{(z)} \otimes g_{0}\left(\frac{1}{r}\right) + \left(K + \frac{4G}{3}\right)g_{2}^{(z)} \otimes g_{0}(r) + K_{F}R_{F}g_{F}$$

$$\mathcal{L}_{3} = Gg_{0}^{(z)} \otimes g_{2}(r) + 4Gg_{0}^{(z)} \otimes g_{0}\left(\frac{1}{r}\right) + \left(K + \frac{4G}{3}\right)g_{2}^{(z)} \otimes g_{0}(r) + K_{F}R_{F}g_{F}$$

$$(4.2.110)$$

# $M_r M_z \times 3 M_{Dr} M_{Dz}$ matrices:

$$\mathcal{Z}_{0} = \int_{z_{T}}^{z_{B}} \int_{R_{B}}^{R_{F}} \left[ \hat{\mathbf{b}}(z) \otimes \left( r \, \bar{\mathbf{b}}_{r}(r) + \bar{\mathbf{b}}(r) \right) \right] \vec{b}_{1}^{T} \left( \hat{b}(z)^{T} \otimes B(r) \right) \mathrm{d}r \, \mathrm{d}z$$
$$= \left[ \int_{z_{T}}^{z_{B}} \hat{\mathbf{b}}(z) \hat{b}(z)^{T} \, \mathrm{d}z \right] \otimes \left[ \int_{R_{B}}^{R_{F}} \left( r \, \bar{\mathbf{b}}_{r}(r) + \bar{\mathbf{b}}(r) \right) \vec{b}_{1}^{T} B(r) \, \mathrm{d}r \right]$$
(4.2.111)

$$\mathcal{Z}_{1} = \int_{z_{T}}^{z_{B}} \int_{R_{B}}^{R_{F}} \left[ \hat{\mathbf{b}}(z) \otimes \bar{\mathbf{b}}(r) \right] \vec{b}_{1}^{T} \left( \hat{b}(z)^{T} \otimes B(r) \right) \mathrm{d}r \,\mathrm{d}z$$
$$= \left[ \int_{z_{T}}^{z_{B}} \hat{\mathbf{b}}(z) \hat{b}(z)^{T} \,\mathrm{d}z \right] \otimes \left[ \int_{R_{B}}^{R_{F}} \bar{\mathbf{b}}(r) \vec{b}_{1}^{T} B(r) \,\mathrm{d}r \right]$$
(4.2.112)

$$\mathcal{Z}_{2} = \int_{z_{T}}^{z_{B}} \int_{R_{B}}^{R_{F}} \left[ \hat{\mathbf{b}}_{z}(z) \otimes r \,\bar{\mathbf{b}}(r) \right] \vec{b}_{1}^{T} \left( \hat{b}(z)^{T} \otimes B(r) \right) \mathrm{d}r \,\mathrm{d}z$$
$$= \left[ \int_{z_{T}}^{z_{B}} \hat{\mathbf{b}}_{z}(z) \hat{b}(z)^{T} \,\mathrm{d}z \right] \otimes \left[ \int_{R_{B}}^{R_{F}} r \,\bar{\mathbf{b}}(r) \vec{b}_{1}^{T} B(r) \,\mathrm{d}r \right]$$
(4.2.113)

 $M_r M_z \times 3 M_{Dz}$  matrices:

$$Y_{B0} = \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ \hat{\mathbf{b}}(z) \otimes \left( r \, \bar{\mathbf{b}}_r(r) + \bar{\mathbf{b}}(r) \right) \right] \frac{R_F - r}{R_F - R_B} \vec{b}_1^T$$

$$\cdot \left[ \hat{B}(z) - \frac{z - z_T}{z_B - z_T} \hat{B}(z_B) - \frac{z_B - z}{z_B - z_T} \hat{B}(z_T) \right] dr dz$$

$$= \left[ \int_{z_T}^{z_B} \hat{\mathbf{b}}(z) \vec{b}_1^T \left( \hat{B}(z) - \frac{z - z_T}{z_B - z_T} \hat{B}(z_B) - \frac{z_B - z}{z_B - z_T} \hat{B}(z_T) \right) dz \right]$$

$$\otimes \left[ \int_{R_B}^{R_F} \left( r \, \bar{\mathbf{b}}_r(r) + \bar{\mathbf{b}}(r) \right) \frac{R_F - r}{R_F - R_B} dr \right]$$
(4.2.114)

Similarly,

$$Y_{F0} = \left[ \int_{z_T}^{z_B} \hat{\mathbf{b}}(z) \vec{b}_1^T \left( \hat{B}(z) - \frac{z - z_T}{z_B - z_T} \hat{B}(z_B) - \frac{z_B - z}{z_B - z_T} \hat{B}(z_T) \right) dz \right]$$
$$\otimes \left[ \int_{R_B}^{R_F} \left( r \, \bar{\mathbf{b}}_r(r) + \bar{\mathbf{b}}(r) \right) \frac{r - R_B}{R_F - R_B} dr \right]$$
(4.2.115)

$$Y_{B1} = \left[ \int_{z_T}^{z_B} \hat{\mathbf{b}}(z) \vec{b}_1^T \left( \hat{B}(z) - \frac{z - z_T}{z_B - z_T} \hat{B}(z_B) - \frac{z_B - z}{z_B - z_T} \hat{B}(z_T) \right) dz \right] \\ \otimes \left[ \int_{R_B}^{R_F} \bar{\mathbf{b}}(r) \frac{R_F - r}{R_F - R_B} dr \right]$$
(4.2.116)

$$Y_{F1} = \left[ \int_{z_T}^{z_B} \hat{\mathbf{b}}(z) \vec{b}_1^T \left( \hat{B}(z) - \frac{z - z_T}{z_B - z_T} \hat{B}(z_B) - \frac{z_B - z}{z_B - z_T} \hat{B}(z_T) \right) dz \right] \\ \otimes \left[ \int_{R_B}^{R_F} \bar{\mathbf{b}}(r) \frac{r - R_B}{R_F - R_B} dr \right]$$
(4.2.117)

$$Y_{B2} = \left[ \int_{z_T}^{z_B} \hat{\mathbf{b}}_z(z) \vec{b}_1^T \left( \hat{B}(z) - \frac{z - z_T}{z_B - z_T} \hat{B}(z_B) - \frac{z_B - z}{z_B - z_T} \hat{B}(z_T) \right) dz \right] \\ \otimes \left[ \int_{R_B}^{R_F} r \, \bar{\mathbf{b}}(r) \frac{R_F - r}{R_F - R_B} dr \right]$$
(4.2.118)

$$Y_{F2} = \left[ \int_{z_T}^{z_B} \hat{\mathbf{b}}_z(z) \vec{b}_1^T \left( \hat{B}(z) - \frac{z - z_T}{z_B - z_T} \hat{B}(z_B) - \frac{z_B - z}{z_B - z_T} \hat{B}(z_T) \right) dz \right]$$
$$\otimes \left[ \int_{R_B}^{R_F} r \, \bar{\mathbf{b}}(r) \frac{r - R_B}{R_F - R_B} dr \right]$$
(4.2.119)

$$S_{0} = R_{B} \int_{z_{T}}^{z_{B}} \left( \hat{\mathbf{b}}(z) \otimes \bar{\mathbf{b}}(R_{B}) P_{3}^{T} \hat{B}(z) \, \mathrm{d}z \right)$$
$$= R_{B} \left[ \int_{z_{T}}^{z_{B}} \hat{\mathbf{b}}(z) P_{3}^{T} \hat{B}(z) \, \mathrm{d}z \right] \otimes \bar{\mathbf{b}}(R_{B})$$
(4.2.120)

 $M_r M_z \times 3 M_{Dr}$  matrices:

$$\tilde{Y}_{B0} = \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left[ \hat{\mathbf{b}}(z) \otimes \left( r \, \bar{\mathbf{b}}_r(r) + \, \bar{\mathbf{b}}(r) \right) \right] \frac{z - z_T}{z_B - z_T} \vec{b}_1^T B(r) \, \mathrm{d}r \, \mathrm{d}z$$
$$= \left[ \int_{z_T}^{z_B} \hat{\mathbf{b}}(z) \frac{z - z_T}{z_B - z_T} \, \mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} \left( r \, \bar{\mathbf{b}}_r(r) + \, \bar{\mathbf{b}}(r) \right) \vec{b}_1^T B(r) \, \mathrm{d}r \right]$$
(4.2.121)

Similarly,

$$\tilde{Y}_{T0} = \left[ \int_{z_T}^{z_B} \hat{\mathbf{b}}(z) \frac{z_B - z}{z_B - z_T} \, \mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} \left( r \, \bar{\mathbf{b}}_r(r) + \, \bar{\mathbf{b}}(r) \right) \vec{b}_1^T B(r) \, \mathrm{d}r \right]$$
(4.2.122)

$$\tilde{Y}_{B1} = \left[\int_{z_T}^{z_B} \hat{\mathbf{b}}(z) \frac{z - z_T}{z_B - z_T} \,\mathrm{d}z\right] \otimes \left[\int_{R_B}^{R_F} \bar{\mathbf{b}}(r) \vec{b}_1^T B(r) \,\mathrm{d}r\right] \tag{4.2.123}$$

$$\tilde{Y}_{T1} = \left[\int_{z_T}^{z_B} \hat{\mathbf{b}}(z) \frac{z_B - z}{z_B - z_T} \,\mathrm{d}z\right] \otimes \left[\int_{R_B}^{R_F} \bar{\mathbf{b}}(r) \vec{b}_1^T B(r) \,\mathrm{d}r\right]$$
(4.2.124)

$$\tilde{Y}_{B2} = \left[ \int_{z_T}^{z_B} \hat{\mathbf{b}}_z(z) \frac{z - z_T}{z_B - z_T} \, \mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} r \, \bar{\mathbf{b}}_r(r) \vec{b}_1^T B(r) \, \mathrm{d}r \right]$$
(4.2.125)

$$\tilde{Y}_{T2} = \left[\int_{z_T}^{z_B} \hat{\mathbf{b}}_z(z) \frac{z_B - z}{z_B - z_T} \,\mathrm{d}z\right] \otimes \left[\int_{R_B}^{R_F} r \,\bar{\mathbf{b}}_r(r) \vec{b}_1^T B(r) \,\mathrm{d}r\right]$$
(4.2.126)

## $M_r M_z \times M_{Dz}$ matrices:

$$S_{RF} = R_F g_0^{(z)} \otimes \bar{\mathbf{b}}(R_F) \tag{4.2.127}$$

$$S_{1} = \left(\hat{\mathbf{b}}(z_{B})\,\hat{\mathbf{b}}(z_{B})^{T} - \,\hat{\mathbf{b}}(z_{T})\,\hat{\mathbf{b}}(z_{T})^{T}\right) \otimes \left[\int_{R_{B}}^{R_{F}} \left(1 - \frac{R_{B}^{2}}{r^{2}}\right) r\,\bar{\mathbf{b}}(r)\,\mathrm{d}r\right]$$
(4.2.128)

$$S_{2} = -\left(\hat{\mathbf{b}}(z_{B})\,\hat{\mathbf{b}}(z_{B})^{T} - \,\hat{\mathbf{b}}(z_{T})\,\hat{\mathbf{b}}(z_{T})^{T}\right) \otimes \left[\int_{R_{B}}^{R_{F}} \left(1 + \frac{R_{B}^{2}}{r^{2}}\right) r\,\bar{\mathbf{b}}(r)\,\mathrm{d}r\right] \quad (4.2.129)$$

$$S_3 = \left(\hat{\mathbf{b}}(z_B)\,\hat{\mathbf{b}}(z_B)^T - \,\hat{\mathbf{b}}(z_T)\,\hat{\mathbf{b}}(z_T)^T\right) \otimes \left[\int_{R_B}^{R_F} r\,\bar{\mathbf{b}}(r)\,\mathrm{d}r\right]$$
(4.2.130)

$$S_4 = -4\nu R_B^2 \Big( \hat{\mathbf{b}}(z_B) \, \hat{\mathbf{b}}(z_B)^T - \, \hat{\mathbf{b}}(z_T) \, \hat{\mathbf{b}}(z_T)^T \Big) \otimes \left[ \int_{R_B}^{R_F} r \, \bar{\mathbf{b}}(r) \, \mathrm{d}r \right]$$
(4.2.131)

Applying (4.2.110)-(4.2.131) to (4.2.96)-(4.2.109), we obtain the following matrix equations:

$$\begin{bmatrix} \mathcal{M}_{0} & \mathcal{M}_{3} \\ \mathcal{M}_{3}^{T} & \mathcal{L}_{0} \end{bmatrix} \begin{bmatrix} \bar{\mu}_{0}(t) \\ \bar{\lambda}_{0}(t) \end{bmatrix} = \begin{bmatrix} \mathcal{Z}_{0} \\ \mathcal{Z}_{2} \end{bmatrix} \bar{c}_{0}(t) + \begin{bmatrix} S_{RF} \\ 0 \end{bmatrix} \bar{\sigma}_{R} + \begin{bmatrix} 0 \\ S_{3} \end{bmatrix} \bar{\sigma}_{zz}$$

$$+ \begin{bmatrix} Y_{B0} - S_{0} \\ Y_{B2} \end{bmatrix} \bar{\nu}_{B0}(t) + \begin{bmatrix} Y_{F0} \\ Y_{F2} \end{bmatrix} \bar{\nu}_{F0} + \begin{bmatrix} \tilde{Y}_{B0} \\ \tilde{Y}_{B2} \end{bmatrix} \bar{\nu}_{\bar{B}0}(t) + \begin{bmatrix} \tilde{Y}_{T0} \\ \tilde{Y}_{T2} \end{bmatrix} \bar{\nu}_{\bar{T}0}(t) \quad (4.2.132)$$

$$\begin{bmatrix} \mathcal{M}_{1} & -\mathcal{N}_{1} & \mathcal{M}_{3} \\ -\mathcal{N}_{1}^{T} & \mathcal{N}_{2} & -\mathcal{L}_{1} \\ \mathcal{M}_{3}^{T} & -\mathcal{L}_{1}^{T} & \mathcal{L}_{2} \end{bmatrix} \begin{bmatrix} \bar{\mu}_{1}(t) \\ \bar{\eta}_{2}(t) \\ \bar{\lambda}_{1}(t) \end{bmatrix} = \begin{bmatrix} \mathcal{Z}_{0} \\ -\mathcal{Z}_{1} \\ \mathcal{Z}_{2} \end{bmatrix} \bar{c}_{1}(t) + \begin{bmatrix} S_{1} \\ -S_{2} \\ S_{RF} \end{bmatrix} \bar{\sigma}_{yz}$$

$$+ \begin{bmatrix} Y_{F0} \\ -Y_{F1} \\ Y_{F2} \end{bmatrix} \bar{\nu}_{F1} + \begin{bmatrix} \tilde{Y}_{B0} \\ -\tilde{Y}_{B1} \\ \tilde{Y}_{B2} \end{bmatrix} \bar{\nu}_{\bar{B}1}(t) + \begin{bmatrix} \tilde{Y}_{T0} \\ -\tilde{Y}_{T1} \\ \tilde{Y}_{T2} \end{bmatrix} \bar{\nu}_{\bar{T}1}(t) \quad (4.2.133)$$

$$\begin{bmatrix} \mathcal{M}_{1} & \mathcal{M}_{1} & \mathcal{M}_{3} \\ \mathcal{N}_{1}^{T} & \mathcal{N}_{2} & \mathcal{L}_{1} \\ \mathcal{M}_{3}^{T} & \mathcal{L}_{1}^{T} & \mathcal{L}_{2} \end{bmatrix} \begin{bmatrix} \bar{\mu}_{2}(t) \\ \bar{\eta}_{1}(t) \\ \bar{\lambda}_{2}(t) \end{bmatrix} = \begin{bmatrix} \mathcal{Z}_{0} \\ \mathcal{Z}_{1} \\ \mathcal{Z}_{2} \end{bmatrix} \bar{c}_{2}(t) + \begin{bmatrix} S_{1} \\ S_{2} \\ S_{2} \end{bmatrix} \bar{\sigma}_{xz}$$
$$+ \begin{bmatrix} Y_{F0} \\ Y_{F1} \\ Y_{F2} \end{bmatrix} \bar{\nu}_{F2} + \begin{bmatrix} \tilde{Y}_{B0} \\ \tilde{Y}_{B1} \\ \tilde{Y}_{B2} \end{bmatrix} \bar{\nu}_{\tilde{B}2}(t) + \begin{bmatrix} \tilde{Y}_{T0} \\ \tilde{Y}_{T1} \\ \tilde{Y}_{T2} \end{bmatrix} \bar{\nu}_{\tilde{T}2}(t) \qquad (4.2.134)$$

$$\begin{bmatrix} \mathcal{M}_{2} & -2\mathcal{N}_{1} & \mathcal{M}_{3} \\ -2\mathcal{N}_{1}^{T} & \mathcal{N}_{3} & -2\mathcal{L}_{1} \\ \mathcal{M}_{3}^{T} & -2\mathcal{L}_{1}^{T} & \mathcal{L}_{3} \end{bmatrix} \begin{bmatrix} \bar{\mu}_{3}(t) \\ \bar{\eta}_{4}(t) \\ \bar{\lambda}_{3}(t) \end{bmatrix} = \begin{bmatrix} \mathcal{Z}_{0} \\ -2\mathcal{Z}_{1} \\ \mathcal{Z}_{2} \end{bmatrix} \bar{c}_{3}(t) + \begin{bmatrix} S_{RF} \\ S_{RF} \\ S_{4} \end{bmatrix} \bar{\sigma}_{xy}$$

$$+ \begin{bmatrix} Y_{F0} \\ -2Y_{F1} \\ Y_{F2} \end{bmatrix} \bar{\nu}_{F3} + \begin{bmatrix} \tilde{Y}_{B0} \\ -2\tilde{Y}_{B1} \\ \tilde{Y}_{B2} \end{bmatrix} \bar{\nu}_{\tilde{B}3}(t) + \begin{bmatrix} \tilde{Y}_{T0} \\ -2\tilde{Y}_{T1} \\ \tilde{Y}_{T2} \end{bmatrix} \bar{\nu}_{\tilde{T}3}(t) \qquad (4.2.135)$$

$$\begin{bmatrix} \mathcal{M}_{2} & 2\mathcal{N}_{1} & \mathcal{M}_{3} \\ 2\mathcal{N}_{1}^{T} & \mathcal{N}_{3} & 2\mathcal{L}_{1} \\ \mathcal{M}_{3}^{T} & 2\mathcal{L}_{1}^{T} & \mathcal{L}_{3} \end{bmatrix} \begin{bmatrix} \bar{\mu}_{4}(t) \\ \bar{\eta}_{3}(t) \\ \bar{\lambda}_{4}(t) \end{bmatrix} = \begin{bmatrix} \mathcal{Z}_{0} \\ 2\mathcal{Z}_{1} \\ \mathcal{Z}_{2} \end{bmatrix} \bar{c}_{4}(t) + \begin{bmatrix} S_{RF} \\ S_{RF} \\ S_{4} \end{bmatrix} \bar{\sigma}_{D}$$
$$+ \begin{bmatrix} Y_{F0} \\ 2Y_{F1} \\ Y_{F2} \end{bmatrix} \bar{\nu}_{F4} + \begin{bmatrix} \tilde{Y}_{B0} \\ 2\tilde{Y}_{B1} \\ \tilde{Y}_{B2} \end{bmatrix} \bar{\nu}_{\tilde{B}4}(t) + \begin{bmatrix} \tilde{Y}_{T0} \\ 2\tilde{Y}_{T1} \\ \tilde{Y}_{T2} \end{bmatrix} \bar{\nu}_{\tilde{T}4}(t) \qquad (4.2.136)$$

Define the following matrices:

$$\begin{aligned} \mathcal{Q}_{0} &= \begin{bmatrix} \mathcal{M}_{0} & \mathcal{M}_{3} \\ \mathcal{M}_{3}^{T} & \mathcal{L}_{0} \end{bmatrix} \\ J &= \begin{bmatrix} I_{M_{r}M_{z}} & 0 & 0 \\ 0 & -I_{M_{r}M_{z}} & 0 \\ 0 & 0 & I_{M_{r}M_{z}} \end{bmatrix} \\ \mathcal{Q}_{1} &= \begin{bmatrix} \mathcal{M}_{1} & \mathcal{N}_{1} & \mathcal{M}_{3} \\ \mathcal{N}_{1}^{T} & \mathcal{N}_{2} & \mathcal{L}_{1} \\ \mathcal{M}_{3}^{T} & \mathcal{L}_{1}^{T} & \mathcal{L}_{2} \end{bmatrix} \Rightarrow \begin{bmatrix} \mathcal{M}_{1} & -\mathcal{N}_{1} & \mathcal{M}_{3} \\ -\mathcal{N}_{1}^{T} & \mathcal{N}_{2} & -\mathcal{L}_{1} \\ \mathcal{M}_{3}^{T} & -\mathcal{L}_{1}^{T} & \mathcal{L}_{2} \end{bmatrix} = J\mathcal{Q}_{1}J \\ \mathcal{Q}_{2} &= \begin{bmatrix} \mathcal{M}_{2} & 2\mathcal{N}_{1} & \mathcal{M}_{3} \\ 2\mathcal{N}_{1}^{T} & \mathcal{N}_{3} & 2\mathcal{L}_{1} \\ \mathcal{M}_{3}^{T} & 2\mathcal{L}_{1}^{T} & \mathcal{L}_{3} \end{bmatrix} \Rightarrow \begin{bmatrix} \mathcal{M}_{2} & -2\mathcal{N}_{1} & \mathcal{M}_{3} \\ -2\mathcal{N}_{1}^{T} & \mathcal{N}_{3} & -2\mathcal{L}_{1} \\ \mathcal{M}_{3}^{T} & -2\mathcal{L}_{1}^{T} & \mathcal{L}_{3} \end{bmatrix} = J\mathcal{Q}_{2}J \\ \tilde{\mathcal{L}}_{0} &= \begin{bmatrix} \mathcal{Z}_{0} \\ \mathcal{Z}_{2} \\ \mathcal{Z}_{1} \end{bmatrix}, \qquad \tilde{\mathcal{Z}}_{1} &= \begin{bmatrix} \mathcal{Z}_{0} \\ \mathcal{Z}_{1} \\ \mathcal{Z}_{2} \end{bmatrix} = J\mathcal{Q}_{2}J \\ \tilde{\mathcal{L}}_{0} &= \begin{bmatrix} \mathcal{Z}_{0} \\ -\mathcal{Z}_{1} \\ \mathcal{Z}_{2} \end{bmatrix} = J\tilde{\mathcal{Z}}_{1}, \qquad \begin{bmatrix} \mathcal{Z}_{0} \\ -2\mathcal{Z}_{1} \\ \mathcal{Z}_{2} \end{bmatrix} = J\tilde{\mathcal{Z}} \\ \mathcal{X}_{B0} &= \begin{bmatrix} Y_{B0} - \mathcal{S}_{0} \\ Y_{B2} \end{bmatrix}, \qquad X_{F0} &= \begin{bmatrix} Y_{F0} \\ Y_{F2} \\ Y_{F2} \end{bmatrix}, \qquad X_{F1} &= \begin{bmatrix} Y_{F0} \\ Y_{F1} \\ Y_{F2} \end{bmatrix}, \qquad X_{F2} &= \begin{bmatrix} Y_{F0} \\ 2Y_{F1} \\ Y_{F2} \end{bmatrix} \Rightarrow \\ \begin{bmatrix} Y_{F0} \\ -Y_{F1} \\ Y_{F2} \end{bmatrix} = JX_{F1}, \qquad \begin{bmatrix} Y_{F0} \\ -2Y_{F1} \\ Y_{F2} \end{bmatrix} = JX_{F2} \end{aligned}$$

$$\begin{split} \tilde{X}_{B0} &= \begin{bmatrix} \tilde{Y}_{B0} \\ \tilde{Y}_{B2} \end{bmatrix}, \qquad \tilde{X}_{B1} = \begin{bmatrix} \tilde{Y}_{B0} \\ \tilde{Y}_{B1} \\ \tilde{Y}_{B2} \end{bmatrix}, \qquad \tilde{X}_{B2} = \begin{bmatrix} \tilde{Y}_{B0} \\ 2\tilde{Y}_{B1} \\ \tilde{Y}_{B2} \end{bmatrix} \Rightarrow \\ \begin{bmatrix} \tilde{Y}_{B0} \\ -\tilde{Y}_{B1} \\ \tilde{Y}_{B2} \end{bmatrix} = J\tilde{X}_{B1}, \qquad \begin{bmatrix} \tilde{Y}_{B0} \\ -2\tilde{Y}_{B1} \\ \tilde{Y}_{B2} \end{bmatrix} = J\tilde{X}_{B2} \\ \tilde{X}_{T0} &= \begin{bmatrix} \tilde{Y}_{T0} \\ \tilde{Y}_{T2} \end{bmatrix}, \qquad \tilde{X}_{T1} = \begin{bmatrix} \tilde{Y}_{T0} \\ \tilde{Y}_{T1} \\ \tilde{Y}_{T2} \end{bmatrix}, \qquad \tilde{X}_{T2} = \begin{bmatrix} \tilde{Y}_{T0} \\ 2\tilde{Y}_{T1} \\ \tilde{Y}_{T2} \end{bmatrix} \Rightarrow \\ \begin{bmatrix} \tilde{Y}_{T0} \\ -\tilde{Y}_{T1} \\ \tilde{Y}_{T2} \end{bmatrix} = J\tilde{X}_{T1}, \qquad \begin{bmatrix} \tilde{Y}_{T0} \\ -2\tilde{Y}_{T1} \\ \tilde{Y}_{T2} \end{bmatrix} = J\tilde{X}_{T2} \\ \begin{bmatrix} \tilde{Y}_{T0} \\ -\tilde{Y}_{T1} \\ \tilde{Y}_{T2} \end{bmatrix} = J\tilde{X}_{T1}, \qquad \begin{bmatrix} \tilde{Y}_{T0} \\ -2\tilde{Y}_{T1} \\ \tilde{Y}_{T2} \end{bmatrix} = J\tilde{X}_{T2} \\ \bar{X}_{T2} \end{bmatrix} \\ \bar{S}_{0} &= \begin{bmatrix} S_{RF} \\ 0 \end{bmatrix}, \quad \bar{S}_{1} = \begin{bmatrix} S_{1} \\ S_{2} \\ S_{RF} \end{bmatrix} \Rightarrow \begin{bmatrix} S_{1} \\ -S_{2} \\ S_{RF} \end{bmatrix} = J\bar{S}_{1}, \quad \bar{S}_{2} = \begin{bmatrix} S_{RF} \\ S_{RF} \\ S_{4} \end{bmatrix}, \quad \bar{S}_{3} = \begin{bmatrix} 0 \\ S_{3} \end{bmatrix} \end{split}$$

 $P_{100} = \begin{bmatrix} 1 & 0 & 0 \end{bmatrix} \otimes I_{M_r M_z \times M_r M_z}$   $P_{010} = \begin{bmatrix} 0 & 1 & 0 \end{bmatrix} \otimes I_{M_r M_z \times M_r M_z}$   $P_{001} = \begin{bmatrix} 0 & 0 & 1 \end{bmatrix} \otimes I_{M_r M_z \times M_r M_z}$   $P_{10} = \begin{bmatrix} 1 & 0 \end{bmatrix} \otimes I_{M_r M_z \times M_r M_z}$   $P_{01} = \begin{bmatrix} 0 & 1 \end{bmatrix} \otimes I_{M_r M_z \times M_r M_z}$ 

From (4.2.132)-(4.2.136) we have

$$\mathcal{Q}_{0}\begin{bmatrix}\bar{\mu}_{0}(t)\\\bar{\lambda}_{0}(t)\end{bmatrix} = \bar{\mathcal{Z}}_{0}\bar{c}_{0}(t) + \bar{S}_{0}\bar{\sigma}_{R} + \bar{S}_{3}\bar{\sigma}_{zz} + X_{B0}\bar{\nu}_{B0}(t) + X_{F0}\bar{\nu}_{F0} + \tilde{X}_{B0}\bar{\nu}_{\tilde{B}0}(t) + \tilde{X}_{T0}\bar{\nu}_{\tilde{T}0}(t) \qquad (4.2.137)$$

$$JQ_{1}J\begin{bmatrix}\bar{\mu}_{1}(t)\\\bar{\eta}_{2}(t)\\\bar{\lambda}_{1}(t)\end{bmatrix} = J\bar{Z}_{1}\bar{c}_{1}(t) + J\bar{S}_{1}\bar{\sigma}_{yz} + JX_{F1}\bar{\nu}_{F1} + J\tilde{X}_{B1}\bar{\nu}_{\tilde{B}1}(t) + J\tilde{X}_{T1}\bar{\nu}_{\tilde{T}1}(t)$$
(4.2.138)

$$\mathcal{Q}_{1} \begin{bmatrix} \bar{\mu}_{2}(t) \\ \bar{\eta}_{1}(t) \\ \bar{\lambda}_{2}(t) \end{bmatrix} = \bar{\mathcal{Z}}_{1} \bar{c}_{2}(t) + \bar{S}_{1} \bar{\sigma}_{xz} \\ + X_{F1} \bar{\nu}_{F2} + \tilde{X}_{B1} \bar{\nu}_{\tilde{B}2}(t) + \tilde{X}_{T1} \bar{\nu}_{\tilde{T}2}(t)$$
(4.2.139)

$$JQ_{2}J\begin{bmatrix}\bar{\mu}_{3}(t)\\\bar{\eta}_{4}(t)\\\bar{\lambda}_{3}(t)\end{bmatrix} = J\bar{Z}_{2}\bar{c}_{3}(t) + JJ\bar{S}_{2}\bar{\sigma}_{xy} + JX_{F2}\bar{\nu}_{F3} + J\tilde{X}_{B2}\bar{\nu}_{\tilde{B}3}(t) + J\tilde{X}_{T2}\bar{\nu}_{\tilde{T}3}(t)$$

$$(4.2.140)$$

$$\mathcal{Q}_{2} \begin{bmatrix} \mu_{4}(t) \\ \bar{\eta}_{3}(t) \\ \bar{\lambda}_{4}(t) \end{bmatrix} = \bar{\mathcal{Z}}_{2} \bar{c}_{4}(t) + \bar{S}_{2} \bar{\sigma}_{D} \\ + X_{F2} \bar{\nu}_{F4} + \tilde{X}_{B2} \bar{\nu}_{\tilde{B}4}(t) + \tilde{X}_{T2} \bar{\nu}_{\tilde{T}4}(t)$$
(4.2.141)

From (4.2.137)-(4.2.141) we obtain

$$\bar{\mu}_{0}(t) = P_{10}\mathcal{Q}_{0}^{-1} \Big[ \bar{\mathcal{Z}}_{0}\bar{c}_{0}(t) + \bar{S}_{0}\bar{\sigma}_{R} + \bar{S}_{3}\bar{\sigma}_{zz} + X_{B0}\bar{\nu}_{B0}(t) + X_{F0}\bar{\nu}_{F0} + \tilde{X}_{B0}\bar{\nu}_{\tilde{B}0}(t) + \tilde{X}_{T0}\bar{\nu}_{\tilde{T}0}(t) \Big]$$
(4.2.142)

$$\bar{\lambda}_{0}(t) = P_{01} \mathcal{Q}_{0}^{-1} \Big[ \bar{\mathcal{Z}}_{0} \bar{c}_{0}(t) + \bar{S}_{0} \bar{\sigma}_{R} + \bar{S}_{3} \bar{\sigma}_{zz} + X_{B0} \bar{\nu}_{B0}(t) + X_{F0} \bar{\nu}_{F0} + \tilde{X}_{B0} \bar{\nu}_{\tilde{B}0}(t) + \tilde{X}_{T0} \bar{\nu}_{\tilde{T}0}(t) \Big]$$
(4.2.143)

$$\bar{\mu}_{1}(t) = P_{100} \mathcal{Q}_{1}^{-1} \Big[ \bar{\mathcal{Z}}_{1} \bar{c}_{1}(t) + \bar{S}_{1} \bar{\sigma}_{yz} \\ + X_{F1} \bar{\nu}_{F1} + \tilde{X}_{B1} \bar{\nu}_{\tilde{B}1}(t) + \tilde{X}_{T1} \bar{\nu}_{\tilde{T}1}(t) \Big]$$
(4.2.144)

$$\bar{\eta}_{2}(t) = -P_{010}\mathcal{Q}_{1}^{-1} \Big[ \bar{\mathcal{Z}}_{1}\bar{c}_{1}(t) + \bar{S}_{1}\bar{\sigma}_{yz} + X_{F1}\bar{\nu}_{F1} + \tilde{X}_{B1}\bar{\nu}_{\tilde{B}1}(t) + \tilde{X}_{T1}\bar{\nu}_{\tilde{T}1}(t) \Big]$$
(4.2.145)

$$\bar{\lambda}_{1}(t) = P_{001} \mathcal{Q}_{1}^{-1} \Big[ \bar{\mathcal{Z}}_{1} \bar{c}_{1}(t) + \bar{S}_{1} \bar{\sigma}_{yz} + X_{F1} \bar{\nu}_{F1} + \tilde{X}_{B1} \bar{\nu}_{\tilde{B}1}(t) + \tilde{X}_{T1} \bar{\nu}_{\tilde{T}1}(t) \Big]$$
(4.2.146)

$$\bar{\mu}_{2}(t) = P_{100} \mathcal{Q}_{1}^{-1} \Big[ \bar{\mathcal{Z}}_{1} \bar{c}_{2}(t) + \bar{S}_{1} \bar{\sigma}_{xz} + X_{F1} \bar{\nu}_{F2} + \tilde{X}_{B1} \bar{\nu}_{\tilde{B}2}(t) + \tilde{X}_{T1} \bar{\nu}_{\tilde{T}2}(t) \Big]$$
(4.2.147)

$$\bar{\eta}_{1}(t) = P_{010} \mathcal{Q}_{1}^{-1} \Big[ \bar{\mathcal{Z}}_{1} \bar{c}_{2}(t) + \bar{S}_{1} \bar{\sigma}_{xz} + X_{F1} \bar{\nu}_{F2} + \tilde{X}_{B1} \bar{\nu}_{\tilde{B}2}(t) + \tilde{X}_{T1} \bar{\nu}_{\tilde{T}2}(t) \Big]$$
(4.2.148)

$$\bar{\lambda}_{2}(t) = P_{001} \mathcal{Q}_{1}^{-1} \Big[ \bar{\mathcal{Z}}_{1} \bar{c}_{2}(t) + \bar{S}_{1} \bar{\sigma}_{xz} + X_{F1} \bar{\nu}_{F2} + \tilde{X}_{B1} \bar{\nu}_{\tilde{B}2}(t) + \tilde{X}_{T1} \bar{\nu}_{\tilde{T}2}(t) \Big]$$
(4.2.149)

$$\bar{\mu}_{3}(t) = P_{100} \mathcal{Q}_{2}^{-1} \Big[ \bar{\mathcal{Z}}_{2} \bar{c}_{3}(t) + J \bar{S}_{2} \bar{\sigma}_{xy} + X_{F2} \bar{\nu}_{F3} + \tilde{X}_{B2} \bar{\nu}_{\tilde{B}3}(t) + \tilde{X}_{T2} \bar{\nu}_{\tilde{T}3}(t) \Big]$$
(4.2.150)

$$\bar{\eta}_4(t) = -P_{010}\mathcal{Q}_2^{-1} \Big[ \bar{\mathcal{Z}}_2 \bar{c}_3(t) + J \bar{S}_2 \bar{\sigma}_{xy} + X_{F2} \bar{\nu}_{F3} + \tilde{X}_{B2} \bar{\nu}_{\tilde{B}3}(t) + \tilde{X}_{T2} \bar{\nu}_{\tilde{T}3}(t) \Big]$$
(4.2.151)

$$\bar{\lambda}_{3}(t) = P_{001} \mathcal{Q}_{2}^{-1} \Big[ \bar{\mathcal{Z}}_{2} \bar{c}_{3}(t) + J \bar{S}_{2} \bar{\sigma}_{xy} + X_{F2} \bar{\nu}_{F3} + \tilde{X}_{B2} \bar{\nu}_{\tilde{B}3}(t) + \tilde{X}_{T2} \bar{\nu}_{\tilde{T}3}(t) \Big]$$
(4.2.152)

$$\bar{\mu}_{4}(t) = P_{100} \mathcal{Q}_{2}^{-1} \Big[ \bar{\mathcal{Z}}_{2} \bar{c}_{4}(t) + \bar{S}_{2} \bar{\sigma}_{D} + X_{F2} \bar{\nu}_{F4} + \tilde{X}_{B2} \bar{\nu}_{\tilde{B}4}(t) + \tilde{X}_{T2} \bar{\nu}_{\tilde{T}4}(t) \Big]$$
(4.2.153)

$$\bar{\eta}_{3}(t) = P_{010} \mathcal{Q}_{2}^{-1} \Big[ \bar{\mathcal{Z}}_{2} \bar{c}_{4}(t) + \bar{S}_{2} \bar{\sigma}_{D} + X_{F2} \bar{\nu}_{F4} + \tilde{X}_{B2} \bar{\nu}_{\tilde{B}4}(t) + \tilde{X}_{T2} \bar{\nu}_{\tilde{T}4}(t) \Big]$$
(4.2.154)

$$\bar{\lambda}_4(t) = P_{001} \mathcal{Q}_2^{-1} \Big[ \bar{\mathcal{Z}}_2 \bar{c}_4(t) + \bar{S}_2 \bar{\sigma}_D + X_{F2} \bar{\nu}_{F4} + \tilde{X}_{B2} \bar{\nu}_{\tilde{B}4}(t) + \tilde{X}_{T2} \bar{\nu}_{\tilde{T}4}(t) \Big]$$
(4.2.155)

#### Finite element approximation of the TCP system

System I (radially symmetric case)

System I is given by (4.2.28)-(4.2.30):

$$M\dot{V}_{0} - A\left[\frac{1}{r}(rV_{0_{r}})_{r} + (V_{0_{z}})_{z}\right] = -M\dot{W}_{0} + A\left(\frac{1}{r}(rW_{0_{r}})_{r} + (W_{0_{z}})_{z}\right)$$
$$-\vec{b}_{0}\left(\dot{u}_{0_{r}} + \frac{1}{r}\dot{u}_{0} + \dot{w}_{0_{z}}\right)$$
(4.2.28)

with boundary conditions

$$V_0(r, z, t) = 0$$
 on  $\Gamma \times [0, t_f)$  (4.2.29)

and initial conditions

$$V_0(r, z, 0) = V_{I0}(r, z) - W_0(r, z, 0)$$
(4.2.30)

Let

$$\mathcal{A} = M^{-1}A$$

Then (4.2.28) has the form

$$\dot{V}_{0} - \mathcal{A}\left[\frac{1}{r}(rV_{0_{r}})_{r} + (V_{0_{z}})_{z}\right] = -\dot{W}_{0} + \mathcal{A}\left[\frac{1}{r}(rW_{0_{r}})_{r} + (W_{0_{z}})_{z}\right]$$
$$-M^{-1}\vec{b}_{0}\left(\dot{u}_{0_{r}} + \frac{1}{r}\dot{u}_{0} + \dot{w}_{0_{z}}\right)$$
(4.2.156)

Using (4.2.88), the left-hand side of (4.2.156) is

$$\left(\hat{b}(z)^T \otimes B(r)\right) \dot{\bar{c}}_0(t) - \mathcal{A} \frac{1}{r} \left(\hat{b}(z)^T \otimes r B_r(r)\right)_r \bar{c}_0(t) - \mathcal{A} \left(\hat{b}_z(z)^T \otimes B(r)\right)_z \bar{c}_0(t) \quad (4.2.157)$$

Multiplying (4.2.157) by  $(\hat{b}(z)^T \otimes B(r))^T = (\hat{b}(z) \otimes B(r)^T)$  and integrating with respect to r and z, we have

$$\Big[\int_{z_T}^{z_B}\int_{R_B}^{R_F} \left(\hat{b}(z)\otimes B(r)^T\right) \left(\hat{b}(z)^T\otimes B(r)\right) r\,\mathrm{d}r\,\mathrm{d}z\Big]\dot{\bar{c}}_0(t)$$

$$-\Big[\int_{z_T}^{z_B}\int_{R_B}^{R_F} \left(\hat{b}(z)\otimes B(r)^T\right)\mathcal{A}\frac{1}{r}\left(\hat{b}(z)^T\otimes rB_r(r)\right)_r r\,\mathrm{d}r\,\mathrm{d}z\Big]\bar{c}_0(t)$$
$$-\Big[\int_{z_T}^{z_B}\int_{R_B}^{R_F} \left(\hat{b}(z)\otimes B(r)^T\right)\mathcal{A}\left(\hat{b}_z(z)^T\otimes B(r)\right)_z r\,\mathrm{d}r\,\mathrm{d}z\Big]\bar{c}_0(t)$$

$$= \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z) \hat{b}(z)^T \, \mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T B(r) r \, \mathrm{d}r \right] \right\} \dot{\bar{c}}_0(t) \\ + \left[ \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left( \hat{b}(z) \otimes B_r(r)^T \right) \mathcal{A} \left( \hat{b}(z)^T \otimes r B_r(r)^T \right) \, \mathrm{d}r \, \mathrm{d}z \right] \bar{c}_0(t) \\ + \left[ \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left( \hat{b}_z(z) \otimes B(r)^T \right) \mathcal{A} \left( \hat{b}_z(z)^T \otimes B_r(r)^T \right) r \, \mathrm{d}r \, \mathrm{d}z \right] \bar{c}_0(t) \right\}$$

$$= \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z) \hat{b}(z)^T \, \mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T B(r) r \, \mathrm{d}r \right] \right\} \dot{\bar{c}}_0(t) \\ + \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z) \hat{b}(z)^T \, \mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} B_r(r)^T \mathcal{A} B_r(r) r \, \mathrm{d}r \right] \right\} \bar{c}_0(t) \\ + \left\{ \left[ \int_{z_T}^{z_B} \hat{b}_z(z) \hat{b}_z(z)^T \, \mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T \mathcal{A} B(r) r \, \mathrm{d}r \right] \right\} \bar{c}_0(t)$$

$$= \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z)\hat{b}(z)^T \,\mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T B(r)r \,\mathrm{d}r \right] \right\} \dot{\bar{c}}_0(t) \\ + \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z)\hat{b}(z)^T \,\mathrm{d}z \right] \otimes \mathcal{A} \otimes \left[ \int_{R_B}^{R_F} \bar{b}_r(r)\bar{b}_r(r)^T r \,\mathrm{d}r \right] \right\} \bar{c}_0(t) \\ + \left\{ \left[ \int_{z_T}^{z_B} \hat{b}_z(z)\hat{b}_z(z)^T \,\mathrm{d}z \right] \otimes \mathcal{A} \otimes \left[ \int_{R_B}^{R_F} \bar{b}(r)\bar{b}(r)^T r \,\mathrm{d}r \right] \right\} \bar{c}_0(t)$$
(4.2.158)

Using (4.2.90)-(4.2.92), the last term of the right-hand side of (4.2.156) is

$$-M^{-1}\vec{b}_0\left(\dot{u}_{0r} + \frac{1}{r}\dot{u}_0 + \dot{w}_{0z}\right)$$
$$= -M^{-1}\vec{b}_0\left[\left(\hat{\mathbf{b}}(z)^T \otimes \bar{\mathbf{b}}_r(r)^T\right)\dot{\mu}_0(t) + \left(\hat{\mathbf{b}}(z)^T \otimes \frac{1}{r}\bar{\mathbf{b}}(r)^T\right)\dot{\mu}_0(t) + \left(\hat{\mathbf{b}}_z(z)^T \otimes \bar{\mathbf{b}}(r)^T\right)\dot{\lambda}_0(t)\right]$$

Multiplying by  $(\hat{b}(z) \otimes B(r)^T)$  and integrating with respect to r and z, we have

$$-\left[\int_{z_T}^{z_B}\int_{R_B}^{R_F} \left(\hat{b}(z)\otimes B(r)^T\right)M^{-1}\vec{b}_0\left(\hat{b}(z)^T\otimes\bar{b}_r(r)^T\right)r\,\mathrm{d}r\,\mathrm{d}z\right.\\ \left.+\int_{z_T}^{z_B}\int_{R_B}^{R_F} \left(\hat{b}(z)\otimes B(r)^T\right)M^{-1}\vec{b}_0\left(\hat{b}(z)^T\otimes\frac{1}{r}\,\bar{b}(r)^T\right)r\,\mathrm{d}r\,\mathrm{d}z\right]\dot{\mu}_0(t)\\ \left.-\left[\int_{z_T}^{z_B}\int_{R_B}^{R_F} \left(\hat{b}(z)\otimes B(r)^T\right)M^{-1}\vec{b}_0\left(\hat{b}_z(z)^T\otimes\bar{b}(r)^T\right)r\,\mathrm{d}r\,\mathrm{d}z\right]\dot{\lambda}_0(t)\right]$$

$$= -\left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z) \,\hat{\mathbf{b}}(z)^T \,\mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T M^{-1} \vec{b}_0(r \,\bar{\mathbf{b}}_r(r) + \bar{\mathbf{b}}(r))^T \,\mathrm{d}r \right] \right\} \dot{\mu}_0(t) \\ -\left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z) \,\hat{\mathbf{b}}_z(z)^T \,\mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T M^{-1} \vec{b}_0 \,\bar{\mathbf{b}}(r)^T r \,\mathrm{d}r \right] \right\} \dot{\lambda}_0(t) \quad (4.2.159)$$

From (4.2.20) and (4.2.89)

$$W_{i}(r, z, t) = \frac{R_{F} - r}{R_{F} - R_{B}} \hat{B}(z)\bar{\nu}_{Bi}(t) + \frac{r - R_{B}}{R_{F} - R_{B}} \hat{B}(z)\bar{\nu}_{Fi}$$
  
+  $\frac{z - z_{T}}{z_{B} - z_{T}} \Big( B(r)\bar{\nu}_{\tilde{B}i}(t) - \frac{R_{F} - r}{R_{F} - R_{B}} \hat{B}(z_{B})\bar{\nu}_{Bi}(t) - \frac{r - R_{B}}{R_{F} - R_{B}} \hat{B}(z_{B})\bar{\nu}_{Fi} \Big)$   
+  $\frac{z_{B} - z}{z_{B} - z_{T}} \Big( B(r)\bar{\nu}_{\tilde{T}i}(t) - \frac{R_{F} - r}{R_{F} - R_{B}} \hat{B}(z_{T})\bar{\nu}_{Bi}(t) - \frac{r - R_{B}}{R_{F} - R_{B}} \hat{B}(z_{T})\bar{\nu}_{Fi} \Big)$ 

$$= \frac{R_F - r}{R_F - R_B} \Big( \hat{B}(z) - \frac{z - z_T}{z_B - z_T} \hat{B}(z_B) - \frac{z_B - z}{z_B - z_T} \hat{B}(z_T) \Big) \bar{\nu}_{Bi}(t) + \frac{r - R_B}{R_F - R_B} \Big( \hat{B}(z) - \frac{z - z_T}{z_B - z_T} \hat{B}(z_B) - \frac{z_B - z}{z_B - z_T} \hat{B}(z_T) \Big) \bar{\nu}_{Fi} + \frac{z - z_T}{z_B - z_T} B(r) \bar{\nu}_{\tilde{B}i}(t) + \frac{z_B - z}{z_B - z_T} B(r) \bar{\nu}_{\tilde{T}i}(t), \quad i = 0, \dots, 4$$
(4.2.160)

and  $\bar{\nu}_{Bi}(t) = 0$ ,  $i = 1, \dots, 4$ .

Multiplying (4.2.160) by  $(\hat{b}(z) \otimes B(r)^T) = (\hat{b}(z) \otimes I_{3\times 3}) \otimes \bar{b}(r)$  and integrating with respect to r and z, we have

$$\begin{split} \left[ \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left( \hat{b}(z) \otimes B(r)^T \right) \frac{R_F - r}{R_F - R_B} \left( \hat{B}(z) - \frac{z - z_T}{z_B - z_T} \hat{B}(z_B) \right. \\ & \left. - \frac{z_B - z}{z_B - z_T} \hat{B}(z_T) \right) r \, \mathrm{d}r \, \mathrm{d}z \right] \bar{\nu}_{Bi}(t) \\ + \left[ \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left( \hat{b}(z) \otimes B(r)^T \right) \frac{r - R_B}{R_F - R_B} \left( \hat{B}(z) - \frac{z - z_T}{z_B - z_T} \hat{B}(z_B) \right. \\ & \left. - \frac{z_B - z}{z_B - z_T} \hat{B}(z_T) \right) r \, \mathrm{d}r \, \mathrm{d}z \right] \bar{\nu}_{Fi} \\ + \left[ \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left( \hat{b}(z) \otimes B(r)^T \right) \frac{z - z_T}{z_B - z_T} B(r) r \, \mathrm{d}r \, \mathrm{d}z \right] \bar{\nu}_{\tilde{B}i}(t) \\ + \left[ \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left( \hat{b}(z) \otimes B(r)^T \right) \frac{z_B - z}{z_B - z_T} B(r) r \, \mathrm{d}r \, \mathrm{d}z \right] \bar{\nu}_{\tilde{T}i}(t) \end{split}$$

$$= \left[ \int_{z_T}^{z_B} \hat{b}(z) \otimes \left( \hat{B}(z) - \frac{z - z_T}{z_B - z_T} \hat{B}(z_B) - \frac{z_B - z}{z_B - z_T} \hat{B}(z_T) \right) dz \right]$$
$$\otimes \left\{ \left[ \int_{R_B}^{R_F} \bar{b}(r) \frac{R_F - r}{R_F - R_B} r dr \right] \bar{\nu}_{Bi}(t) + \left[ \int_{R_B}^{R_F} \bar{b}(r) \frac{r - R_B}{R_F - R_B} r dr \right] \bar{\nu}_{Fi} \right\}$$
$$+ \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z) \frac{z - z_T}{z_B - z_T} dz \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T B(r) r dr \right] \right\} \bar{\nu}_{\tilde{B}i}(t)$$
$$+ \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z) \frac{z_B - z}{z_B - z_T} dz \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T B(r) r dr \right] \right\} \bar{\nu}_{\tilde{T}i}(t)$$
(4.2.161)

From (4.2.23) and (4.2.89)

$$W_{i_{r}}(r,z,t) = -\frac{1}{R_{F} - R_{B}} \Big( \hat{B}(z) - \frac{z - z_{T}}{z_{B} - z_{T}} \hat{B}(z_{B}) - \frac{z_{B} - z}{z_{B} - z_{T}} \hat{B}(z_{T}) \Big) \bar{\nu}_{Bi}(t) + \frac{1}{R_{F} - R_{B}} \Big( \hat{B}(z) - \frac{z - z_{T}}{z_{B} - z_{T}} \hat{B}(z_{B}) - \frac{z_{B} - z}{z_{B} - z_{T}} \hat{B}(z_{T}) \Big) \bar{\nu}_{Fi} + \frac{z - z_{T}}{z_{B} - z_{T}} B_{r}(r) \bar{\nu}_{\tilde{B}i}(t) + \frac{z_{B} - z}{z_{B} - z_{T}} B_{r}(r) \bar{\nu}_{\tilde{T}i}(t), \quad i = 0, \dots, 4 \quad (4.2.162)$$

and  $\bar{\nu}_{Bi}(t) = 0$ ,  $i = 1, \dots, 4$ .

Multiplying  $\mathcal{A}\frac{1}{r}(rW_{i_r})_r$  by  $(\hat{b}(z)\otimes B(r)^T) = (\hat{b}(z)\otimes I_{3\times 3})\otimes \bar{b}(r)$  and integrating with respect to r and z, we have

with respect to r and z, we have

$$\begin{bmatrix} \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left( \hat{b}(z) \otimes B(r)^T \right) \mathcal{A} \frac{1}{R_F - R_B} \left( \hat{B}(z) - \frac{z - z_T}{z_B - z_T} \hat{B}(z_B) - \frac{z_B - z}{z_B - z_T} \hat{B}(z_T) \right) dr dz \end{bmatrix} \left( -\bar{\nu}_{Bi}(t) + \bar{\nu}_{Fi} \right)$$
$$- \begin{bmatrix} \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left( \hat{b}(z) \otimes B_r(r)^T \right) \mathcal{A} \frac{z - z_T}{z_B - z_T} B_r(r) r dr dz \end{bmatrix} \bar{\nu}_{\tilde{B}i}(t)$$
$$- \begin{bmatrix} \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left( \hat{b}(z) \otimes B_r(r)^T \right) \mathcal{A} \frac{z_B - z}{z_B - z_T} B_r(r) r dr dz \end{bmatrix} \bar{\nu}_{\tilde{T}i}(t)$$

$$= \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z) \otimes \left[ \mathcal{A} \left( \hat{B}(z) - \frac{z - z_T}{z_B - z_T} \hat{B}(z_B) - \frac{z_B - z}{z_B - z_T} \hat{B}(z_T) \right) \right] dz \right] \\ \otimes \left[ \int_{R_B}^{R_F} \bar{b}(r) \frac{1}{R_F - R_B} dr \right] \right\} \left( -\bar{\nu}_{Bi}(t) + \bar{\nu}_{Fi} \right) \\ - \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z) \frac{z - z_T}{z_B - z_T} dz \right] \otimes \mathcal{A} \otimes \left[ \int_{R_B}^{R_F} \bar{b}_r(r) \bar{b}_r(r)^T r dr \right] \right\} \bar{\nu}_{\tilde{B}i}(t) \\ - \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z) \frac{z_B - z}{z_B - z_T} dz \right] \otimes \mathcal{A} \otimes \left[ \int_{R_B}^{R_F} \bar{b}_r(r) \bar{b}_r(r)^T r dr \right] \right\} \bar{\nu}_{\tilde{T}i}(t), \quad i = 0, \dots, 4$$

$$(4.2.163)$$

and  $\bar{\nu}_{Bi}(t) = 0$ ,  $i = 1, \dots, 4$ .

From (4.2.26) and (4.2.89)

$$W_{i_{z}}(r,z,t) = \frac{R_{F} - r}{R_{F} - R_{B}} \Big( \hat{B}_{z}(z) - \frac{1}{z_{B} - z_{T}} \hat{B}(z_{B}) + \frac{1}{z_{B} - z_{T}} \hat{B}(z_{T}) \Big) \bar{\nu}_{Bi}(t) + \frac{r - R_{B}}{R_{F} - R_{B}} \Big( \hat{B}_{z}(z) - \frac{1}{z_{B} - z_{T}} \hat{B}(z_{B}) + \frac{1}{z_{B} - z_{T}} \hat{B}(z_{T}) \Big) \bar{\nu}_{Fi} + \frac{1}{z_{B} - z_{T}} B(r) \Big( \bar{\nu}_{\tilde{B}i}(t) - \bar{\nu}_{\tilde{T}i}(t) \Big), \quad i = 0, \dots, 4$$
(4.2.164)

and  $\bar{\nu}_{Bi}(t) = 0$ ,  $i = 1, \dots, 4$ .

Multiplying  $\mathcal{A}(W_{i_z})_z$  by  $(\hat{b}(z) \otimes B(r)^T) = (\hat{b}(z) \otimes I_{3\times 3}) \otimes \bar{b}(r)$  and integrating with respect to r and z, we have

 $-\left[\int_{z_T}^{z_B}\int_{R_B}^{R_F} \left(\hat{b}_z(z)\otimes B(r)^T\right)\mathcal{A}\frac{R_F-r}{R_F-R_B}\left(\hat{B}_z(z)-\frac{1}{z_B-z_T}\hat{B}(z_B)\right)\right.\\\left.+\frac{1}{z_B-z_T}\hat{B}(z_T)\right)r\,\mathrm{d}r\,\mathrm{d}z\right]\bar{\nu}_{Bi}(t)$  $-\left[\int_{z_T}^{z_B}\int_{R_B}^{R_F} \left(\hat{b}_z(z)\otimes B(r)^T\right)\mathcal{A}\frac{r-R_B}{R_F-R_B}\left(\hat{B}_z(z)-\frac{1}{z_B-z_T}\hat{B}(z_B)\right)\right.\\\left.+\frac{1}{z_B-z_T}\hat{B}(z_T)\right)r\,\mathrm{d}r\,\mathrm{d}z\right]\bar{\nu}_{Fi}$  $-\left[\int_{z_T}^{z_B}\int_{R_B}^{R_F} \left(\hat{b}_z(z)\otimes B(r)^T\right)\mathcal{A}\frac{1}{z_B-z_T}B(r)r\,\mathrm{d}r\,\mathrm{d}z\right]\bar{\nu}_{Bi}(t)\right.\\\left.+\left[\int_{z_T}^{z_B}\int_{R_B}^{R_F} \left(\hat{b}_z(z)\otimes B(r)^T\right)\mathcal{A}\frac{1}{z_B-z_T}B_r(r)r\,\mathrm{d}r\,\mathrm{d}z\right]\bar{\nu}_{Ti}(t)\right]$ 

$$= -\left[\int_{z_T}^{z_B} \hat{b}_z(z) \otimes \left[\mathcal{A}\left(\hat{B}_z(z) - \frac{1}{z_B - z_T}\hat{B}(z_B) + \frac{1}{z_B - z_T}\hat{B}(z_T)\right)\right] dz\right]$$
$$\otimes \left\{ \left[\int_{R_B}^{R_F} \bar{b}(r)\frac{R_F - r}{R_F - R_B}r dr\right]\bar{\nu}_{Bi}(t) + \left[\int_{R_B}^{R_F} \bar{b}(r)\frac{r - R_B}{R_F - R_B}r dr\right]\bar{\nu}_{Fi}\right\}$$
$$- \left\{ \left[\int_{z_T}^{z_B} \hat{b}_z(z)\frac{1}{z_B - z_T} dz\right] \otimes \mathcal{A} \otimes \left[\int_{R_B}^{R_F} \bar{b}(r)\bar{b}(r)^T r dr\right]\right\} \left(\bar{\nu}_{\tilde{B}i}(t) - \bar{\nu}_{\tilde{T}i}(t)\right)$$
$$(4.2.165)$$

Define the following matrices:

 $3M_{Dr}M_{Dz} \times 3M_{Dr}M_{Dz}$  matrices:

$$G_{0} = \left[\int_{z_{T}}^{z_{B}} \hat{b}(z)\hat{b}(z)^{T} dz\right] \otimes \left[\int_{R_{B}}^{R_{F}} B(r)^{T}B(r)r dr\right]$$
$$G_{2}(r) = \left[\int_{z_{T}}^{z_{B}} \hat{b}(z)\hat{b}(z)^{T} dz\right] \otimes \mathcal{A} \otimes \left[\int_{R_{B}}^{R_{F}} \bar{b}_{r}(r)\bar{b}_{r}(r)^{T}r dr\right]$$
$$G_{2}(z) = \left[\int_{z_{T}}^{z_{B}} \hat{b}_{z}(z)\hat{b}_{z}(z)^{T} dz\right] \otimes \mathcal{A} \otimes \left[\int_{R_{B}}^{R_{F}} \bar{b}(r)\bar{b}(r)^{T}r dr\right]$$

## $3M_{Dr}M_{Dz} \times M_rM_z$ matrices:

$$H_0 = \left[ \int_{z_T}^{z_B} \hat{b}(z) \,\hat{\mathbf{b}}(z)^T \,\mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T M^{-1} \vec{b}_0 (r \,\bar{\mathbf{b}}_r(r) + \bar{\mathbf{b}}(r))^T \,\mathrm{d}r \right]$$
$$H_1 = \left[ \int_{z_T}^{z_B} \hat{b}(z) \,\hat{\mathbf{b}}_z(z)^T \,\mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T M^{-1} \vec{b}_0 \,\bar{\mathbf{b}}(r)^T r \,\mathrm{d}r \right]$$

 $3M_{Dr}M_{Dz} \times 3M_{Dz}$  matrices:

$$\mathcal{R}_B = \int_{z_T}^{z_B} \hat{b}(z) \otimes \left(\hat{B}(z) - \frac{z - z_T}{z_B - z_T}\hat{B}(z_B) - \frac{z_B - z}{z_B - z_T}\hat{B}(z_T)\right) dz \bigg]$$
$$\otimes \bigg[\int_{R_B}^{R_F} \bar{b}(r) \frac{R_F - r}{R_F - R_B} r dr\bigg]$$

$$\mathcal{R}_F = \int_{z_T}^{z_B} \hat{b}(z) \otimes \left(\hat{B}(z) - \frac{z - z_T}{z_B - z_T}\hat{B}(z_B) - \frac{z_B - z}{z_B - z_T}\hat{B}(z_T)\right) dz \bigg]$$
$$\otimes \bigg[\int_{R_B}^{R_F} \bar{b}(r) \frac{r - R_B}{R_F - R_B} r dr\bigg]$$

$$\mathcal{R}_F(a) = \left[ \int_{z_T}^{z_B} \hat{b}(z) \otimes \left[ \mathcal{A} \left( \hat{B}(z) - \frac{z - z_T}{z_B - z_T} \hat{B}(z_B) - \frac{z_B - z}{z_B - z_T} \hat{B}(z_T) \right) \right] dz \right]$$
$$\otimes \left[ \int_{R_B}^{R_F} \bar{b}(r) \frac{1}{R_F - R_B} dr \right]$$

$$\mathcal{R}_B(z) = \left[ \int_{z_T}^{z_B} \hat{b}_z(z) \otimes \left[ \mathcal{A} \left( \hat{B}_z(z) - \frac{1}{z_B - z_T} \hat{B}(z_B) + \frac{1}{z_B - z_T} \hat{B}(z_T) \right) \right] dz \right]$$
$$\otimes \left[ \int_{R_B}^{R_F} \bar{b}(r) \frac{R_F - r}{R_F - R_B} r \, dr \right]$$

$$\mathcal{R}_F(z) = \left[ \int_{z_T}^{z_B} \hat{b}_z(z) \otimes \left[ \mathcal{A} \left( \hat{B}_z(z) - \frac{1}{z_B - z_T} \hat{B}(z_B) + \frac{1}{z_B - z_T} \hat{B}(z_T) \right) \right] dz \right]$$
$$\otimes \left[ \int_{R_B}^{R_F} \bar{b}(r) \frac{r - R_B}{R_F - R_B} r \, dr \right]$$

 $3M_{Dr}M_{Dz} \times 3M_{Dr}$  matrices:

$$\tilde{\mathcal{R}}_{B} = \left[ \int_{z_{T}}^{z_{B}} \hat{b}(z) \frac{z - z_{T}}{z_{B} - z_{T}} dz \right] \otimes \left[ \int_{R_{B}}^{R_{F}} B(r)^{T} B(r) r dr \right]$$

$$\tilde{\mathcal{R}}_{T} = \left[ \int_{z_{T}}^{z_{B}} \hat{b}(z) \frac{z_{B} - z}{z_{B} - z_{T}} dz \right] \otimes \left[ \int_{R_{B}}^{R_{F}} B(r)^{T} B(r) r dr \right]$$

$$\tilde{\mathcal{R}}_{B}(a) = \left[ \int_{z_{T}}^{z_{B}} \hat{b}(z) \frac{z - z_{T}}{z_{B} - z_{T}} dz \right] \otimes \mathcal{A} \otimes \left[ \int_{R_{B}}^{R_{F}} \bar{b}_{r}(r) \bar{b}_{r}(r)^{T} r dr \right]$$

$$\tilde{\mathcal{R}}_{T}(a) = \left[ \int_{z_{T}}^{z_{B}} \hat{b}(z) \frac{z_{B} - z}{z_{B} - z_{T}} dz \right] \otimes \mathcal{A} \otimes \left[ \int_{R_{B}}^{R_{F}} \bar{b}_{r}(r) \bar{b}_{r}(r)^{T} r dr \right]$$

$$\tilde{\mathcal{R}}_{0}(z) = \left[ \int_{z_{T}}^{z_{B}} \hat{b}_{z}(z) \frac{1}{z_{B} - z_{T}} dz \right] \otimes \mathcal{A} \otimes \left[ \int_{R_{B}}^{R_{F}} \bar{b}(r) \bar{b}(r)^{T} r dr \right]$$

Using these matrices, (4.2.156) multiplied by  $(\hat{b}(z) \otimes B(r)^T)$  and integrated with respect to r and z, has the form

$$G_{0}\dot{\bar{c}}_{0}(t) + (G_{2}(r) + G_{2}(z))\bar{c}_{0}(t) = -H_{0}\dot{\bar{\mu}}_{0}(t) - H_{1}\dot{\bar{\lambda}}_{0}(t) - \mathcal{R}_{B}\dot{\bar{\nu}}_{B0}(t) - \tilde{\mathcal{R}}_{B}\dot{\bar{\nu}}_{\tilde{B}0}(t) - \tilde{\mathcal{R}}_{T}\dot{\bar{\nu}}_{\tilde{T}0}(t) - \mathcal{R}_{F}(a)\bar{\nu}_{B0}(t) + \mathcal{R}_{F}(a)\bar{\nu}_{F0} - \tilde{\mathcal{R}}_{B}(a)\bar{\nu}_{\tilde{B}0}(t) - \tilde{\mathcal{R}}_{T}(a)\bar{\nu}_{\tilde{T}0}(t) - \mathcal{R}_{B}(z)\bar{\nu}_{B0}(t) - \mathcal{R}_{F}(z)\bar{\nu}_{F0} - \tilde{\mathcal{R}}_{0}(z)\bar{\nu}_{\tilde{B}0}(t) + \tilde{\mathcal{R}}_{0}(z)\bar{\nu}_{\tilde{T}0}(t)$$
(4.2.166)

From (4.2.142), (4.2.143)

$$\dot{\bar{\mu}}_{0}(t) = P_{10}\mathcal{Q}_{0}^{-1} \Big[ \bar{\mathcal{Z}}_{0} \dot{\bar{c}}_{0}(t) + X_{B0} \dot{\bar{\nu}}_{B0}(t) + \tilde{X}_{B0} \dot{\bar{\nu}}_{\tilde{B}0}(t) + \tilde{X}_{T0} \dot{\bar{\nu}}_{\tilde{T}0}(t) \Big]$$
(4.2.167)

$$\dot{\bar{\lambda}}_{0}(t) = P_{01} \mathcal{Q}_{0}^{-1} \Big[ \bar{\mathcal{Z}}_{0} \dot{\bar{c}}_{0}(t) + X_{B0} \dot{\bar{\nu}}_{B0}(t) + \tilde{X}_{B0} \dot{\bar{\nu}}_{\tilde{B}0}(t) + \tilde{X}_{T0} \dot{\bar{\nu}}_{\tilde{T}0}(t) \Big]$$
(4.2.168)

Applying (4.2.167) and (4.2.168) to (4.2.166), we have

$$\mathcal{H}_{0}\dot{\bar{c}}_{0}(t) + \mathcal{G}_{0}\bar{c}_{0}(t) = \mathcal{D}_{B0}\dot{\bar{\nu}}_{B0}(t) + \mathcal{D}_{\tilde{B}0}\dot{\bar{\nu}}_{\tilde{B}0}(t) + \mathcal{D}_{\tilde{T}0}\dot{\bar{\nu}}_{\tilde{T}0}(t) + \mathcal{F}_{B0}\bar{\bar{\nu}}_{B0}(t) + \mathcal{F}_{F0}\bar{\bar{\nu}}_{F0} + \mathcal{F}_{\tilde{B}0}\bar{\bar{\nu}}_{\tilde{B}0}(t) + \mathcal{F}_{\tilde{T}0}\bar{\bar{\nu}}_{\tilde{T}0}(t)$$
(4.2.169)

where  $\mathcal{H}_0 = G_0 + (H_0 P_{10} + H_1 P_{01}) \mathcal{Q}_0^{-1} \bar{\mathcal{Z}}_0$ 

$$\mathcal{G}_{0} = G_{2}(r) + G_{2}(z)$$

$$\mathcal{D}_{B0} = -(H_{0}P_{10} + H_{1}P_{01})\mathcal{Q}_{0}^{-1}X_{B0} - \mathcal{R}_{B}$$

$$\mathcal{D}_{\tilde{B}0} = -(H_{0}P_{10} + H_{1}P_{01})\mathcal{Q}_{0}^{-1}\tilde{X}_{B0} - \tilde{\mathcal{R}}_{B}$$

$$\mathcal{D}_{\tilde{T}0} = -(H_{0}P_{10} + H_{1}P_{01})\mathcal{Q}_{0}^{-1}\tilde{X}_{T0} - \tilde{\mathcal{R}}_{T}$$

$$\mathcal{F}_{B0} = -\mathcal{R}_{F}(a) - \mathcal{R}_{B}(z)$$

$$\mathcal{F}_{F0} = \mathcal{R}_{F}(a) - \mathcal{R}_{F}(z)$$

$$\mathcal{F}_{\tilde{B}0} = -\tilde{\mathcal{R}}_{B}(a) - \tilde{\mathcal{R}}_{0}(z)$$

$$\mathcal{F}_{\tilde{T}0} = -\tilde{\mathcal{R}}_{T}(a) + \tilde{\mathcal{R}}_{0}(z)$$

Now we approximate initial conditions. From (4.2.20), (4.2.30) and (4.2.89), the coefficients  $V_i(r, z, 0)$  of  $\theta$ -approximation of initial conditions have the form

$$V_i(r, z, 0) = \tilde{V}_{Ii}(r, z) - W_i(r, z, 0)$$

where

$$\tilde{V}_{Ii}(r,z) = \left(\hat{b}(z)^T \otimes B(r)\right) \bar{\nu}_{Ii}$$

$$W_{i}(r, z, 0) = \frac{R_{F} - r}{R_{F} - R_{B}} \Big( \hat{B}(z) - \frac{z - z_{T}}{z_{B} - z_{T}} \hat{B}(z_{B}) - \frac{z_{B} - z}{z_{B} - z_{T}} \hat{B}(z_{T}) \Big) \bar{\nu}_{Bi}(0) + \frac{r - R_{B}}{R_{F} - R_{B}} \Big( \hat{B}(z) - \frac{z - z_{T}}{z_{B} - z_{T}} \hat{B}(z_{B}) - \frac{z_{B} - z}{z_{B} - z_{T}} \hat{B}(z_{T}) \Big) \bar{\nu}_{Fi} + \frac{z - z_{T}}{z_{B} - z_{T}} B(r) \bar{\nu}_{\tilde{B}i}(0) + \frac{z_{B} - z}{z_{B} - z_{T}} B(r) \bar{\nu}_{\tilde{T}i}(0), \quad i = 0, \dots, 4 \quad (4.2.170)$$

and  $\bar{\nu}_{Bi}(0) = 0$ ,  $i = 1, \dots, 4$ .

Multiplying  $V_i(r, z, 0)$  by  $\hat{b}(z) \otimes B(r)^T$  and integrating with respect to r and z, we have

$$\left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z)\hat{b}(z)^T \,\mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T B(r)r \,\mathrm{d}r \right] \right\} \bar{c}_i(0)$$
$$= \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z)\hat{b}(z)^T \,\mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T B(r)r \,\mathrm{d}r \right] \right\} \bar{\nu}_{Ii}$$
$$- \left[ \int_{z_T}^{z_B} \hat{b}(z) \otimes \left( \hat{B}(z) - \frac{z - z_T}{z_B - z_T} \hat{B}(z_B) - \frac{z_B - z}{z_B - z_T} \hat{B}(z_T) \right) \,\mathrm{d}z \right]$$

$$\otimes \left\{ \left[ \int_{R_B}^{R_F} \bar{b}(r) \frac{R_F - r}{R_F - R_B} r \, \mathrm{d}r \right] \bar{\nu}_{Bi}(0) + \left[ \int_{R_B}^{R_F} \bar{b}(r) \frac{r - R_B}{R_F - R_B} r \, \mathrm{d}r \right] \bar{\nu}_{Fi} \right\} - \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z) \frac{z - z_T}{z_B - z_T} \, \mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T B(r) r \, \mathrm{d}r \right] \right\} \bar{\nu}_{\tilde{B}i}(0) - \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z) \frac{z_B - z}{z_B - z_T} \, \mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T B(r) r \, \mathrm{d}r \right] \right\} \bar{\nu}_{\tilde{T}i}(0)$$

$$(4.2.171)$$

Using matrices defined above, (4.2.171) multiplied by  $(\hat{b}(z) \otimes B(r)^T)$  and integrated with respect to r and z, has the form

$$G_0 \bar{c}_i(0) = G_0 \bar{\nu}_{Ii} - \mathcal{R}_B \bar{\nu}_{Bi}(0) - \mathcal{R}_F \bar{\nu}_{Fi} - \tilde{\mathcal{R}}_B \bar{\nu}_{\tilde{B}i}(0) - \tilde{\mathcal{R}}_T \bar{\nu}_{\tilde{T}i}(0)$$
(4.2.172)

From (4.2.169) and (4.2.172) with i = 0 we obtain the following system.

System of ODEs for TCP I (radially symmetric case):

$$\mathcal{H}_{0}\bar{c}_{0}(t) + \mathcal{G}_{0}\bar{c}_{0}(t) = \mathcal{D}_{B0}\bar{\nu}_{B0}(t) + \mathcal{D}_{\tilde{B}0}\bar{\nu}_{\tilde{B}0}(t) + \mathcal{D}_{\tilde{T}0}\bar{\nu}_{\tilde{T}0}(t) + \mathcal{F}_{B0}\bar{\nu}_{B0}(t) + \mathcal{F}_{F0}\bar{\nu}_{F0} + \mathcal{F}_{\tilde{B}0}\bar{\nu}_{\tilde{B}0}(t) + \mathcal{F}_{\tilde{T}0}\bar{\nu}_{\tilde{T}0}(t)$$
(4.2.169)

with initial conditions

$$\bar{c}_0(0) = \bar{\nu}_{I0} - G_0^{-1} \left[ \mathcal{R}_B \bar{\nu}_{B0}(0) + \mathcal{R}_F \bar{\nu}_{F0} + \tilde{\mathcal{R}}_B \bar{\nu}_{\tilde{B}0}(0) + \tilde{\mathcal{R}}_T \bar{\nu}_{\tilde{T}0}(0) \right]$$
(4.2.173)

#### System II (corresponds to $\sin \theta$ approximation)

System II is given by (4.2.31)-(4.2.33):

$$M\dot{V}_{1} - A\left[\frac{1}{r}(rV_{1_{r}})_{r} - \frac{1}{r^{2}}V_{1} + (V_{1_{z}})_{z}\right] = -M\dot{W}_{1} + A\left(\frac{1}{r}(rW_{1_{r}})_{r} - \frac{1}{r^{2}}W_{1} + (W_{1_{z}})_{z}\right)$$
$$-\vec{b}_{0}\left(\dot{u}_{1_{r}} + \frac{1}{r}(\dot{u}_{1} - \dot{v}_{2}) + \dot{w}_{1_{z}}\right)$$
(4.2.31)

with boundary conditions

$$V_1(r, z, t) = 0$$
 on  $\Gamma \times [0, t_f)$  (4.2.32)

and initial conditions

$$V_1(r, z, 0) = \tilde{V}_{I1}(r, z) - W_1(r, z, 0)$$
(4.2.33)

Using  $\mathcal{A} = M^{-1}A$ , (4.2.31) has the form

$$\dot{V}_{1} - \mathcal{A}\left[\frac{1}{r}(rV_{1_{r}})_{r} - \frac{1}{r^{2}}V_{1} + (V_{1_{z}})_{z}\right] = -M^{-1}\vec{b}_{0}\left(\dot{u}_{1_{r}} + \frac{1}{r}\dot{u}_{1} - \frac{1}{r}\dot{v}_{2} + \dot{w}_{1_{z}}\right)$$
$$-\dot{W}_{1} + \mathcal{A}\left[\frac{1}{r}(rW_{1_{r}})_{r} - \frac{1}{r^{2}}W_{1} + (W_{1_{z}})_{z}\right]$$
(4.2.174)

Using (4.2.87), the left-hand side of (4.2.174) is

$$\left(\hat{b}(z)^T \otimes B(r)\right) \dot{\bar{c}}_1(t) - \mathcal{A} \frac{1}{r} \left(\hat{b}(z)^T \otimes r B_r(r)\right)_r \bar{c}_1(t) + \mathcal{A} \frac{1}{r^2} \left(\hat{b}(z)^T \otimes B(r)\right) \bar{c}_1(t) - \mathcal{A} \left(\hat{b}_z(z)^T \otimes B(r)\right)_z \bar{c}_0(t)$$

$$(4.2.175)$$

Multiplying (4.2.175) by  $(\hat{b}(z) \otimes B(r)^T)$  and integrating with respect to r and z we have

$$\left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z)\hat{b}(z)^T \,\mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T B(r)r \,\mathrm{d}r \right] \right\} \dot{\bar{c}}_1(t) \\
+ \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z)\hat{b}(z)^T \,\mathrm{d}z \right] \otimes \mathcal{A} \otimes \left( \left[ \int_{R_B}^{R_F} \bar{b}_r(r)\bar{b}_r(r)^T r \,\mathrm{d}r \right] + \left[ \int_{R_B}^{R_F} \bar{b}(r)\bar{b}(r)^T \frac{1}{r} \,\mathrm{d}r \right] \right) \\
+ \left[ \int_{z_T}^{z_B} \hat{b}_z(z)\hat{b}_z(z)^T \,\mathrm{d}z \right] \otimes \mathcal{A} \otimes \left[ \int_{R_B}^{R_F} \bar{b}(r)\bar{b}(r)^T r \,\mathrm{d}r \right] \right\} \bar{c}_1(t) \qquad (4.2.176)$$

Using (4.2.90)-(4.2.92), the first term of the right-hand side of (4.2.174) is

$$-M^{-1}\vec{b}_0 \Big[ \Big( \hat{\mathbf{b}}(z)^T \otimes \bar{\mathbf{b}}_r(r)^T \Big) \dot{\bar{\mu}}_1(t) + \Big( \hat{\mathbf{b}}(z)^T \otimes \frac{1}{r} \bar{\mathbf{b}}(r)^T \Big) \dot{\bar{\mu}}_1(t) \\ - \Big( \hat{\mathbf{b}}(z)^T \otimes \frac{1}{r} \bar{\mathbf{b}}(r)^T \Big) \dot{\bar{\eta}}_2(t) + \Big( \hat{\mathbf{b}}_z(z)^T \otimes \bar{\mathbf{b}}(r)^T \Big) \dot{\bar{\lambda}}_1(t) \Big]$$

Multiplying by  $(\hat{b}(z) \otimes B(r)^T)$  and integrating with respect to r and z, we have

$$-\left\{\left[\int_{z_T}^{z_B}\hat{b}(z)\,\hat{\mathbf{b}}(z)^T\,\mathrm{d}z\right]\otimes\left[\int_{R_B}^{R_F}B(r)^TM^{-1}\vec{b}_0(r\,\bar{\mathbf{b}}_r(r)+\,\bar{\mathbf{b}}(r))^T\,\mathrm{d}r\right]\right\}\dot{\bar{\mu}}_1(t)$$

$$+ \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z) \,\hat{\mathbf{b}}(z)^T \,\mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T M^{-1} \vec{b}_0 \,\bar{\mathbf{b}}(r)^T \,\mathrm{d}r \right] \right\} \dot{\bar{\eta}}_2(t) \\ - \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z) \,\hat{\mathbf{b}}_z(z)^T \,\mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T M^{-1} \vec{b}_0 \,\bar{\mathbf{b}}(r)^T r \,\mathrm{d}r \right] \right\} \dot{\bar{\lambda}}_1(t)$$
(4.2.177)

Multiplying the corresponding term in (4.2.174) by  $(\hat{b}(z) \otimes B(r)^T)$  and integrating with respect to r and z, we have the following results

for 
$$\dot{W}_{1}$$
: (4.2.161) with  $\dot{\nu}_{B1}(t) = 0$   
for  $\mathcal{A}\frac{1}{r}(rW_{1r})_{r}$ : (4.2.163) with  $\bar{\nu}_{B1}(t) = 0$   
for  $\mathcal{A}(rW_{1z})_{z}$ : (4.2.165) with  $\bar{\nu}_{B1}(t) = 0$ .  
 $W_{i}(r, z, t)$  is given by (4.2.160) with  $\bar{\nu}_{Bi}(t) = 0$ ,  $i = 1, ..., 4$ .

Multiplying  $\mathcal{A}\frac{1}{r^2}W_i$  by  $(\hat{b}(z)\otimes B(r)^T)$  and integrating with respect to r and z, we have

$$\begin{split} \left[ \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left( \hat{b}(z) \otimes B(r)^T \right) \mathcal{A} \frac{1}{r^2} \frac{r - R_B}{R_F - R_B} \left( \hat{B}(z) - \frac{z - z_T}{z_B - z_T} \hat{B}(z_B) \right. \\ \left. - \frac{z_B - z}{z_B - z_T} \hat{B}(z_T) \right) r \, \mathrm{d}r \, \mathrm{d}z \right] \bar{\nu}_{Fi} \\ + \left[ \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left( \hat{b}(z) \otimes B(r)^T \right) \mathcal{A} \frac{1}{r^2} \frac{z - z_T}{z_B - z_T} B(r) r \, \mathrm{d}r \, \mathrm{d}z \right] \bar{\nu}_{\bar{B}i}(t) \\ + \left[ \int_{z_T}^{z_B} \int_{R_B}^{R_F} \left( \hat{b}(z) \otimes B(r)^T \right) \mathcal{A} \frac{1}{r^2} \frac{z_B - z}{z_B - z_T} B(r) r \, \mathrm{d}r \, \mathrm{d}z \right] \bar{\nu}_{\bar{T}i}(t) \\ = \left[ \int_{z_T}^{z_B} \hat{b}(z) \otimes \left( \mathcal{A} \left( \hat{B}(z) - \frac{z - z_T}{z_B - z_T} \hat{B}(z_B) - \frac{z_B - z}{z_B - z_T} \hat{B}(z_T) \right) \right) \mathrm{d}z \right] \\ \otimes \left[ \int_{R_B}^{R_F} \bar{b}(r) \frac{r - R_B}{R_F - R_B} \frac{1}{r} \, \mathrm{d}r \right] \bar{\nu}_{Fi} \\ + \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z) \frac{z - z_T}{z_B - z_T} \, \mathrm{d}z \right] \otimes \mathcal{A} \otimes \left[ \int_{R_B}^{R_F} \bar{b}(r) \bar{b}(r)^T \frac{1}{r} \, \mathrm{d}r \right] \right\} \bar{\nu}_{\bar{B}i}(t) \\ + \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z) \frac{z - z_T}{z_B - z_T} \, \mathrm{d}z \right] \otimes \mathcal{A} \otimes \left[ \int_{R_B}^{R_F} \bar{b}(r) \bar{b}(r)^T \frac{1}{r} \, \mathrm{d}r \right] \right\} \bar{\nu}_{\bar{B}i}(t) \\ + \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z) \frac{z - z_T}{z_B - z_T} \, \mathrm{d}z \right] \otimes \mathcal{A} \otimes \left[ \int_{R_B}^{R_F} \bar{b}(r) \bar{b}(r)^T \frac{1}{r} \, \mathrm{d}r \right] \right\} \bar{\nu}_{\bar{B}i}(t), \quad i = 1, \dots, 4. \\ (4.2.178) \end{aligned} \right] \end{aligned}$$

Define the following matrices:

$$3M_{Dr}M_{Dz} \times 3M_{Dr}M_{Dz}$$
 matrix:

$$G_0\left(\frac{1}{r}\right) = \left[\int_{z_T}^{z_B} \hat{b}(z)\hat{b}(z)^T \,\mathrm{d}z\right] \otimes \mathcal{A} \otimes \left[\int_{R_B}^{R_F} \bar{b}(r)\bar{b}(r)^T \frac{1}{r} \,\mathrm{d}r\right]$$

 $3M_{Dr}M_{Dz} \times M_rM_z$  matrix:

$$H_0(r) = \left[\int_{z_T}^{z_B} \hat{b}(z) \,\hat{\mathbf{b}}(z)^T \,\mathrm{d}z\right] \otimes \left[\int_{R_B}^{R_F} B(r)^T M^{-1} \vec{b}_0 \,\bar{\mathbf{b}}(r)^T \,\mathrm{d}r\right]$$

 $3M_{Dr}M_{Dz} \times 3M_{Dz}$  matrix:

$$\mathcal{R}_{F}^{(a)}\left(\frac{1}{r}\right) = \left\{ \int_{z_{T}}^{z_{B}} \hat{b}(z) \otimes \left[ \mathcal{A}\left(\hat{B}(z) - \frac{z - z_{T}}{z_{B} - z_{T}}\hat{B}(z_{B}) - \frac{z_{B} - z}{z_{B} - z_{T}}\hat{B}(z_{T}) \right) \right] \mathrm{d}z \right\}$$
$$\otimes \left[ \int_{R_{B}}^{R_{F}} \bar{b}(r) \frac{r - R_{B}}{R_{F} - R_{B}} \frac{1}{r} \mathrm{d}r \right]$$

$$3M_{Dr}M_{Dz} \times 3M_{Dr}$$
 matrices:

$$\tilde{\mathcal{R}}_{B}^{(a)}\left(\frac{1}{r}\right) = \left[\int_{z_{T}}^{z_{B}} \hat{b}(z) \frac{z - z_{T}}{z_{B} - z_{T}} \,\mathrm{d}z\right] \otimes \mathcal{A} \otimes \left[\int_{R_{B}}^{R_{F}} \bar{b}(r) \bar{b}(r)^{T} \frac{1}{r} \,\mathrm{d}r\right]$$
$$\tilde{\mathcal{R}}_{T}^{(a)}\left(\frac{1}{r}\right) = \left[\int_{z_{T}}^{z_{B}} \hat{b}(z) \frac{z_{B} - z}{z_{B} - z_{T}} \,\mathrm{d}z\right] \otimes \mathcal{A} \otimes \left[\int_{R_{B}}^{R_{F}} \bar{b}(r) \bar{b}(r)^{T} \frac{1}{r} \,\mathrm{d}r\right]$$

Using these matrices together with matrices defined for System I, (4.2.174) multiplied by  $(\hat{b}(z) \otimes B(r)^T)$  and integrated with respect to r and z, has the form

$$G_{0}\dot{\bar{c}}_{1}(t) + \left(G_{2}(r) + G_{0}\left(\frac{1}{r}\right) + G_{2}(z)\right)\bar{c}_{1}(t) =$$

$$= -H_{0}\dot{\bar{\mu}}_{1}(t) + H_{0}(r)\dot{\bar{\eta}}_{2}(t) - H_{1}\dot{\bar{\lambda}}_{1}(t) - \tilde{\mathcal{R}}_{B}\dot{\bar{\nu}}_{\tilde{B}1}(t) - \tilde{\mathcal{R}}_{T}\dot{\bar{\nu}}_{\tilde{T}1}(t)$$

$$+ \mathcal{R}_{F}(a)\bar{\nu}_{F1} - \tilde{\mathcal{R}}_{B}(a)\bar{\nu}_{\tilde{B}1}(t) - \tilde{\mathcal{R}}_{T}(a)\bar{\nu}_{\tilde{T}1}(t)$$

$$- \mathcal{R}_{F}^{(a)}\left(\frac{1}{r}\right)\bar{\nu}_{F1} - \tilde{\mathcal{R}}_{B}^{(a)}\left(\frac{1}{r}\right)\bar{\nu}_{\tilde{B}1}(t) - \tilde{\mathcal{R}}_{T}^{(a)}\left(\frac{1}{r}\right)\bar{\nu}_{\tilde{T}1}(t)$$

$$- \mathcal{R}_{F}(z)\bar{\nu}_{F1} - \tilde{\mathcal{R}}_{0}(z)\bar{\nu}_{\tilde{B}1}(t) + \tilde{\mathcal{R}}_{0}(z)\bar{\nu}_{\tilde{T}1}(t) \qquad (4.2.179)$$
From (4.2.144)-(4.2.146)

$$\dot{\bar{\mu}}_1(t) = P_{100} \mathcal{Q}_1^{-1} \Big[ \bar{\mathcal{Z}}_1 \dot{\bar{c}}_1(t) + \tilde{X}_{B1} \dot{\bar{\nu}}_{\tilde{B}1}(t) + \tilde{X}_{T1} \dot{\bar{\nu}}_{\tilde{T}1}(t) \Big]$$
(4.2.180)

$$\dot{\bar{\eta}}_2(t) = -P_{010}\mathcal{Q}_1^{-1} \Big[ \bar{\mathcal{Z}}_1 \dot{\bar{c}}_1(t) + \tilde{X}_{B1} \dot{\bar{\nu}}_{\tilde{B}1}(t) + \tilde{X}_{T1} \dot{\bar{\nu}}_{\tilde{T}1}(t) \Big]$$
(4.2.181)

$$\dot{\bar{\lambda}}_{1}(t) = P_{001} \mathcal{Q}_{1}^{-1} \Big[ \bar{\mathcal{Z}}_{1} \dot{\bar{c}}_{1}(t) + \tilde{X}_{B1} \dot{\bar{\nu}}_{\tilde{B}1}(t) + \tilde{X}_{T1} \dot{\bar{\nu}}_{\tilde{T}1}(t) \Big]$$
(4.2.182)

Applying (4.2.180)-(4.2.182) to (4.2.179), we have

$$\mathcal{H}_{1}\bar{c}_{1}(t) + \mathcal{G}_{1}\bar{c}_{1}(t) = \mathcal{D}_{\tilde{B}1}\bar{\nu}_{\tilde{B}1}(t) + \mathcal{D}_{\tilde{T}1}\bar{\nu}_{\tilde{T}1}(t) + \mathcal{F}_{F1}\bar{\nu}_{F1} + \mathcal{F}_{\tilde{B}1}\bar{\nu}_{\tilde{B}1}(t) + \mathcal{F}_{\tilde{T}1}\bar{\nu}_{\tilde{T}1}(t)$$
(4.2.183)

where

$$\mathcal{H}_{1} = G_{0} + \left(H_{0}P_{100} + H_{0}(r)P_{010} + H_{1}P_{001}\right)\mathcal{A}_{1}^{-1}\bar{\mathcal{Z}}_{1}$$

$$\mathcal{G}_{1} = G_{2}(r) + G_{0}\left(\frac{1}{r}\right) + G_{2}(z)$$

$$\mathcal{D}_{\tilde{B}1} = -\left(H_{0}P_{100} + H_{0}(r)P_{010} + H_{1}P_{001}\right)\mathcal{A}_{1}^{-1}\tilde{X}_{B1} - \tilde{\mathcal{R}}_{B}$$

$$\mathcal{D}_{\tilde{T}1} = -\left(H_{0}P_{100} + H_{0}(r)P_{010} + H_{1}P_{001}\right)\mathcal{A}_{1}^{-1}\tilde{X}_{T1} - \tilde{\mathcal{R}}_{T}$$

$$\mathcal{F}_{F1} = -\left(-\mathcal{R}_{F}(a) + \mathcal{R}_{F}^{(a)}\left(\frac{1}{r}\right) + \mathcal{R}_{F}(z)\right)$$

$$\mathcal{F}_{\tilde{B}1} = -\left(\mathcal{R}_{B}(a) + \mathcal{R}_{B}^{(a)}\left(\frac{1}{r}\right) + \tilde{\mathcal{R}}_{0}(z)\right)$$

$$\mathcal{F}_{\tilde{T}1} = -\left(\mathcal{R}_{T}(a) + \mathcal{R}_{T}^{(a)}\left(\frac{1}{r}\right) + \tilde{\mathcal{R}}_{0}(z)\right)$$

$$(4.2.184)$$

Initial conditions are given by (4.2.172) with i = 1 and  $\bar{\nu}_{B1}(0) = 0$ .

From (4.2.172) and (4.2.183) with i = 1 and  $\bar{\nu}_{B1}(0) = 0$  we obtain the following system.

System of ODEs for TCP II (corresponds to  $\sin \theta$  approximation):

$$\mathcal{H}_{1}\dot{\bar{c}}_{1}(t) + \mathcal{G}_{1}\bar{c}_{1}(t) = \mathcal{D}_{\tilde{B}1}\dot{\bar{\nu}}_{\tilde{B}1}(t) + \mathcal{D}_{\tilde{T}1}\dot{\bar{\nu}}_{\tilde{T}1}(t) + \mathcal{F}_{F1}\bar{\nu}_{F1} + \mathcal{F}_{\tilde{B}1}\bar{\nu}_{\tilde{B}1}(t) + \mathcal{F}_{\tilde{T}1}\bar{\nu}_{\tilde{T}1}(t)$$
(4.2.183)

with initial conditions

$$\bar{c}_1(0) = \bar{\nu}_{I1} + G_0^{-1} \left[ \mathcal{R}_F \bar{\nu}_{F1} + \tilde{\mathcal{R}}_B \bar{\nu}_{\tilde{B}1}(0) + \tilde{\mathcal{R}}_T \bar{\nu}_{\tilde{T}1}(0) \right]$$
(4.2.185)

System III (corresponds to  $\cos \theta$  approximation)

System III is given by (4.2.34)-(4.2.36):

$$M\dot{V}_{2} - A\left[\frac{1}{r}(rV_{2_{r}})_{r} - \frac{1}{r^{2}}V_{2} + (V_{2_{z}})_{z}\right] = -M\dot{W}_{2} + A\left(\frac{1}{r}(rW_{2_{r}})_{r} - \frac{1}{r^{2}}W_{2} + (W_{2_{z}})_{z}\right)$$
$$-\vec{b}_{0}\left(\dot{u}_{2_{r}} + \frac{1}{r}(\dot{u}_{2} + \dot{v}_{1}) + \dot{w}_{2_{z}}\right)$$
(4.2.34)

with boundary conditions

$$V_2(r, z, t) = 0$$
 on  $\Gamma \times [0, t_f)$  (4.2.35)

and initial conditions

$$V_2(r, z, 0) = \tilde{V}_{I2}(r, z) - W_2(r, z, 0)$$
(4.2.36)

Using  $\mathcal{A} = M^{-1}A$ , (4.2.34) has the form

$$\dot{V}_{2} - \mathcal{A}\left[\frac{1}{r}(rV_{2_{r}})_{r} - \frac{1}{r^{2}}V_{2} + (V_{2_{z}})_{z}\right] = -M^{-1}\vec{b}_{0}\left(\dot{u}_{2_{r}} + \frac{1}{r}\dot{u}_{2} + \frac{1}{r}\dot{v}_{1} + \dot{w}_{2_{z}}\right)$$
$$-\dot{W}_{2} + \mathcal{A}\left[\frac{1}{r}(rW_{2_{r}})_{r} - \frac{1}{r^{2}}W_{2} + (W_{2_{z}})_{z}\right]$$
(4.2.186)

Applying (4.2.87) to the left-hand side of (4.2.186), multiplying the result by  $(\hat{b}(z) \otimes B(r)^T)$  and integrating with respect to r and z, we obtain

$$\left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z) \hat{b}(z)^T \, \mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T B(r) r \, \mathrm{d}r \right] \right\} \dot{\bar{c}}_2(t) \\
+ \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z) \hat{b}(z)^T \, \mathrm{d}z \right] \otimes \mathcal{A} \otimes \left( \left[ \int_{R_B}^{R_F} \bar{b}_r(r) \bar{b}_r(r)^T r \, \mathrm{d}r \right] + \left[ \int_{R_B}^{R_F} \bar{b}(r) \bar{b}(r)^T \frac{1}{r} \, \mathrm{d}r \right] \right) \\
+ \left[ \int_{z_T}^{z_B} \hat{b}_z(z) \hat{b}_z(z)^T \, \mathrm{d}z \right] \otimes \mathcal{A} \otimes \left[ \int_{R_B}^{R_F} \bar{b}(r) \bar{b}(r)^T r \, \mathrm{d}r \right] \right\} \bar{c}_2(t) \tag{4.2.187}$$

Using (4.2.90)-(4.2.92), the last term of the right-hand side of (4.2.186) is

$$-M^{-1}\vec{b}_{0}\left(\dot{u}_{2r}+\frac{1}{r}\dot{u}_{2}+\frac{1}{r}\dot{v}_{1}+\dot{w}_{2z}\right)$$
  
=  $-M^{-1}\vec{b}_{0}\left[\left(\hat{\mathbf{b}}(z)^{T}\otimes\bar{\mathbf{b}}_{r}(r)^{T}\right)\dot{\mu}_{2}(t)+\left(\hat{\mathbf{b}}(z)^{T}\otimes\frac{1}{r}\bar{\mathbf{b}}(r)^{T}\right)\dot{\mu}_{2}(t)\right]$   
+  $\left(\hat{\mathbf{b}}(z)^{T}\otimes\frac{1}{r}\bar{\mathbf{b}}(r)^{T}\right)\dot{\eta}_{1}(t)+\left(\hat{\mathbf{b}}_{z}(z)^{T}\otimes\bar{\mathbf{b}}(r)^{T}\right)\dot{\lambda}_{2}(t)\right]$ 

Multiplying by  $(\hat{b}(z) \otimes B(r)^T)$  and integrating with respect to r and z, we have

$$-\left\{ \left[ \int_{z_{T}}^{z_{B}} \hat{b}(z) \,\hat{\mathbf{b}}(z)^{T} \,\mathrm{d}z \right] \otimes \left[ \int_{R_{B}}^{R_{F}} B(r)^{T} M^{-1} \vec{b}_{0}(r \,\bar{\mathbf{b}}_{r}(r) + \bar{\mathbf{b}}(r))^{T} \,\mathrm{d}r \right] \right\} \dot{\mu}_{2}(t) \\ -\left\{ \left[ \int_{z_{T}}^{z_{B}} \hat{b}(z) \,\hat{\mathbf{b}}(z)^{T} \,\mathrm{d}z \right] \otimes \left[ \int_{R_{B}}^{R_{F}} B(r)^{T} M^{-1} \vec{b}_{0} \,\bar{\mathbf{b}}(r)^{T} \,\mathrm{d}r \right] \right\} \dot{\eta}_{1}(t) \\ -\left\{ \left[ \int_{z_{T}}^{z_{B}} \hat{b}(z) \,\hat{\mathbf{b}}_{z}(z)^{T} \,\mathrm{d}z \right] \otimes \left[ \int_{R_{B}}^{R_{F}} B(r)^{T} M^{-1} \vec{b}_{0} \,\bar{\mathbf{b}}(r)^{T} r \,\mathrm{d}r \right] \right\} \dot{\lambda}_{2}(t)$$
(4.2.188)

Multiplying the corresponding term in (4.2.186) by  $(\hat{b}(z) \otimes B(r)^T)$  and integrating with respect to r and z, we have the following results

for 
$$\dot{W}_{2}$$
: (4.2.161) with  $\dot{\bar{\nu}}_{B2}(t) = 0$   
for  $\mathcal{A}\frac{1}{r}(rW_{2_{r}})_{r}$ : (4.2.163) with  $\bar{\nu}_{B2}(t) = 0$   
for  $\mathcal{A}(rW_{2_{z}})_{z}$ : (4.2.165) with  $\bar{\nu}_{B2}(t) = 0$   
for  $\mathcal{A}\frac{1}{r^{2}}W_{2}$ : (4.2.178).

Using matrices defined for Systems I and II, (4.2.186) multiplied by  $(\hat{b}(z) \otimes B(r)^T)$ and integrated with respect to r and z, has the form

$$G_{0}\dot{\bar{c}}_{2}(t) + \left(G_{2}(r) + G_{0}\left(\frac{1}{r}\right) + G_{2}(z)\right)\bar{c}_{2}(t) =$$

$$= -H_{0}\dot{\bar{\mu}}_{2}(t) - H_{0}(r)\dot{\bar{\eta}}_{1}(t) - H_{1}\dot{\bar{\lambda}}_{2}(t) - \tilde{\mathcal{R}}_{B}\dot{\bar{\nu}}_{\tilde{B}2}(t) - \tilde{\mathcal{R}}_{T}\dot{\bar{\nu}}_{\tilde{T}2}(t)$$

$$+ \mathcal{R}_{F}(a)\bar{\nu}_{F2} - \tilde{\mathcal{R}}_{B}(a)\bar{\nu}_{\tilde{B}2}(t) - \tilde{\mathcal{R}}_{T}(a)\bar{\nu}_{\tilde{T}2}(t)$$

$$- \mathcal{R}_{F}^{(a)}\left(\frac{1}{r}\right)\bar{\nu}_{F2} - \tilde{\mathcal{R}}_{B}^{(a)}\left(\frac{1}{r}\right)\bar{\nu}_{\tilde{B}2}(t) - \tilde{\mathcal{R}}_{T}^{(a)}\left(\frac{1}{r}\right)\bar{\nu}_{\tilde{T}2}(t)$$

$$- \mathcal{R}_{F}(z)\bar{\nu}_{F2} - \tilde{\mathcal{R}}_{0}(z)\bar{\nu}_{\tilde{B}2}(t) + \tilde{\mathcal{R}}_{0}(z)\bar{\nu}_{\tilde{T}2}(t) \qquad (4.2.189)$$

From (4.2.147)-(4.2.149)

$$\dot{\bar{\mu}}_{2}(t) = P_{100} \mathcal{Q}_{1}^{-1} \Big[ \bar{\mathcal{Z}}_{1} \dot{\bar{c}}_{2}(t) + \tilde{X}_{B1} \dot{\bar{\nu}}_{\tilde{B}2}(t) + \tilde{X}_{T1} \dot{\bar{\nu}}_{\tilde{T}2}(t) \Big]$$
(4.2.190)

$$\dot{\bar{\eta}}_1(t) = P_{010} \mathcal{Q}_1^{-1} \Big[ \bar{\mathcal{Z}}_1 \dot{\bar{c}}_2(t) + \tilde{X}_{B1} \dot{\bar{\nu}}_{\tilde{B}2}(t) + \tilde{X}_{T1} \dot{\bar{\nu}}_{\tilde{T}2}(t) \Big]$$
(4.2.191)

$$\dot{\bar{\lambda}}_{2}(t) = P_{001} \mathcal{Q}_{1}^{-1} \left[ \bar{\mathcal{Z}}_{1} \dot{\bar{c}}_{2}(t) + \tilde{X}_{B1} \dot{\bar{\nu}}_{\tilde{B}2}(t) + \tilde{X}_{T1} \dot{\bar{\nu}}_{\tilde{T}2}(t) \right]$$
(4.2.192)

Applying (4.2.190)-(4.2.192) to (4.2.189), we have

$$\mathcal{H}_{1}\dot{\bar{c}}_{2}(t) + \mathcal{G}_{1}\bar{c}_{2}(t) = \mathcal{D}_{\tilde{B}1}\dot{\bar{\nu}}_{\tilde{B}2}(t) + \mathcal{D}_{\tilde{T}1}\dot{\bar{\nu}}_{\tilde{T}2}(t) + \mathcal{F}_{F1}\bar{\nu}_{F2} + \mathcal{F}_{\tilde{B}1}\bar{\nu}_{\tilde{B}2}(t) + \mathcal{F}_{\tilde{T}1}\bar{\nu}_{\tilde{T}2}(t)$$
(4.2.193)

where matrices  $\mathcal{H}_1, \mathcal{G}_1, \mathcal{D}_{\tilde{B}1}, \mathcal{D}_{\tilde{T}1}, \mathcal{F}_{F1}, \mathcal{F}_{\tilde{B}1}, \mathcal{F}_{\tilde{T}1}$  are given by (4.2.183).

Initial conditions are given by (4.2.172) with i = 2 and  $\bar{\nu}_{B2}(0) = 0$ .

From (4.2.193) and (4.2.172) with i = 2 and  $\bar{\nu}_{B2}(0) = 0$  we obtain the following system.

System of ODEs for TCP III (corresponds to  $\cos \theta$  approximation):

$$\mathcal{H}_{1}\dot{\bar{c}}_{2}(t) + \mathcal{G}_{1}\bar{c}_{2}(t) = \mathcal{D}_{\tilde{B}1}\dot{\bar{\nu}}_{\tilde{B}2}(t) + \mathcal{D}_{\tilde{T}1}\dot{\bar{\nu}}_{\tilde{T}2}(t) + \mathcal{F}_{F1}\bar{\nu}_{F2} + \mathcal{F}_{\tilde{B}1}\bar{\nu}_{\tilde{B}2}(t) + \mathcal{F}_{\tilde{T}1}\bar{\nu}_{\tilde{T}2}(t)$$
(4.2.193)

with initial conditions

$$\bar{c}_2(0) = \bar{\nu}_{I2} + G_0^{-1} \left[ \mathcal{R}_F \bar{\nu}_{F2} + \tilde{\mathcal{R}}_B \bar{\nu}_{\tilde{B}2}(0) + \tilde{\mathcal{R}}_T \bar{\nu}_{\tilde{T}2}(0) \right]$$
(4.2.194)

System IV (corresponds to  $\sin 2\theta$  approximation)

System IV is given by (4.2.37)-(4.2.39):

$$M\dot{V}_{3} - A\left[\frac{1}{r}(rV_{3r})_{r} - \frac{4}{r^{2}}V_{3} + (V_{3z})_{z}\right] = -M\dot{W}_{3} + A\left(\frac{1}{r}(rW_{3r})_{r} - \frac{4}{r^{2}}W_{3} + (W_{3z})_{z}\right)$$
$$-\vec{b}_{0}\left(\dot{u}_{3r} + \frac{1}{r}(\dot{u}_{3} - 2\dot{v}_{4}) + \dot{w}_{3z}\right)$$
(4.2.37)

with boundary conditions

$$V_3(r, z, t) = 0$$
 on  $\Gamma \times [0, t_f)$  (4.2.38)

and initial conditions

$$V_3(r,z,0) = \tilde{V}_{I3}(r,z) - W_3(r,z,0)$$
(4.2.39)

Using  $\mathcal{A} = M^{-1}A$ , (4.2.37) has the form

$$\dot{V}_{3} - \mathcal{A}\left[\frac{1}{r}(rV_{3_{r}})_{r} - \frac{4}{r^{2}}V_{3} + (V_{3_{z}})_{z}\right] = -M^{-1}\vec{b}_{0}\left(\dot{u}_{3_{r}} + \frac{1}{r}\dot{u}_{3} - \frac{2}{r}\dot{v}_{4} + \dot{w}_{3_{z}}\right)$$
$$-\dot{W}_{3} + \mathcal{A}\left[\frac{1}{r}(rW_{3_{r}})_{r} - \frac{4}{r^{2}}W_{3} + (W_{3_{z}})_{z}\right]$$
(4.2.195)

Applying (4.2.87) to the left-hand side of (4.2.195), multiplying the result by  $(\hat{b}(z) \otimes B(r)^T)$  and integrating with respect to r and z, we obtain

$$\left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z)\hat{b}(z)^T \,\mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T B(r)r \,\mathrm{d}r \right] \right\} \dot{\bar{c}}_3(t) \\
+ \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z)\hat{b}(z)^T \,\mathrm{d}z \right] \otimes \mathcal{A} \otimes \left( \left[ \int_{R_B}^{R_F} \bar{b}_r(r)\bar{b}_r(r)^T r \,\mathrm{d}r \right] + 4 \left[ \int_{R_B}^{R_F} \bar{b}(r)\bar{b}(r)^T \frac{1}{r} \,\mathrm{d}r \right] \right) \\
+ \left[ \int_{z_T}^{z_B} \hat{b}_z(z)\hat{b}_z(z)^T \,\mathrm{d}z \right] \otimes \mathcal{A} \otimes \left[ \int_{R_B}^{R_F} \bar{b}(r)\bar{b}(r)^T r \,\mathrm{d}r \right] \right\} \bar{c}_2(t) \qquad (4.2.196)$$

Using (4.2.90)-(4.2.92), the first term of the right-hand side of (4.2.195) is

$$-M^{-1}\vec{b}_{0}\left(\dot{u}_{3r}+\frac{1}{r}\dot{u}_{3}-\frac{2}{r}\dot{v}_{4}+\dot{w}_{3z}\right)$$
  
=  $-M^{-1}\vec{b}_{0}\left[\left(\hat{\mathbf{b}}(z)^{T}\otimes\bar{\mathbf{b}}_{r}(r)^{T}\right)\dot{\mu}_{3}(t)+\left(\hat{\mathbf{b}}(z)^{T}\otimes\frac{1}{r}\bar{\mathbf{b}}(r)^{T}\right)\dot{\mu}_{3}(t)-\left(\hat{\mathbf{b}}(z)^{T}\otimes\frac{2}{r}\bar{\mathbf{b}}(r)^{T}\right)\dot{\eta}_{4}(t)+\left(\hat{\mathbf{b}}_{z}(z)^{T}\otimes\bar{\mathbf{b}}(r)^{T}\right)\dot{\lambda}_{3}(t)\right]$ 

Multiplying by  $(\hat{b}(z) \otimes B(r)^T)$  and integrating with respect to r and z, we have

$$-\left\{\left[\int_{z_T}^{z_B}\hat{b}(z)\,\hat{\mathbf{b}}(z)^T\,\mathrm{d}z\right]\otimes\left[\int_{R_B}^{R_F}B(r)^TM^{-1}\vec{b}_0(r\,\bar{\mathbf{b}}_r(r)+\,\bar{\mathbf{b}}(r))^T\,\mathrm{d}r\right]\right\}\dot{\bar{\mu}}_3(t)$$

$$+2\left\{\left[\int_{z_{T}}^{z_{B}}\hat{b}(z)\,\hat{\mathbf{b}}(z)^{T}\,\mathrm{d}z\right]\otimes\left[\int_{R_{B}}^{R_{F}}B(r)^{T}M^{-1}\vec{b}_{0}\,\bar{\mathbf{b}}(r)^{T}\,\mathrm{d}r\right]\right\}\dot{\eta}_{4}(t) \\ -\left\{\left[\int_{z_{T}}^{z_{B}}\hat{b}(z)\,\hat{\mathbf{b}}_{z}(z)^{T}\,\mathrm{d}z\right]\otimes\left[\int_{R_{B}}^{R_{F}}B(r)^{T}M^{-1}\vec{b}_{0}\,\bar{\mathbf{b}}(r)^{T}r\,\mathrm{d}r\right]\right\}\dot{\lambda}_{3}(t) \qquad (4.2.197)$$

Multiplying the corresponding term in (4.2.195) by  $(\hat{b}(z) \otimes B(r)^T)$  and integrating with respect to r and z, we have the following results

for 
$$\dot{W}_3$$
: (4.2.161) with  $\dot{\bar{\nu}}_{B3}(t) = 0$   
for  $\mathcal{A}\frac{1}{r}(rW_{3r})_r$ : (4.2.163) with  $\bar{\nu}_{B3}(t) = 0$   
for  $\mathcal{A}(rW_{3z})_z$ : (4.2.165) with  $\bar{\nu}_{B3}(t) = 0$   
for  $\mathcal{A}\frac{4}{r^2}W_3 = 4\mathcal{A}\frac{1}{r^2}W_3$ : (4.2.178).

Using matrices defined for Systems I and II, (4.2.195) multiplied by  $(\hat{b}(z) \otimes B(r)^T)$ and integrated with respect to r and z, has the form

$$G_{0}\dot{\bar{c}}_{3}(t) + \left(G_{2}(r) + 4G_{0}\left(\frac{1}{r}\right) + G_{2}(z)\right)\bar{c}_{3}(t) =$$

$$= -H_{0}\dot{\mu}_{3}(t) + 2H_{0}(r)\dot{\bar{\eta}}_{4}(t) - H_{1}\dot{\bar{\lambda}}_{3}(t) - \tilde{\mathcal{R}}_{B}\dot{\bar{\nu}}_{\bar{B}3}(t) - \tilde{\mathcal{R}}_{T}\dot{\bar{\nu}}_{\bar{T}3}(t)$$

$$+ \mathcal{R}_{F}(a)\bar{\nu}_{F3} - \tilde{\mathcal{R}}_{B}(a)\bar{\nu}_{\bar{B}3}(t) - \tilde{\mathcal{R}}_{T}(a)\bar{\nu}_{\bar{T}3}(t)$$

$$- 4\mathcal{R}_{F}^{(a)}\left(\frac{1}{r}\right)\bar{\nu}_{F3} - 4\tilde{\mathcal{R}}_{B}^{(a)}\left(\frac{1}{r}\right)\bar{\nu}_{\bar{B}3}(t) - 4\tilde{\mathcal{R}}_{T}^{(a)}\left(\frac{1}{r}\right)\bar{\nu}_{\bar{T}3}(t)$$

$$- \mathcal{R}_{F}(z)\bar{\nu}_{F3} - \tilde{\mathcal{R}}_{0}(z)\bar{\nu}_{\bar{B}3}(t) + \tilde{\mathcal{R}}_{0}(z)\bar{\nu}_{\bar{T}3}(t) \qquad (4.2.198)$$

From (4.2.150)-(4.2.152)

$$\dot{\mu}_{3}(t) = P_{100} \mathcal{Q}_{2}^{-1} \Big[ \bar{\mathcal{Z}}_{2} \dot{\bar{c}}_{3}(t) + \tilde{X}_{B2} \dot{\bar{\nu}}_{\tilde{B}3}(t) + \tilde{X}_{T2} \dot{\bar{\nu}}_{\tilde{T}3}(t) \Big]$$
(4.2.199)

$$\dot{\bar{\eta}}_4(t) = -P_{010}\mathcal{Q}_2^{-1} \Big[ \bar{\mathcal{Z}}_2 \dot{\bar{c}}_3(t) + \tilde{X}_{B2} \dot{\bar{\nu}}_{\tilde{B}3}(t) + \tilde{X}_{T2} \dot{\bar{\nu}}_{\tilde{T}3}(t) \Big]$$
(4.2.200)

$$\dot{\bar{\lambda}}_{3}(t) = P_{001} \mathcal{Q}_{2}^{-1} \Big[ \bar{\mathcal{Z}}_{2} \dot{\bar{c}}_{3}(t) + \tilde{X}_{B2} \dot{\bar{\nu}}_{\tilde{B}3}(t) + \tilde{X}_{T2} \dot{\bar{\nu}}_{\tilde{T}3}(t) \Big]$$
(4.2.201)

Applying (4.2.199)-(4.2.201) to (4.2.198), we have

$$\mathcal{H}_{2}\bar{c}_{3}(t) + \mathcal{G}_{2}\bar{c}_{3}(t) = \mathcal{D}_{\tilde{B}2}\bar{\nu}_{\tilde{B}3}(t) + \mathcal{D}_{\tilde{T}2}\bar{\nu}_{\tilde{T}3}(t) + \mathcal{F}_{F2}\bar{\nu}_{F3} + \mathcal{F}_{\tilde{B}2}\bar{\nu}_{\tilde{B}3}(t) + \mathcal{F}_{\tilde{T}2}\bar{\nu}_{\tilde{T}3}(t)$$
(4.2.202)

where

$$\mathcal{H}_{2} = G_{0} + \left(H_{0}P_{100} + 2H_{0}(r)P_{010} + H_{1}P_{001}\right)\mathcal{A}_{2}^{-1}\bar{\mathcal{Z}}_{2}$$

$$\mathcal{G}_{2} = G_{2}(r) + 4G_{0}\left(\frac{1}{r}\right) + G_{2}(z)$$

$$\mathcal{D}_{\tilde{B}2} = -\left(H_{0}P_{100} + 2H_{0}(r)P_{010} + H_{1}P_{001}\right)\mathcal{A}_{2}^{-1}\tilde{X}_{B2} - \tilde{\mathcal{R}}_{B}$$

$$\mathcal{D}_{\tilde{T}2} = -\left(H_{0}P_{100} + 2H_{0}(r)P_{010} + H_{1}P_{001}\right)\mathcal{A}_{2}^{-1}\tilde{X}_{T2} - \tilde{\mathcal{R}}_{T}$$

$$\mathcal{F}_{F2} = -\left(-\mathcal{R}_{F}(a) + 4\mathcal{R}_{F}^{(a)}\left(\frac{1}{r}\right) + \mathcal{R}_{F}(z)\right)$$

$$\mathcal{F}_{\tilde{B}2} = -\left(\mathcal{R}_{B}(a) + 4\mathcal{R}_{B}^{(a)}\left(\frac{1}{r}\right) + \tilde{\mathcal{R}}_{0}(z)\right)$$

$$\mathcal{F}_{\tilde{T}2} = -\left(\mathcal{R}_{T}(a) + 4\mathcal{R}_{T}^{(a)}\left(\frac{1}{r}\right) + \tilde{\mathcal{R}}_{0}(z)\right)$$

$$(4.2.203)$$

Initial conditions are given by (4.2.172) with i = 3 and  $\bar{\nu}_{B3}(0) = 0$ .

From (4.2.202) and (4.2.172) with i = 3 and  $\bar{\nu}_{B3}(0) = 0$  we obtain the following system.

System of ODEs for TCP IV (corresponds to  $\sin 2\theta$  approximation):

$$\mathcal{H}_{2}\dot{\bar{c}}_{3}(t) + \mathcal{G}_{2}\bar{c}_{3}(t) = \mathcal{D}_{\tilde{B}2}\dot{\bar{\nu}}_{\tilde{B}3}(t) + \mathcal{D}_{\tilde{T}2}\dot{\bar{\nu}}_{\tilde{T}3}(t) + \mathcal{F}_{F2}\bar{\nu}_{F3} + \mathcal{F}_{\tilde{B}2}\bar{\nu}_{\tilde{B}3}(t) + \mathcal{F}_{\tilde{T}2}\bar{\nu}_{\tilde{T}3}(t)$$
(4.2.202)

with initial conditions

$$\bar{c}_3(0) = \bar{\nu}_{I3} + G_0^{-1} \left[ \mathcal{R}_F \bar{\nu}_{F3} + \tilde{\mathcal{R}}_B \bar{\nu}_{\tilde{B}3}(0) + \tilde{\mathcal{R}}_T \bar{\nu}_{\tilde{T}3}(0) \right]$$
(4.2.204)

System V (corresponds to  $\cos 2\theta$  approximation)

System V is given by (4.2.40)-(4.2.42):

$$M\dot{V}_{4} - A\left[\frac{1}{r}(rV_{4_{r}})_{r} - \frac{4}{r^{2}}V_{4} + (V_{4_{z}})_{z}\right] = -M\dot{W}_{4} + A\left(\frac{1}{r}(rW_{4_{r}})_{r} - \frac{4}{r^{2}}W_{4} + (W_{4_{z}})_{z}\right)$$
$$-\vec{b}_{0}\left(\dot{u}_{4_{r}} + \frac{1}{r}(\dot{u}_{4} + 2\dot{v}_{3}) + \dot{w}_{4_{z}}\right)$$
(4.2.40)

with boundary conditions

$$V_4(r, z, t) = 0$$
 on  $\Gamma \times [0, t_f)$  (4.2.41)

and initial conditions

$$V_4(r, z, 0) = \tilde{V}_{I4}(r, z) - W_4(r, z, 0)$$
(4.2.42)

Using  $\mathcal{A} = M^{-1}A$ , (4.2.40) has the form

$$\dot{V}_4 - \mathcal{A}\Big[\frac{1}{r}(rV_{4_r})_r - \frac{4}{r^2}V_4 + (V_{4_z})_z\Big] = -M^{-1}\vec{b}_0\Big(\dot{u}_{4_r} + \frac{1}{r}\dot{u}_4 + \frac{2}{r}\dot{v}_3 + \dot{w}_{4_z}\Big) -\dot{W}_4 + \mathcal{A}\Big[\frac{1}{r}(rW_{4_r})_r - \frac{4}{r^2}W_4 + (W_{4_z})_z\Big]$$
(4.2.205)

Applying (4.2.87) to the left-hand side of (4.2.205), multiplying the result by  $(\hat{b}(z) \otimes B(r)^T)$  and integrating with respect to r and z, we obtain

$$\left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z)\hat{b}(z)^T \,\mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T B(r)r \,\mathrm{d}r \right] \right\} \dot{\bar{c}}_4(t) \\
+ \left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z)\hat{b}(z)^T \,\mathrm{d}z \right] \otimes \mathcal{A} \otimes \left( \left[ \int_{R_B}^{R_F} \bar{b}_r(r)\bar{b}_r(r)^T r \,\mathrm{d}r \right] + 4 \left[ \int_{R_B}^{R_F} \bar{b}(r)\bar{b}(r)^T \frac{1}{r} \,\mathrm{d}r \right] \right) \\
+ \left[ \int_{z_T}^{z_B} \hat{b}_z(z)\hat{b}_z(z)^T \,\mathrm{d}z \right] \otimes \mathcal{A} \otimes \left[ \int_{R_B}^{R_F} \bar{b}(r)\bar{b}(r)^T r \,\mathrm{d}r \right] \right\} \bar{c}_4(t) \qquad (4.2.206)$$

Using (4.2.90)-(4.2.92), the last term of the right-hand side of (4.2.205) is

$$-M^{-1}\vec{b}_{0}\left(\dot{u}_{4_{r}}+\frac{1}{r}\dot{u}_{4}+\frac{2}{r}\dot{v}_{3}+\dot{w}_{4_{z}}\right)$$
  
=  $-M^{-1}\vec{b}_{0}\left[\left(\hat{\mathbf{b}}(z)^{T}\otimes\bar{\mathbf{b}}_{r}(r)^{T}\right)\dot{\mu}_{4}(t)+\left(\hat{\mathbf{b}}(z)^{T}\otimes\frac{1}{r}\bar{\mathbf{b}}(r)^{T}\right)\dot{\mu}_{4}(t)-\left(\hat{\mathbf{b}}(z)^{T}\otimes\frac{2}{r}\bar{\mathbf{b}}(r)^{T}\right)\dot{\eta}_{3}(t)+\left(\hat{\mathbf{b}}_{z}(z)^{T}\otimes\bar{\mathbf{b}}(r)^{T}\right)\dot{\lambda}_{4}(t)\right]$ 

Multiplying by  $(\hat{b}(z) \otimes B(r)^T)$  and integrating with respect to r and z, we have

$$-\left\{\left[\int_{z_T}^{z_B}\hat{b}(z)\,\hat{\mathbf{b}}(z)^T\,\mathrm{d}z\right]\otimes\left[\int_{R_B}^{R_F}B(r)^TM^{-1}\vec{b}_0(r\,\bar{\mathbf{b}}_r(r)+\,\bar{\mathbf{b}}(r))^T\,\mathrm{d}r\right]\right\}\dot{\bar{\mu}}_4(t)$$

$$-2\left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z) \,\hat{\mathbf{b}}(z)^T \,\mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T M^{-1} \vec{b}_0 \,\bar{\mathbf{b}}(r)^T \,\mathrm{d}r \right] \right\} \dot{\bar{\eta}}_3(t) \\ -\left\{ \left[ \int_{z_T}^{z_B} \hat{b}(z) \,\hat{\mathbf{b}}_z(z)^T \,\mathrm{d}z \right] \otimes \left[ \int_{R_B}^{R_F} B(r)^T M^{-1} \vec{b}_0 \,\bar{\mathbf{b}}(r)^T r \,\mathrm{d}r \right] \right\} \dot{\bar{\lambda}}_4(t) \qquad (4.2.207)$$

Multiplying the corresponding term in (4.2.205) by  $(\hat{b}(z) \otimes B(r)^T)$  and integrating with respect to r and z, we have the following results

for 
$$\dot{W}_4$$
: (4.2.161) with  $\dot{\bar{\nu}}_{B4}(t) = 0$   
for  $\mathcal{A}\frac{1}{r}(rW_{4_r})_r$ : (4.2.163) with  $\bar{\nu}_{B4}(t) = 0$   
for  $\mathcal{A}(rW_{4_z})_z$ : (4.2.165) with  $\bar{\nu}_{B4}(t) = 0$   
for  $\mathcal{A}\frac{4}{r^2}W_4 = 4\mathcal{A}\frac{1}{r^2}W_4$ : (4.2.178).

Using matrices defined for Systems I and II, (4.2.195) multiplied by  $(\hat{b}(z) \otimes B(r)^T)$ and integrated with respect to r and z, has the form

$$G_{0}\dot{\bar{c}}_{4}(t) + \left(G_{2}(r) + 4G_{0}\left(\frac{1}{r}\right) + G_{2}(z)\right)\bar{c}_{4}(t) =$$

$$= -H_{0}\dot{\bar{\mu}}_{4}(t) - 2H_{0}(r)\dot{\bar{\eta}}_{3}(t) - H_{1}\dot{\bar{\lambda}}_{4}(t) - \tilde{\mathcal{R}}_{B}\dot{\bar{\nu}}_{\bar{B}4}(t) - \tilde{\mathcal{R}}_{T}\dot{\bar{\nu}}_{\bar{T}4}(t)$$

$$+ \mathcal{R}_{F}(a)\bar{\nu}_{F4} - \tilde{\mathcal{R}}_{B}(a)\bar{\nu}_{\bar{B}4}(t) - \tilde{\mathcal{R}}_{T}(a)\bar{\nu}_{\bar{T}4}(t)$$

$$- 4\mathcal{R}_{F}^{(a)}\left(\frac{1}{r}\right)\bar{\nu}_{F4} - 4\tilde{\mathcal{R}}_{B}^{(a)}\left(\frac{1}{r}\right)\bar{\nu}_{\bar{B}4}(t) - 4\tilde{\mathcal{R}}_{T}^{(a)}\left(\frac{1}{r}\right)\bar{\nu}_{\bar{T}4}(t)$$

$$- \mathcal{R}_{F}(z)\bar{\nu}_{F4} - \tilde{\mathcal{R}}_{0}(z)\bar{\nu}_{\bar{B}4}(t) + \tilde{\mathcal{R}}_{0}(z)\bar{\nu}_{\bar{T}4}(t) \qquad (4.2.208)$$

From (4.2.153)-(4.2.155)

$$\dot{\bar{\mu}}_4(t) = P_{100} \mathcal{Q}_2^{-1} \left[ \bar{\mathcal{Z}}_2 \dot{\bar{c}}_4(t) + \tilde{X}_{B2} \dot{\bar{\nu}}_{\tilde{B}4}(t) + \tilde{X}_{T2} \dot{\bar{\nu}}_{\tilde{T}4}(t) \right]$$
(4.2.209)

$$\dot{\bar{\eta}}_{3}(t) = -P_{010}\mathcal{Q}_{2}^{-1} \Big[ \bar{\mathcal{Z}}_{2} \dot{\bar{c}}_{4}(t) + \tilde{X}_{B2} \dot{\bar{\nu}}_{\tilde{B}4}(t) + \tilde{X}_{T2} \dot{\bar{\nu}}_{\tilde{T}4}(t) \Big]$$
(4.2.210)

$$\dot{\bar{\lambda}}_4(t) = P_{001} \mathcal{Q}_2^{-1} \Big[ \bar{\mathcal{Z}}_2 \dot{\bar{c}}_4(t) + \tilde{X}_{B2} \dot{\bar{\nu}}_{\tilde{B}4}(t) + \tilde{X}_{T2} \dot{\bar{\nu}}_{\tilde{T}4}(t) \Big]$$
(4.2.211)

Applying (4.2.209)-(4.2.211) to (4.2.208), we have

$$\mathcal{H}_{2}\bar{c}_{4}(t) + \mathcal{G}_{2}\bar{c}_{4}(t) = \mathcal{D}_{\tilde{B}2}\bar{\nu}_{\tilde{B}4}(t) + \mathcal{D}_{\tilde{T}2}\bar{\nu}_{\tilde{T}4}(t) + \mathcal{F}_{F2}\bar{\nu}_{F4} + \mathcal{F}_{\tilde{B}2}\bar{\nu}_{\tilde{B}4}(t) + \mathcal{F}_{\tilde{T}2}\bar{\nu}_{\tilde{T}4}(t)$$
(4.2.212)

where matrices  $\mathcal{H}_2, \mathcal{G}_2, \mathcal{D}_{\tilde{B}2}, \mathcal{D}_{\tilde{T}2}, \mathcal{F}_{F2}, \mathcal{F}_{\tilde{B}2}, \mathcal{F}_{\tilde{T}2}$  are given by (4.2.203). Initial conditions are given by (4.2.172) with i = 4 and  $\bar{\nu}_{B4}(0) = 0$ . From (4.2.212) and (4.2.172) with i = 4 and  $\bar{\nu}_{B4}(0) = 0$  we obtain the following system.

System of ODEs for TCP V (corresponds to  $\cos 2\theta$  approximation):

$$\mathcal{H}_{2}\bar{c}_{4}(t) + \mathcal{G}_{2}\bar{c}_{4}(t) = \mathcal{D}_{\tilde{B}2}\bar{\nu}_{\tilde{B}4}(t) + \mathcal{D}_{\tilde{T}2}\bar{\nu}_{\tilde{T}4}(t) + \mathcal{F}_{F2}\bar{\nu}_{F4} + \mathcal{F}_{\tilde{B}2}\bar{\nu}_{\tilde{B}4}(t) + \mathcal{F}_{\tilde{T}2}\bar{\nu}_{\tilde{T}4}(t)$$
(4.2.212)

with initial conditions

$$\bar{c}_4(0) = \bar{\nu}_{I4} + G_0^{-1} \left[ \mathcal{R}_F \bar{\nu}_{F4} + \tilde{\mathcal{R}}_B \bar{\nu}_{\tilde{B}4}(0) + \tilde{\mathcal{R}}_T \bar{\nu}_{\tilde{T}4}(0) \right]$$
(4.2.213)

We have the following five systems of ODEs:

System of ODEs for TCP I (radially symmetric case):

$$\mathcal{H}_{0}\dot{\bar{c}}_{0}(t) + \mathcal{G}_{0}\bar{c}_{0}(t) = \mathcal{D}_{B0}\dot{\bar{\nu}}_{B0}(t) + \mathcal{D}_{\tilde{B}0}\dot{\bar{\nu}}_{\tilde{B}0}(t) + \mathcal{D}_{\tilde{T}0}\dot{\bar{\nu}}_{\tilde{T}0}(t) + \mathcal{F}_{B0}\bar{\bar{\nu}}_{B0}(t) + \mathcal{F}_{F0}\bar{\bar{\nu}}_{F0} + \mathcal{F}_{\tilde{B}0}\bar{\bar{\nu}}_{\tilde{B}0}(t) + \mathcal{F}_{\tilde{T}0}\bar{\bar{\nu}}_{\tilde{T}0}(t)$$
(4.2.169)

with initial conditions

$$\bar{c}_0(0) = \bar{\nu}_{I0} + G_0^{-1} \left[ \mathcal{R}_B \bar{\nu}_{B0}(0) + \mathcal{R}_F \bar{\nu}_{F0} + \tilde{\mathcal{R}}_B \bar{\nu}_{\tilde{B}0}(0) + \tilde{\mathcal{R}}_T \bar{\nu}_{\tilde{T}0}(0) \right]$$
(4.2.173)

System I gives coefficients  $V_0$ ,  $u_0$ ,  $\omega_0$  of the  $\theta$ -approximation (4.2.18).

System of ODEs for TCP II (corresponds to  $\sin \theta$  approximation):

$$\mathcal{H}_{1}\dot{\bar{c}}_{1}(t) + \mathcal{G}_{1}\bar{c}_{1}(t) = \mathcal{D}_{\tilde{B}1}\dot{\bar{\nu}}_{\tilde{B}1}(t) + \mathcal{D}_{\tilde{T}1}\dot{\bar{\nu}}_{\tilde{T}1}(t) + \mathcal{F}_{F1}\bar{\nu}_{F1} + \mathcal{F}_{\tilde{B}1}\bar{\nu}_{\tilde{B}1}(t) + \mathcal{F}_{\tilde{T}1}\bar{\nu}_{\tilde{T}1}(t)$$
(4.2.183)

with initial conditions

$$\bar{c}_1(0) = \bar{\nu}_{I1} + G_0^{-1} \left[ \mathcal{R}_F \bar{\nu}_{F1} + \tilde{\mathcal{R}}_B \bar{\nu}_{\tilde{B}1}(0) + \tilde{\mathcal{R}}_T \bar{\nu}_{\tilde{T}1}(0) \right]$$
(4.2.185)

System II gives  $\sin \theta$ -coefficients  $V_1$ ,  $u_1$ ,  $w_1$  and  $\cos \theta$ -coefficient  $v_2$  of the  $\theta$ -approximation (4.2.18).

System of ODEs for TCP III (corresponds to  $\cos \theta$  approximation):

$$\mathcal{H}_{1}\dot{\bar{c}}_{2}(t) + \mathcal{G}_{1}\bar{c}_{2}(t) = \mathcal{D}_{\tilde{B}1}\dot{\bar{\nu}}_{\tilde{B}2}(t) + \mathcal{D}_{\tilde{T}1}\dot{\bar{\nu}}_{\tilde{T}2}(t) + \mathcal{F}_{F1}\bar{\nu}_{F2} + \mathcal{F}_{\tilde{B}1}\bar{\nu}_{\tilde{B}2}(t) + \mathcal{F}_{\tilde{T}1}\bar{\nu}_{\tilde{T}2}(t)$$
(4.2.193)

with initial conditions

$$\bar{c}_2(0) = \bar{\nu}_{I2} + G_0^{-1} \left[ \mathcal{R}_F \bar{\nu}_{F2} + \tilde{\mathcal{R}}_B \bar{\nu}_{\tilde{B}2}(0) + \tilde{\mathcal{R}}_T \bar{\nu}_{\tilde{T}2}(0) \right]$$
(4.2.194)

System III gives  $\sin \theta$ -coefficient  $v_1$  and  $\cos \theta$ -coefficients  $V_2$ ,  $u_2$ ,  $w_2$  of the  $\theta$ -approximation (4.2.18).

System of ODEs for TCP IV (corresponds to  $\sin 2\theta$  approximation):

$$\mathcal{H}_{2}\dot{\bar{c}}_{3}(t) + \mathcal{G}_{2}\bar{c}_{3}(t) = \mathcal{D}_{\tilde{B}2}\dot{\bar{\nu}}_{\tilde{B}3}(t) + \mathcal{D}_{\tilde{T}2}\dot{\bar{\nu}}_{\tilde{T}3}(t) + \mathcal{F}_{F2}\bar{\nu}_{F3} + \mathcal{F}_{\tilde{B}2}\bar{\nu}_{\tilde{B}3}(t) + \mathcal{F}_{\tilde{T}2}\bar{\nu}_{\tilde{T}3}(t)$$
(4.2.202)

with initial conditions

$$\bar{c}_3(0) = \bar{\nu}_{I3} + G_0^{-1} \left[ \mathcal{R}_F \bar{\nu}_{F3} + \tilde{\mathcal{R}}_B \bar{\nu}_{\tilde{B}3}(0) + \tilde{\mathcal{R}}_T \bar{\nu}_{\tilde{T}3}(0) \right]$$
(4.2.204)

System IV gives  $\sin 2\theta$ -coefficients  $V_3$ ,  $u_3$ ,  $w_3$  and  $\cos 2\theta$ -coefficient  $v_4$  of the  $\theta$ -approximation (4.2.18).

System of ODEs for TCP V (corresponds to  $\cos 2\theta$  approximation):

$$\mathcal{H}_{2}\dot{\bar{c}}_{4}(t) + \mathcal{G}_{2}\bar{c}_{4}(t) = \mathcal{D}_{\tilde{B}2}\dot{\bar{\nu}}_{\tilde{B}4}(t) + \mathcal{D}_{\tilde{T}2}\dot{\bar{\nu}}_{\tilde{T}4}(t) + \mathcal{F}_{F2}\bar{\nu}_{F4} + \mathcal{F}_{\tilde{B}2}\bar{\nu}_{\tilde{B}4}(t) + \mathcal{F}_{\tilde{T}2}\bar{\nu}_{\tilde{T}4}(t)$$
(4.2.212)

with initial conditions

$$\bar{c}_4(0) = \bar{\nu}_{I4} + G_0^{-1} \left[ \mathcal{R}_F \bar{\nu}_{F4} + \tilde{\mathcal{R}}_B \bar{\nu}_{\tilde{B}4}(0) + \tilde{\mathcal{R}}_T \bar{\nu}_{\tilde{T}4}(0) \right]$$
(4.2.213)

System V gives  $\sin 2\theta$ -coefficient  $v_3$  and  $\cos 2\theta$ -coefficients  $V_4$ ,  $u_4$ ,  $w_4$  of the  $\theta$ -approximation (4.2.18).

Thus, in this chapter, numerical methods for solving fully coupled parabolicelliptic initial-boundary value problems for two- and three-dimensional TCPu processes, including radially non-symmetric processes in an inclined borehole were developed. These methods are based on a hybrid Fourier-finite-element approximation technique proposed in this research. The technique also involves a boundary penalization strategy and captures borehole geometry and mechanics. The proposed numerical methods reduce the solution of fully coupled parabolic-elliptic initial-boundary value problems to the solution of systems of time-dependent ODEs supplemented by initial conditions. The systems of ODEs explicitly contain data of the underlying boundary conditions, thereby allowing numerical analysis of a control problem in which the temperature and pressure on the borehole boundary are considered as control parameters used to achieve desirable stresses in the neighborhood of the borehole.

#### Chapter 5

# Experimental Validation of the Fourier-Finite Element Method

In this chapter, we experimentally validate the Fourier-finite element method for the TCPu models developed in Chapter 4. Numerical experiments are performed using a MATLAB code that we have written to implement the above numerical method. In this code, the backward Euler approximation is used to discretize the temporal domain. The validation is done by comparing experimental results to other available analytical and previously validated numerical solutions for thermo-chemo-poroelastic wellbore problems.

For our numerical experiments, we focus on the two-dimensional radially symmetric fully coupled TCPu model. In this case, the temperature T, the solute mass fraction C, and the pore pressure p on the far-field boundary are considered as independent of the polar angle  $\theta$ . However, applied far-field boundary stresses remain  $\theta$ -dependent in accordance with realistic geomechanical behavior around the borehole. We assume that an annular region exterior to the borehole has the inner (wellbore) radius  $R_B = 0.1$  m and the outer (far-field) radius  $R_F = 1$  m. The values of physical constants used in the computations are presented in Appendix A.

Below, we present the experimental results for the temperature T, the solute mass fraction C, and the pore pressure p, as well as for total radial stress  $\tau_r$  and total tangential stress  $\tau_{\theta}$  at different polar angles  $\theta$ . Each figure contains four solution curves that correspond to time intervals of 1 hour, 12 hours, 24 hours, and 120 hours.



Figure 5.1: Temperature distribution around the borehole



Figure 5.2: Solute mass fraction distribution around the borehole



Figure 5.3: Pore pressure distribution around the borehole



Figure 5.4: Total radial stress distribution around the borehole at the polar angle of  $\theta=0$ 



Figure 5.5: Total tangential stress distribution around the borehole at the polar angle of  $\theta=0$ 



Figure 5.6: Total radial stress distribution around the borehole at the polar angle of  $\theta=30$ 



Figure 5.7: Total tangential stress distribution around the borehole at the polar angle of  $\theta = 30$ 



Figure 5.8: Total radial stress distribution around the borehole at the polar angle of  $\theta=45$ 



Figure 5.9: Total tangential stress distribution around the borehole at the polar angle of  $\theta=45$ 



Figure 5.10: Total radial stress distribution around the borehole at the polar angle of  $\theta=60$ 



Figure 5.11: Total tangential stress distribution around the borehole at the polar angle of  $\theta=60$ 



Figure 5.12: Total radial stress distribution around the borehole at the polar angle of  $\theta=90$ 



Figure 5.13: Total tangential stress distribution around the borehole at the polar angle of  $\theta = 90$ 

The obtained experimental results are consistent with available analytical and previously validated numerical solutions for thermo-chemo-poroelastic wellbore problems presented, for instance, in Zhou and Ghassemi [46] or Rafieepour at el. [34]. The existing discrepancy between our results and those presented in [34, 46] is due to the essential difference between the TCPu model and the models considered in [34, 46]. The TCPu model includes all coupling mechanisms, whereas both models [34, 46] deal with fully decoupled thermal diffusion, and [34] neglects couplings in the solute diffusion equation, as well. This allows us to conclude that the developed Fourier-finite element method provides correct and satisfactory numerical approximation to the coupled TCPu models.

#### Chapter 6

## **Conclusions and Future Work**

This dissertation has addressed the questions of well-posedness and approximation for a specific class of coupled parabolic-elliptic initial-boundary value problems with applications to geomechanics. The underlying thermo-chemo-poroelastic (TCPu) model constitutes the general theory of chemical thermo-poro-elasticity that fully couples thermal, chemical, hydrolic, and mechanical processes in homogeneous isotropic fluid-saturated porous media. The main results obtained within the dissertation research are summarized below. These results are fundamental to posing optimal control problems using borehole parameters as controls to achieve desired stress distributions in the neighborhood of the borehole.

A sufficient condition for well-posedness in the weak sense of the coupled parabolicelliptic initial-boundary value problems for the two-and three-dimensional TCPu models has been established. This condition depends on physical parameters of the system, coupling vectors, and the Korn's constant. For a two-dimensional annular region that does not experience pure rotation and has the ratio of inner to outer radii not exceeding  $e^{-1}$ , the sufficient condition for the well-posedness of the problem has been formulated in terms of physical parameters, coupling vectors, and the ratio of the inner and outer radii. The obtained results have provided a mathematical basis for the general theory of fully coupled chemo-thermo-poroelasticity in homogeneous isotropic porous media saturated by a fluid.

Numerical methods for solving fully coupled parabolic-elliptic initial-boundary value problems for two- and three-dimensional TCPu processes, including radially non-symmetric processes in an inclined borehole have been developed. These methods are based on a hybrid Fourier-finite-element approximation technique proposed in this research. The technique also involves a boundary penalization strategy and captures borehole geometry and mechanics. The developed numerical methods allow numerical analysis of a control problem in which the temperature and pressure on the borehole boundary are considered as control parameters used to achieve desirable stresses in the neighborhood of the borehole.

The experimental validation of the developed numerical method was performed in MATLAB using the two-dimensional radially symmetric fully coupled TCPu model with real data. The experimental results have confirmed that the hybrid Fourier-finite-element method provides correct and satisfactory numerical approximation to the coupled TCPu models for the purpose of boundary control and comprehensive numerical investigation of the corresponding fully coupled reservoir-geomechanical processes that impact borehole stability.

There are several promising directions for further research on coupled parabolicelliptic systems with applications to geomechanics:

- 1. An open question that immediately arises from this dissertation is establishing necessary conditions for well-posedness of the coupled parabolic-elliptic initialboundary value problems for the two- and three-dimensional TCPu models.
- 2. The second direction is the development of a well-posedness theory and numerical methods for coupled parabolic-elliptic systems with variable coefficients. The variability of coefficients results in nonlinearity of the coupled system but allows modeling fully coupled reservoir-geomechanical processes in inhomogeneous fluidsaturated porous media.
- 3. The other project is the development of the optimal boundary control theory for systems governed by coupled linear and nonlinear parabolic-elliptic partial

differential equations. This research direction is motivated by geomechanics and wellbore stability problems in which the temperature and pressure at the borehole are considered as control parameters used to obtain desirable stresses in the neighborhood of the borehole.

4. Another project is also dictated by the geomechanical nature of the problem and includes the development of the mathematical theory of fully coupled thermochemo-poroelastic fracture initiation and propagation processes in fluid-saturated porous media.

### Bibliography

- Y. Abousleiman and S. Ekbote. 2005. Solutions for the Inclined Borehole in a Porothermoelastic Transversely Isotropic Medium. ASME J. Appl. Mech. 72 (1): 102-114.
- [2] J.-L. Auriault, E. Sanchez-Palencia. 1977. Étude du comportement macroscopique d'un milieu poreux saturedéformable. *Journal de Mécanique*. 16: 575-603.
- [3] J. R. Barber. 2010. Elasticity. Dordrecht New York: Springer-Verlag.
- [4] Anastasia Belotserkovets, Jean H. Prevost. 2011. Thermoporoelastic Response of a Fluid-Saturated Porous Sphere: an Analytical Solution. Int. J. Eng. Sci. 49: 1415-1423.
- [5] M. A. Biot. 1941. General Theory of Three-Dimensional Consolidation. J. Appl. Phys. 12: 155-169.
- [6] M. A. Biot. 1956. Thermoelasticity and Irreversible Thermodynamics. J. Appl. Phys. 28: 240-253.
- [7] M. A. Biot. 1962. Mechanics of Deformation and Acoustic Propagation in Porous Media. Indiana Univ. Math. J. 21: 597-620.
- [8] M. A. Biot. 1972. Theory of Finite Deformations of Porous Solids. J. Appl. Phys. 33: 1482-1498.
- [9] W. B. Bradley. 1979. Failure of Inclined Borehole. J. Energy Resour. Technol., Trans. ASME 101: 233-239.
- [10] Coussy, O. 2004. Poromechanics. Chichester: John Wiley & Sons Ltd.
- [11] A. Diek, L. White, J.-C. Roegiers, K. Bartko, and F. Chang. 2011. A Fully Coupled Thermoporoelastic Model for Drilling in Chemically Active Formations. ARMA 11-406, 45th US Rock Mechanics / Geomechanics Symposium, San Francisco, June 26-29, 2011.
- [12] Diek, A., White, L. and Roegiers, J.-C. 2011. A Fully Coupled Thermoporoelastic Model for Drilling in Geological Formations. *Geothermal Resources Council 35th* Annual Meeting October 23-26, 2011 in San Diego, California, paper 176.
- [13] A. Diek, L. White, J.-C. Roegiers, D. Blankenship. 2011. A Fully Coupled Thermoporoelastic Model for Drilling in HPHT Formations. 12th International Congress on Rock Mechanics in Beijing, ISRM-2011: paper 184.
- [14] A. Diek, L. White, J.-C. Roegiers, J. Moore & J. D. McLennan. 2012. Borehole Preconditioning Of Geothermal Wells for Enhanced Geothermal System Reservoir Development. Proceedings, Thirty-Seventh Workshop on Geothermal

Reservoir Engineering, Stanford University, Stanford, California, January 30 -February 1, 2012. SGP-TR-194.

- [15] Dietrich Braess. 2007. Finite Elements. Cambridge: Cambridge University Press.
- [16] G. Duvaut and J.-L. Lions. 1976. Inequalities in Mechanics and Physics. Berlin: Springer.
- [17] Alexandre Ern and Sébastien Meunier. 2009. A Posteriori Error Analysis of Eulier-Galerkin Approximations to Coupled Elliptic-Parabolic Problems. M2AN Math. Model. Numer. Anal. 43: 353-375.
- [18] A. Ghassemi, A. Diek. 2002. Porothermoelasticity for Swelling Shales. J. Petrol. Sci. Eng. 34: 123-135.
- [19] A. Ghassemi, Q. Zhang. 2004. A Transient Fictitious Stress Boundary Element Method for Porothermoelastic Media. Eng. Anal. Bound. Elem. 28: 1363-1373.
- [20] A. Ghassemi, A. Diek. 2003. Linear Chemo-Poroelasticity for Swelling Shales: Theory and Application. J. Petrol. Sci. Eng. 38: 199-212.
- [21] C. O. Horgan. 1995. Korn's Inequalities and Their Applications in Continuum Mechanics. SIAM Review Vol. 37 (4): 491-511.
- [22] J. C. Jaeger, N. G. W. Cook, R. W. Zimmerman. 2007. Fundamentals of Rock Mechanics. Malden: Blackwell Publishing.
- [23] Tsuyoshi Kodashima and Michio Kurashige. 1996. Thermal Stresses in a Fluid-Saturated Poroelastic Hollow Sphere. J. Therm. Stresses. 19 (2): 139-151.
- [24] Tsuyoshi Kodashima and Michio Kurashige. 1997. Active Cooling and Thermal Stresses Reduction in a Poroelastic Hollow Sphere. J. Therm. Stresses. 20 (3-4): 389-405.
- [25] Vadim Koshelev, Ahmad Ghassemi. 2004 Complex Variable BEM for Thermoand Poroelasticity. Eng. Anal. Bound. Elem. 28: 825-832.
- [26] Kurashige, M. 1989. A Thermoelastic Theory of Fluid-Filled Porous Materials. Int. J. Solids Struct. 25 (9): 1039-1052.
- [27] Lions J.-L. 1971. Optimal Control of Systems Governed by Partial Differential Equations. New York - Heidelberg - Berlin: Springer-Verlag.
- [28] McTigue, D. 1986. Thermoelastic Response of Fluid-Saturated Porous Rock. J. Geophys. Res. 91 (B9): 9533-9542.
- [29] A. Mourits and F. M. Sattari. 1994. Coupling of Geomechanics and Reservoir Simulation Models. Computer Methods and Advances in Geomechanics: 2151-2158.

- [30] M. A. Murad and A. F. D. Loula. 1992. Improved Accuracy in Finite Element Analysis of Biots Consolidation Problem. *Comput. Meth. Appl. Mech. Engrg.* 95: 359-382.
- [31] M. A. Murad and A. F. D. Loula. 1994. On Stability and Convergence of Finite Element Approximations of Biots Consolidation Problem. *Internat. J. Numer. Methods Engrg.* 37: 645-667.
- [32] M. A. Murad, V. Thomée and A. F. D. Loula. 1996. Asymptotic Behavior of Semidiscrete Finite-Element Approximations of Biots Consolidation Problem. SIAM J. Numer. Anal. 33: 1065-1083.
- [33] W.K.S. Pao, R.W. Lewis and I. Masters. 2001. A Fully Coupled Hydro-Thermo-Poro-Mechanical Model for Black Oil Reservoir Simulation. Int. J. Numer. Anal. Meth. Geomech. 25: 1229-1256.
- [34] Saeed Rafieepour & Haniyeh Jalayeri & Cyrus Ghotbi & Mahmoud Reza Pishvaie. 2015. Simulation of Wellbore Stability with Thermo-Hydro-Chemo-Mechanical Coupling in Troublesome Formations: an Example from Ahwaz Oil Field, SW Iran Arab. J. Geosci. 8: 379-396.
- [35] R. L. Schiffman. 1971. A Thermoelastic Theory of Consolidation. *Environmental and Geophysical Heat Transfer.* C.J. Cremers, et al., (eds.). New York: ASME. Vol. 4: 78-84.
- [36] Showalter, R. E. 1977. Hilbert Space Methods for Partial Differential Equations. London - San Francisco: Pitman.
- [37] Showalter, R. E. 2000. Diffusion in Deformable Media. *IMA Volumes in Mathematics and its Applications* 131: 115-130.
- [38] Showalter, R. E. 2000. Diffusion in Poro-Elastic Media. J. Math. Anal. Appl. 251: 310-340.
- [39] Tao, Q. and Ghassemi, A. 2010. Poro-Thermoelastic Borehole Stress Analysis for Determination of the in-situ Stress and Rock Strength. *Geothermics.* 39 (3): 250-259.
- [40] R. Temam. 1977. Navier-Stokes Equations: Theory and Numerical Analysis. Amsterdam - New York - Oxford: North-Holland.
- [41] Terzaghi, K. 1923. Die berechnung der durchlassigkeitzifer des tones aus dem verlauf der hydrodynamischen spannungserscheinungen. Akademie der Wissenschaften, Mathematish-naturwissenschaftliche, Klasse Vienna. Part IIa. vol. 132: 125-138.
- [42] K. von Terzaghi. 1943. Theoretical Soil Mechanics. New York: Wiley.

- [43] Xiaonan Wang and Ahmad Ghassemi. 2012. A 3D Termal-Poroelastic Model for Geothermal Reservoir Stimulation. Proceedings, Thirty-Seventh Workshop on Geothermal Reservoir Engineering, Stanford University, Stanford, California, January 30 - February 1, 2012. SGP-TR-194.
- [44] Y. Yoloo, K. Yamagata and H. Nagoaka. 1971. Finite Element Method Applied to Biot's Consolidation Theory. Soils and Foundations. 11: 29-46.
- [45] A. Zenisek. 1984. The Existence and Uniqueness Theorem in Biot's Consolidation Theory. Apl. Mat. 29: 194-211.
- [46] Zhou X., Ghassemi A. 2009. Finite Element Analysis of Coupled Chemo-Poro-Thermo-Mechanical Effects around a Wellbore in Swelling Shale. Int. J. Rock Mech. Min. Sci. 46: 769-778.
- [47] O. C. Zienkiewicz, A. H. C. Chan, M. Pastor, B. A. Schrefler, T. Shiomi. 1999. Computational Geomechanics. Chichester: Wiley.

# Appendix A

# Physical Parameters of the Coupled TCPu System

The coefficients in the matrices and vectors (2.1.17) and (2.1.30),

$$M = \begin{bmatrix} \Lambda & \Sigma & \Phi \\ 0 & \phi & 0 \\ \Gamma & \chi & \Psi \end{bmatrix}, \qquad A = \begin{bmatrix} \frac{k^T}{T_F} & \frac{\rho_f R T_F D^T}{M^s (1 - C_F)} & -K^T \\ C_F D^T & D & -\frac{k \Re}{\eta} \\ -K^T & -\frac{\rho_f \Re \Omega k}{\eta} & \frac{k}{\eta} \end{bmatrix}$$
$$\vec{b}_0 = \begin{bmatrix} \zeta \\ 0 \\ \alpha \end{bmatrix}, \qquad \vec{b}_1 = \begin{bmatrix} \tilde{\zeta} \\ -\xi \\ \tilde{\alpha} \end{bmatrix}$$

are the following:

$$\begin{split} \Lambda &= \frac{c}{T_F} - s_F \beta_s \tilde{\omega}, \qquad c = (1 - \phi) \rho_s c_s + \phi \rho_f c_f, \quad s_F = c_f - \frac{\beta_f p_F}{\rho_f}, \quad \tilde{\omega} = \frac{\omega}{1 - C_F} \\ \Sigma &= \xi \beta_s, \qquad \xi = \frac{R T_F \tilde{\omega} (1 - 2 C_F)}{M^s C_F} \\ \Phi &= \Xi - \phi \beta_f + \frac{\beta_s \tilde{\omega}}{\rho_f}, \qquad \Xi = (\alpha - \phi) \beta_s, \qquad \alpha = 1 - \frac{K}{K_s} \\ \Gamma &= \Xi - \phi \beta_f - \frac{s_F \tilde{\omega}}{K_s}, \qquad \chi = -\frac{\xi}{K_s} \\ \Psi &= \psi + \frac{\phi}{K_f} - \frac{\tilde{\omega}}{\rho_f K_s}, \qquad \psi = \frac{\alpha - \phi}{K_s} \\ \Omega &= \frac{R T_F}{M^s (1 - C_F) C_F} \\ \zeta &= K \beta_s, \qquad \tilde{\zeta} = \zeta + s_F \tilde{\omega}, \qquad \tilde{\alpha} = \alpha - \frac{\tilde{\omega}}{\rho_f} \end{split}$$

where  $s_F$  is specific fluid entropy, respectively, R is the universal gas constant,  $\alpha$  is Biot effective stress coefficient, and K is the rock's bulk modulus. Other physical

properties of the rock/fluid system and input parameters for examples are given in Table A.1 and Table A.2, respectively.

Drained elastic modulus, $E$	$45 \times 10^9$ Pa
Solid bulk modulus, $K_s$	$65 \times 10^9$ Pa
Drained Poisson's ratio, $\nu$	0.27
Permeability/Fluid viscosity ratios, $k/\eta$	$10^{-17} \text{ m}^2/(\text{Pa}\cdot\text{s})$
Reference porosity, $\phi$	0.15
Fluid density, $\rho_f$	$1111 \ {\rm kg/m^3}$
Fluid specific heat capacity, $c_f$	$4186 \text{ J/(Kg^{\circ}K)}$
Fluid bulk modulus, $K_f$	$3.3 \times 10^9$ Pa
Fluid volumetric thermal expansion coefficient, $\beta_f$	$3 \times 10^{-4} ^{\circ}\mathrm{K}^{-1}$
Solid density, $\rho_s$	$2.83 \times 10^3 \text{ kg/m}^3$
Solid specific heat capacity, $c_s$	$920 \text{ J/(Kg^{\circ}K)}$
Solid volumetric thermal expansion coefficient, $\beta_s$	$2.4 \times 10^{-5} ^{\circ}\mathrm{K}^{-1}$
Rock thermal conductivity coefficient, $k^T$	$4 \text{ W/(m^{\circ}K)}$
Thermal osmosis coefficient, $K^T$	$10^{-11} \text{ m}^2/(\text{s}^{\circ}\text{K})$
Solute molar mass, $M^s$	$0.1111 \ \rm kg/m^3$
Solute reflected fraction, $\mathfrak{R}$	0.01
Solute chemical diffusion coefficient, $D$	$10^{-9} \text{ m}^2/\text{s}$
Solute thermal diffusion coefficient, $D^T$	$10^{-10} \text{ m}^2/(\text{s}^{\circ}\text{K})$
Chemical stress coupling parameter, $\omega$	$100 \text{ kg/m}^3$
Universal gas constant, $R$	$8.3 \ \mathrm{JK^{-1}/mol^{-1}}$

Table A.1: Physical properties of the rock/fluid system

Table A.2: Input parameters for experimental validation

In situ hydrostatic stress,	91MPa (Therm. ex.)
$\sigma_x = \sigma_y$	88.5 MPa (Chem. ex.)
In situ non-hydrostatic stress	$\sigma_x = 91 \text{ MPa}$
	$\sigma_y = 86 \text{ MPa}$
In situ formation pore pressure, $p_F$	52 MPa
Mud pressure, $p_B$	60 MPa
Mud solute mass fraction, $C_B$	0.2
Reference formation solute mass fraction, $C_F$	0.1
In situ formation temperature, $T_F$	135°C
Mud temperature, $T_B$	$85 \ ^{\circ}C$
Wellbore radius, $R_B$	0.1 m